

## Adiabatic control of quantum systems

Nicolas Augier

### ▶ To cite this version:

Nicolas Augier. Adiabatic control of quantum systems. Optimization and Control [math.OC]. Université Paris Saclay (COmUE), 2019. English. NNT: 2019SACLX042. tel-02434725

## HAL Id: tel-02434725 https://theses.hal.science/tel-02434725v1

Submitted on 10 Jan2020

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# Contrôle adiabatique des systèmes quantiques

Thèse de doctorat de l'Université Paris-Saclay préparée à l'École Polytechnique

Ecole doctorale n°574 Ecole doctorale de mathématiques Hadamard (EDMH) Spécialité de doctorat : Mathématiques appliquées

Thèse présentée et soutenue à Palaiseau, le 27 septembre 2019, par

### **NICOLAS AUGIER**

Composition du Jury :

Alain Sarlette	
Chargé de recherche, INRIA (QUANTIC)	Examinateur
Anne De Bouard	
Directrice de Recherche, CNRS (CMAP)	Examinatrice
Francesca Chittaro	
Maître de Conférence, Université de Toulon	Examinatrice
Ugo Boscain	
Directeur de Recherche, CNRS (LJLL)	Directeur de thèse
Witold Respondek	
Professeur, INSA Rouen	Rapporteur
Mario Sigalotti	
Directeur de recherche, INRIA (LJLL)	Co-directeur de thèse
Hans-Rudolf Jauslin	
Professeur, Université Bourgogne Franche Comté (ICB)	Rapporteur
Jean-Paul Gauthier	
Professeur émérite, Université de Toulon	Président du jury

# Contrôle adiabatique des sytèmes quantiques

## Adiabatic control of quantum systems

Nicolas Augier

CMAP, Equipe INRIA CAGE (LJLL) École Polytechnique

Thèse présentée pour l'obtention du grade de Docteur en mathématiques de l'Université Paris-Saclay

Université Paris-Saclay

Septembre 2019

#### Résumé

Le but principal de la thèse est d'étudier les liens entre les singularités du spectre d'un Hamiltonien quantique contrôlé et les questions de contrôlabilité de l'équation Schrödinger associée.

La question majeure qui se pose est de savoir comment contrôler une famille de systèmes quantiques dépendant des paramètres avec une entrée de commande commune. Ce problème de contrôlabilité d'ensemble est lié à la conception d'une stratégie de contrôle robuste lorsqu'un paramètre (une fréquence de résonance ou une inhomogénéité de champ de contrôle par exemple) est inconnu, et constitue un enjeu important pour les expérimentateurs. Grâce à l'étude des familles à un paramètre de Hamiltoniens et de leurs singularités génériques, nous donnons une stratégie de contrôle explicite pour le problème de contrôlabilité d'ensemble lorsque certaines conditions géométriques sur le spectre des Hamiltoniens sont satisfaites. Le résultat est basé sur la théorie de l'approximation adiabatique et sur la présence de courbes d'intersections coniques de valeurs propres du Hamiltonien contrôlé. La technique proposée fonctionne à la fois pour des systèmes évoluant dans des espaces de Hilbert de dimension finie et de dimension infinie. Nous étudions ensuite le problème de la contrôlabilité d'ensemble sous des hypothèses moins restrictives sur le spectre, à savoir la présence de singularités non-coniques. Sous des conditions génériques, de telles singularités n'apparaissent pas pour des systèmes uniques, mais apparaissent pour des familles de systèmes à un paramètre.

Pour l'étude d'un système unique, nous nous concentrons sur une classe de courbes dans l'espace des contrôles, appelées les courbes non-mixantes (définies dans [22]), qui peuvent optimiser la dynamique adiabatique près des intersections coniques et non coniques. Elles sont liées à la géométrie des espaces propres du Hamiltonien contrôlé et l'approximation adiabatique possède une meilleure précision le long de celles-ci.

Nous proposons d'étudier la compatibilité de l'approximation adiabatique avec la Rotating Wave Approximation. De telles approximations sont généralement combinées par les physiciens. On montre que cela ne se justifie pour les systèmes quantiques à dimensions finies que dans certaines conditions sur les échelles de temps. Nous étudions également les questions de contrôle d'ensemble dans ce cas.

#### Abstract

The main purpose of the thesis is to study the links between the singularities of the spectrum of a controlled quantum Hamiltonian and the controllability issues of the associated Schrödinger equation.

The principal issue that is developed is how to control a parameter-dependent family of quantum systems with a common control input. This problem of ensemble controllability is linked to the design of a robust control strategy when a parameter (a resonance frequency or a control field inhomogeneity for instance) is unknown, and is an important issue for experimentalists. Thanks to the study one-parametric families of Hamiltonians and their generic singularities, we give an explicit control strategy for the ensemble controllability problem when geometric conditions on the spectrum of the Hamiltonian are satisfied. The result is based on adiabatic approximation theory and on the presence of curves of conical eigenvalue intersections of the controlled Hamiltonian. The proposed technique works for systems evolving both in finite-dimensional and infinite-dimensional Hilbert spaces. Then we study the problem of ensemble controllability under less restrictive hypotheses on the spectrum, namely the presence of non-conical singularities. Under generic conditions such non-conical singularities are not present for single systems, but appear for one-parametric families of systems. , For the study of a single system, we focus on a class of curves in the space of controls, called the non-mixing curves (defined in [22]), that can optimize the adiabatic dynamics near conical and non-conical intersections. They are linked to the geometry of the eigenspaces of the controlled Hamiltonian and the adiabatic approximation holds with higher precision along them.

We propose to study the compatibility of the adiabatic approximation with the rotating wave approximation. Such approximations are usually done in cascade by physicists. It is shown in this work that this is justified for finite dimensional quantum systems only under certain conditions on the time scales. We also study ensemble control issues in this case.

# Remerciements

Dans un premier temps, je tiens à remercier vivement mes directeurs de thèse, Mario Sigalotti et Ugo Boscain, qui m'ont offert l'opportunité de travailler sur ce sujet vaste, riche et passionnant, à la frontière de nombreux domaines. Je suis conscient de l'immense chance que j'ai eue de travailler avec Mario et Ugo. Je les remercie d'avoir toujours été disponibles, attentifs et de m'avoir toujours encouragé. Ils m'ont permis d'avancer sur le chemin exigeant de la recherche, parsemé de singularités, lentement mais sûrement, tout en me prévenant lorsque je prenais une mauvaise direction (non-conique). Merci pour leur confiance en mon travail, leur sympathie et leur profonde bienveillance.

Je souhaite ensuite remercier les membres du jury, dont la présence m'honore. Merci aux rapporteurs Witold Respondek et Hans-Rudolf Jauslin pour leur travail de relecture et leurs remarques constructives. Merci à Francesca Chittaro d'avoir accepté d'examiner ma thèse, j'espère que le décalage horaire ne nous empêchera pas de parler de courbes nonmixantes cette fois. Merci à Jean-Paul Gauthier, dont les travaux en théorie du contrôle, quantique ou non, ont inspiré notre approche. Je remercie Alain Sarlette, dont les recherches en contrôle quantique ainsi que les présentations auxquelles j'ai assisté ont enrichi notre étude. Merci enfin à Anne De Bouard d'avoir accepté de représenter le CMAP pour cette soutenance.

Je remercie tous les membres du CMAP, en particulier Nasséra et Alexandra qui ont toujours facilité les démarches et rendu possibles les déplacements. Je remercie Emmanuel Fullenwarth et Thierry Bodineau d'avoir permis à cette soutenance d'avoir lieu. Merci enfin à Aline Lefèbvre-Lepot de m'avoir intégré à l'équipe des moniteurs du département de mathématiques appliquées.

Merci à Pierre Rouchon qui a organisé le trimestre IHP "Measurement and control of quantum systems" et m'a permis d'effectuer une présentation en juillet 2018.

Merci aux membres de la communauté très stimulante du contrôle géométrique, que j'ai pu rencontrer à Porquerolles en 2017, et à ceux du contrôle quantique, en particulier Paolo Mason, Thomas Chambrion, Marco Caponigro, Nabile Boussaïd et Dominique Sugny, que j'ai pu rencontrer durant le trimestre IHP et à Besançon en 2018. Merci à toutes les personnes du LJLL, qui m'ont accueilli durant la deuxième partie de ma thèse, en particulier Salima Lounici, Luis Almeida, Catherine Drouet, Malika Larcher, Laurent Boudin, aux membres de l'équipe Inria Cage, en particulier à Emmanuel Trélat, qui m'a dirigé vers Ugo et Mario lors de ma recherche de stage de Master 2.

Merci à mes amis/collègues Mathieu Kohli et Ludovic Sacchelli. Merci à toi Bro de m'avoir souvent écouté patiemment et donné des idées. Merci à toi Ludovic, j'espère qu'on aura la peau de ces champs de direction singuliers, merci de m'y avoir initié en tous cas.

Je remercie tous les doctorants/Postdocs du LJLL : Gontran (qui m'a permis de découvrir un bon resto, on t'enlève bientôt les petites roues ne t'en fais pas), Ana (j'espère que l'attente à Aix n'a pas été trop longue), Allen, Lucile, Gabriela, Christophe, Ludovic, Amaury, Lydie, Alex.R, Nicolas C., Federica, Idriss, Olivier, Lilian, Ivan, Eugenio, Jules, Elise, Francesco, Rémi, et tous ceux que j'oublie. Je remercie aussi bien les doctorants du CMAP, Kevish, Tristan, Rémi, Fedor, Alexey, Romain, Aymeric, Léa, Hadrien, Perle, Florian, Jean-Bernard, Ludovic et tous les autres.

Un grand merci à mes cobureaux du LJLL : Merci à Katia et Cécile qui ont eu à supporter mon désordre, des OVNIs, et surtout la musique de Léo. Merci à Léo malgré tout. Je remercie Julia et Alexandre qui ont su me remettre à ma place lorsque je volais leurs chaises. Merci aussi à Fabien et Nacime.

Je remercie ma famille, mes parents, Daniel et Odile pour m'avoir tant de fois épuisé en montagne, Denis, Elisete, Michèle, Tata, Thierry, les cousin-e-s Aurore, Max, Seb, Paul et Thomas, et puis Marc, Christine et tous les autres. Je remercie aussi la famille d'Ariane au sens large. Un grand merci va aussi à Caro, Pierre T., Emilie, Fafa, Anaïs, Cristina et Pierre L.

Merci à tous les potes de Cachan, Edwin, Plancha, Bro, Mathilde, Michaël, Yann, Léa, Amine, Léa et tous les autres, et au T. J'ai aussi l'honneur de remercier la troupe de Jussieu 2015/2016 (Anouk, Marc, Clément M., From, Louis). Rémi, mon acolyte de toujours, qui connaît tout sur les vis en carbone maintenant, merci d'être là. Vient ensuite le tour du  $\text{Troll}^{TM}$ , dont la voix douce et l'humour décapant ont toujours résonné en moi, de Valence à Mésage, de Mésage à Valence, de Toulon à Lille, Marseille, depuis la Corse, San Peyre, j'espère qu'il y en aura beaucoup d'autres. Anouk et Alex vous êtes géniaux, j'vous jure, vous m'apportez tant. Marine, je ne louperai pas ton anniversaire l'an prochain promis, je salue toute la team Riousset au passage. Max, je n'oublie pas ta promesse de ballade en moto, hein! Nico, que dire, tu es toujours là depuis tout ce temps, de Toulon à Paris, des vacances, une amitié en cuivre comme on dit, merci! Le Groum, j'attends toujours que tu me cuisines des aubergines. Leïla, Hugo, à bientôt j'espère! Antoine, n'oublie pas qu'on doit se mettre au tandem... Louis, dont le sens du débat est aiguisé, même du haut de la Marmolada, merci de m'avoir accueilli dans ton antre aux mille diamants. A la team Pisane, Louise, Oury, Julia, Sandro, dont le Lambrusco est toujours le bienvenu. Charles, j'espère que ton club fétiche gagnera un jour! Nicoto, j'aurais tant aimé continuer nos ballades endiablées, tu sais... Pierre, Thom, Ben, Romu, Marie, j'ai hâte de vous retrouver, à la mer, en musique, à Lyon, ou à la montagne. Le Pat, qui pour des raisons peu recevables ne sera pas là aujourd'hui, tu es nul à Fifa mais je t'aime bien quand même, tu as parfois de bons goûts musicaux, bonne soutenance à toi ! HugoW qui est nul en escalade, mais sympa quand même ! Merci aux gens du groupe de musique, vous vous reconnaîtrez, c'était que du bonheur ! Alex T., ton jambon à l'os restera gravé dans mes papilles à tout jamais. Merci à Nicolas D., pour qui j'ai été un colloc ephémère. Merci à Sean, qui me considère, à tort ou à raison, comme un master chief. Luce, merci pour ces bons coeurs de canard inoubliables. Marco, on se retrouvera à Sainte-Hélène, ou plus près, je n'en doute pas !

Merci à Ariane, qui m'offre un bonheur unique!

Et bien sûr merci à Mémé,

# Table des matières

1	Intr	oducti	- fr	1
	1.1	Prélim	inaires de physique quantique, théorème adiabatique	1
	1.2	Thème	es d'études	6
		1.2.1	Contrôle d'ensemble avec deux contrôles réels	7
		1.2.2	Classification des singularités du champ non-mixant	15
		1.2.3	Vers le contrôle d'ensemble avec un seul contrôle	21
<b>2</b>	Intr	oducti	on	<b>25</b>
	2.1	Quant	um physics preliminaries, adiabatic theorem	25
	2.2	Topics	of investigation	30
		2.2.1	Ensemble control with two real controls	31
		2.2.2	Classification of the singularities of the non-mixing field	38
		2.2.3	Towards ensemble control with a single input	44
3	Ens	emble	control of quantum systems with two controls: conical case	49
	3.1	Introd	uction	50
	3.2	Basic o	definitions and statement of the main results in the finite-dimensional	
		case		50
	3.3	Proof	of the ensemble controllability result in the finite-dimensional case $\ .$	54
	3.4	Exam	ple 1: Two-level system driven by a chirped pulse	56
	3.5	Permu	tations	60
	3.6	Generi	icity	63
	3.7	Multic	limensional set of parameters	64
		3.7.1	Chirped pulses for two-level systems with two parameters	65
		3.7.2	Example 2: STIRAP	65
	3.8	Extens	sion to the infinite-dimensional case	68
		3.8.1	Example 3: Eberly–Law-like models	70
	3.9	Appen	dix	73

4	Ens	emble	control of quantum systems with two controls: non-conica	al
	case	Э		77
	4.1	Introd	luction	. 78
	4.2	Basic	facts and normal forms	. 82
		4.2.1	Generic families of 2-level Hamiltonians	. 82
		4.2.2	Admissible transformations	. 87
		4.2.3	Normal forms for the non parametric case $\ldots \ldots \ldots \ldots \ldots$	. 91
		4.2.4	Normal forms for the parametric case	. 96
	4.3	Gener	ic global properties of the singular locus	. 99
		4.3.1	Proof of Lemma 4.1.1 and Theorem 4.1.2	. 99
		4.3.2	Generic self-intersections of $\pi(f)$	. 99
	4.4	Adiaba	atic control through a semi-conical intersection of eigenvalues	. 100
		4.4.1	Adiabatic dynamics	. 101
		4.4.2	Regularity of the eigenpairs along smooth control paths	. 102
		4.4.3	Dynamical properties at semi-conical intersections of eigenvalues .	. 104
	4.5	Contro	ol of a continuum of systems	. 106
		4.5.1	Ensemble adiabatic dynamics	. 106
		4.5.2	Controllability properties between the eigenstates for the normal	
			forms	. 107
		4.5.3	The control path $(u, v)$ exits from $\pi(f)$ .	. 110
		4.5.4	Proof of Theorem 4.1.3	. 111
	4.6	Extens	sion to $n$ -level systems	. 111
		4.6.1	Generic assumptions on n-level Hamiltonians and adiabatic decou-	
			pling	. 111
		4.6.2	Adiabatic decoupling	. 113
		4.6.3	Semi-conical intersections for $n$ -level quantum systems $\ldots \ldots$	. 115
		4.6.4	Controllability result	. 117
	4.7	Appen	ndix	. 117
		4.7.1	Averaging theorems and estimates of oscillatory integrals $\ldots$ .	. 117
		4.7.2	Two useful lemmas	. 119
<b>5</b>	Cla	ssificat	ion of the singularities of the non-mixing field	121
	5.1	Genera	al definition of the non-mixing field	. 122
	5.2	The ne	on-mixing field for two level systems	. 123
		5.2.1	Classification of the singularities of the non-mixing field of a generic	
			two-level system	. 124
		5.2.2	Bifurcations of the non-mixing field for two-level systems and the	
			avoided crossing problem	. 125
		5.2.3	Parametric families of real Hamiltonians	. 128

	5.3	The ne	on-mixing field for general quantum systems	134
		5.3.1	Useful results about line fields	134
		5.3.2	Non-mixing field	138
		5.3.3	Singularities of $\chi_{j-1,j}$ at intersections $(j, j+1)$	139
6	Con	trol of	f quantum systems with a single input	147
	6.1	Gener	al framework and main results	148
		6.1.1	Problem formulation	148
		6.1.2	Main results	150
	6.2	Appro	ximation results	153
		6.2.1	Variation formula	153
		6.2.2	Regularity of the eigenstates	153
		6.2.3	Averaging of quantum systems	154
		6.2.4	Perturbation of an adiabatic trajectory	155
		6.2.5	Parametric case	158
	6.3	Contro	ol of two-level systems	159
		6.3.1	Control strategy for two-level systems and simulations	159
		6.3.2	Robustness of the control strategy with respect to amplitude of	
			control inhomogeneities	160
	6.4	Contro	ol of STIRAP Process	162
Ré	éférei	nces		165

Table des matières

# Table des figures

1.1	Deux valeurs propres de $H(u)$ au voisinage d'une intersection conique, tra-	
	cées en fonction de $u \in \mathbb{R}^k$ avec $k = 2$	4
1.2	Chemin de transition à la vitesse $\epsilon$ when $k = 2. \ldots \ldots \ldots \ldots$	6
1.3	Les deux valeurs propres de $H_f(u, v)$ au voisinage d'une intersection semi-	
	conique, tracées en fonction of $(u, v)$	10
1.4	Intersection semi-conique pour un processus STIRAP dégénéré	11
1.5	Chemin de contrôle passant à travers la singularité dans la direction non-	
	conique.	12
1.6	Un contrôle réalisant une transition d'ensemble entre $\phi_1^z(u_0, v_0)$ et $\phi_4^z(u_0, v_0)$ .	13
1.7	Une courbe de contrôles $(u, v)$ réalisant des transitions uniformes en $z \in$	
	$[z_0, z_1]$ entre deux niveaux comportant des intersections F-semi-coniques	15
1.8	Un noeud critique	16
1.9	Un exemple de singularité semi-conique hyperbolique. L'indice est 0. La	
	direction non-conique est $(0, 1)$ . Deux droites (en rouge) traversent la sin-	
	gularité dans des directions coniques	17
1.10	Un exemple de singularité semi-conique elliptique. L'indice est 2. La direc-	
	tion non-conique est $(0, 1)$ . Toute courbe non-mixante passe à la singularité	
	dans la direction non-conique.	17
1.11	Combinaison de singularités d'indice demi-entier et entier dans les cristaux	
	liquides, image issue de $[8]$	19
1.12	La singularité Lemon, d'indice $\frac{1}{2}$	20
1.13	La singularité Monstar, d'indice $\frac{1}{2}$	20
1.14	La singularité Star, d'indice $-\frac{1}{2}$	20
1.15	Champ non-mixant entre $\lambda_1$ et $\lambda_2$ pour le STIRAP du Corollaire 1.2.10	22
2.1	Two eigenvalues of $H(u)$ in a neighborhood of a conical intersection when	
	$k = 2$ , as a function of $u \in \mathbb{R}^2$	28
2.2	Climbing path at speed $\epsilon$ when $k = 2$	30

## Table des figures

2.3	The two eigenvalues of $H_f(u, v)$ in a neighborhood of a semi-conical inter- section as a function of $(u, v)$	33
2.4	Semi-conical intersection for a degenerate STIRAP process	35
2.5	Control path passing in the non-conical direction.	36
2.6	A control realizing an ensemble transition between $\phi_z^z(u_0, v_0)$ and $\phi_z^z(u_0, v_0)$	37
2.7	A curve of controls $(u, v)$ achieving uniform transitions w.r.t. $z \in [z_0, z_1]$	0.
	between two levels having F-semi-conical intersections $\ldots \ldots \ldots \ldots$	39
2.8	A critical node	40
2.9	A hyperbolic semi-conical singularity. The non-conical direction is $(0, 1)$ . The red curves are the only non-mixing curves passing through the singularity in conical directions. The index is $0, \ldots, \ldots, \ldots, \ldots$ .	41
2.10	An <i>elliptic semi-conical singularity</i> . The non-conical direction is $(0, 1)$ . Every non-mixing curve passes through the singularity in the non-conical direction. The index is 2.	41
2.11	Combination of integer and half-integer index singularities in nematic crys-	
	tals, picture taken from [8]	42
2.12	The Lemon singularity, of index $\frac{1}{2}$	43
2.13	The Monstar singularity, of index $\frac{1}{2}$	43
2.14	The Star singularity, of index $-\frac{1}{2}$	43
2.15	Non mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP of Corollary 2.2.10. $~$ .	45
3.1	An example of conical intersection in the case $d = N = 2$	51
3.2	The component $\hat{\gamma}_j$ is such that Assumption $A_j$ is satisfied	52
3.3	A control $u(\cdot)$ as in the statement of Theorem 3.2.5	53
3.4	A control realizing an ensemble transition between $\phi_1^z(u_0)$ and $\phi_4^z(u_0)$	54
3.5	The control $u(\cdot)$ .	58
3.6	Two-level system driven by a chirped pulse. Here $E = 2, z \in [-1, 1],$ $u_1(t) = \frac{3}{2}(1 - \cos(4\pi t)), u_2(t) = -3\cos(2\pi t), \text{ for } t \in [0, \frac{1}{2}] \text{ and } \varepsilon = 0.004.$ The frequency of the pulse is $2E + u_2(\varepsilon t) = 4 - 3\cos(2\pi\varepsilon t)$ which varies monotonically between 1 and 7	59
3.7	A situation in which Assumption $\mathbf{P}$ is verified	61
3.8	On the left the functions $f_i$ and the times $t_i$ . On the right the control	01
0.0	$u(\cdot)$ . In this case $N = 3$ , $\mu = 3$ , $\tau(1) = 1$ , $\tau(2) = 2$ , $\tau(3) = 1$ . Hence, $u(\cdot)$	
	passes through $\pi(\hat{\gamma}_1)$ , $\pi(\hat{\gamma}_2)$ , and again $\pi(\hat{\gamma}_1)$ while connecting $u_0$ to $u_1$ .	62
3.9	The sets $\pi(\gamma_1)$ and $\pi(\gamma_2)$ for the STIRAP process	66

3.10	Example of STIRAP process with $E_1 = -1, E_2 = 0, E_3 = 1, \alpha_1, \alpha_3 \in$
	$[-0.1, 0.1], \beta_1, \beta_2 \in [0.8, 1.2]$ . In (a) we give the parametric plot of the
	control, in (b) the shape of its components as functions of time and in (c)
	the components of the wave function for the three choices of parameters
	$(\alpha_1, \alpha_3, \beta_1, \beta_2) = (0, 0, 1, 1), (0.1, -0.1, 0.8, 1.2), (-0.1, 0.1, 1.2, 0.8).$ For this
	simulation we used $\varepsilon = 0.05$
3.11	The eigenvalues $\lambda_1^z(u) \leq \lambda_2^z(u) \leq \lambda_3^z(u)$ of $H^z(u)$ separated from the rest
	of the spectrum (which is contained in the shaded regions)
3.12	The eigenvalues of $H^{z}(u_{1}, 0)$ (left) and $H^{z}(0, u_{2})$ (right)
3.13	A loop in the plane $(u_1, u_2)$ inducing a population transfer from the first
	to the fourth eigenstate of the drift Hamiltonian
4.1	Conical intersection as a function of the controls $(u, v) \in \mathbb{R}^2$
4.2	Semi-conical intersection of eigenvalues as a function of the controls $(u, v) \in$
	$\mathbb{R}^2. \ldots \ldots$
4.3	A curve $(u, v)$ as in the statement of Theorem 4.1.3
4.4	A control path passing at a semi-conical intersection in the non-conical
	direction as a function of the controls $(u, v) \in \mathbb{R}^2$
5.1	Non-mixing curves at a conical intersection
5.2	An example of <i>elliptic semi-conical singularity</i> with $h(u) = 1 - u$ for every $u$ .
	The non-conical direction is $(0, 1)$ . Every non-mixing curve passes through
	the singularity in the non-conical direction. The index is 2
5.3	A example of hyperbolic semi-conical singularity with $h(u) = 1 + u$ for
	every $u$ . The non-conical direction is $(0,1)$ . The red curves are the only
	non-mixing curves passing through the singularity in conical directions.
	The index is 0. $\ldots \ldots \ldots$
5.4	Elliptic semi-conical singularity in the coordinates $(r, \theta) \in \mathbb{R} \times \mathbb{S}^1$ after
	desingularization with $h(u) = 1 - u$ for every $u$
5.5	Hyperbolic semi-conical singularity in the coordinates $(r, \theta) \in \mathbb{R} \times \mathbb{S}^1$ after
	desingularization with $h(u) = 1 + u$ for every $u$
5.6	Energy levels of $\tilde{H}_z(u, v)$ with $z < 0$ , as a function of $(u, v)$
5.7	Energy levels of $\tilde{H}_0(u, v)$ , as a function of $(u, v)$
5.8	Energy levels of $\tilde{H}_z(u, v)$ with $z > 0$ , as a function of $(u, v)$
5.9	Saddle-center bifurcation for $z < \overline{z}$ , $\overline{z}$ being the value of z for which the
	bifurcation occurs.
5.10	Saddle-center bifurcation for $z = \overline{z}$
5.11	Hyperbolic non-mixing curves for $z < 0.$
5.12	Elliptic non-mixing curves for $z < 0$

5.13	Hyperbolic non-mixing curves for $z > 0$
5.14	Elliptic non-mixing curves for $z > 0$
5.15	The Lemon singularity, of index $\frac{1}{2}$
5.16	The Monstar singularity, of index $\frac{1}{2}$
5.17	The Star singularity, of index $-\frac{1}{2}$
5.18	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP of Corollary 5.3.13. $$ . 143 $$
5.19	Spectrum as a function of $(u, v)$ for the STIRAP, where $E_1 < E_2 < E_3$ 143
5.20	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the degenerate STIRAP, where
	$E_1 = E_2 < E_3.  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $
5.21	Spectrum as a function of $(u, v)$ for the degenerate STIRAP, where $E_1 =$
	$E_2 < E_3. \ldots \ldots$
5.22	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP, where $E_2 < E_1 < E_3$ . 144
5.23	Spectrum as a function of $(u, v)$ for the STIRAP, where $E_2 < E_1 < E_3$ 144
5.24	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP, where $E_1 < E_2 = E_3$ . 144
5.25	Spectrum as a function of $(u, v)$ for the STIRAP, where $E_1 < E_2 = E_3$ 144
5.26	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP, where $E_1 < E_3$ ,
	$E_2 > E_3.  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $
5.27	Spectrum as a function of $(u, v)$ for the STIRAP, where $E_1 < E_3$ and
	$E_2 > E_3.  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  \dots  $
5.28	Non-mixing field between $\lambda_1$ and $\lambda_2$ for the STIRAP, where $E_1 = E_2 = E_3$ . 145
5.29	Spectrum as a function of $(u, v)$ for the STIRAP, where $E_1 = E_2 = E_3$ 145
6.1	Fidelity $ \langle \psi_{\epsilon}(\tau), e_2 \rangle ^2$ as a function of the time variable $\tau \in [0, 1]$ with
	$\epsilon = 0.01, \alpha = 1.5, \text{ and } E = 1.$
6.2	Squared norm of the difference between $\psi_{\epsilon}$ and $\hat{\psi}_{\epsilon}$ as a function of the time
	variable $\tau \in [0, 1]$ with $\epsilon = 0.01$ , $\alpha = 1.5$ , and $E = 1.$
6.3	Fidelity $ \langle \psi_{\epsilon}(\delta, 1), e_2 \rangle ^2$ as a function of the amplitude inhomogeneity $\delta$ ,
	with $\epsilon = 0.01$ , $\alpha = 1.5$ , and $E = 1$
6.4	Fidelity $ \langle \psi_{\epsilon}(E,1), e_2 \rangle ^2$ as a function of the drift term E, with $\epsilon = 0.01$
	and $\alpha = 1.5$
6.5	Time-evolution of $ \langle \psi_{\epsilon}(\delta,\tau), e_3 \rangle ^2$ where $\tau$ is the renormalized time, with
	$\delta_1 = 0.5$ in blue, $\delta_1 = 1$ in orange, $\delta_1 = 1.5$ in green, with $\epsilon = 0.02$ and
	$\alpha = 1.5. \dots $
6.6	Time-evolution of $v_{12}(\tau)$ in blue, and of $v_{23}(\tau)$ in orange $\ldots \ldots \ldots$
6.7	Time-evolution of $ \langle \psi_{\epsilon}(\delta,\tau), e_3 \rangle ^2$ where $\tau$ is the renormalized time, with
	$\delta_1 = \delta_2 = 1, \ \epsilon = 0.02 \text{ and } \alpha = 0.5. \dots $
6.8	Time-evolution of $ \langle \psi_{\epsilon}(\delta,\tau), e_3 \rangle ^2$ where $\tau$ is the renormalized time, with
	$\delta_1 = \delta_2 = 1, \epsilon = 0.02$ and $\alpha = 1.5$ , in the resonant case $ E_1 - E_2  =  E_2 - E_3 .164$

## Chapitre

# Introduction

#### Contents

1.1	Préli	iminaires de physique quantique, théorème adiabatique .	1
1.2	2 Thèmes d'études		6
	1.2.1	Contrôle d'ensemble avec deux contrôles réels	7
	1.2.2	Classification des singularités du champ non-mixant $\ . \ . \ .$	15
	1.2.3	Vers le contrôle d'ensemble avec un seul contrôle	21

## 1.1 Préliminaires de physique quantique, théorème adiabatique

Pour un système quantique fermé général sous l'action d'un contrôle u à valeurs dans un ouvert U de  $\mathbb{R}^m$ , nous sommes intéressés par l'équation contrôlée de la forme

$$i\frac{d\psi}{dt}(t) = H(u(t))\psi(t), \qquad \psi(t) \in \mathcal{H},$$
(1.1)

avec H(u) essentiellement auto-adjoint sur l'espace de Hilbert complexe séparable  $\mathcal{H}$  pour chaque valeur de  $u \in U$ , avec un domaine dense commun pour tout  $u \in U$ . Pour  $\mathcal{H}$  de dimension finie, la contrôlabilité de l'équation (2.9) a été largement étudiée par des méthodes d'algèbre de Lie (voir [6, 9, 34]). Pour  $\mathcal{H}$  est de dimension infinie, l'équation (2.9) n'est en général pas contrôlable de façon exacte [11]. De nombreux résultats de contrôlabilité plus faibles ont été prouvés [17, 25, 22]. Cependant, dans les deux cas, il existe peu de résultats théoriques sur la robustesse des stratégies de contrôle par rapport aux variations des paramètres du système, sauf pour les systèmes à deux niveaux (voir [13, 63]). Le contrôle adiabatique est réputé pour être une bonne technique pour gérer les perturbations et les incertitudes. Dans sa forme la plus simple, le théorème adiabatique affirme que dans une condition de séparation sur les niveaux d'énergie du Hamiltonien contrôlé,

#### 1.1. Préliminaires de physique quantique, théorème adiabatique

les probabilités d'occupation des niveaux d'énergie sont approximativement conservées lorsque les contrôles varient lentement. L'un de ses principaux avantages est qu'il fournit des lois de contrôle explicites et régulières. Il a donc été appliqué avec succès pour obtenir des stratégies de contrôle telles que les chirped pulses (voir par exemple [27, 74, 43]) pour les systèmes spin 1/2 à fréquence de Larmor dispersée. Une autre application classique du contrôle adiabatique à la contrôlabilité d'ensemble est ce qu'on appelle les impulsions contre-intuitives pour le processus STIRAP [38, 73, 83]. Une généralisation de cette approche a été proposée dans [62], et a soulevé de nombreuses questions intéressantes, en particulier sur la validité de l'approximation Rotating Wave dans ce cadre. Une théorie générale est alors nécessaire pour un contrôle robuste dans le cadre quantique. La principale contribution de la thèse est la compréhension de la robustesse des stratégies de contrôle adiabatique, en étudiant le problème du contrôle de familles paramétrées de systèmes quantiques avec une entrée de contrôle commune, c'est-à-dire le problème de la contrôlabilité d'ensemble. Nos principaux résultats à ce sujet sont présentés dans l'introduction de la section 2.2.1 avec deux contrôles réels et des Hamiltoniens réels, la plupart des arguments étant valables lorsque  $\mathcal{H}$  est de dimension finie ou infinie selon des hypothèses appropriées sur le Hamiltonien. Une extension de ceci avec une seule entrée de commande réelle utilisant l'approximation Rotating Wave (voir par exemple [53]) est exposée dans la section 2.2.3. Nous développerons ces questions dans les chapitres 3 et 6. Le point de départ de la thèse est le théorème adiabatique (voir par exemple [16, 55, 54, 47]), que nous énonçons dans sa forme générale, en adaptant les résultats exposés dans [78, Chapitre 2] à nos questions de théorie du contrôle.

**Théorème adiabatique général** Soit  $\mathcal{H}$  un espace de Hilbert séparable complexe et  $k \in \mathbb{N}$ . Designons l'ensemble des opérateurs linéaires essentiellement auto-adjoints sur  $\mathcal{H}$  par  $\mathcal{L}_{sa}(\mathcal{H})$ , et l'ensemble des fonctions  $C^j$  bornées entre deux espaces topologiques X et Y par  $C_b^j(X, Y)$ . Soit  $H(\cdot) \in C_b(\mathbb{R}^k, \mathcal{L}_{sa}(\mathcal{H}))$ . On dit que le spectre  $\sigma(\cdot)$  de  $H(\cdot)$  contient une partie séparée  $\sigma_*(\cdot) \subset \sigma(\cdot)$  localement autour de  $\bar{u} \in \mathbb{R}^k$  s'il existe deux fonctions continues et bornées  $f, g \in C_b(\mathbb{R}^k, \mathbb{R})$  et un voisinage V de  $\bar{u}$  dans  $\mathbb{R}^k$  satisfaisant, pour chaque  $u \in V$ ,  $\sigma_*(u) \subset [f(u), g(u)]$  et  $\inf_{u \in V} \text{dist}([f(u), g(u)], \sigma(u) \setminus \sigma_*(u)) > 0$ . Pour  $u \in V$ , définissons  $P_*(u)$  comme la projection spectrale de H(u) sur l'espace propre de H(u) associé à  $\sigma_*(u)$ . Soit  $\gamma: [0, 1] \to V$  un chemin de contrôle régulier  $C^2$ .

Nous sommes intéressés par les trajectoires  $\psi^{\epsilon}(t) \in \mathcal{H}$  de l'Equation de Schödinger Adiabatique :

$$i\dot{\psi}^{\epsilon}(t) = H(\gamma(\epsilon t))\psi^{\epsilon}(t), \quad \psi^{\epsilon}(0) = \psi_0,$$

où  $\psi_0 \in \mathcal{H}, t \in [0, \frac{1}{\epsilon}]$ , et  $\epsilon > 0$ . Dans la variable  $\tau = \epsilon t \in [0, 1]$ , la trajectoire reparamétrée  $\tilde{\psi}^{\epsilon}(\tau) = \psi^{\epsilon}(\tau \epsilon)$  satisfait

$$i\tilde{\psi}^{\epsilon}(\tau) = \frac{1}{\epsilon}H(\gamma(\tau))\tilde{\psi}^{\epsilon}(\tau), \quad \tilde{\psi}^{\epsilon}(0) = \psi_0, \qquad (1.2)$$

où  $\tau \in [0, 1]$ .

Définissons le Hamiltonien adiabatique, pour tout  $\tau \in [0, 1]$ , par

$$H_a^{\epsilon}(\tau) = H(\tau) - i\epsilon P_*(\gamma(\tau))\dot{P}_*(\gamma(\tau)) + i\epsilon(Id - P_*(\gamma(\tau)))\dot{P}_*(\gamma(\tau)),$$

où  $P_*(\gamma(\tau))$  est la dérivée de  $P_*$  le long du chemin de contrôle  $\gamma$ . Lorqu'ils sont bien définis, définissons les deux propagateurs unitaires  $U^{\epsilon}(\tau)$  et  $U_a^{\epsilon}(\tau)$  sur  $\mathcal{H}$  de, respectivement, l'Equation (1.2) et  $i\psi^{\epsilon}(\tau) = \frac{1}{\epsilon}H_a^{\epsilon}(\gamma(\tau))\psi^{\epsilon}(\tau)$ , pour  $\tau \in [0, 1]$ . Une propriété essentielle de  $U_a^{\epsilon}$  est que pour tout  $\tau \in [0, 1]$ ,  $P_*(\gamma(\tau))U_a^{\epsilon}(\tau) = U_a^{\epsilon}(\tau)P_*(\gamma(0))$ .

**Theorem 1.1.1.** [78, Theorems 2.1, 2.2] Soit V un ouvert de  $\mathbb{R}^k$  et  $\gamma$  :  $[0,1] \to V$ chemin de contrôle régulier  $C^2$ . Soit  $H(\cdot) \in C_b^2(V, \mathcal{L}_{sa}(\mathcal{H}))$  tel que les opérateurs H(u)ont un domaine dense commun  $D \subset \mathcal{H}$  pour tout  $u \in V$ , et sont minorés, uniformément par rapport à  $u \in V$ . Supposons que  $\sigma_*(u) \subset \sigma(H(u))$  est une partie localement séparée de  $\sigma$ , pour  $u \in V$ . Alors  $P_* \in C_b^2(V, \mathcal{L}_{sa}(\mathcal{H})))$ , les propagateurs unitaires  $U_{\epsilon}$  et  $U_a^{\epsilon}$  sont bien définis, et il existe une constante  $C < \infty$  telle que pour tout  $\tau \in [0, 1]$ ,

$$||U^{\epsilon}(\tau) - U^{\epsilon}_{a}(\tau)|| \le C\epsilon.$$

Adiabatic decoupling Soit  $j \in \mathbb{N}$ . Supposons que les valeurs propres  $\lambda_j$  et  $\lambda_{j+1}$  de Hsont séparées du reste du spectre pour u dans un voisinage V de  $\bar{u} \in \mathbb{R}^k$ . Remarquons qu'on ne suppose pas ici l'existence d'un gap entre  $\lambda_j$  et  $\lambda_{j+1}$ . En effet, le but de ce qui suit est d'avoir une approximation de la dynamique au voisinage d'un point  $\bar{u}$  tel que  $\lambda_j(\bar{u}) = \lambda_{j+1}(\bar{u})$ , c'est à dire une *intersection de valeurs propres*. Définissons pour  $u \in V$ , la projection spectrale  $P_{j,j+1}(u)$  de H(u) sur l'espace propre associé à  $\sigma_*(u) =$  $\{\lambda_j(u), \lambda_{j+1}(u)\}$ . Considérons, pour  $u \in V$ , une base orthonormale  $(\psi_j(u), \psi_{j+1}(u))$  de  $\operatorname{Im} P_{j,j+1}(u)$  qui est  $C^2$ . Pour tout  $u \in V$ , considérons une application unitaire I(u) de  $\mathbb{C}^2$ à  $\operatorname{Im} P_{j,j+1}(u)$ , qui est  $C^2$  en  $u \in V$ , telle que  $I(u)(e_1) = \psi_j(u)$  et  $I(u)(e_2) = \psi_{j+1}(u)$ , où  $(e_1, e_2)$  est la base canonique de  $\mathbb{C}^2$ . Le long d'un chemin régulier de contrôle  $C^2$ , noté  $(\gamma(\tau))_{\tau \in [0,1]}$  dans V, nous définissons le *Hamiltonien effectif* pour tout  $\tau \in [0, 1]$ , par

$$H_{\text{eff}}(\gamma(\tau)) = h(\gamma(\tau)) - i\epsilon \begin{pmatrix} 0 & \langle \dot{\psi}_j(\gamma(\tau)), \psi_{j+1}(\gamma(\tau)) \rangle \\ \langle \dot{\psi}_{j+1}(\gamma(\tau)), \psi_j(\gamma(\tau)) \rangle & 0 \end{pmatrix}, \quad (1.3)$$

où, pour tout  $u \in V$ ,  $h(u) = I^{-1}(u)H(u)I(u)$  est une matrice hermitienne de dimension 2,  $C^2$  en  $u \in V$ , et  $\dot{\psi}_q(\gamma(\tau))$ ,  $q \in \{j, j+1\}$  est la dérivée de  $\psi_q$  le long du chemin  $\gamma$ . En notant le propagateur associé à  $H_{\text{eff}}(\gamma(\tau))$  par  $U_{\text{eff}}^{\epsilon}(\tau)$ , le théorèm 1.1.1 prouve que :

$$\left|\left|\left(U^{\epsilon}(\tau) - I(\gamma(\tau))U^{\epsilon}_{\text{eff}}(\tau)I^{-1}(\gamma(0))\right)P_{j,j+1}(0)\right|\right| \le C\epsilon,\tag{1.4}$$

pour tout  $\tau \in [0,1]$ .

Une propriété importante obtenue en négligeant le second terme de  $H_{\text{eff}}$  par une procédure classique de moyennisation (voir le théorème 4.6.4 prouvé dans le chapitre 4 pour



FIGURE 1.1 – Deux valeurs propres de H(u) au voisinage d'une intersection conique, tracées en fonction de  $u \in \mathbb{R}^k$  avec k = 2.

plus de précisions), est la suivante, qui est vérifiée sous des hypothèses convenables sur les valeurs propres de H, qui sont celles de h, et des vecteurs propres de H, qui sont ceux de h, modulo la transformation I unitaire. Pour un chemin de contrôle régulier  $\gamma : [0,1] \to V$ et  $\tilde{\psi}_0 \in \mathbb{C}^2$ , les solutions  $\psi$  et  $\tilde{\psi}$  de, respectivement,  $i\frac{d\psi}{dt} = H(\gamma(\epsilon t))\psi, \psi(0) = I(\gamma(0))\tilde{\psi}_0$ , et  $i\frac{d\tilde{\psi}}{dt} = h(\gamma(\epsilon t))\tilde{\psi}, \tilde{\psi}(0) = \tilde{\psi}_0$  sont telles que  $\psi(1/\epsilon)$  est  $O(\epsilon^{\alpha})$  près de  $I(\gamma(1))\tilde{\psi}(1/\epsilon)$ . La valeur de  $\alpha \in \mathbb{R}$  est positive et dépend de l'ordre du gap entre les valeurs propres  $\lambda_j$  et  $\lambda_{j+1}$  de H le long de  $\gamma$ , c'est à dire de la quantité  $\lambda_{j+1} - \lambda_j$  et de ses dérivées (voir le théorème 4.6.4).

Cette approximation est intéressante car elle préserve la structure adiabatique des Hamiltoniens, ceci étant important pour obtenir des estimations uniformes de la dynamique le long de tout chemin de contrôle suffisamment régulier dans V. En outre, la connaissance de h nous donne toute l'information requise à propos de la régularité locale des valeurs propres et vecteurs propres de H. Remarquons que h, appelé *Hamiltonien réduit* dépend du choix de la base de  $\pi_{H(u)} = \text{Im}P_{j,j+1}(u)$ , pour  $u \in V$ . Cependant, nous verrons en Section 4.6.3 que les propriétés géométriques et par conséquent les propriétés dynamiques de l'équation de Schrödinger qui nous intéressent sont invariantes par un tel changement de base. Nous avons prouvé en Section 4.6.2 que lorsque  $\mathcal{H}$  est de dimension finie, des propriétés génériques sur h et ses jets jusqu'à l'ordre 2 fournissent des propriétés génériques sur le Hamiltonien H et ses jets jusqu'à l'ordre 2.

**Intersections coniques** Une *intersection conique* (aussi appelée point diabolo) est une singularité de type cône du spectre de H(u), vu comme une fonction du contrôle  $u \in \mathbb{R}^k$ . (voir la figure 1.1)

Les intersections coniques sont génériques au sens où elles sont les singularités les moins dégénérées du spectre d'un Hamiltonien. Elles ont été étudiées depuis les débuts de la théorie de la mécanique quantique [82]. Elle jouent un rôle important dans le contexte de l'analyse semi-classique [32, 33]. Les chemins adiabatiques passant à travers les intersections coniques peuvent être utilisés pour induire des superpositions d'états propres, comme montré dans [22, 46], ainsi que pour obtenir des tests pour la contrôlabilité exacte lorsque  $\mathcal{H}$  est de dimension finie et de contrôlabilité approchée lorsque  $\mathcal{H}$  est de dimension finie et de systèmes, elles sont génériques et structurellement stables, au sens où elles ne peuvent pas être effacées par de petites perturbations du Hamiltonien.

Soit U un ouvert connexe de  $\mathbb{R}^2$ . Par un léger abus de notations, pour  $u \in \mathbb{R}^2$ , désignons les composantes de u par (u, v). Supposons que  $H(u, v) = H_0 + uH_1 + vH_2$  où  $H_0, H_1, H_2$ sont des opérateurs essentiellement auto-adjoints sur  $\mathcal{H}$  ayant un domaine dense commun  $\mathcal{D}$  telles que les conditions suivantes, que nous appellons ( $\mathcal{R}$ ) sont satisfaites :

- $-H_0$  a un spectre discret;
- $-H_1, H_2$  sont bornés;
- Il existe une base orthonormal  $(b_j)_j$  de l'espace de Hilbert  $\mathcal{H}$  tel que  $\langle b_j, H_0 b_q \rangle$ ,  $\langle b_j, H_1 b_q \rangle$ ,  $\langle b_j, H_2 b_q \rangle$  sont réels pour tout j, q.

Sous ces hypothèses, on dit que  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  est une intersection conique entre les valeurs propres  $\lambda_j$  et  $\lambda_{j+1}$  si  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$  a une multiplicité égale à 2 et il existe C > 0 tel que pour tout vecteur unitaire  $\eta \in \mathbb{R}^2$  et t > 0 assez petit,  $\lambda_{j+1}((\bar{u}, \bar{v}) + t\eta) - \lambda_j((\bar{u}, \bar{v}) + t\eta) > Ct$ .

Soit  $(\gamma(t))_{t\in[0,1]} = (u(t), v(t))_{t\in[0,1]}$  un chemin régulier de contrôle  $C^3$  dans  $\mathbb{R}^2$ . Considérons l'équation

$$i\frac{d\psi(t)}{dt} = H(\gamma(\epsilon t))\psi(t), \qquad (1.5)$$

où  $\psi(t) \in \mathcal{H}$ . Supposons que H a une intersection conique en (0,0). Supposons  $\gamma(t_0) = (0,0)$  pour un  $t_0 \in (0,1), \ \gamma(t) \neq 0$  pour tout  $t \neq t_0$ . Considérons une solution  $\psi_{\epsilon}$  de l'équation (1.5) telle que  $\psi_{\epsilon}(0) = \phi_j(\gamma(0))$ . Alors

$$\|\psi_{\epsilon}(\frac{1}{\epsilon}) - e^{i\theta}\phi_{j+1}(\gamma(1))\| = O(\epsilon^{1/2}), \qquad (1.6)$$

où  $\theta \in \mathbb{R}$  dépend possiblement de  $\epsilon$ . Nous pouvons alors implémenter une stratégie pour effectuer des transitions d'états passant au travers des intersections coniques, comme exposé dans [22] (see Figure 1.2).

Nous nous intéressons à la notion suivante de contrôlabilité On dit que l'équation (1.5) est contrôlable de façon approchée si pour tout  $\epsilon > 0$  et  $\psi_1 \in \mathcal{H}$ , il existe un contrôle  $\gamma(\cdot) = (u(\cdot), v(\cdot)) : [0, T] \rightarrow U$  tel que la solution de l'équation (1.5) avec condition initiale  $\psi(0) = \psi_0$  satisfait  $||\psi(T) - \psi_1|| < \epsilon$ . On dit que le spectre  $\sigma(\cdot)$  de  $H(\cdot)$  est coniquement connexe si toutes les intersections de valeurs propres sont coniques et si pour

#### 1.2. Thèmes d'études



FIGURE 1.2 – Chemin de transition à la vitesse  $\epsilon$  when k = 2.

tout j, il existe une intersection conique  $(\bar{u}_j, \bar{v}_j)$  entre les valeurs propres  $\lambda_j, \lambda_{j+1}$ , avec  $\lambda_l(\bar{u}_j, \bar{v}_j)$  simple si  $l \neq j, j+1$ . Nous avons les résultats généraux suivants.

**Theorem 1.1.2** ([19]). Supposons que le spectre de  $H(\cdot)$  est coniquement connexe. Alors l'équation (1.5) est contrôlable de façon approchée.

**Theorem 1.1.3** ([19]). Supposons que  $\mathcal{H}$  est de dimension finie. Supposons de plus que le spectre de  $H(\cdot)$  est coniquement connexe. Alors l'équation (1.5) est contrôlable de façon exacte.

**Proposition 1.1.4** (Genericité). Soit  $H_0$  une matrice symétrique réelle de dimension n. Alors, génériquement par rapport à  $(H_1, H_2)$  dans l'espace des matrices symétriques réelles de dimension n, les intersections de valeurs propres de  $H_0 + uH_1 + vH_2$  sont coniques.

Certains résultats de généricité lorsque  $\mathcal{H}$  est de dimension infinie ont été prouvés dans [31, 22], dans le cas de potentiels réels contrôlés.

### 1.2 Thèmes d'études

Notre but est de continuer à explorer les liens entre les singularités du spectre des Hamiltoniens, la géométrie des espaces propres et les propriétés de contrôlabilité du système quantique associé. En particulier, comme annoncé précédemment, nous étudions le problème du contrôle des familles paramétrées de systèmes quantiques avec une entrée de contrôle commune. C'est une tâche importante pour de nombreuses applications en contrôle quantique (voir [39] et ses références), notamment en résonance magnétique nucléaire [40]. Ce problème de contrôlabilité d'ensemble a été formulé mathématiquement par Li et Khaneja dans [63] et par Beauchard, Coron et Rouchon dans [13]. Les résultats pour le contrôle d'ensemble au-delà du cadre de contrôle quantique peuvent être trouvés dans [4, 50, 64, 72].

#### 1.2.1 Contrôle d'ensemble avec deux contrôles réels

Pour un système quantique fermé général soumis à l'action d'un contrôle  $u \in \mathbb{R}^m$  et dépendant d'un paramètre z, l'équation contrôlée associée est de la forme

$$i\frac{d\psi}{dt}(t) = H(u(t), z)\psi(t), \qquad \psi(t) \in \mathcal{H},$$
(1.7)

où H(u, z) est essentiellement auto-adjoint sur l'espace de Hilbert  $\mathcal{H}$  pour toute valeur de u et z. Le paramètre z peut être utilisé soit pour décrire une famille de systèmes physiques sur lesquels agit un champ commun u ou bien un système physique pour lequel la valeur d'un des paramètres n'est pas connue précisément.

Les propriétés de contrôlabilité de systèmes de cette forme ont été étudiées à la fois pour des ensembles de paramètres discrets et continus. Le cas d'un ensemble fini de systèmes est caractérisé dans [14, 35]. Dans [29] la stabilisation asymptotique d'ensemble est étudiée pour des ensembles de paramètres dénombrables. Dans [63], [13] une preuve d'une forte notion de contrôlabilité d'ensemble a été obtenue pour un système à deux niveaux. Le contrôle d'ensemble numérique dans le cas d'un continuum de paramètres a fait l'objet d'études approfondies pour les systèmes à deux niveaux [79, 28, 75]. Notre but est de généraliser les théorèmes 2.1.2 et 2.1.3 à la question du contrôle d'ensemble.

Nous limitons notre étude au problème de la contrôlabilité d'ensemble approchée entre états propres. Sauf lorsque explicitement mentionné, nous étudierons les systèmes quantiques réels de dimension finie, c'est-à-dire les systèmes dont le Hamiltonien appartient à l'ensemble  $S_n(\mathbb{R})$  des matrices réelles symétriques de dimension n. Cependant, les résultats principaux de contrôlabilité restent valides lorsque  $H(u, v) = H_0 + uH_1 + vH_2$  où  $H_0, H_1, H_2$  sont essentiellements auto-adjoints sur  $\mathcal{H}$  avec un domaine dense commun, et satisfont la condition  $(\mathcal{R}), (u, v) \in U$ , où U est un ouvert connexe de  $\mathbb{R}^2$ , et une adaptation de ces conditions au cas parametrique (voir le chapitre 2.2.1).

#### Notations

— Soit  $H \in C^k(U, S_n(\mathbb{R}))$ . Pour  $u \in U$ , designons le spectre de H(u) by  $(\lambda_j(u))_{j=1}^n$ , où  $j \mapsto \lambda_j(u)$  est la suite des valeurs propres de H(u) dans l'ordre croissant, répétées selon leur multiplicités. Définissons  $(\phi_1(u), \ldots, \phi_n(u))$  une base orthonormale des vecteurs propres associés.

— Soit  $H \in C^k(U \times [z_0, z_1]), S_n(\mathbb{R}))$ . Pour  $u \in U$ , et  $z \in [z_0, z_1]$ , désignons le spectre de H(u, z) par  $(\lambda_j(u, z))_{j=1}^n$ , où  $j \mapsto \lambda_j(u, z)$  est la suite des valeurs propres de H(u, z) dans l'ordre croissant, répétées selon leur multiplicités. Définissons  $(\phi_1^z(u), \ldots, \phi_n^z(u))$  une base orthonormale des vecteurs propres associés.

**Definition 1.2.1.** Le système (1.7) est dit ensemble contrôlable de façon approchée entre les vecteurs propres si pour tout  $\epsilon > 0$ ,  $j, k \in \{1, ..., n\}$ , et  $u_0, u_1 \in U$  tels que  $\lambda_j(u_0, z)$ et  $\lambda_k(u_0, z)$  sont simples pour tout  $z \in [z_0, z_1]$ , il existe un contrôle  $u(\cdot) : [0, T] \to U$  tel que pour tout  $z \in [z_0, z_1]$  la solution de (1.7) de condition initiale  $\psi^z(0) = \phi_j^z(u_0)$  satisfait  $\|\psi^z(T) - e^{i\theta}\phi_k^z(u_1)\| < \epsilon$ , avec  $\theta \in \mathbb{R}$  (possiblement dépendant de z et  $\epsilon$ ).

En factorisant la dynamique par la trace, le Hamiltonien h défini en Section 1.1 peut être réduit à un Hamiltonien de trace nulle (à une phase globale près dans la dynamique). Dans le cas k = 2, on peut supposer sans perte de généralité que pour tout  $(u, v) \in \mathbb{R}^2$ dans un voisinage de (0,0),  $h(u,v) = H_f(u,v)$ , où  $H_f$  est défini par

$$H_f(u,v) = \begin{pmatrix} f_1(u,v) & f_2(u,v) \\ f_2(u,v) & -f_1(u,v) \end{pmatrix},$$

avec  $f = (f_1, f_1) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Dans le cas paramétrique, pour tout  $(u, v, z) \in \mathbb{R}^3$  au voisinage de (0, 0, 0), on peut supposer  $h(u, v, z) = H_f(u, v, z)$ , où  $H_f$  est défini par

$$H_f(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix},$$

avec  $f = (f_1, f_1) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Des intersections de valeurs propres transverses dans toutes les directions de  $\mathbb{R}^2$  et tangentes dans une direction spécifique, appelée la *direction non conique*, apparaissent génériquement. Celles-ci sont appelées *intersections semiconiques* (see Figure 2.3), et sont définies rigoureusement en Section 4.2.1. Nous allons étudier des formes normales pour ce cas générique.

#### Formes normales

Transformations admissibles de systèmes à deux niveaux Nous définissons les transformations admissibles qui nous permettent d'obtenir des formes normales pour les Hamiltoniens  $H_f$ . Considérons l'équation de Schrödinger, définie pour  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ par

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t))\psi(t), \qquad (1.8)$$

où  $\psi(t) \in \mathbb{C}^2$ , et l'équation de Schrödinger d'ensemble, définie pour  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  par

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t), z)\psi(t), \qquad (1.9)$$

où  $\psi(t) \in \mathbb{C}^2$ . Les fonctions de contrôle u, v sont dans  $L^{\infty}(\mathbb{R}, \mathbb{R})$  et le paramètre z appartient à  $[z_0, z_1]$  où  $z_0, z_1 \in \mathbb{R}$ .

Les trois transformations correspondent aux représentations équivalentes des systèmes dynamiques (1.8) et (1.9) respectivement, par la reparamétrisation temporelle, par difféomorphisme de l'espace des états, et des transformations difféomorphes indépendantes de l'espace des contrôles et de l'espace du paramètre z.

**Definition 1.2.2.** On dit que f et  $\tilde{f}$  dans  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  sont equivalentes en 0 s'il existe  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  où  $\phi$  est un difféomorphisme satisfaisant  $\phi(0) = 0$ , et  $\zeta = \pm 1$ , tel que pout tout (u, v) dans un voisinage de 0,

$$\begin{cases} \tilde{f}_1(u,v) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v) - \zeta\sin(2\theta)f_2 \circ \phi(u,v)), \\ \tilde{f}_2(u,v) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v) + \zeta\cos(2\theta)f_2 \circ \phi(u,v)). \end{cases}$$

**Definition 1.2.3.** On dit que f et  $\tilde{f}$  dans  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  sont equivalentes en 0 s'il existe  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^3) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  où  $\phi$  est un diffeomorphisme de la forme  $\phi : (u, v, z) \mapsto (\phi_1(u, v), \phi_2(u, v), \phi_3(z)),$  où  $\phi_1, \phi_2 \in C^{\infty}(\mathbb{R}^2, \mathbb{R})$  et  $\phi_3 \in C^{\infty}(\mathbb{R}, \mathbb{R}),$  satisfont  $\phi(0) = 0$ , et  $\zeta = \pm 1$ , tel que pour tout (u, v, z) dans un voisinage de 0,

$$\begin{cases} \tilde{f}_1(u,v,z) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v,z) - \zeta\sin(2\theta)f_2 \circ \phi(u,v,z)), \\ \tilde{f}_2(u,v,z) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v,z) + \zeta\cos(2\theta)f_2 \circ \phi(u,v,z)). \end{cases}$$

Ceci permet de classifier les comportements génériques des systèmes quantiques par rapport à (u, v).

**Cas non-paramétrique** Considérons  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Alors  $H_f$  a une intersection conique en (0,0) si et seulement si f est équivalent à  $(u,v) \mapsto \begin{pmatrix} u & v \\ v & -u \end{pmatrix}$ . Un Hamiltonien générique n'admet que des singularités coniques.

On dit que  $H_f$  a une intersection semi-conique (voir Figure 2.3) en (0,0) si et seulement si f est équivalent à  $(u,v) \mapsto \begin{pmatrix} h(u)u \\ u+v^2 \end{pmatrix}$  où  $h : \mathbb{R} \to \mathbb{R}$  est une fonction lisse satisfaisant h(0) = 1. Les intersections semi-coniques ne sont pas génériques pour  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , cependant, nous allons voir qu'elles peuvent apparaître en des points isolés lorsque l'on considère le cas paramétrique. Le Hamiltonien correspondant  $\begin{pmatrix} h(u)u & u+v^2 \\ u+v^2 & -h(u)u \end{pmatrix}$  est tel que ses valeurs propres sont tangentes le long d'un chemin de contrôle tangent à  $e_2 = (0, 1)$ , c'est à dire à la direction non-conique en (0, 0), alors qu'elles sont transverses dans toutes les autres directions.

Nous classifions maintenant les comportements génériques des systèmes quantiques par rapport à (u, v, z).

#### 1.2. Thèmes d'études



FIGURE 1.3 – Les deux valeurs propres de  $H_f(u, v)$  au voisinage d'une intersection semiconique, tracées en fonction of (u, v).

**Cas paramétrique** Nous définissons les intersections F-coniques et F-semi-coniques, qui sont les deux cas generiques pour les Hamiltoniens parametriques. Ils correspondent, pour  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , au cas où  $H_f(\cdot, 0)$  a une intersection conique ou semi-conique, combiné à des hypothèses additionelles de régularité en le paramètre z.

- Soit  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .
- On dit que 0 est *F-conical* pour f si et seulement si il existe  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$ satisfaisant  $h_1(0) = h_2(0) = 1$ , tels que f est équivalente à

$$(u,v,z) \mapsto \begin{pmatrix} h_1(u,v,z)(z-u) \\ h_2(u,v,z)(z-v) \end{pmatrix};$$

— On dit que 0 est *F-semi-conique* pour  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  si et seulement si il existe  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfaisant  $h_1(0) = h_2(0) = 1$  et  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfaisant  $m(0) \notin \{-1, 0\}$  tel que f est équivalent à

$$(u,v,z) \mapsto \begin{pmatrix} h_1(u,v,z)(z-m(u)u) \\ h_2(u,v,z)(z+u+v^2) \end{pmatrix}$$

Systèmes à *n*-niveaux Soit  $H \in C^{\infty}(\mathbb{R}^2, S_n(\mathbb{R}))$ . On prouve que H a une intersection conique en (0,0) entre les niveaux  $\lambda_j$  et  $\lambda_{j+1}$  si et seulement si il existe un Hamiltonien réduit h de H localement en (0,0) ayant une intersection conique en (0,0). Par conséquent, il est cohérent de donner la définition suivante d'intersection semi-conique pour H.

**Definition 1.2.4.** On dit que H a une intersection semi-conique en (0,0) entre les niveaux  $\lambda_j$  et  $\lambda_{j+1}$  s'il existe un Hamiltonien réduit h de H ayant une intersection semi-conique en (0,0).



FIGURE 1.4 – Intersection semi-conique pour un processus STIRAP dégénéré

Nous prouvons en Section 4.6.3 que ces propriétés ne dépendent pas de l'application unitaire I(u, v), pour (u, v) dans un voisinage de (0, 0), utilisé pour le découplage adiabatique comme en Section 2.1 de l'introduction.

Dans le cas paramétrique, on dit que H a une intersection F-conique (respectivement, F-semi-conique) en (0,0,0) s'il existe un Hamiltonien réduit h of H ayant une intersection F-conique (respectivement, F-semi-conique) en (0,0,0). En Section 4.6.3, on montre que, generiquement par rapport à  $H \in C^{\infty}(\mathbb{R}^3, S_n(\mathbb{R}))$ , une intersection de valeur propre  $(u, v, z) \in \mathbb{R}^3$  de H est soit F-conique soit F-semi-conique.

Les intersections semi-coniques apparaissent notamment lorsque, pour  $(u, v) \in \mathbb{R}^2$ ,

 $H(u,v) = \begin{pmatrix} E & u & 0 \\ u & E & v \\ 0 & v & E' \end{pmatrix}, \text{ avec } E < E', \text{ cas que nous appelons } STIRAP \ dégénéré. \text{ Sur la}$ 

Figure 2.4, nous avons tracé le spectre de H(u, v) en fonction de (u, v) pour un Hamiltonien de ce type. Nous remarquons la présence d'une intersection semi-conique entre les deux niveaux les plus bas et deux intersections coniques entre les deux niveaux les plus hauts.

Propriétés dynamiques aux intersections semi-coniques Grâce aux formes normales, nous pouvons estimer l'approximation adiabatique pour des systèmes à *n*niveaux. Soit  $H \in C^{\infty}(\mathbb{R}^2, S_n(\mathbb{R}))$ , ayant une intersection semi-coniqueen (0, 0). Désignons par  $\eta$  la direction non-conique.. Soit  $(\gamma(t))_{t \in [0,1]} = (u(t), v(t))_{t \in [0,1]}$  un chemin de contrôle régulier lisse de  $\mathbb{R}^2$  tel que  $\gamma(t_0) = (0, 0)$  pour  $t_0 \in (0, 1), \gamma(t) \neq 0$  pour tout  $t \neq t_0$ , et  $\gamma$ est tangent à  $\eta$  en  $t = t_0$ . Considérons une solution  $\psi_{\epsilon}$  de l'équation

$$i\frac{d\psi(t)}{dt} = H(u(\epsilon t), v(\epsilon t))\psi(t), \qquad (1.10)$$

#### 1.2. Thèmes d'études



FIGURE 1.5 – Chemin de contrôle passant à travers la singularité dans la direction nonconique.

où  $\psi(t) \in \mathbb{C}^n$ , tel que  $\psi_{\epsilon}(0) = \phi_j(\gamma(0))$ . Alors  $\|\psi_{\epsilon}(\frac{1}{\epsilon}) - e^{i\theta}\phi_j(\gamma(1))\| = O(\epsilon^{1/3})$ , où  $\theta \in \mathbb{R}$  est possiblement dependant de  $\epsilon$  (voir Figure 1.5 pour une représentation graphique du chemin de contrôle passant à travers la singularité dans la direction non-conique).

#### Lorsque la conicité est préservée quand le paramètre z varie

Dans le chapitre 3 nous proposons un cadre pour le contrôle adiabatique d'ensemble d'un continuum de systèmes à *n*-niveaux avec un Hamiltonien reél, pilotés par deux contrôles et ayant des intersections coniques entre les valeurs propres. L'idée principale est que, si un système correspondant à un paramètre fixe a des intersections coniques entre deux valeurs propres, alors une petite perturbation du paramètre donne une courbe d'intersections coniques, chaque point de la courbe correspondant exactement à une valeur du paramètre. On peut alors suivre adiabatiquement de telles courbes dans l'espace de contrôle et obtenir un transfert de population entre les deux niveaux pour l'ensemble des systèmes. Ce résultat a été écrit sans recourir aux formes normales en raison de l'estimation déjà bien connue de la dynamique adiabatique aux intersections coniques dans la littérature. Cependant, le même résultat peut aussi être déduit de l'étude locale des formes normales.

Soit U un ouvert connexe de  $\mathbb{R}^2$ . Pour  $j \in \{1 \dots n\}$ , désignons par  $\gamma_j$  l'ensemble  $\{(u, v, z) \in U \times [z_0, z_1] \mid \lambda_j(u, v, z) = \lambda_{j+1}(u, v, z)\}$ , où par convention  $\gamma_0 = \gamma_n = \emptyset$ .

Chapitre 1. Introduction -fr –



FIGURE 1.6 – Un contrôle réalisant une transition d'ensemble entre  $\phi_1^z(u_0, v_0)$  et  $\phi_4^z(u_0, v_0)$ .

Notons la projection de  $(u, v, z) \in \mathbb{R}^3$  sur la composante (u, v) par  $\pi$ .

ASSUMPTION  $A_j$ . Il existe une composante connexe  $\hat{\gamma}_j$  de  $\gamma_j$  et une application  $\beta_j : [z_0, z_1] \to U$  telle que  $\beta_j$  est un plongement  $C^3$  et

-  $\pi(\hat{\gamma}_j)$  est inclus dans U ×  $[z_0, z_1] \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1}));$ 

$$- \pi(\hat{\gamma}_j) = \beta_j([z_0, z_1]);$$

— Pour tout  $z \in [z_0, z_1]$ ,  $\lambda_j(\cdot, z)$  et  $\lambda_{j+1}(\cdot, z)$  ont une unique intersection dans  $\pi(\hat{\gamma}_j)$ , qui est conique et a lieu en  $\beta_j(z)$ .

En outre, l'ensemble  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  est connexe par arcs.

Remarquons que les secondes et troisièmes conditions sont équivalentes à dire que les courbes  $\hat{\gamma}_j$  formées que d'intersections F-coniques et que  $\pi(\hat{\gamma}_j)$  est une courbe plongée dans  $\mathbb{R}^2$  sans auto-intersection. Le résultat qui suit est le résultat principal de l'article [10] et du chapitre 3.

**Theorem 1.2.5.** Considérons une application  $C^3$ ,  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$ . Supposons que Assumption  $A_j$  est satisfaite pour tout  $j \in \{1 \dots n\}$ . Alors l'équation  $i\frac{d\psi}{dt} = H(u, v, z)\psi$  est ensemble contrôlable entre états propres de façon approchée.

Méthode de contrôle Notre méthode de contrôle est présentée sur la figure 1.6, avec un chemin  $(u(t), v(t))_{t \in [0,1]}$ , parcouru à la vitesse  $\epsilon$ . L'erreur dans le régime adiabatique est d'ordre  $\sqrt{\epsilon}$ .

#### Lorsque qu'une intersection semi-conique apparait quand le paramètre z varie

L'argument esquissé ci-dessus repose sur l'hypothèse que pour toutes les valeurs du paramètre, les intersections de valeurs propres restent coniques et décrivent une courbe lisse. Ces hypothèses sont satisfaites pour de petites perturbations génériques du paramètre. Pour de grandes perturbations génériques, il peut arriver que la conicité des intersections de valeurs propres soit perdue en des points isolés de la courbe. L'objectif est d'étendre l'analyse à ce cas.

Dans le cas où des intersections F-semi-coniques sont présentes sur  $\gamma_j$ , la question que l'on pose est celle de la validité de l'approximation adiabatique lorsque le chemin de contrôle passe par de tels points. En utilisant les formes normales, nous remarquons que  $\pi(\hat{\gamma}_j)$  est tangent à la direction non-conique en une intersection F-semi-conique. Par conséquent, lorsqu'un chemin de contrôle suit  $\pi(\hat{\gamma}_j)$ , il passe dans l'intersection semiconique pour  $H(\cdot, z)$  dans la direction non-conique, et il n'y a alors pas de transition d'état pour le système de paramètre z. Cependant, nous montrons que sous l'hypothèse appelée Assumption  $\tilde{A}_j$  qui suit, l'approximation adiabatique est uniforme par rapport au paramètre  $z \in [z_0, z_1]$  et le système est alors ensemble contrôlable entre les états propres de façon approchée. Dans ce cas, nous ne pouvons considérer un plongement de  $[z_0, z_1]$ dans  $\pi(\hat{\gamma}_j)$  puisque la même valeur du paramètre z correspond à différentes intersections de valeurs propres de  $H(\cdot, z)$ . C'est pourquoi nous faisons les hypothèses suivantes.

ASSUMPTION  $\hat{A}_j$ . Il existe une composante connexe  $\hat{\gamma}_j$  de  $\gamma_j$  telle que

- $\hat{\gamma}_j$  est une sous-variété de dimension 1 de  $\mathbb{R}^3$  composée uniquement d'intersections F-coniques et F-semi-coniques.
- Il existe  $(u_0, v_0) \in U$  et  $(u_1, v_1) \in U$  tel que  $(u_0, v_0, z_0), (u_1, v_1, z_1) \in \hat{\gamma}_j$  sont des intersections F-coniques pour H;
- $\pi(\hat{\gamma}_j)$  est une courbe  $C^{\infty}$  plongée dans  $\mathbb{R}^2$  sans auto-intersection, contenue dans  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1})).$

En outre, l'ensemble  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  est connexe par arcs.

**Theorem 1.2.6.** Considérons une application  $C^4$ ,  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$ . Supposons que Assumption  $\tilde{A}_j$  est satisfaite pour tout  $j \in \{1 \dots n\}$ . Alors l'équation  $i\frac{d\psi}{dt} = H(u, v, z)\psi$  est ensemble contrôlable entre les états propres de façon approchée.

Méthode de contrôle Sur la figure 1.7, nous proposons un chemin de contrôle lisse (u, v) (voir Theorem 4.1.3 pour plus de précisions) réalisant des transitions entre les niveaux j et j + 1 uniformément par rapport au paramètre  $z \in [z_0, z_1]$ . Le chemin de contrôle (u, v) entre dans  $\pi(\gamma)$  en un point  $(u_0, v_0)$  tel que  $(u_0, v_0, z_0)$  est une intersection de valeurs propres F-conique de  $H(u, v, z_0)$  entre les niveaux j et j + 1, et quittant  $\pi(\gamma)$  en un point  $(u_1, v_1)$  tel que  $(u_1, v_1, z_1)$  est une intersection F-conique de valeurs propres de  $H(u, v, z_1)$  entre les niveaux j et j+1. De façon à garantir que les transitions sont effectuées

pour tout  $z \in [z_0, z_1]$ , le chemin de contrôle doit passer un nombre impair de fois à travers des intersections F-coniques de H(u, v, z) entre les niveaux j et j+1, pour tout  $z \in [z_0, z_1]$ . En outre, l'uniformité de l'approximation adiabatique par rapport au paramètre z est assurée lorsque le chemin de contrôle est choisi de façon à avoir  $z \in [z_0, z_1]$  pour tous z et t tels que  $(u(t), v(t), z) \in \gamma$ . L'erreur commise dans l'approximation adiabatique pour un tel chemin de contrôle parcouru à la vitesse  $\epsilon$  est d'ordre  $\epsilon^{\frac{1}{3}}$ .



FIGURE 1.7 – Une courbe de contrôles (u, v) réalisant des transitions uniformes en  $z \in [z_0, z_1]$  entre deux niveaux comportant des intersections F-semi-coniques

#### 1.2.2 Classification des singularités du champ non-mixant

Pour un Hamiltonien H dépendant de deux contrôles, les courbes non-mixantes entre  $\lambda_{j-1}$  et  $\lambda_j$  pour  $j \geq 2$  ont été définies dans [22] comme les courbes  $\gamma = (\gamma(t))_{t \in [0,1]}$  de  $\mathbb{R}^2$  le long desquelles  $\dot{\phi}_{j-1}(\gamma(t))$  est orthogonal à  $\phi_j(\gamma(t))$ , pour tout  $t \in [0,1]$ . Alors, par le théorème 1.1.1, la précision de l'approximation adiabatique suivant une telle courbe est améliorée. En particulier, l'erreur d'ordre  $\sqrt{\epsilon}$  obtenue pour la dynamique de l'équation (??) pour un chemin de contrôle passant à travers une intersection conique devient d'ordre  $\epsilon$  en suivant une courbe non-mixante. Cette propriété a été utilisée dans [22] pour un contrôle précis de l'équation de Schödinger pour un Hamiltonien réel soumis à deux contrôles réels et une généralisation a été publiée dans [31] pour un Hamiltonien complexe soumis à trois contrôles réels.



FIGURE 1.8 – Un noeud critique

Dans cette section, nous proposons d'étudier les singularités des courbes non-mixantes d'abord pour des systèmes à deux niveaux, puis pour des systèmes quantiques plus généraux.

#### Le champ non-mixant pour les systèmes à deux niveaux et ses singularités dans ce cas

Pour f dans  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , étudions la dynamique de

$$i\frac{d\psi_{\epsilon}(t)}{dt} = H_f(u(\epsilon t), v(\epsilon t))\psi_{\epsilon}(t), \quad \psi_{\epsilon}(0) = \tilde{\psi}_0 \in \mathbb{C}^2,$$
(1.11)

où  $t \in [0, \frac{1}{\epsilon}]$  et  $\tilde{\psi}_0$  est indépendant de  $\epsilon$ .

Le long de  $\gamma$ , l'erreur commise dans le théorème adiabatique est nulle, c'est à dire que les courbes non-mixantes peuvent être suivies à une vitesse arbitraire, la dynamique suit de façon exacte l'évolution des vecteurs propres (à phase près) le long de  $\gamma$ , quelle que soit la vitesse de parcours. Nous montrons dans un premier temps que de telles courbes sont courbes intégrales d'un champ de vecteur lisse  $\chi(f)$ , qu'on appelle *champ non-mixant*.

**Theorem 1.2.7.** Génériquement par rapport à  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ ,  $\chi(f)$  a des singularités de trois types, à un changement de coordonnées  $C^{\infty}$ -difféomorphe de  $\mathbb{R}^2$  dans  $\mathbb{R}^2$  près :

- Des noeuds critiques (voir Figure 1.8) aux points d'intersections de valeurs propres de H<sub>f</sub> (c'est à dire des zéros de f),
- Des selles et des centres aux points qui ne sont pas des intersections de valeurs propres de  $H_f$ .

Une famille générique à un paramètre de Hamiltoniens peut être mise, au voisinage d'une intersection de valeurs propres, sous une des formes normales suivantes par une



FIGURE 1.9 – Un exemple de sin-FIGURE 1.10 – Un exemple de gularité semi-conique hyperbolique. singularité semi-conique elliptique. L'indice est 0. La direction non-L'indice est 2. La direction non-conique est (0, 1). Deux droites (en conique est (0, 1). Toute courbe rouge) traversent la singularité dans non-mixante passe à la singularité des directions coniques dans la direction non-conique.

transformation admissible :

$$H(u, v, z) = \begin{pmatrix} h_1(u, v, z)(z - u) & h_2(u, v, z)(z - v) \\ h_2(u, v, z)(z - v) & -h_1(u, v, z)(z - u) \end{pmatrix},$$

ou

$$H(u, v, z) = \begin{pmatrix} h_1(u, v, z)(z - m(u)u) & h_2(u, v, z)(z + u + v^2) \\ h_2(u, v, z)(z + u + v^2) & -h_1(u, v, z)(z - m(u)u) \end{pmatrix},$$

où  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  satisfont  $h_1(0) = h_2(0) = 1$ , et  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfait  $m(0) \notin \{-1, 0\}$ .

Cette propriété nous permet d'étudier les bifurcations des singularités du champ nonmixant données dans le théorème 1.2.7 lorsqu'un paramètre z varie. Pour tout  $z \in \mathbb{R}$ , soit  $\chi_z(f)$  le champ non-mixant associé à  $f(\cdot, \cdot, z) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ .

**Theorem 1.2.8.** Génériquement par rapport à  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , pour tout  $z \in \mathbb{R}$ ,  $\chi_z(f)$  possède les singularités suivantes à un changement de coordonnées  $C^{\infty}$ -diffeomorphe de  $\mathbb{R}^2$  dans  $\mathbb{R}^2$  près :

- Des noeuds critiques aux points d'intersections de valeurs propres de  $H_{f}$ ,
- Des singularités semi-coniques elliptiques ou hyperboliques aux intersections semiconiques de valeurs propres de H<sub>f</sub> (voir le chapitre 5 Section 5.2.2 pour une définition précise, et les figures 5.3 et 5.2),
- Des selles, des centres ou des cusps aux points qui ne sont pas des intersections de valeurs propres de  $H_f$ .
#### Le champ non-mixant pour des systèmes plus généraux

Pour des systèmes de plus grande dimension, les courbes non-mixantes ont été étudiées dans [22] Dans ce qui suit, nous considérons  $H_0, H_1, H_2 \in S_n(\mathbb{R})$ , et nous définissons  $H(u, v) = H_0 + uH_1 + vH_2$ . Dans ce cas, il a été montré dans [22] que les courbes non-mixantes sont les courbes intégrales d'un champ de direction sur  $\mathbb{R}^2$ , défini par  $\chi_{j-1,j}(u, v) = \begin{pmatrix} -\langle H_2\phi_{j-1}(u, v), \phi_j(u, v) \rangle \\ \langle H_1\phi_{j-1}(u, v), \phi_j(u, v) \rangle \end{pmatrix}$  où  $\langle,\rangle$  est le produit scalaire Euclidien sur  $\mathbb{R}^n$ .

Remarquons que  $\chi_{j-1,j}$  est défini au signe près aux points  $(u,v) \in \mathbb{R}^2$  tels que  $\lambda_{j-1}(u,v) \neq \lambda_j(u,v)$ . Le long de ces courbes, l'approximation adiabatique a une plus grande précision, c'est-à-dire que l'erreur est de l'ordre de  $\epsilon$  le long d'un chemin de contrôle de vitesse  $\epsilon$ , même quand la courbe passe par des intersections de valeurs propres. Contrairement à ce que l'on pourrait penser, les courbes non-mixantes d'un Hamiltonien de niveau n avec  $n \geq 3$  ne correspondent pas localement à celles d'un Hamiltonien réduit de deux niveaux tel que défini dans Section 2.1. Il s'agit d'une structure plus complexe qui présente des propriétés topologiques intéressantes. En effet, nous allons voir que le champ non-mixant peut avoir à la fois des singularités d'indice entier qui sont génériques pour un champ de vecteurs de  $\mathbb{R}^2$  et des singularités d'indice demi-entier qui sont génériques pour un champ de directions de  $\mathbb{R}^2$ , alors que l'espace des champs de vecteurs et des champs de directions sont munis de topologies différentes, telles que les combinaisons de ces deux types de singularités semble à première vue être impossible génériquement. Ce type de structures a déjà été observé dans des cristaux liquides (voir [26] pour une revue générale de la théorie des cristaux liquides et la Figure 1.11 pour une illustration de telles singularités). D'autres recherches sur ce type de structures seront menées dans le cadre de travaux futurs. Notre but est de classifier les singularités du champ de direction  $\chi_{i-1,i}$ , nous avons d'abord besoin de rappeler quelques outils topologiques.

**Propriétés des champs de directions** Soit (M, g) une variété Riemannienne de dimension 2, et K un sous-ensemble fermé de M. Un champ de direction L sur M est une section de  $PT(M \setminus K)$ . L'ensemble K est l'ensemble singulier de L. Comme prouvé dans [21], tout champ de direction peut être obtenu par bissection angulaire (définie pour la métrique g) de deux champs de vecteurs X et Y sur M. Lorsqu'un champ de directions L est bissection de deux champs de vecteurs X et Y, on dit que L est défini comme un proto champ de direction. Les zéros de X et Y correspondent aux singularités de L. Avec cette définition, la topologie de Whitney sur les paires de champs de vecteurs sur M définissent une topologie sur les champs de directions sur M. Il a été prouvé dans [21] que génériquement, un proto champ de direction n'a que des singularités Darbouriennes, c'est à dire que ses courbes intégrales sont localement homéomorphes à celles des proto



FIGURE 1.11 – Combinaison de singularités d'indice demi-entier et entier dans les cristaux liquides, image issue de [8]

champs de directions définis par  $(X_L, Y_L)$ ,  $(X_M, Y_M)$  and  $(X_S, Y_S)$  comme suit :

— Le Lemon proto champ de direction (voit la figure 5.15) est la paire de champs de vecteurs sur  $(\mathbb{R}^2, Eucl)$  définis par

$$X_L(x,y) = \begin{pmatrix} x+y\\ y-x \end{pmatrix}, \quad Y_L(x,y) = \begin{pmatrix} 1\\ 1 \end{pmatrix}.$$

— Le Monstar proto champ de direction (voir la figure 5.16) est la paire de champs de vecteurs sur  $(\mathbb{R}^2, Eucl)$  définis par

$$X_M(x,y) = \begin{pmatrix} x \\ 3y \end{pmatrix}, \quad Y_M(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$$

— Le Star proto champ de direction (voir la figure 5.17) est la paire de champs de vecteurs sur  $(\mathbb{R}^2, Eucl)$  definis par

$$X_S(x,y) = \begin{pmatrix} x \\ -y \end{pmatrix}, \quad Y_S(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}.$$



FIGURE 1.12 – La singularité Lemon, d'indice  $\frac{1}{2}$ .



FIGURE 1.13 – La singularité Monstar, d'indice  $\frac{1}{2}$ .



FIGURE 1.14 – La singularité Star, d'indice  $-\frac{1}{2}$ .

Courbes non-mixantes pour systèmes à n-niveaux aux intersections entre les **niveaux** j-1 et j Au voisinage d'une intersection conique  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  entre les niveaux j-1 et j,

- Il existe un choix lisse du signe des vecteurs propres  $\phi_{i-1}(u, v)$  et  $\phi_i(u, v)$  tel que  $\chi_{j-1,j}$  définit un champ de vecteur  $C^{\infty}$  dans un voisinage épointé de  $(\bar{u}, \bar{v})$ ,
- Les courbes intégrales de  $\chi_{j-1,j}$  sont  $C^{\infty}$  et les vecteurs propres  $\phi_{j-1}$  et  $\phi_j$  sont  $C^{\infty}$ le long de ces courbes,
- Pour toute direction  $\eta$  de  $\mathbb{R}^2$ , il existe une courbe intégrale  $\gamma : [0,1] \to \mathbb{R}^2$  de  $\chi_{j-1,j}$ telle que  $\gamma(1) = (\bar{u}, \bar{v}), \lim_{t \to 1^{-}} \frac{\dot{\gamma}(t)}{||\dot{\gamma}(t)||} = \eta.$ Dans ce cas, on dit que  $\chi_{j,j-1}$  a une singularité de type **(N)**.

Courbes non-mixantes pour systèmes à n-niveaux aux intersections entre les niveaux j et j + 1 (respectivement j - 2 et j - 1) Pour compléter la classification des singularités du champ non-mixant aux intersections de valeurs propres, nous nous concentrons maintenant sur les singularités de  $\chi_{j-1,j}$  aux intersections coniques entre les niveaux j et j + 1.

Soit  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  une intersection conique entre  $\lambda_j$  et  $\lambda_{j+1}$ . On introduit la condition suivante :

**Condition (C)** : Les vecteurs  $P_{j,j+1}H_1\phi_{j-1}(u,v)$  et  $P_{j,j+1}H_2\phi_{j-1}(u,v)$  ne sont pas colinéaires.

Nous prouvons le théorème suivant.

**Theorem 1.2.9.** Soit  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  tel que  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$  est une intersection conique. Supposons que Condition (C) est satisfaite en  $(\bar{u}, \bar{v})$ . Alors  $\chi_{j-1,j}$  a une singularité Darbouxienne en  $(\bar{u}, \bar{v})$ .

Corollary 1.2.10. Soit  $E_1, E_2, E_3 \in \mathbb{R}$  tels que  $E_1 < E_2 < E_3$  et  $H(u, v) = \begin{pmatrix} E_1 & u & 0 \\ u & E_2 & v \\ 0 & v & E_3 \end{pmatrix}$ .

Alors  $\chi_{1,2}$  (respectivement  $\chi_{2,3}$ ) possède des singularités Darbouxiennes aux points  $(u, v) \in \mathbb{R}^2$  tels que  $\lambda_2(u, v) = \lambda_3(u, v)$  (respectivement,  $\lambda_1(u, v) = \lambda_2(u, v)$ ) qui sont

$$(u,v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0) \ (respectivement \ (\pm \sqrt{(E_1 - E_3)(E_1 - E_2)}, 0)).$$

Le tracé des courbes intégrales de  $\chi_{1,2}$  est effectué en Figure 1.15. Nous remarquons qu'il y a une singularité de Type (**N**) aux intersections coniques entre  $\lambda_1$  et  $\lambda_2$ , alors qu'il y a des singularités Star aux intersections coniques entre  $\lambda_2$  and  $\lambda_3$ .

#### 1.2.3 Vers le contrôle d'ensemble avec un seul contrôle

Un enjeu important du contrôle quantique est de concevoir des lois de contrôle explicites pour le problème de l'équation de Schrödinger bilinéaire avec un seul contrôle, à savoir

$$i\frac{d\psi}{dt} = (H_0 + uH_1)\psi \tag{1.12}$$

où  $\psi$  appartient à la sphère unité d'un espace de Hilbert  $\mathcal{H}$ ,  $H_0$  est un opérateur autoadjoint représentant un terme de dérive appelé Hamiltonien libre,  $H_1$  est un opérateur auto-adjoint représentant le couplage de contrôle et  $u : [0, T] \to \mathbb{R}, T > 0$ . D'importants résultats théoriques de contrôlabilité ont été prouvés avec différentes techniques (voir [7, 12, 17] et les références qui y figurent). mais les contrôles explicites sont souvent difficiles à calculer. Pour le problème avec deux entrées ou plus, les méthodes adiabatiques sont considérées aujourd'hui comme façon classique d'obtenir une expression explicite des contrôles et peuvent être utilisées sous des conditions géométriques sur le spectre du Hamiltonien contrôlé (voir les articles [10, 22, 62] et leurs références), et nos résultats sur ces méthodes sont présentés dans la section 2.2.1 de l'Introduction et sont développés

#### 1.2. Thèmes d'études



FIGURE 1.15 – Champ non-mixant entre  $\lambda_1$  et  $\lambda_2$  pour le STIRAP du Corollaire 1.2.10.

dans le Chapitre 3. Cependant, ces méthodes sont efficaces pour des systèmes ayant au moins eux contrôles et ne peuvent pas être appliquées au cas de l'équation (2.12). Notre objectif est alors d'étendre une équation de Schrödinger bilinéaire avec un seul contrôle en une équation de Schrödinger bilinéaire avec deux contrôles dans le même esprit que les extensions de Lie introduites par Sussmann et Liu dans [66] et [77], puis d'appliquer au système étendu les techniques adiabatiques connues. La première étape de cette procédure est l'approximation Rotating Wave (RWA, en abrégé), et est bien connue des physiciens. C'est une approximation de découplage qui permet de se débarrasser des termes très oscillants lorsque le système est commandé par un contrôle réel. Cette approximation est basée sur une procédure de moyennisation du premier ordre (voir [71, 77, 66, 23] pour plus d'informations sur la moyennisation des systèmes dynamiques). Cette approximation est connue pour bien fonctionner pour un faible désaccord par rapport à fréquence de résonance du système et une petite amplitude. Pour un récapitulatif de la RWA et de ses limites, voir [37] et [44, 45, 51]. Dans [24], un cadre mathématique a été établi pour les systèmes quantiques de dimension infinie, formalisant ce que les physiciens appellent oscillations de Rabi généralisées et montrant que la RWA est valide pour une large classe de systèmes quantiques. L'approximation adiabatique et la RWA impliquent des échelles de temps différentes, et il est naturel de se demander si elles peuvent ou non être utilisées en cascade. Le but du chapitre 2.2.3 est de montrer la validité d'une telle approximation sous une certaine condition sur les échelles de temps impliquées dans la dynamique, en utilisant une procédure de calcul de moyennisation. Ensuite, les résultats bien connus de la théorie adiabatique (voir [22, 18, 78]) peuvent être appliqués afin d'obtenir des transitions entre les états propres du Hamiltonien libre  $H_0$ . Cela nous amène à concevoir des lois de contrôle permettant l'inversion d'une particule Spin- $\frac{1}{2}$  et des transferts de population dans le processus STIRAP qui sont robustes par rapport aux inhomogénéités de l'amplitude de l'entrée de contrôle (voir [81, 10] et le chapitre 3). Comme conséquence de l'utilisation d'un contrôle oscillant avec un petit désaccord de fréquence, la méthode proposée n'est pas robuste par rapport aux inhomogénéités des fréquences de résonance, c'est-à-dire les inhomogénéités du terme de drift  $H_0$ .

Pour de tels systèmes, lorsque  $\mathcal{H}$  est de dimension finie, nous proposons d'utiliser des contrôles sous la forme

$$u_{\epsilon}(t) = 2\epsilon^{\alpha} \sum_{(j,k)\in\{1\dots N\}^2} v_{jk}(\epsilon^{\alpha+1}t)\cos(\beta_{jk}t + \frac{1}{\epsilon}\phi_{jk}(\epsilon^{\alpha+1}t)),$$

où, pour tous  $j, k \in \{1 \dots N\}, v_{jk}, \phi_{jk} \in C^{\infty}([0,1], \mathbb{R})$  tels que  $\phi_{jk}(0) = 0$  et  $\beta_{jk} \in \mathbb{R}, N \in \mathbb{N} \setminus \{0\}$ , puis nous étudions le comportement asymptotique du système quantique associé (1.12) lorsque  $\epsilon \to 0$  sur l'intervalle de temps  $t \in [0, \frac{1}{\epsilon^{\alpha+1}}]$ . En supposant  $\alpha > 1$ , par un bon choix de  $(v_{kj}, \phi_{kj})$ , et  $\beta_{kj}$  pour tous k, j, nous montrons un théorème général d'approximation (voir les théorèmes 6.1.3 et 6.2.11) qui permet de concevoir des lois de contrôle explicites qui sont robustes par rapport aux inhomogénéités de l'amplitude du contrôle. Ce résultat nous a permis de prouver les résultats suivants.

Soit U un intervalle ouvert de  $\mathbb{R}$  contenant 0.

**Theorem 1.2.11.** Soit  $E \in \mathbb{R} \setminus \{0\}$ . L'équation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E & \delta u\\ \delta u & -E \end{pmatrix}\psi$$
(1.13)

est ensemble contrôlable de façon approchée entre les états propres de  $H_0 = \begin{pmatrix} E & 0 \\ 0 & -E \end{pmatrix}$ uniformément par rapport à  $\delta \in [a,b] \subset (0,+\infty)$  et  $u \in U$ .

**Theorem 1.2.12.** Soient  $E_1, E_2, E_3 \in \mathbb{R}$  tels que  $|E_k - E_q| \neq |E_j - E_l|$ , pour tous  $k, q, j, l \in \{1 \dots 3\}$  tels que  $(k, q) \notin \{(j, l), (l, j)\}$ . L'équation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E_1 & \delta_1 u & 0\\ \delta_1 u & E_2 & \delta_2 u\\ 0 & \delta_2 u & E_3 \end{pmatrix} \psi$$
(1.14)

est ensemble contrôlable de façon approchée entre les états propres de  $H_0 = \begin{pmatrix} E_1 & 0 & 0 \\ 0 & E_2 & 0 \\ 0 & 0 & E_3 \end{pmatrix} \psi$ uniformément par rapport à  $\delta_1 \in [a_1, b_1] \subset (0, +\infty), \ \delta_2 \in [a_2, b_2] \subset (0, +\infty)$  et  $u \in U$ .

#### 1.2. Thèmes d'études

Une question ouverte est celle de la contrôlabilité d'ensemble par rapport à  $\delta \in [a, b] \subset (0, +\infty)$  de l'équation

$$i\frac{d\psi}{dt} = (H_0 + \delta u H_1)\psi,$$

où  $H_0, H_1 \in S_n(\mathbb{R})$ , en supposant que  $H_0, H_1$  sont tels que l'équation (2.12) est contrôlable. Des questions similaires en dimension infinie seraient aussi très intéressantes à étudier. Un autre projet d'étude est de comprendre comment contrôler de façon robuste une équation où le terme de drift est incertain.

# Chapitre

# Introduction

#### Contents

2.1 Qua	ntum physics preliminaries, adiabatic theorem	<b>25</b>
2.2 Topics of investigation		30
2.2.1	Ensemble control with two real controls	31
2.2.2	Classification of the singularities of the non-mixing field $\ . \ . \ .$	38
2.2.3	Towards ensemble control with a single input $\ldots \ldots \ldots$	44

### 2.1 Quantum physics preliminaries, adiabatic theorem

For a general closed quantum system under the action of a control u having values in an open set U of  $\mathbb{R}^m$ , we are interested in the controlled equation of the form

$$i\frac{d\psi}{dt}(t) = H(u(t))\psi(t), \qquad \psi(t) \in \mathcal{H},$$
(2.1)

with H(u) essentially self-adjoint on the complex separable Hilbert space  $\mathcal{H}$  for every value of u, with a common dense domain for every  $u \in U$ . If  $\mathcal{H}$  is finite dimensional, the controllability of Equation (2.9) has been extensively studied by Lie-algebraic methods (see [6, 9, 34]). If  $\mathcal{H}$  is infinite dimensional, then Equation (2.9) is in general not exactly controllable [11]. Many results concerning weaker notions of controllability have been proved [17, 25, 22]. However, in both cases, there exist few theoretical results about the robustness of the control strategies with respect to variations of the parameters of the system, except for two level systems (see [13, 63]). Adiabatic control is known as a powerful technique which can be used to handle perturbations and uncertainties. The adiabatic theorem states, in its simplest form, that under a separation condition on the energy levels of the controled Hamiltonian, the occupation probabilities of the energy levels are approximately conserved when the controls are slowly varying. One of its main advantages

#### 2.1. Quantum physics preliminaries, adiabatic theorem

is that it provides explicit and regular control laws. It has hence been successfully applied to obtain control strategies such as the chirp pulses (see, for instance [27, 74, 43]) for spin 1/2 systems with dispersed Larmor frequency. Another nowadays classical application of adiabatic control to ensemble controllability are the so-called counterintuitive pulses for the STIRAP process [38, 73, 83]. A generalization of this approach has been proposed in [62], and has raised many interesting questions, in particular about the validity of the Rotating Wave Approximation in this framework. Hence, a general theory is needed for the robust control in the quantum framework. The main contribution of the thesis is the understanding of the robustness of adiabatic control strategies, by studying the problem of controlling parametrized families of quantum systems with a common control input, that is the *Ensemble controllability* problem. Our main results about this topic are presented in the introduction in Section 2.2.1 with two real control inputs and real Hamiltonians, most of the arguments are valid both when  $\mathcal{H}$  is finite dimensional or infinite dimensional under suitable assumptions on the Hamiltonian. An extension of this with a single real control input using the Rotating Wave Approximation (see, for instance [53]) is exposed in Section 2.2.3. We will develop these issues in the chapters 3 and 6.

The starting point of the thesis is the adiabatic theorem (see, for instance [16, 55, 54, 47]), that we state in its general form, adapting the results exposed in [78, Chapter 2] to our issues of control theory.

General adiabatic theorem Let  $\mathcal{H}$  be a complex separable Hilbert space and  $k \in \mathbb{N}$ . Denote the set of essentially self-adjoint linear operators on  $\mathcal{H}$  by  $\mathcal{L}_{sa}(\mathcal{H})$ , and the set of  $C^{j}$  and bounded functions between two topological spaces X and Y by  $C_{b}^{j}(X,Y)$ . Let  $H(\cdot) \in C_{b}(\mathbb{R}^{k}, \mathcal{L}_{sa}(\mathcal{H}))$ . We say that the spectrum  $\sigma(\cdot)$  of  $H(\cdot)$  contains a separated part  $\sigma_{*}(\cdot) \subset \sigma(\cdot)$  locally around  $\bar{u} \in \mathbb{R}^{k}$  if there exists two bounded and continuous functions  $f, g \in C_{b}(\mathbb{R}^{k}, \mathbb{R})$  and a neighborhood V of  $\bar{u}$  in  $\mathbb{R}^{k}$  satisfying, for every  $u \in V$ ,  $\sigma_{*}(u) \subset [f(u), g(u)]$  and  $\inf_{u \in V} \operatorname{dist}([f(u), g(u)], \sigma(u) \setminus \sigma_{*}(u)) > 0$ . For  $u \in V$ , define  $P_{*}(u)$ as the spectral projection of H(u) on the eigenspace of H(u) associated with  $\sigma_{*}(u)$ . Let  $\gamma : [0, 1] \to V$  be a regular  $C^{2}$  control path. We are interested in the trajectories  $\psi^{\epsilon}(t) \in \mathcal{H}$ of the Adiabatic Schödinger Equation:

$$i\dot{\psi}^{\epsilon}(t) = H(\gamma(\epsilon t))\psi^{\epsilon}(t), \quad \psi^{\epsilon}(0) = \psi_0,$$

where  $\psi_0 \in \mathcal{H}, t \in [0, \frac{1}{\epsilon}]$ , and  $\epsilon > 0$ . In the variable  $\tau = \epsilon t \in [0, 1]$ , the reparameterized trajectory  $\tilde{\psi}^{\epsilon}(\tau) = \psi^{\epsilon}(\tau/\epsilon)$  satisfies

$$i\dot{\tilde{\psi}^{\epsilon}}(\tau) = \frac{1}{\epsilon} H(\gamma(\tau))\tilde{\psi}^{\epsilon}(\tau), \quad \tilde{\psi}^{\epsilon}(0) = \psi_0, \qquad (2.2)$$

where  $\tau \in [0, 1]$ .

Define the *adiabatic Hamiltonian*, for every  $\tau \in [0, 1]$ , by

$$H_a^{\epsilon}(\tau) = H(\tau) - i\epsilon P_*(\gamma(\tau))\dot{P}_*(\gamma(\tau)) + i\epsilon (Id - P_*(\gamma(\tau)))\dot{P}_*(\gamma(\tau)).$$

where  $\dot{P}_*(\gamma(\tau))$  is the derivative of  $P_*$  along the control path  $\gamma$ . When they are well defined, define the two unitary propagators  $U^{\epsilon}(\tau)$  and  $U^{\epsilon}_a(\tau)$  on  $\mathcal{H}$  of, respectively, Equation (2.2) and  $i\dot{\psi}^{\epsilon}(\tau) = \frac{1}{\epsilon}H^{\epsilon}_a(\gamma(\tau))\psi^{\epsilon}(\tau)$ , for  $\tau \in [0, 1]$ . An essential property of  $U^{\epsilon}_a$  is that for every  $\tau \in [0, 1], P_*(\gamma(\tau))U^{\epsilon}_a(\tau) = U^{\epsilon}_a(\tau)P_*(\gamma(0)).$ 

**Theorem 2.1.1.** [78, Theorems 2.1, 2.2] Let V be an open set of  $\mathbb{R}^k$  and  $\gamma : [0,1] \to V$ be a regular  $C^2$  control path. Let  $H(\cdot) \in C_b^2(V, \mathcal{L}_{sa}(\mathcal{H}))$  such that the operators H(u) have a common dense domain  $D \subset \mathcal{H}$  for every  $u \in V$ , and are bounded from below, uniformly with respect to  $u \in V$ . Assume that  $\sigma_*(u) \subset \sigma(H(u))$  is a locally separated part of  $\sigma$ , for  $u \in V$ .

Then  $P_* \in C_b^2(V, \mathcal{L}_{sa}(\mathcal{H})))$ , the unitary propagators  $U_{\epsilon}$  and  $U_a^{\epsilon}$  are well defined, and there is a constant  $C < \infty$  such that for all  $\tau \in [0, 1]$ ,

$$||U^{\epsilon}(\tau) - U^{\epsilon}_{a}(\tau)|| \le C\epsilon.$$

Adiabatic decoupling Let  $j \in \mathbb{N}$ . Assume that the eigenvalues  $\lambda_j$  and  $\lambda_{j+1}$  of H are separated from the rest of the spectrum for u in a neighborhood V of  $\bar{u} \in \mathbb{R}^k$ . Notice that we do not assume here the existence of a gap between  $\lambda_j$  and  $\lambda_{j+1}$ . Indeed, the aim of what follows is to get an appropriate approximation of the dynamics in a neighborhood of a point  $\bar{u}$  such that  $\lambda_j(\bar{u}) = \lambda_{j+1}(\bar{u})$ , that is, an *intersection of eigenvalues*. Set for every  $u \in V$ , the spectral projection  $P_{j,j+1}(u)$  of H(u) on the eigenspace associated with  $\sigma_*(u) =$  $\{\lambda_j(u), \lambda_{j+1}(u)\}$ . Consider, for every  $u \in V$ , a  $C^2$  orthonormal basis  $(\psi_j(u), \psi_{j+1}(u))$  of  $\operatorname{Im} P_{j,j+1}(u)$ . For every  $u \in V$ , consider a unitary mapping I(u) from  $\mathbb{C}^2$  to  $\operatorname{Im} P_{j,j+1}(u)$ , which is  $C^2$  with respect to  $u \in V$ , such that  $I(u)(e_1) = \psi_j(u)$  and  $I(u)(e_2) = \psi_{j+1}(u)$ , where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{C}^2$ . Along every regular  $C^2$  control path  $(\gamma(\tau))_{\tau \in [0,1]}$ in V we define the *effective Hamiltonian* for every  $\tau \in [0, 1]$ , by

$$H_{\text{eff}}(\gamma(\tau)) = h(\gamma(\tau)) - i\epsilon \begin{pmatrix} 0 & \langle \dot{\psi}_j(\gamma(\tau)), \psi_{j+1}(\gamma(\tau)) \rangle \\ \langle \dot{\psi}_{j+1}(\gamma(\tau)), \psi_j(\gamma(\tau)) \rangle & 0 \end{pmatrix}, \quad (2.3)$$

where, for every  $u \in V$ ,  $h(u) = I^{-1}(u)H(u)I(u)$  is a two-dimensional hermitian matrix,  $C^2$  in  $u \in V$ , and  $\dot{\psi}_q(\gamma(\tau))$ ,  $q \in \{j, j+1\}$  is the derivative of  $\psi_q$  along the path  $\gamma$ . Denoting the propagator associated with  $H_{\text{eff}}(\gamma(\tau))$  by  $U_{\text{eff}}^{\epsilon}(\tau)$ , Theorem 2.1.1 yields

$$\left|\left(U^{\epsilon}(\tau) - I(\gamma(\tau))U^{\epsilon}_{\text{eff}}(\tau)I^{-1}(\gamma(0))\right)P_{j,j+1}(0)\right|\right| \le C\epsilon,$$
(2.4)

for every  $\tau \in [0, 1]$ .

An important property, obtained by deleting the second term of  $H_{\text{eff}}$  by a standard averaging procedure (see Theorem 4.6.4 proved in Chapter 4 for a precise statement), is the following, which is valid under suitable assumptions on the eigenvalues of H, which are those of h, and the eigenvectors of H, which are those of h up to the transformation



Figure 2.1 – Two eigenvalues of H(u) in a neighborhood of a conical intersection when k = 2, as a function of  $u \in \mathbb{R}^2$ .

I. Given a sufficiently regular path  $\gamma : [0,1] \to V$  and  $\tilde{\psi}_0 \in \mathbb{C}^2$ , the solutions  $\psi$  and  $\tilde{\psi}$ of, respectively,  $i\frac{d\psi}{dt} = H(\gamma(\epsilon t))\psi, \psi(0) = I(\gamma(0))\tilde{\psi}_0$ , and  $i\frac{d\tilde{\psi}}{dt} = h(\gamma(\epsilon t))\tilde{\psi}, \tilde{\psi}(0) = \tilde{\psi}_0$  are such that  $\psi(1/\epsilon)$  is  $\epsilon^{\alpha}$ -close, up to phases, to  $I(\gamma(1))\tilde{\psi}(1/\epsilon)$ . The value of  $\alpha \in \mathbb{R}$  is positive and depends on the order of the gap between the eigenvalues  $\lambda_j$  and  $\lambda_{j+1}$  of H along the path  $\gamma$ , that is on the quantity  $\lambda_{j+1} - \lambda_j$  and its derivatives (see Theorem 4.6.4).

This approximation is interesting because it preserves the adiabatic structure of the Hamiltonians, it is important to get uniform estimations of the dynamics holding along every sufficiently regular control path of  $\mathbb{R}^k$ . Moreover, the knowledge of h provides all the information needed about the local regularity of the eigenpairs of H. Notice that h, called *reduced Hamiltonian* depends on the choice of a basis of  $\pi_{H(u)} = \text{Im}P_{j,j+1}(u)$ , for  $u \in V$ . However, we will see in Section 4.6.3 that the geometric properties and hence the dynamical properties of the Schrödinger equation that we are are interested in are invariant by such a change of basis. We have proved (see Section 4.6.2) that, when  $\mathcal{H}$  is finite dimensional, generic properties on h and its jet until order 2 provide generic properties on the Hamiltonian  $\mathcal{H}$  and its jet until order 2.

**Conical intersections** A conical intersection (also called diabolic point) is a conelike singularity of the spectrum of H(u), seen as a function of the control  $u \in \mathbb{R}^k$  (see Figure 2.1)

In the case where H(u) is real and k = 2, they are generic in the sense that they are the least degenerate singularities of the spectrum of a Hamiltonian and have been studied since the beginning of quantum mechanics [82]. They play an important role in the context of semiclassical analysis [32, 33]. Adiabatic paths through conical intersections can be used to induce superpositions of eigenstates, as shown in [22, 46] and to obtain tests for exact controllability when  $\mathcal{H}$  is finite-dimensional and approximate controllability when  $\mathcal{H}$  is infinite-dimensional [19]. For a large class of systems, they are generic and structurally stable, in the sense that they are not removed by small perturbations of the Hamiltonian.

Let U be a connected open set of  $\mathbb{R}^2$ . By a slight abuse of notations, for  $u \in \mathbb{R}^2$ , denote the components of u by (u, v). Assume that  $H(u, v) = H_0 + uH_1 + vH_2$  where  $H_0, H_1, H_2$  are self-adjoint operators on  $\mathcal{H}$  having a common domain  $\mathcal{D}$  such that the following conditions, that we refer as  $(\mathcal{R})$  are satisfied:

- $H_0$  has a discrete spectrum;
- $H_1, H_2$  are bounded;
- There exists an orthonormal basis  $(b_j)_j$  of the Hilbert space  $\mathcal{H}$  such that  $\langle b_j, H_0 b_q \rangle$ ,  $\langle b_j, H_1 b_q \rangle$ ,  $\langle b_j, H_2 b_q \rangle$  are real for every j, q.

We say that  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  is a conical intersection between two eigenvalues  $\lambda_j$  and  $\lambda_{j+1}$ if  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$  has multiplicity two and there exists C > 0 such that for every unit vector  $\eta \in \mathbb{R}^2$  and t > 0 small enough,  $\lambda_{j+1}((\bar{u}, \bar{v}) + t\eta) - \lambda_j((\bar{u}, \bar{v}) + t\eta) > Ct$ .

Let  $(\gamma(t))_{t\in[0,1]} = (u(t), v(t))_{t\in[0,1]}$  be a  $C^3$  regular path of  $\mathbb{R}^2$ . Consider the equation

$$i\frac{d\psi(t)}{dt} = H(\gamma(\epsilon t))\psi(t), \qquad (2.5)$$

where  $\psi(t) \in \mathcal{H}$ . Assume that H has a conical intersection at (0,0). Assume that  $\gamma(t_0) = (0,0)$  for  $t_0 \in (0,1)$ ,  $\gamma(t) \neq 0$  for every  $t \neq t_0$ . Consider a solution  $\psi_{\epsilon}$  of Equation (2.5) such that  $\psi_{\epsilon}(0) = \phi_j(\gamma(0))$ . Then

$$\|\psi_{\epsilon}(\frac{1}{\epsilon}) - e^{i\theta}\phi_{j+1}(\gamma(1))\| = O(\epsilon^{1/2}), \qquad (2.6)$$

where  $\theta \in \mathbb{R}$  is possibly depending on  $\epsilon$ . We can then implement a strategy in order to make transitions of states passing through conical intersections, as it has been exposed in [22] (see Figure 2.2).

We are interested in the following notion of controllability. We say that Equation (2.5) is approximately controllable if for every  $\epsilon > 0$  and  $\psi_1 \in \mathcal{H}$ , there exists a control  $\gamma(\cdot) = (u(\cdot), v(\cdot)) : [0, T] \to U$  such that the solution of Equation (2.5) with initial condition  $\psi(0) = \psi_0$  satisfies  $\|\psi(T) - \psi_1\| < \epsilon$ . We say that the spectrum  $\sigma(\cdot)$  of  $H(\cdot)$  is conically connected if all eigenvalue intersections are conical and for every j, there exists a conical intersection  $(\bar{u}_j, \bar{v}_j)$  between the eigenvalues  $\lambda_j, \lambda_{j+1}$ , with  $\lambda_l(\bar{u}_j, \bar{v}_j)$  simple if  $l \neq j, j + 1$ . We have the following general results.

**Theorem 2.1.2** ([19]). Assume that the spectrum of  $H(\cdot)$  is conically connected. Then Equation (2.5) is approximately controllable.

**Theorem 2.1.3** ([19]). Assume that  $\mathcal{H}$  is finite dimensional. Assume that the spectrum of  $H(\cdot)$  is conically connected. Then Equation (2.5) is exactly controllable.

#### 2.2. Topics of investigation



Figure 2.2 – Climbing path at speed  $\epsilon$  when k = 2.

**Proposition 2.1.4** (Genericity). Let  $H_0$  be a n-dimensional real symmetric matrix. Then, generically with respect to  $(H_1, H_2)$  in the space of n-dimensional real symmetric matrices, all intersections of eigenvalues of  $H_0 + uH_1 + vH_2$  are conical.

Some results of genericity of conical intersections when  $\mathcal{H}$  is infinite dimensional have been proved in [31, 22], in the cases of real controlled potentials.

## 2.2 Topics of investigation

Our aim is to continue exploring the links between singularities of the spectrum of Hamiltonians, the geometry of the eigenspaces, and controllability properties of the associated quantum system. In particular, as announced previously, we study the problem of controlling parametrized families of quantum systems with a common control input. It is a critical task for many applications in quantum control (see [39] and references therein), notably in Nuclear Magnetic Resonance [40]. This issue of ensemble controllability has been formulated mathematically by Li and Khaneja in [63] and by Beauchard, Coron and Rouchon in [13]. Results for ensemble control beyond the quantum control setting can be found in [4, 50, 64, 72].

#### 2.2.1 Ensemble control with two real controls

For a general closed quantum system under the action of a control  $u \in \mathbb{R}^m$  and depending on a parameter z, the corresponding controlled equation is of the form

$$i\frac{d\psi}{dt}(t) = H(u(t), z)\psi(t), \qquad \psi(t) \in \mathcal{H},$$
(2.7)

with H(u, z) essentially self-adjoint on the separable complex Hilbert space  $\mathcal{H}$  for every value of u and z. The parameter z can be used either to describe a family of physical systems on which acts a common field driven by u or a physical systems for which the value of one parameter is not known precisely.

The controllability properties of systems of this form has been studied both for discrete and continuous sets of parameters. The case of a finite set of systems is characterized in [14, 35]. In [29] the asymptotic ensemble stabilization is studied for countable sets of parameters. In [63], [13] a proof of a strong notion of ensemble controllability has been obtained for a two-level system. Numerical ensemble control in the case of a continuum of parameters has been throughly studied for two-level systems [79, 28, 75]. Our aim is to generalize the theorems 2.1.2 and 2.1.3 to the issue of Ensemble control.

We restrict our study to the problem of ensemble approximate controllability between eigenstates. Excepted when it is explicitly mentioned, we will study finite dimensional real quantum systems, that is systems whose Hamiltonian belong to the set  $S_n(\mathbb{R})$  of real symmetric *n*-dimensional matrices. However the main results of controllability remain valid when  $H(u, v) = H_0 + uH_1 + vH_2$  where  $H_0, H_1, H_2$  are essentially self-ajoint operators on  $\mathcal{H}$  with a common dense domain, and satisfying Condition  $(\mathcal{R}), (u, v) \in U$ , where Uis a connected open set of  $\mathbb{R}^2$ , and an adaptation of this condition to the parametric case (see Chapter 2.2.1).

#### Notations

- Let  $H \in C^k(U, S_n(\mathbb{R}))$ . For  $u \in U$ , denote the spectrum of H(u) by  $(\lambda_j(u))_{j=1}^n$ , where  $j \mapsto \lambda_j(u)$  is the nondecreasing sequence of eigenvalues of H(u) repeated according to their multiplicities. We write  $(\phi_1(u), \ldots, \phi_n(u))$  to denote an orthonormal basis of associated eigenvectors.
- Let  $H \in C^k(U \times [z_0, z_1]), S_n(\mathbb{R})$ ). For  $u \in U$ , and  $z \in [z_0, z_1]$ , denote the spectrum of H(u, z) by  $(\lambda_j(u, z))_{j=1}^n$ , where  $j \mapsto \lambda_j(u, z)$  is the nondecreasing sequence of eigenvalues of H(u, z) repeated according to their multiplicities. We write  $(\phi_1^z(u), \ldots, \phi_n^z(u))$  to denote an orthonormal basis of associated eigenvectors.

**Definition 2.2.1.** We say that system (2.7) is ensemble approximately controllable between eigenstates if for every  $\epsilon > 0$ ,  $j, k \in \{1, ..., n\}$ , and  $u_0, u_1 \in U$  such that  $\lambda_j(u_0, z)$ and  $\lambda_k(u_0, z)$  are simple for every  $z \in [z_0, z_1]$ , there exists a control  $u(\cdot) : [0, T] \to U$  such that for every  $z \in [z_0, z_1]$  the solution of (2.7) with initial condition  $\psi^z(0) = \phi_j^z(u_0)$ satisfies  $\|\psi^z(T) - e^{i\theta}\phi_k^z(u_1)\| < \epsilon$  for some  $\theta \in \mathbb{R}$  (possibly depending on z and  $\epsilon$ ).

By a factorization of the dynamics by the trace, the reduced Hamiltonian h defined in Section 2.1 can be reduced to a zero trace Hamiltonian. In the case k = 2, we can assume, without loss of generality that for every  $(u, v) \in \mathbb{R}^2$  in a neighborhood of (0, 0),  $h(u, v) = H_f(u, v)$ , where  $H_f$  is defined by

$$H_f(u,v) = \begin{pmatrix} f_1(u,v) & f_2(u,v) \\ f_2(u,v) & -f_1(u,v) \end{pmatrix},$$

with  $f = (f_1, f_1) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . In the parametric case, for every  $(u, v, z) \in \mathbb{R}^3$  in a neighborhood of (0, 0, 0), we can assume  $h(u, v, z) = H_f(u, v, z)$ , where  $H_f$  is defined by

$$H_f(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix},$$

with  $f = (f_1, f_1) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . In this case, intersections of eigenvalues which are transverse in every direction and tangent in a specific direction, called the *non-conical direction*, appear generically. They are called *semi-conical* intersections (see Figure 2.3), and are defined rigorously in Section 4.2.1. We are going to study normal forms for this generic case.

#### Normal forms

Admissible transformations of two-level systems We define admissible transformations in order to get normal forms for the Hamiltonians  $H_f$ . Consider the Schrödinger Equation, defined for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  by

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t))\psi(t), \qquad (2.8)$$

where  $\psi(t) \in \mathbb{C}^2$ , and the Ensemble Schrödinger Equation, defined for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  by

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t), z)\psi(t), \qquad (2.9)$$

where  $\psi(t) \in \mathbb{C}^2$ . The control functions u, v are in  $L^{\infty}(\mathbb{R}, \mathbb{R})$  and the parameter z belongs to  $[z_0, z_1]$  where  $z_0, z_1 \in \mathbb{R}$ .

The three transformations correspond to equivalent representations of the dynamical systems (2.8) and (2.9) achieved, respectively, by time-reparameterization, state-space diffeomorphism, and independent diffeomorphic transformations of both the space of controls and the space of the parameter z.

**Definition 2.2.2.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  are equivalent at 0 if there exists  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  where  $\phi$  is a diffeomorphism satisfying  $\phi(0) = 0$ , and  $\zeta = \pm 1$ , such that for every (u, v) in a neighborhood of 0,

$$\begin{cases} \tilde{f}_1(u,v) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v) - \zeta\sin(2\theta)f_2 \circ \phi(u,v)), \\ \tilde{f}_2(u,v) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v) + \zeta\cos(2\theta)f_2 \circ \phi(u,v)). \end{cases}$$



Figure 2.3 – The two eigenvalues of  $H_f(u, v)$  in a neighborhood of a semi-conical intersection, as a function of (u, v).

**Definition 2.2.3.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  are equivalent at 0 if there exists  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^3) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  where  $\phi$  is a diffeomorphism of the form  $\phi : (u, v, z) \mapsto (\phi_1(u, v), \phi_2(u, v), \phi_3(z))$ , where  $\phi_1, \phi_2 \in C^{\infty}(\mathbb{R}^2, \mathbb{R})$  and  $\phi_3 \in C^{\infty}(\mathbb{R}, \mathbb{R})$ , satisfying  $\phi(0) = 0$ , and  $\zeta = \pm 1$ , such that for every (u, v, z) in a neighborhood of 0,

$$\begin{cases} \tilde{f}_1(u,v,z) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v,z) - \zeta\sin(2\theta)f_2 \circ \phi(u,v,z)), \\ \tilde{f}_2(u,v,z) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v,z) + \zeta\cos(2\theta)f_2 \circ \phi(u,v,z)). \end{cases}$$

This allows to classify the generic local behaviours of quantum systems with respect to (u, v).

**Non-parametric case** Consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Then  $H_f$  has a conical intersection at (0,0) if and only if f is equivalent to  $(u,v) \mapsto \begin{pmatrix} u & v \\ v & -u \end{pmatrix}$ . A generic Hamiltonian has only conical singularities.

We say that  $H_f$  has a *semi-conical intersection* (see Figure 2.3) at (0,0) if and only if f is equivalent to  $(u,v) \mapsto \begin{pmatrix} h(u)u \\ u+v^2 \end{pmatrix}$  where  $h : \mathbb{R} \to \mathbb{R}$  is a smooth function satisfying h(0) = 1. Semi-conical intersections are not generic for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , however, we are going to see that they may appear at isolated points when considering the parametric case. The corresponding Hamiltonian  $\begin{pmatrix} h(u)u & u+v^2 \\ u+v^2 & -h(u)u \end{pmatrix}$  is such that its eigenvalues are tangent along a control path tangent to  $e_2 = (0, 1)$ , that is, the non-conical direction at (0, 0), while they are transverse in every other directions. Then we classify the different generic local behaviours of quantum systems with respect to (u, v, z).

**Parametric case** Now we can define F-conical and F-semi-conical intersections, which are the two generic cases for parametric Hamiltonians. They correspond, for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , to the case where  $H_f(\cdot, 0)$  has conical or semi-conical intersections, combined with additional regularity hypothesis with respect to the parameter z.

Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

— We say that 0 is *F*-conical for f if and only if there exist  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$ satisfying  $h_1(0) = h_2(0) = 1$ , such that f is equivalent to

$$(u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z - u) \\ h_2(u, v, z)(z - v) \end{pmatrix};$$

— We say that 0 is *F-semi-conical* for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  if and only if there exist  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfying  $h_1(0) = h_2(0) = 1$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfying  $m(0) \notin \{-1, 0\}$  such that f is equivalent to

$$(u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z - m(u)u) \\ h_2(u, v, z)(z + u + v^2) \end{pmatrix}$$

*n*-level systems Let  $H \in C^{\infty}(\mathbb{R}^2, S_n(\mathbb{R}))$ . We prove that H has a conical intersection at (0,0) between the levels  $\lambda_j$  and  $\lambda_{j+1}$  if and only if there exists a reduced Hamiltonian h of H locally around (0,0) having a conical intersection at (0,0). As a consequence, it is coherent to give the following definition of semi-conical intersections for H.

**Definition 2.2.4.** We say that H has a semi-conical intersection at (0,0) between the levels  $\lambda_j$  and  $\lambda_{j+1}$  if there exists a reduced Hamiltonian h of H having a semi-conical intersection at (0,0).

We prove in Section 4.6.3 that these properties do not depend on the mapping I(u, v), for (u, v) in a neighborhood of (0, 0), used for the adiabatic decoupling as in Section 2.1 of the introduction.

In the parametric case, we say that H has a F-conical intersection (respectively, F-semiconical intersection) at (0,0,0) if there exists a reduced Hamiltonian h of H having a Fconical intersection (respectively, F-semi-conical intersection) at (0,0,0). In Section 4.6.3, we show that, generically with respect to  $H \in C^{\infty}(\mathbb{R}^3, S_n(\mathbb{R}))$ , an eigenvalue intersection  $(u, v, z) \in \mathbb{R}^3$  of H is either F-conical ot F-semi-conical.

Actually, semi-conical intersections are not so rare. For instance, they can appear when, for  $(u, v) \in \mathbb{R}^2$ ,  $H(u, v) = \begin{pmatrix} E & u & 0 \\ u & E & v \\ 0 & v & E' \end{pmatrix}$ , where E < E', we call this case the *degenerate* 



Figure 2.4 – Semi-conical intersection for a degenerate STIRAP process

STIRAP. On Figure 2.4, we have plotted the spectrum of H(u, v) as a function of (u, v) for such a Hamiltonian. We can notice that there is a semi-conical intersection between the first and second levels, and two conical intersections between the second and third levels.

**Dynamical properties at semi-conical intersections** Thanks to the normal forms, we can estimate the adiabatic approximation for general *n*-level Hamiltonians. Let  $H \in C^{\infty}(\mathbb{R}^2, S_n(\mathbb{R}))$ , having a semi-conical intersection at (0, 0). Denote by  $\eta$  the non-conical direction. Let  $(\gamma(t))_{t \in [0,1]} = (u(t), v(t))_{t \in [0,1]}$  be a smooth regular path of  $\mathbb{R}^2$  such that  $\gamma(t_0) = (0, 0)$  for  $t_0 \in (0, 1), \gamma(t) \neq 0$  for every  $t \neq t_0$ , and  $\gamma$  is tangent to  $\eta$  at  $t = t_0$ . Consider a solution  $\psi_{\epsilon}$  of the equation

$$i\frac{d\psi(t)}{dt} = H(u(\epsilon t), v(\epsilon t))\psi(t), \qquad (2.10)$$

where  $\psi(t) \in \mathbb{C}^n$ , such that  $\psi_{\epsilon}(0) = \phi_j(\gamma(0))$ . Then  $\|\psi_{\epsilon}(\frac{1}{\epsilon}) - e^{i\theta}\phi_j(\gamma(1))\| = O(\epsilon^{1/3})$ , where  $\theta \in \mathbb{R}$  is possibly depending on  $\epsilon$  (see Figure 2.5 for a graphic representation of the control path passing through the singularity in the non-conical direction).

#### When conicity is preserved when the parameter z varies

In Chapter 3 we propose a framework for the adiabatic ensemble control of a continuum of n-level systems with real Hamiltonian, driven by two controls and having conical intersections between the eigenvalues. The main idea is that, if a system corresponding to a fixed parameter has conical intersections between two eigenvalues, then a small perturbation of the parameter yields a curve of conical intersections, each point of the curve



Figure 2.5 – Control path passing in the non-conical direction.

corresponding to exactly one value of the parameter. One can then follow adiabatically such curves in the space of controls and obtain a population transfer between the two levels for the whole ensemble of systems. This result has been written without resorting to the normal forms because of the already well-known estimation of the adiabatic dynamics at conical intersections in the litterature. However, the same result can be deduced also from the local study of normal forms.

Let U be an open set of  $\mathbb{R}^2$ . For  $j \in \{1 \dots n\}$ , let us denote by  $\gamma_j$  the set  $\{(u, v, z) \in U \times [z_0, z_1] \mid \lambda_j(u, v, z) = \lambda_{j+1}(u, v, z)\}$ , where by convention  $\gamma_0 = \gamma_n = \emptyset$ . Denote the projection of  $(u, v, z) \in \mathbb{R}^3$  onto the (u, v)-component by  $\pi$ .

ASSUMPTION  $A_j$ . There exist a connected component  $\hat{\gamma}_j$  of  $\gamma_j$  and a map such that  $\beta_j : [z_0, z_1] \to U$  such that  $\beta_j$  is a  $C^3$  embedding and

-  $\pi(\hat{\gamma}_j)$  is contained in U ×  $[z_0, z_1] \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1}));$ 

$$- \pi(\hat{\gamma}_j) = \beta_j([z_0, z_1]);$$

— for every  $z \in [z_0, z_1]$ ,  $\lambda_j(\cdot, z)$  and  $\lambda_{j+1}(\cdot, z)$  have a unique intersection on  $\pi(\hat{\gamma}_j)$ , which is conical and occurs at  $\beta_j(z)$ .

Moreover the set  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  is pathwise connected.

Notice that the second and third conditions are equivalent to say that  $\hat{\gamma}_j$  has F-conical intersections only and  $\pi(\hat{\gamma}_j)$  is an embedded curve of  $\mathbb{R}^2$  without self-intersections. The following result is the main result of the article [10] and of the chapter 3.

**Theorem 2.2.5.** Consider a  $C^3$  map  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$ . Let



Figure 2.6 – A control realizing an ensemble transition between  $\phi_1^z(u_0, v_0)$  and  $\phi_4^z(u_0, v_0)$ .

assumption  $A_j$  be satisfied for every  $j \in \{1 \dots n\}$ . Then the equation  $i\frac{d\psi}{dt} = H(u, v, z)\psi$  is ensemble approximately controllable between eigenstates.

Method of control The method of control is presented on Figure 2.6, with a path  $(u(t), v(t))_{t \in [0,1]}$ , followed at a speed  $\epsilon$ . The error in the adiabatic regime has order  $\sqrt{\epsilon}$  on an interval of time of length  $\frac{1}{\epsilon}$ .

#### When a semi-conical intersection appears when the parameter z varies

The argument sketched above works under the assumption that for all values of the parameter eigenvalue intersections remain conical and describe a smooth curve. These assumptions are satisfied for generic small parametric perturbations. For generic large perturbations it may happen that conicity of eigenvalue intersections is lost at isolated points of the curve. The goal is to extend the analysis to this case.

**Controllability properties** In the case where there are F-semi-conical intersections of eigenvalues on  $\gamma_j$ , the question is whether or not the adiabatic approximation remains valid when the control path passes through such points. Using the normal forms, we notice that  $\pi(\hat{\gamma}_j)$  is tangent to the non-conical direction at a F-semi-conical intersection. Hence, when a control path follows  $\pi(\hat{\gamma}_j)$ , it passes into the semi-conical intersection for  $H(\cdot, z)$ in the non-conical direction, and hence there is no transition between eigenstates for the system with parameter z. However, we show that under the assumption  $\tilde{A}_j$  that follows, the adiabatic approximation is uniform with respect to the parameter  $z \in [z_0, z_1]$  and the system is ensemble approximately controllable between eigenstates. In this case, we cannot consider an embedding from  $[z_0, z_1]$  to  $\pi(\hat{\gamma}_j)$  because the same value of the parameter z corresponds to different intersections of eigenvalues of  $H(\cdot, z)$ . This is why we consider the following assumptions.

**Controllability properties** ASSUMPTION  $\tilde{A}_j$ . There exist a connected component  $\hat{\gamma}_j$  of  $\gamma_j$  such that

- $\hat{\gamma}_j$  is a one-dimensional submanifold of  $\mathbb{R}^3$  made of F-conical intersections and F-semi-conical intersections only;
- There exist  $(u_0, v_0) \in U$  and  $(u_1, v_1) \in U$  such that  $(u_0, v_0, z_0), (u_1, v_1, z_1) \in \hat{\gamma}_j$  are F-conical intersections for H;
- $\pi(\hat{\gamma}_j)$  is a  $C^{\infty}$  embedded curve of  $\mathbb{R}^2$  without self-intersections, which is contained in  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1})).$

Moreover the set  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  is pathwise connected.

**Theorem 2.2.6.** Consider a  $C^4$  map  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$ . Let assumption  $\tilde{A}_j$  be satisfied for every  $j \in \{1 \dots n\}$ . Then the equation  $i\frac{d\psi}{dt} = H(u, v, z)\psi$  is ensemble approximately controllable between eigenstates.

Method of control On figure 2.7, we give a smooth control path (u, v) (see Theorem 4.1.3 for more precisions) achieving transitions between two levels j and j + 1 which are uniform with respect to the parameter  $z \in [z_0, z_1]$ . The control path (u, v) enters in  $\pi(\gamma)$  at a point  $(u_0, v_0)$  such that  $(u_0, v_0, z_0)$  is a F-conical intersection of eigenvalues of  $H(u, v, z_0)$  between the levels j and j + 1, and exiting at a point  $(u_1, v_1)$  such that  $(u_1, v_1, z_1)$  is a F-conical intersection of eigenvalues of  $H(u, v, z_0)$  between the levels j and j + 1, and exiting at a point  $(u_1, v_1)$  such that  $(u_1, v_1, z_1)$  is a F-conical intersection of eigenvalues of  $H(u, v, z_1)$  between the levels j and j + 1. In order to guarantee that transitions are achieved for every  $z \in [z_0, z_1]$ , the control path has to pass an odd number of times through F-conical intersections of H(u, v, z), for every  $z \in [z_0, z_1]$ . Moreover the uniformity of the adiabatic approximation with respect to the parameter z is ensured when the control path is chosen such that we have  $z \in [z_0, z_1]$  for every z and t such that  $(u(t), v(t), z) \in \gamma$ . The error made in the adiabatic approximation for such a path of speed  $\epsilon$  has order  $\epsilon^{\frac{1}{3}}$ .

#### 2.2.2 Classification of the singularities of the non-mixing field

For a general Hamiltonian H depending on two real controls, the non-mixing curves between  $\lambda_{j-1}$  and  $\lambda_j$  for  $j \geq 2$  have been defined in [22] as the curves  $\gamma = (\gamma(t))_{t \in [0,1]}$  of  $\mathbb{R}^2$  along which  $\dot{\phi}_{j-1}(\gamma(t))$  is orthogonal to  $\phi_j(\gamma(t))$ , for every  $t \in [0, 1]$ . Then, by Theorem 2.1.1, the precision of the adiabatic approximation along such a curve is improved. In



Figure 2.7 – A curve of controls (u, v) achieving uniform transitions w.r.t.  $z \in [z_0, z_1]$  between two levels having F-semi-conical intersections

particular, the error of order  $\sqrt{\epsilon}$  in Equation (2.6) for a control path at a conical intersection is transformed into  $\epsilon$  along a non-mixing curve. This property has been used in [22] for a precise control of the Schödinger Equation with real Hamiltonians and an extension has been presented in [31] for complex Hamiltonians with three real controls.

In this section, we study the singularities of the non-mixing curves for two level systems, then for more general quantum system.

#### Non-mixing field for two-level systems and singularities in this case

For f in  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , let us study the dynamics of

$$i\frac{d\psi_{\epsilon}(t)}{dt} = H_f(u(\epsilon t), v(\epsilon t))\psi_{\epsilon}(t), \quad \psi_{\epsilon}(0) = \tilde{\psi}_0 \in \mathbb{C}^2,$$
(2.11)

where  $t \in [0, \frac{1}{\epsilon}]$  and  $\tilde{\psi}_0$  is independent of  $\epsilon$ .

Along a non-mixing curve  $\gamma$ , the error occuring in the adiabatic theorem is equal to 0, that is, the non-mixing curves can be followed at an arbitrary speed. We show that they are integral curves of a smooth vector field  $\chi(f)$ , namely the non-mixing field.

**Theorem 2.2.7.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ ,  $\chi(f)$  has three types of singularities, up to a  $C^{\infty}$  diffeomorphic coordinate change from  $\mathbb{R}^2$  to  $\mathbb{R}^2$ :

— Critical nodes (see Figure 2.8) at intersections of eigenvalues,



Figure 2.8 – A critical node

- Saddles and centers at points that are not intersections of eigenvalues.

By admissible transformations, generic one-parameter perturbation of a Hamiltonian can be transformed locally around eigenvalue intersections into:

$$H(u, v, z) = \begin{pmatrix} h_1(u, v, z)(z - u) & h_2(u, v, z)(z - v) \\ h_2(u, v, z)(z - v) & -h_1(u, v, z)(z - u) \end{pmatrix},$$

or

$$H(u, v, z) = \begin{pmatrix} h_1(u, v, z)(z - m(u)u) & h_2(u, v, z)(z + u + v^2) \\ h_2(u, v, z)(z + u + v^2) & -h_1(u, v, z)(z - m(u)u) \end{pmatrix},$$

where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  satisfy  $h_1(0) = h_2(0) = 1$ , and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $m(0) \notin \{-1, 0\}$ .

This property allows to study the bifurcations of singularities occuring in Theorem 2.2.7 when a parameter z varies. For every z in  $\mathbb{R}$ , let  $\chi_z(f)$  be the non-mixing field associated with  $f(\cdot, \cdot, z) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ .

**Theorem 2.2.8.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , for every  $z \in \mathbb{R}$ ,  $\chi_z(f)$  has the following singularities, up to a  $C^{\infty}$ -diffeomorphic coordinate change from  $\mathbb{R}^2$  to  $\mathbb{R}^2$ :

- Critical nodes at conical intersections of eigenvalues,
- Hyperbolic or elliptic semi-conical singularities at semi-conical intersections of eigenvalues (see Chapter 5 Section 5.2.2 for a precise definition, and the figures 5.3 and 5.2),
- Saddles, centers or cusps at points that are not intersections of eigenvalues.



Figure 2.9 – A hyperbolic semi-Figure 2.10 – An elliptic semiconical singularity. The non-conical conical singularity. The non-conical direction is (0, 1). The red curves direction is (0, 1). Every non-mixing are the only non-mixing curves curve passes through the singularity passing through the singularity in the non-conical direction. The inconical directions. The index is 0. dex is 2.

#### Non-mixing field for higher dimensional systems

For higher dimensional systems, the non-mixing curves have been studied in [22]. In the following, we consider  $H_0, H_1, H_2 \in S_n(\mathbb{R})$ , and we define

$$H(u, v) = H_0 + uH_1 + vH_2.$$

In this case, it has been shown in [22] that the non-mixing curves are the integral curves of a line field on  $\mathbb{R}^2$ , defined up to a sign, for  $(u, v) \in \mathbb{R}^2$ , by

$$\chi_{j-1,j}(u,v) = \begin{pmatrix} -\langle H_2\phi_{j-1}(u,v), \phi_j(u,v) \rangle \\ \langle H_1\phi_{j-1}(u,v), \phi_j(u,v) \rangle \end{pmatrix},$$

where  $\langle, \rangle$  is the natural Euclidean scalar product on  $\mathbb{R}^n$ . Notice that  $\chi_{j-1,j}$  is defined up to a sign at the points  $(u, v) \in \mathbb{R}^2$  such that  $\lambda_{j-1}(u, v) \neq \lambda_j(u, v)$ . Along these curves the adiabatic approximation has a higher precision, that is the error has an order  $\epsilon$  along a control path of speed  $\epsilon$ , even when it passes through crossings of eigenvalues. Contrarily to what one could think, the non-mixing curves for a *n*-level Hamiltonian with  $n \geq 3$  do not correspond locally to those of a reduced two level Hamiltonian as defined in Section 2.1. They involve a more complex structure that has interesting topological properties. Indeed, we are going to see that the non-mixing field may have both singularities of integer index which are generic for a vector field of  $\mathbb{R}^2$  and singularities of half-integer index which are generic for a line field of  $\mathbb{R}^2$ , while the space of vector fields and of line fields are used to be



Figure 2.11 – Combination of integer and half-integer index singularities in nematic crystals, picture taken from [8]

endowed with different topologies so that at first view such a combination of singularities seems to be impossible generically. These kind of structures have already been observed in nematic liquid crystals (see [26] for a general review on nematic crystals and Figure 2.11 for an illustration of such a behaviour). Further investigations on this type of structures will be made in future works.

We aim at classifying the singularities of the line field  $\chi_{j-1,j}$ , and we first need to recap some topological tools.

**Basic facts about line-fields** Let (M, g) be a 2-dimensional Riemannian manifold, and K be a closed subset of M. A line field L on M is a section of  $PT(M \setminus K)$ . The set K is the set of singularities of L. As it has been proved in [21], any line field with singularities can be realized as the bisection for the angle defined for the Riemannian metric g of two vector fields X and Y on M. When a line field L is a bisection of the vector fields X and Y, we say that L is defined as a *proto-line-field*. The zeros of X and Ybecome singularities of the associated proto-line-field. With this definition, the Whitney topology on pairs of vector fields on M defines a topology on line fields on M. It has been proved in [21] that generically, a proto-line-field has only Darbouxian singularities, that is, its integral manifolds are homeomorphic to those of the proto-line-fields defined by  $(X_L, Y_L)$ ,  $(X_M, Y_M)$  and  $(X_S, Y_S)$  as follows:

— The Lemon proto-line-field (see Figure 5.15) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$  defined by

$$X_L(x,y) = \begin{pmatrix} x+y\\ y-x \end{pmatrix}, \quad Y_L(x,y) = \begin{pmatrix} 1\\ 1 \end{pmatrix}.$$

— The Monstar proto-line-field (see Figure 5.16) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$ 

defined by

$$X_M(x,y) = \begin{pmatrix} x \\ 3y \end{pmatrix}, \quad Y_M(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$$

— The *Star proto-line-field* (see Figure 5.17) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$  defined by

$$X_S(x,y) = \begin{pmatrix} x \\ -y \end{pmatrix}, \quad Y_S(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}.$$



Figure 2.12 – The Lemon singularity, of index  $\frac{1}{2}$ .



Figure 2.13 – The Monstar singularity, of index  $\frac{1}{2}$ .



Figure 2.14 – The Star singularity, of index  $-\frac{1}{2}$ .

Non-mixing curves for *n*-level systems at conical intersections between the levels j - 1 and j In a neighborhood of a conical intersection  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  between the levels j - 1 and j,

— There exists a smooth choice of the sign of the eigenvectors  $\phi_{j-1}(u, v)$  and  $\phi_j(u, v)$ such that  $\chi_{j-1,j}$  defines a  $C^{\infty}$  vector field in a punctured neighborhood of  $(\bar{u}, \bar{v})$ ,

- The integral curves of  $\chi_{j-1,j}$  are  $C^{\infty}$  and the eigenvectors  $\phi_{j-1}$  and  $\phi_j$  are  $C^{\infty}$  along them,
- For every direction  $\eta$  of  $\mathbb{R}^2$ , there exists an integral curve  $\gamma : [0,1] \to \mathbb{R}^2$  of  $\chi_{j-1,j}$ such that  $\gamma(1) = (\bar{u}, \bar{v}), \lim_{t \to 1^-} \frac{\dot{\gamma}(t)}{||\dot{\gamma}(t)||} = \eta$ .

We say that the singularity of  $\chi_{j,j-1}$  has type (**N**), for node.

Non-mixing curves for *n*-level systems at conical intersections between the levels j and j+1 (respectively j-2 and j-1) In order to complete the classification of the singularities of the non-mixing field at intersections of eigenvalues, we focus on the singularities of  $\chi_{j-1,j}$  at conical intersections between the levels j and j+1.

Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be a conical intersection between  $\lambda_j$  and  $\lambda_{j+1}$ . We introduce the following condition:

**Condition (C)**: The vectors  $P_{j,j+1}H_1\phi_{j-1}(u,v)$  and  $P_{j,j+1}H_2\phi_{j-1}(u,v)$  are not colinear.

We prove the following theorem.

**Theorem 2.2.9.** Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be such that  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$  is a conical intersection. Assume that Condition (C) is satisfied at  $(\bar{u}, \bar{v})$ . Then  $\chi_{j-1,j}$  has a Darbouxian singularity at  $(\bar{u}, \bar{v})$ .

Corollary 2.2.10. Let  $E_1, E_2, E_3 \in \mathbb{R}$  such that  $E_1 < E_2 < E_3$  and  $H(u, v) = \begin{pmatrix} E_1 & u & 0 \\ u & E_2 & v \\ 0 & v & E_3 \end{pmatrix}$ .

Then  $\chi_{1,2}$  (respectively  $\chi_{2,3}$ ) has Darbouxian singularities at the points  $(u, v) \in \mathbb{R}^2$  such that  $\lambda_2(u, v) = \lambda_3(u, v)$  (respectively,  $\lambda_1(u, v) = \lambda_2(u, v)$ ) that are

$$(u,v) = (\pm\sqrt{(E_1 - E_3)(E_2 - E_3)}, 0) \ (respectively \ (0, \pm\sqrt{(E_1 - E_3)(E_1 - E_2)})).$$

The plot of  $\chi_{1,2}$  is made on Figure 5.18. We notice that there is a singularity of Type **(N)** at conical intersections between  $\lambda_1$  and  $\lambda_2$ , while there is a Star singularity at conical intersections between  $\lambda_2$  and  $\lambda_3$ .

#### 2.2.3 Towards ensemble control with a single input

An important issue of quantum control is to design explicit control laws for the problem of the single input bilinear Schrödinger equation, that is

$$i\frac{d\psi}{dt} = \left(H_0 + uH_1\right)\psi\tag{2.12}$$

where  $\psi$  belongs to the unit sphere in a Hilbert space  $\mathcal{H}$ ,  $H_0$  is a self adjoint operator representing a drift term called *free Hamiltonian*,  $H_1$  is a self-adjoint operator representing



Figure 2.15 – Non mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP of Corollary 2.2.10.

the control coupling and  $u: [0,T] \to \mathbb{R}, T > 0$ . Important theoretical results of controllability have been proved with different techniques (see [7, 12, 17] and references therein). For the problem with two or more inputs, adiabatic methods are a nowadays classical way to get an explicit expression of the controls and can be used under geometric conditions on the spectrum of the controlled Hamiltonian (see the articles [10, 22, 62] and references therein), and our results about these methods are presented in Section 2.2.1 of the Introduction and are developed in Chapter 3. However, these methods are effective for inputs of dimension at least 2. Our aim is then to extend a single-input bilinear Schrödinger equation into a two-inputs bilinear Schrödinger equation in the same spirit as the Lieextensions introduced by Sussmann and Liu in [66] and [77], then to apply the well-known adiabatic techniques to the extended system. The first step of this procedure is well known by physicists and it is called the rotating-wave approximation (RWA, for short). It is a decoupling approximation to get rid of highly oscillating terms when the system is driven by a real control. This approximation is based on a first-order averaging procedure (see [71, 77, 66, 23] for more informations about averaging of dynamical systems). This approximation is known to work well for a small detuning from the resonance frequency and a small amplitude. For a review of the RWA and its limitations see [37] and [44, 45, 51]. In [24], the mathematical framework has been set for infinite-dimensional quantum systems, formalizing what physicists call *Generalized Rabi oscillations* and showing that the RWA

is valid for a large class of quantum systems. The adiabatic and RWA involve different time scales, and it is natural to ask whether or not they can be used in cascade. The aim of Chapter 2.2.3 is to show the validity of such an approximation under a certain condition on the time scales involved in the dynamics, using an averaging procedure. Then the well-known results of adiabatic theory (see [22, 18, 78]) can be applied in order to get transitions between the eigenstates of the free Hamiltonian. It leads us to design control laws achieving the inversion of a Spin- $\frac{1}{2}$  particule and population transfers in the STIRAP process that are robust with respect to inhomogeneities of the amplitude of the control input (see [81, 10] and Chapter 3). As a byproduct of the use of a control oscillating with a small frequency detuning, the proposed method is not expected to be robust with respect to inhomogeneities of the resonance frequencies, that is inhomogeneities of the drift term  $H_0$ .

For such systems, when  $\mathcal{H}$  is finite dimensional, we use controls under the form

$$u_{\epsilon}(t) = 2\epsilon^{\alpha} \sum_{(j,k)\in\{1\dots N\}^2} v_{jk}(\epsilon^{\alpha+1}t)\cos(\beta_{jk}t + \frac{1}{\epsilon}\phi_{jk}(\epsilon^{\alpha+1}t)),$$

where, for every  $j, k \in \{1 \dots N\}$ ,  $v_{jk}, \phi_{jk} \in C^{\infty}([0, 1], \mathbb{R})$  such that  $\phi_{jk}(0) = 0$  and  $\beta_{jk} \in \mathbb{R}$ ,  $N \in \mathbb{N} \setminus \{0\}$  and we study the asymptotic behaviour of the associated quantum system (2.12) when  $\epsilon \to 0$  on the interval of time  $[0, \frac{1}{\epsilon^{\alpha+1}}]$ . Assuming  $\alpha > 1$ , by a suitable choice of  $(v_{kj}, \phi_{kj})$ , and  $\beta_{kj}$  for every k, j, we show a general approximation result (see the theorems 6.1.3 and 6.2.11) that allows to explicitly design control laws that are robust with respect to inhomogeneities of the amplitude of the control input.

This result has led us to prove the following results. Let U be an open interval of  $\mathbb{R}$  containing 0.

**Theorem 2.2.11.** Let  $E \in \mathbb{R} \setminus \{0\}$ . The equation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E & \delta u\\ \delta u & -E \end{pmatrix} \psi$$
(2.13)

is approximately ensemble controllable between the eigenstates of  $H_0 = \begin{pmatrix} E & 0 \\ 0 & -E \end{pmatrix}$  uniformly with respect to  $\delta \in [a, b] \subset (0, +\infty)$  and  $u \in U$ .

**Theorem 2.2.12.** Let  $E_1, E_2, E_3 \in \mathbb{R}$  such that  $|E_k - E_q| \neq |E_j - E_l|$ , for every  $k, q, j, l \in \{1...3\}$  such that  $(k, q) \notin \{(j, l), (l, j)\}$ . The equation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E_1 & \delta_1 u & 0\\ \delta_1 u & E_2 & \delta_2 u\\ 0 & \delta_2 u & E_3 \end{pmatrix} \psi$$
(2.14)

is ensemble controllable between eigenstates of  $H_0 = \begin{pmatrix} E_1 & 0 & 0 \\ 0 & E_2 & 0 \\ 0 & 0 & E_3 \end{pmatrix} \psi$  uniformly with respect to  $\delta_1 \in [a_1, b_1] \subset (0, +\infty)$  and  $\delta_2 \in [a_2, b_2] \subset (0, +\infty)$  and  $u \in U$ .

An open question is the ensemble controllability w.r.t.  $\delta \in [a, b] \subset (0, +\infty)$  of the equation

$$i\frac{d\psi}{dt} = (H_0 + \delta u H_1)\psi,$$

where  $H_0, H_1 \in S_n(\mathbb{R})$ , assuming that  $H_0, H_1$  are such that Equation (2.12) is controllable. Of course a similar problem in infinite dimension would be interesting to study. Another topic of study would be to understand how to control robustly with a single input a system where the drift term has uncertainities.

# 2.2. Topics of investigation

# Chapter 3

# Ensemble control of quantum systems with two controls: conical case

In this Chapter we discuss how to control a parameter-dependent family of quantum systems. Our technique is based on adiabatic approximation theory and on the presence of curves of conical eigenvalue intersections of the controlled Hamiltonian. As particular cases, we recover chirped pulses for two-level quantum systems and counterintuitive solutions for three-level stimulated Raman adiabatic passages. The proposed technique works for systems evolving both in finite-dimensional and infinite-dimensional Hilbert spaces. We show that the assumptions guaranteeing ensemble controllability are structurally stable with respect to perturbations of the parameterized family of systems. The Chapter is mainly taken from [10].

#### Contents

3.1	Introduction	<b>50</b>
<b>3.2</b>	Basic definitions and statement of the main results in the	
	finite-dimensional case	50
<b>3.3</b>	Proof of the ensemble controllability result in the finite-	
	dimensional case	<b>54</b>
<b>3.4</b>	3.4 Example 1: Two-level system driven by a chirped pulse	
3.5	3.5 Permutations	
3.6	<b>3.6</b> Genericity	
3.7	3.7 Multidimensional set of parameters	
	3.7.1 Chirped pulses for two-level systems with two parameters $\ldots$	65
	3.7.2 Example 2: STIRAP	65
3.8 Extension to the infinite-dimensional case		68
	3.8.1 Example 3: Eberly–Law-like models	70

### 3.1 Introduction

The chapter is organized as follows: In Section 3.2 we set the ensemble controllability problem and we state our sufficient conditions for approximate ensemble controllability in finite-dimensional spaces. The proof of the sufficiency of such conditions in given in Section 3.3. The proof of the technical result on the regularity of eigenpairs is postponed to the appendix. In Section 3.4 we apply the general result to the case of two-level systems, recovering the classical results on chirped pulses. The extension to permutations within a basis of eigenvectors is studied in Section 3.5. The genericity of the conditions appearing in the sufficient conditions for approximate ensemble controllability is discussed in Section 3.6. In Section 3.7 we consider the case in which the Hamiltonian depends on more than one parameter and we illustrate our results on the STIRAP process. Finally, in Section 3.8, we discuss the extensions to the case of infinite-dimensional spaces, presenting as an example a version of the classical Eberly–Law model.

# 3.2 Basic definitions and statement of the main results in the finite-dimensional case

For every  $n \in N$ , let  $[\![1, n]\!]$  denote the set of integers  $\{1, \ldots, n\}$ . Let **U** be an open connected subset of  $\mathbb{R}^d$ ,  $d \geq 2$ .

We consider the controlled Schrödinger equation in  $\mathbb{C}^N$ ,  $N \in \mathbb{N}$ ,

$$i\dot{\psi} = H^z(u(t))\psi \tag{3.1}$$

where  $u : \mathbb{R} \to \mathbf{U}$  is a  $L^{\infty}$  map. Here z is a time-independent parameter belonging to a compact interval  $[z_0, z_1] \subset \mathbb{R}$ . Each matrix  $H^z(u)$  belongs to the set  $\operatorname{Herm}(N)$  of  $N \times N$  Hermitian matrices. The map  $(z, u) \mapsto H^z(u)$  is sufficiently regular, as it will be specified later on.

Denote the spectrum of  $H^{z}(u)$  by  $(\lambda_{j}^{z}(u))_{j=1}^{N}$  where  $j \mapsto \lambda_{j}^{z}(u)$  is the nondecreasing sequence of eigenvalues of  $H^{z}(u)$  repeated according to their multiplicities. We also write  $(\phi_{1}^{z}(u), \ldots, \phi_{N}^{z}(u))$  to denote an orthonormal basis of associated eigenvectors.

**Definition 3.2.1.** We say that system (3.1) is ensemble approximately controllable between eigenstates if for every  $\varepsilon > 0$ ,  $j, k \in [\![1, N]\!]$  and  $u_0, u_1 \in \mathbf{U}$  such that  $\lambda_j^z(u_0)$  and  $\lambda_k^z(u_1)$  are simple for every  $z \in [z_0, z_1]$ , there exists a control  $u(\cdot) : [0, T] \to \mathbf{U}$  such that for every  $z \in [z_0, z_1]$  the solution of (3.1) with initial condition  $\psi^z(0) = \phi_j^z(u_0)$  satisfies  $\|\psi^z(T) - e^{i\theta}\phi_k^z(u_1)\| < \varepsilon$  for some  $\theta \in \mathbb{R}$  (possibly depending on z and  $\varepsilon$ ). **Remark 3.2.2.** The typical case of interest is when  $u = 0 \in \mathbf{U}$  represents an isolated (i.e., uncontrolled) system (the so-called drift Hamiltonian) and one seeks to steer  $\phi_j^z(0)$  towards  $\phi_k^z(0)$  for some  $j, k \in [\![1, N]\!]$ .

**Definition 3.2.3.** Let us fix  $z \in [z_0, z_1]$ . We say that  $\bar{u} \in \mathbf{U}$  is a conical intersection between  $\lambda_j^z$  and  $\lambda_{j+1}^z$  if  $\lambda_{j-1}^z(\bar{u}) < \lambda_j^z(\bar{u}) = \lambda_{j+1}^z(\bar{u}) < \lambda_{j+2}^z(\bar{u})$  and there exists c > 0 such that for every v in a neighborhood of  $\bar{u}$  in  $\mathbf{U}$ , we have

$$\|\lambda_{j}^{z}(v) - \lambda_{j+1}^{z}(v)\| \ge c\|\bar{u} - v\|.$$

(As example of conical intersection is shown in Figure 3.1.)



Figure 3.1 – An example of conical intersection in the case d = N = 2.

For  $j \in \{1 \dots n\}$ , let us denote by  $\gamma_j$  the set  $\{(u, z) \in U \times [z_0, z_1] \mid \lambda_j^z(u) = \lambda_{j+1}^z(u)\}$ , where by convention  $\gamma_0 = \gamma_n = \emptyset$ . Denote the projection of  $(u, z) \in \mathbb{R}^{d+1}$  onto the *u*-component by  $\pi$ .

Then  $\pi(\gamma_j)$  is the projection of the set  $\{(u, z) \in \mathbf{U} \times [z_0, z_1] \mid \lambda_j^z(u) = \lambda_{j+1}^z(u)\}$  onto the *u*-component, that is,

$$\begin{aligned} \pi(\gamma_0) &= \emptyset, \\ \pi(\gamma_j) &= \{ u \in \mathbf{U} \mid \exists z \in [z_0, z_1] \text{ such that } \lambda_j^z(u) = \lambda_{j+1}^z(u) \}, \quad j \in [\![1, N-1]\!], \\ \pi(\gamma_N) &= \emptyset. \end{aligned}$$

Assumption  $A_j$ . There exist a connected component  $\hat{\gamma}_j$  of  $\gamma_j$  and a map  $\beta_j : [z_0, z_1] \to \mathbf{U}$ such that  $\beta_j$  is a  $\mathcal{C}^3$ embedding and 3.2. Basic definitions and statement of the main results in the finite-dimensional case



Figure 3.2 – The component  $\hat{\gamma}_j$  is such that Assumption  $A_j$  is satisfied

- $-\pi(\hat{\gamma}_j) = \beta_j([z_0, z_1])$
- $\pi(\hat{\gamma}_j)$  is contained in  $\mathbf{U} \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1}))$
- For every  $z \in [z_0, z_1]$ ,  $\lambda_j^z$  and  $\lambda_{j+1}^z$  have a unique intersection on  $\pi(\hat{\gamma}_j)$ , which is conical and occurs at  $\beta_j(z)$ .

Moreover the set  $\mathbf{U} \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  is pathwise connected.

**Remark 3.2.4.** The assumption that  $\beta_j$  is an embedding between the set of parameters  $[z_0, z_1]$  and  $\pi(\hat{\gamma}_j)$  is used here for simplicity and can be relaxed. This will be done in the general context of multi-dimensional sets of parameters in Section 3.7 (see Assumption  $A_i^*$ ).

The structural stability and the genericity of Assumption  $A_j$  are discussed in Section 3.6. In the following theorem we show that the curve  $\pi(\hat{\gamma}_j)$  appearing in Assumption  $A_j$  can be used to induce an adiabatic transition between the *j*-th and the (j + 1)-th eigenstate for every *z*.

**Theorem 3.2.5.** Consider a  $C^3$  map  $[z_0, z_1] \times \mathbf{U} \ni (z, u) \mapsto H^z(u) \in \operatorname{Herm}(N)$ . Let  $j \in [\![1, N-1]\!]$  be such that Assumption  $A_j$  is satisfied. Take  $u_0, u_1 \in \mathbf{U} \setminus (\pi(\gamma_{j-1}) \cup \mathcal{U})$ 



Figure 3.3 – A control  $u(\cdot)$  as in the statement of Theorem 3.2.5.

 $\pi(\gamma_j) \cup \pi(\gamma_{j+1})$  and consider a  $\mathcal{C}^3$  path  $u(\cdot) : [0,1] \to \mathbf{U}$  satisfying  $u(0) = u_0$ ,  $u(1) = u_1$ , and such that  $u|_{[t_0,t_1]}$  is a reparameterization of  $\beta_j$  for some  $0 < t_0 < t_1 < 1$  such that  $u(t) \notin \pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1})$  for every  $t \in [0,1] \setminus [t_0,t_1]$ . (See Figure 3.3.) Assume, moreover, that  $\dot{u}(t) \neq 0$  for every  $t \in [0,1]$ .

Then there exists a constant C > 0 such that for every  $z \in [z_0, z_1]$  and  $\varepsilon > 0$  the solutions  $\psi_{\varepsilon}^{z,\pm}$  of

$$i\dot{\psi}^{z,\pm}_{\varepsilon}(t) = H^z(u(\varepsilon t))\psi^{z,\pm}_{\varepsilon}(t)$$

with initial conditions

$$\psi_{\varepsilon}^{z,+}(0) = \phi_j^z(u_0), \qquad \psi_{\varepsilon}^{z,-}(0) = \phi_{j+1}^z(u_0)$$
(3.2)

satisfy

$$\|\psi_{\varepsilon}^{z,+}(1/\varepsilon) - e^{i\theta^{+}}\phi_{j+1}^{z}(u_{1})\| \le C\sqrt{\varepsilon}, \qquad \|\psi_{\varepsilon}^{z,-}(1/\varepsilon) - e^{i\theta^{-}}\phi_{j}^{z}(u_{1})\| \le C\sqrt{\varepsilon}, \qquad (3.3)$$

for some  $\theta^{\pm} \in \mathbb{R}$ .

**Remark 3.2.6.** By unitarity of the evolution, if in the statement of Theorem 3.2.5 we replace the initial conditions (3.2) by

$$\|\psi_{\varepsilon}^{z,+}(0) - \phi_{j}^{z}(u_{0})\| \le c, \qquad \|\psi_{\varepsilon}^{z,-}(0) - \phi_{j+1}^{z}(u_{0})\| \le c,$$

then the conclusion (3.3) becomes

$$\|\psi_{\varepsilon}^{z,+}(1/\varepsilon) - e^{i\theta^+}\phi_{j+1}^z(u_1)\| \le C\sqrt{\varepsilon} + c, \qquad \|\psi_{\varepsilon}^{z,-}(1/\varepsilon) - e^{i\theta^-}\phi_j^z(u_1)\| \le C\sqrt{\varepsilon} + c.$$

**Corollary 3.2.7.** Consider a  $C^3$  map  $[z_0, z_1] \times \mathbf{U} \ni (z, u) \mapsto H^z(u) \in \text{Herm}(N)$ . Let Assumption  $A_j$  be satisfied for every  $j \in [\![1, N-1]\!]$ . Then (3.1) is ensemble approximately controllable between eigenstates.


Figure 3.4 – A control realizing an ensemble transition between  $\phi_1^z(u_0)$  and  $\phi_4^z(u_0)$ .

The corollary follows from an iterated application of Theorem 3.2.5 and its proof works by constructing a  $C^3$  control such that  $u(0) = u_0$  and  $u(T) = u_1$  (where  $u_0$  and  $u_1$  are as in Definition 3.2.1) and going through the curves  $\pi(\hat{\gamma}_j)$  as in Figure 3.4. Starting from the second iteration of the application of Theorem 3.2.5, we use Remark 3.2.6 with c of order  $\sqrt{\varepsilon}$  to guarantee that the adiabatic approximation after each iteration remains of order  $\sqrt{\varepsilon}$ .

**Remark 3.2.8.** The proof of Theorem 3.2.5 is based on a uniform adiabatic theorem, which is recalled in next section (Theorem 3.3.1). Actually, under the additional hypothesis that the curves  $\pi(\hat{\gamma}_j)$  are non-mixing for every  $z \in [z_0, z_1]$ , in the sense of [22, Section V], one can replace the factor  $\sqrt{\varepsilon}$  in (3.3) and (3.4) by  $\varepsilon$ .

# 3.3 Proof of the ensemble controllability result in the finite-dimensional case

**Theorem 3.3.1.** Consider a  $C^3$  map  $[z_0, z_1] \times \mathbf{U} \ni (z, u) \mapsto H^z(u) \in \operatorname{Herm}(N)$ . Let  $u : [0,1] \to \mathbf{U}$  be a  $C^2$  control. For every  $z \in [z_0, z_1]$  and  $t \in [0,1]$ , let  $\Lambda_1^z(t), \ldots, \Lambda_N^z(t)$  be the eigenvalues of  $H^z(u(t))$  repeated according to their multiplicities and denote by  $(\Phi_1^z(t), \ldots, \Phi_N^z(t))$  an orthonormal basis of associated eigenvectors.

Assume that for every  $z \in [z_0, z_1]$  and every  $j \in [\![1, N]\!]$ ,

$$(\Lambda_i^z, \Phi_i^z) \in \mathcal{C}^2([0, 1], \mathbb{R} \times \mathbb{C}^N).$$

Let  $j \in [\![1, N-1]\!]$  and assume that for every  $z \in [z_0, z_1]$ , there exists  $\tau \in (0, 1)$  such that  $\Lambda_i^z(t)$  and  $\Lambda_{i+1}^z(t)$  are simple for every  $t \in [0, 1] \setminus \{\tau\}$  and

$$\Lambda_j^z(\tau) = \Lambda_{j+1}^z(\tau) \neq \Lambda_m^z(\tau) \text{ for } m \neq j, j+1, \qquad \left. \frac{d}{dt} \right|_{t=\tau} \Lambda_j^z(t) \neq \left. \frac{d}{dt} \right|_{t=\tau} \Lambda_{j+1}^z(t).$$

Then there exists C > 0 such that for every  $z \in [z_0, z_1]$ , and every  $\varepsilon > 0$  the solution  $\psi = \psi_{\varepsilon}^z$  of the equation

$$i\dot{\psi}(t) = H^z(u(\varepsilon t))\psi(t)$$

with initial condition  $\psi(0) = \Phi_i^z(0)$  satisfies

$$\|\psi(1/\varepsilon) - e^{i\theta}\Phi_j^z(1)\| \le C\sqrt{\varepsilon} \tag{3.4}$$

for some  $\theta \in \mathbb{R}$ .

The proof of this theorem can be obtained by applying the classical adiabatic theorem for every z, using the continuity of the corresponding constant C(z) and the compactness of the interval  $[z_0, z_1]$  (see for instance [52, Theorem 4] and [78, Theorem 1.2]).

In order to guarantee the regularity of eigenvalues and eigenvectors along a regular but not necessarily analytic path in the domain of admissible controls, we are going to apply the following result, whose proof is given in the appendix. (For a result in the same spirit in the infinite-dimensional setting, see [30].)

**Lemma 3.3.2.** Let I be an interval in  $\mathbb{R}$  and take  $H \in \mathcal{C}^{k+1}(I, \operatorname{Herm}(N)), k \geq 0$ . For every  $t \in I$ , let  $\lambda_1(t) \leq \cdots \leq \lambda_N(t)$  be the eigenvalues of H(t), repeated according to their multiplicities. Assume that for every  $\overline{t} \in I$  and every  $j \in [[1, N - 1]]$  such that  $\lambda_j(\overline{t}) = \lambda_{j+1}(\overline{t})$ , then  $\lambda_h(\overline{t}) \neq \lambda_j(\overline{t})$  for  $h \neq j, j+1$  and there exist c > 0 and a neighborhood  $\overline{I}$  of  $\overline{t}$  in I such that

$$\lambda_{j+1}(t) - \lambda_j(t) \ge c|t - \bar{t}|, \qquad \text{for every } t \in \bar{I}.$$
(3.5)

Then there exist  $\Lambda_1, \ldots, \Lambda_N \in \mathcal{C}^{k+1}(I, \mathbb{R})$  and  $\Phi_1, \ldots, \Phi_N \in \mathcal{C}^k(I, \mathbb{C}^N)$  such that, for every  $t \in I, \Lambda_1(t), \ldots, \Lambda_N(t)$  are the eigenvalues of H(t) repeated according to their multiplicities and  $(\Phi_1(t), \ldots, \Phi_N(t))$  is an orthonormal basis of corresponding eigenvectors. Moreover,  $\Phi_1, \ldots, \Phi_n$  are  $\mathcal{C}^{k+1}(I, \mathbb{C}^N)$  if all eigenvalues of H are simple along I.

Proof of Theorem 3.2.5. We are going to apply Theorem 3.3.1 on the path  $u(\cdot)$ . Notice that for every z the curve  $u(\cdot)$  passes through exactly one point  $\bar{u}^z$  such that  $\lambda_j^z(\bar{u}^z) = \lambda_{j+1}^z(\bar{u}^z)$ and through no point  $v \in \mathbf{U}$  where either  $\lambda_{j-1}^z(v) = \lambda_j^z(v)$  or  $\lambda_{j+1}^z(v) = \lambda_{j+2}^z(v)$ . Denote by  $t_z$  the time such that  $u(t_z) = \bar{u}^z$ .

By Lemma 3.3.2 applied to the map  $t \mapsto H(u(t))$ , we can assume that

$$[0,1] \ni t \mapsto (\Lambda_j^z(t), \Phi_j^z(t)) = \begin{cases} (\lambda_j^z(u(t)), \phi_j^z(u(t))) & \text{if } t < t_z \\ (\lambda_{j+1}^z(u(t)), \phi_{j+1}^z(u(t))) & \text{if } t \ge t_z \end{cases}$$

is  $C^2$  for every  $j \in [\![1, N]\!]$  and every  $z \in [z_0, z_1]$ . The application of Theorem 3.3.1 then concludes the proof of Theorem 3.2.5.

**Remark 3.3.3.** In Theorem 3.2.5 the control  $u(\cdot)$  is assumed to be  $C^3$ . The  $C^3$  regularity could actually be relaxed. The same proof would work, for instance, by assuming  $u(\cdot)$  to be continuous,  $C^3$  in a neighborhood of  $\pi(\hat{\gamma}_i)$  and piecewise  $C^2$  elsewhere.

# 3.4 Example 1: Two-level system driven by a chirped pulse

A classical example of ensemble control by adiabatic evolution is the famous chirped pulse used in two-level systems. It is used, for instance, for controlling via magnetic fields an ensemble of spin systems with slightly different parameters. We illustrate chirped pulses on the following model

$$i\frac{d}{dt} \begin{pmatrix} \psi_1^z \\ \psi_2^z \end{pmatrix} = \begin{pmatrix} E+z & \Omega(t) \\ \Omega^*(t) & -E-z \end{pmatrix} \begin{pmatrix} \psi_1^z \\ \psi_2^z \end{pmatrix},$$
(3.6)

where E > 0 is fixed,  $\Omega(\cdot) \in L^{\infty}([0,T],\mathbb{C})$  is the control,  $\Omega^*(t)$  denotes its complex conjugate, and z is a real parameter that can vary in a fixed range  $z \in [z_0, z_1] \subset (-E, \infty)$ . The quantity E + z is called the *proper frequency* of system (3.6).

Chirped pulses are well studied in the literature (see [27, 74, 62]). We show here below how they naturally show up in the general framework proposed in this paper. Consider the following ensemble approximate controllability problem for (3.6): we want to construct explicitly a control steering, for every value of  $z \in [z_0, z_1]$ , the eigenstate  $\phi_1^z(0) = (1, 0)$  to  $\phi_2^z(0) = (0, 1)$  of the Hamiltonian in (3.6) corresponding to  $\Omega = 0$ .

Theorem 3.2.5 and Corollary 3.2.7 do not directly apply to system (3.6) since the eigenvalues of the Hamiltonian are simple for every value of  $z \in [z_0, z_1]$  and  $\Omega \in \mathbb{C}$ .

Hence we perform a suitable time-dependent change of variables (to recast the system in interaction picture). In order to do so, we consider controls having the form

$$\Omega(t) = u_1(t)e^{-i(2Et + \Delta(t))},$$
(3.7)

where  $u_1(\cdot)$  and  $\Delta(\cdot)$  are real-valued, and we set

$$h^{z}(t) = \begin{pmatrix} E + z & u_{1}(t)e^{-i(2Et + \Delta(t))} \\ u_{1}(t)e^{i(2Et + \Delta(t))} & -E - z \end{pmatrix}.$$

Let us apply the time-dependent unitary change of variable  $\psi^z(t) = U(t)\Phi^z(t)$  with

$$U(t) = \left(\begin{array}{cc} e^{-iEt} & 0\\ 0 & e^{i(Et+\Delta(t))} \end{array}\right).$$

Notice that the transformation U(t) preserves (up to phases) the two eigenstates  $\phi_1^z(0) = (1,0)$  and  $\phi_2^z(0) = (0,1)$ .

Then  $\Phi^z$  satisfies the equation  $i\frac{d}{dt}\Phi^z = \mathcal{H}^z(t)\Phi^z$  with

$$\mathcal{H}^{z}(t) = U^{-1}(t)h^{z}(t)U(t) - iU^{-1}(t)\frac{dU}{dt}(t) = \left(\begin{array}{cc} z & u_{1}(t) \\ u_{1}(t) & -z + \frac{d\Delta}{dt}(t) \end{array}\right).$$

Setting  $u_2(t) = \frac{d\Delta}{dt}(t)$ , we are then left to control the family of systems

$$i\frac{d}{dt}\begin{pmatrix} \Phi_1^z\\ \Phi_2^z \end{pmatrix} = \begin{pmatrix} z & u_1(t)\\ u_1(t) & -z + u_2(t) \end{pmatrix} \begin{pmatrix} \Phi_1^z\\ \Phi_2^z \end{pmatrix}.$$
(3.8)

We apply now Theorem 3.2.5 to system (3.8). To this purpose we have to check that Assumption  $A_1$  is satisfied. Set

$$H^{z}(u_{1}, u_{2}) = \left(\begin{array}{cc} z & u_{1} \\ u_{1} & -z + u_{2} \end{array}\right)$$

The eigenvalues of  $H^{z}(u_1, u_2)$  are

$$\lambda_1^z(u_1, u_2) = \frac{u_2 - \sqrt{(2z - u_2)^2 + 4u_1^2}}{2}, \quad \lambda_2^z(u_1, u_2) = \frac{u_2 + \sqrt{(2z - u_2)^2 + 4u_1^2}}{2},$$

For any fixed  $z \in [z_0, z_1]$ ,  $H^z$  has a unique eigenvalue intersection, which is conical and takes place at

$$(u_1, u_2) = (0, 2z).$$

In other words for each point of the set

$$\pi(\gamma_1) = \{(0, 2z) \in \mathbb{R}^2 \mid z \in [z_0, z_1]\} = \{0\} \times [2z_0, 2z_1]$$

there is exactly one system having an eigenvalue intersection at such a point, which is, moreover, conical. Assumption  $A_1$  is then satisfied.

By applying Corollary 3.2.7 to system (3.8) we deduce the following result.

**Proposition 3.4.1.** System (3.8) is ensemble approximately controllable between eigenstates.

The controls used to achieve the controllability stated in Proposition 3.4.1 can be taken continuous,  $C^3$  in a neighborhood of  $\pi(\gamma_1)$  and piecewise  $C^2$  elsewhere (see Remark 3.3.3). Notice that the same regularity holds for  $\Omega$ , according to (3.7).

As a consequence of Proposition 3.4.1 we can steer (up to phases) system (3.8) (simultaneously with respect to  $z \in [z_0, z_1]$ ) from (1,0) to an arbitrary neighborhood of (0,1), which are eigenstates of  $\mathcal{H}^z(0, u_{20})$  for every  $u_{20} \in \mathbb{R} \setminus [2z_0, 2z_1]$ . Now, since U(t)



Figure 3.5 – The control  $u(\cdot)$ .

preserves (up to phases) (1,0) and (0,1), we can conclude that system (3.6) can be steered simultaneously by adiabatic control from (1,0) to (0,1) for every  $z \in [z_0, z_1]$ . Notice that the control  $u = (0, u_{20})$  corresponds to  $\Omega = 0$  in the original system, independently of  $u_{20}$ .

We now construct explicitly the adiabatic control realizing the transition. We start by doing this for (3.8) and then we translate the result for (3.6).

Let  $u_{21}$  be in the connected component of  $\mathbb{R} \setminus [2z_0, 2z_1]$  not containing  $u_{20}$  (see Figure 3.5). Let  $u : [0, 1/2] \to \mathbb{R}^2$  be a path following a curve contained in the right half-plane connecting  $(0, u_{20})$  to  $(0, u_{21})$ . Complete the path u by letting  $u : [1/2, 1] \to \mathbb{R}^2$  follow a straight segment from  $(0, u_{21})$  to  $(0, u_{20})$ . The path  $u(\cdot)$  is such that  $t \mapsto u(\varepsilon t)$  yields the desired ensemble approximate transition when  $\varepsilon \to 0$ .

In terms of the original system, the control  $\Omega$  obtained from u as in (3.7) is identically equal to zero on [1/2, 1]. Hence, the evolution corresponding to this portion of the control u has no effect on the dynamics and can be ignored. The original system is then controlled by

$$\Omega(\varepsilon t) = u_1(\varepsilon t)e^{-i(2Et + \frac{1}{\varepsilon}\int_0^{\varepsilon t} u_2(s)ds)}$$

with  $t \in [0, \frac{1}{2\varepsilon}]$ . Notice that  $t \mapsto \Omega(\varepsilon t)$  is a complex function with slow-varying modulus and a phase oscillating with a frequency given by

$$\frac{d}{dt}\left(2Et + \frac{1}{\varepsilon}\int_0^{\varepsilon t} u_2(s)ds\right) = 2E + u_2(\varepsilon t).$$

Since  $u_2(0) = u_{20}$  and  $u_2(1/2) = u_{21}$ , such a frequency slowly varies between two values, one smaller and one larger than every proper frequency E+z of system (3.6) (see Figure 3.6 for an example).

Chapter 3. Ensemble control of quantum systems with two controls: conical case



Figure 3.6 – Two-level system driven by a chirped pulse. Here  $E = 2, z \in [-1, 1], u_1(t) = \frac{3}{2}(1 - \cos(4\pi t)), u_2(t) = -3\cos(2\pi t), \text{ for } t \in [0, \frac{1}{2}] \text{ and } \varepsilon = 0.004$ . The frequency of the pulse is  $2E + u_2(\varepsilon t) = 4 - 3\cos(2\pi\varepsilon t)$  which varies monotonically between 1 and 7.

Let us conclude by a remark on the precision of the adiabatic estimate. According to Theorem 3.2.5, the precision of the transition of system (3.8) corresponding to  $u : [0, 1] \rightarrow \mathbb{R}^2$  is of order  $\sqrt{\varepsilon}$ . Taking into account, moreover, Remark 3.2.8, the precision is of order  $\varepsilon$ . For what concerns system (3.6), we are just interested, as we have seen, to the restriction of u on the interval [0, 1/2] and its corresponding control  $\Omega$ . Hence, since u crosses no eigenvalue intersection on [0, 1/2], one can conclude that the precision of the adiabatic transition is of order  $\varepsilon$  even without recalling Remark 3.2.8.

# 3.5 Permutations

Inspired by [62], we refine in this section our approach in order to select adiabatic controls performing a prescribed permutation of the eigenvectors of the controlled Hamiltonian.

The set of assumptions  $A_j$ ,  $j \in [[1, N-1]]$ , is replaced by the slightly stronger assumption **P** below (see Figure 3.7 for an illustration).

**Assumption P.** For every  $j \in [\![1, N-1]\!]$  there exist a connected component  $\hat{\gamma}_j$  of  $\gamma_j$ and a map  $\beta_j : [z_0, z_1] \to \mathbf{U}$  such that  $\beta_j$  is a  $\mathcal{C}^3$  embedding and

- $\pi(\hat{\gamma}_j) = \beta_j([z_0, z_1]).$
- $\pi(\hat{\gamma}_j) \text{ is contained in } \mathbf{U} \setminus (\cup_{\substack{k \in [\![1,N]\!]\\k \neq j}} \pi(\gamma_k))$
- For every  $z \in [z_0, z_1]$ ,  $\lambda_j^z$  and  $\lambda_{j+1}^z$  have a unique intersection on  $\hat{\gamma}_j$ , which is conical and occurs at  $\beta_j(z)$ .

Moreover the set  $\mathbf{U} \setminus (\bigcup_{k \in [\![1,N]\!]} \pi(\gamma_k))$  is pathwise connected.

Under Assumption P, Corollary 3.2.7 can be refined as follows.

**Theorem 3.5.1.** Assume that **P** is satisfied. Then for every  $u_0, u_1 \in \mathbf{U} \setminus (\bigcup_{k \in [\![1,N]\!]} \pi(\gamma_k))$ , every  $p_1, \ldots, p_N \in \mathbb{C}$  such that  $|p_1|^2 + \cdots + |p_N|^2 = 1$  and every permutation  $\sigma : [\![1,N]\!] \rightarrow [\![1,N]\!]$ , there exists a  $\mathcal{C}^3$  path  $u(\cdot) : [0,1] \rightarrow \mathbf{U}$  satisfying  $u(0) = u_0$  and  $u(1) = u_1$  and a constant C > 0 such that for every  $z \in [z_0, z_1]$  and  $\varepsilon > 0$  the solution  $\psi_{\varepsilon}^z$  of

$$i\dot{\psi}^z_{\varepsilon} = H^z(u(\varepsilon t))\psi^z_{\varepsilon}$$

with initial condition  $\psi_{\varepsilon}^{z}(0) = \sum_{j=1}^{N} p_{j} \phi_{j}^{z}(u_{0})$ , satisfies

$$\|\psi_{\varepsilon}^{z}(1/\varepsilon) - \sum_{j=1}^{N} p_{j} e^{i\theta_{j}} \phi_{\sigma(j)}^{z}(u_{1})\| \leq C\sqrt{\varepsilon},$$

for some  $\theta_1, \ldots, \theta_N \in \mathbb{R}$ .

*Proof.* The path  $u(\cdot)$  is constructively obtained by requiring it to pass through the curves  $\hat{\gamma}_i$  in a suitable order. Such an order is identified by the algorithm described below.





Figure 3.7 – A situation in which Assumption  $\mathbf{P}$  is verified.

Let  $h \in \llbracket 1, N \rrbracket$ . We say that the functions  $f_1, \ldots, f_h : [0, 1] \to \mathbb{R}$  satisfy property  $(\Pi_h)$ if they are continuous, piecewise affine, such that  $f_j(0) = j$ ,  $f_j(1) = \sigma(j)$  for  $j \in \llbracket 1, h \rrbracket$ and, moreover, if for every  $t \in (0, 1)$  and  $j \neq k$ ,  $j, k \in \llbracket 1, h \rrbracket$  such that  $f_j(t) = f_k(t)$  we have

 $- f_l(t) \neq f_m(t) \text{ for every } l \in \llbracket 1, h \rrbracket \setminus \{j, k\}, m \in \llbracket 1, h \rrbracket \setminus \{l\},\$ 

 $- (f_i(t+\varepsilon) - f_k(t+\varepsilon))(f_i(t-\varepsilon) - f_k(t-\varepsilon)) < 0 \text{ for every } \varepsilon > 0 \text{ small enough.}$ 

We now construct recursively a set of functions  $f_1, \ldots, f_N$  satisfying  $(\Pi_N)$ .

Let  $f_1$  and  $f_2$  be the affine functions uniquely determined by  $f_1(0) = 1$ ,  $f_2(0) = 2$ ,  $f_1(1) = \sigma(1)$ ,  $f_2(1) = \sigma(2)$ . Notice that they satisfy  $(\Pi_2)$ .

By induction, assume to have selected  $f_1, \ldots, f_h$  satisfying  $(\Pi_h)$ .

Let  $f_{h+1}$  be the affine function satisfying  $f_{h+1}(0) = h + 1$ ,  $f_{h+1}(1) = \sigma(h+1)$ . If  $(\Pi_{h+1})$  is satisfied then the recursion step is complete. Otherwise modify  $f_{h+1}$  into a continuous function that is constantly equal to h+1 in an interval  $[0,\eta]$  and affine on  $[\eta, 1]$ , with the same boundary conditions. For every positive small enough  $\eta$  one has that  $f_1, \ldots, f_{h+1}$  satisfy  $(\Pi_{h+1})$  and this concludes the induction step.

For every  $t \in [0,1]$  let  $\sigma_t : [\![1,N]\!] \to [\![1,N]\!]$  be a permutation such that

$$f_{\sigma_t(1)}(t) \le f_{\sigma_t(2)}(t) \le \dots \le f_{\sigma_t(N)}(t).$$

Notice that  $t \mapsto \sigma_t$  is piecewise constant and  $\sigma_0 = \text{Id}, \sigma_1 = \sigma$ .

Let  $t_1 < t_2 < \cdots < t_{\mu}$  be the values in (0,1) at which the graphs of the functions



Figure 3.8 – On the left, the functions  $f_j$  and the times  $t_k$ . On the right, the control  $u(\cdot)$ . In this case N = 3,  $\mu = 3$ ,  $\tau(1) = 1$ ,  $\tau(2) = 2$ ,  $\tau(3) = 1$ . Hence,  $u(\cdot)$  passes through  $\pi(\hat{\gamma}_1)$ ,  $\pi(\hat{\gamma}_2)$ , and again  $\pi(\hat{\gamma}_1)$  while connecting  $u_0$  to  $u_1$ .

 $f_1, \ldots, f_n$  intersect (see Figure 3.8). Then  $t_1, \ldots, t_\mu$  are the discontinuity points of  $t \mapsto \sigma_t$ . For every  $j \in [\![1,\mu]\!]$ , let  $\tau(j) \in [\![1,N-1]\!]$  be defined by

$$f_{\sigma_{t_j}(\tau(j))}(t_j) = f_{\sigma_{t_j}(\tau(j)+1)}(t_j).$$

The control  $u(\cdot)$  is constructed in such a way that it passes through  $\pi(\hat{\gamma}_{\tau(1)}), \ldots, \pi(\hat{\gamma}_{\tau(\mu)})$ . More precisely let  $s_1, \ldots, s_{\mu}$  be such that  $0 =: t_0 < s_0 < t_1 < s_1 < t_2 < \cdots < s_{\mu-1} < t_{\mu} < s_{\mu} := 1$ . Let  $u(\cdot) : [0, 1] \to \mathbf{U}$  be such that  $\dot{u}(t) \neq 0$  for every  $t \in [0, 1], u(\cdot) \in \mathcal{C}^3$ ,  $u(0) = u_0, u(1) = u_1$ . Moreover for every  $j = 0, \ldots, \mu$ , the restriction  $u|_{(t_j, s_j)}$  has values in  $\mathbf{U} \setminus (\bigcup_{k \in [\![1,N]\!]} \pi(\gamma_k))$  and for  $j = 1, \ldots, \mu$  the curve  $u|_{[s_{j-1}, t_j]}$  is a reparameterization of  $\beta_{\tau(j)}$  (see Figure 3.8).

By construction and by a repeated application of Theorem 3.2.5 and Remark 3.2.6, the solution of  $i\dot{\psi}_{\varepsilon}^{z} = H^{z}(u(\varepsilon t))\psi_{\varepsilon}^{z}$  with initial condition  $\psi_{\varepsilon}^{z}(0) = \phi_{j}^{z}(u_{0})$ , satisfies for every  $t \in \bigcup_{k=0}^{\mu}(t_{k}, s_{k})$ 

$$\|\psi_{\varepsilon}^{z}(t/\varepsilon) - e^{i\theta_{j}}\phi_{\sigma_{t}(j)}^{z}(u(\varepsilon t))\| \leq C\sqrt{\varepsilon},$$

for some  $\theta_i \in \mathbb{R}$ . The proof of Theorem 3.5.1 is then concluded arguing by linearity.  $\Box$ 

**Remark 3.5.2.** Let us comment on the difference between Assumption **P** and the set of hypotheses  $A_1, \ldots, A_N$  appearing in the statement of Corollary 3.2.7. Assumption **P** is stronger in the sense that each  $\pi(\hat{\gamma}_j)$  is required to have empty intersection not only with  $\pi(\gamma_{j-1})$  and  $\pi(\gamma_{j+1})$ , but also all with  $\pi(\gamma_k)$  for  $k \neq j$ . Assumption **P** guarantees that while inducing a transition between the levels j and j + 1 the other energy levels are untouched.

# 3.6 Genericity

We discuss in this section the genericity of Assumption  $A_j$  which appears in Theorem 3.2.5 and Corollary 3.2.7. Recall that, by the Wigner-von Neumann theorem [82], the set of  $N \times N$  Hermitian matrices of rank equal to N-1 (i.e., with one degenerate eigenvalue of multiplicity 2 and all other eigenvalues simple) is the finite union of submanifolds of codimension 3. More generally the set  $\Upsilon$  of all  $N \times N$  Hermitian matrices with degenerate eigenvalues is a Whitney stratified set (see for instance [42, Section 1.2]) of codimension 3. Similarly, symmetric real matrices with degenerate eigenvalues form a Whitney stratified set of codimension 2 of the space of all symmetric matrices. Let d = 3,  $H: \mathbf{U} \to \operatorname{Herm}(N)$  be a  $\mathcal{C}^3$  map and  $h = H(u) \in \operatorname{Herm}(N)$  have rank equal to N-1. Then H(u) intersects  $\Upsilon$  transversally at h = H(u) if and only if u is a conical intersection. By standard transversality arguments (see, for instance, [1, Proposition 19.1] and [42, Section 1.3.2]), it follows that there exists a residual set  $\mathcal{R}$  in  $\mathcal{C}^3(\mathbf{U}, \operatorname{Herm}(N))$  such that for every  $H \in \mathcal{R}$ , all intersections between the eigenvalues of H are conical. (In particular for every  $H \in \mathcal{R}$  and every  $u \in \mathbf{U}$  such that  $H(u) \in \Upsilon$ , the rank of H(u) is N-1, since the strata of  $\Upsilon$  corresponding to matrices of lower rank are of codimension larger than 3.)

Moreover, a conical intersection is structurally stable in the sense that, if u is a conical intersection for  $H \in C^3(\mathbf{U}, \operatorname{Herm}(N))$ , then any small perturbation of H has a conical intersection near u. Similar results hold for d = 2 in the case of real symmetric Hamiltonians.

The following two results, whose proof can be directly derived from the above considerations, establish that a small one-parameter perturbation of a Hamiltonian with conically connected spectrum satisfies the assumptions of Theorem 3.2.5 and Corollary 3.2.7. Here we impose  $\mathbf{U}$  to be bounded and the Hamiltonian H to be  $\mathcal{C}^3$  on  $\overline{\mathbf{U}}$ , meaning that it admits a  $\mathcal{C}^3$  extension on a neighborhood of  $\overline{\mathbf{U}}$ . We also require H not to have eigenvalue intersections on  $\partial \mathbf{U}$ . This prevents the occurrence of a sequence of eigenvalue intersections converging to  $\partial \mathbf{U}$ .

**Theorem 3.6.1.** Let  $\mathbf{U}$  be an open, connected and bounded subset of  $\mathbb{R}^3$ . Let  $H : \overline{\mathbf{U}} \to \operatorname{Herm}(N)$  be a  $\mathcal{C}^3$  map and  $(\lambda_j(u))_{j \in [\![1,N]\!]}$  be the increasing sequence of eigenvalues of  $H(u), u \in \overline{\mathbf{U}}$ , repeated according to their multiplicities. Fix  $j \in [\![1, N - 1]\!]$ . Assume that the levels  $\lambda_j$  and  $\lambda_{j+1}$  intersect and that all intersections between them are conical and correspond to controls  $u \in \mathbf{U}$ . If j > 1 (respectively, j < N - 1), assume, moreover, that all intersections between the levels  $\lambda_{j-1}$  and  $\lambda_j$  (respectively,  $\lambda_{j+1}$  and  $\lambda_{j+2}$ ) are conical and correspond to controls  $u \in \mathbf{U}$ .

Let us define

$$\Xi = \{ h \in \mathcal{C}^3(\overline{\mathbf{U}} \times \mathbb{R}, \operatorname{Herm}(N)) \mid h(\cdot, 0) = H(\cdot) \}$$

endowed with the  $\mathcal{C}^3$  Whitney topology induced by  $\mathcal{C}^3(\overline{\mathbf{U}} \times \mathbb{R}, \operatorname{Herm}(N))$ .

Then there exists an open and dense subset  $\hat{\Xi}$  of  $\Xi$ , such that for every  $h \in \hat{\Xi}$ , the restriction  $h: \mathbf{U} \times [-\delta, \delta] \to \operatorname{Herm}(N)$  satisfies Assumption  $A_j$  for some  $\delta > 0$ .

**Corollary 3.6.2.** Let  $\mathbf{U}$  be an open, connected and bounded subset of  $\mathbb{R}^3$ . Let  $H : \overline{\mathbf{U}} \to \operatorname{Herm}(N)$  be a  $\mathcal{C}^3$  map and  $(\lambda_j(u))_{j \in [\![1,N]\!]}$  be the increasing sequence of eigenvalues of H(u),  $u \in \overline{\mathbf{U}}$ , repeated according to their multiplicities. Assume that all eigenvalue intersections are conical and correspond to controls  $u \in \mathbf{U}$ . Assume moreover that, for every  $j \in [\![1, N-1]\!]$ ,  $\lambda_j$  and  $\lambda_{j+1}$  intersect.

Let  $\Xi$  be defined as in Theorem 3.6.1. Then there exists a open and dense subset  $\hat{\Xi}$ of  $\Xi$ , such that for every  $h \in \hat{\Xi}$ , the restriction  $h : \mathbf{U} \times [-\delta, \delta] \to \operatorname{Herm}(N)$  satisfies Assumption  $\mathbf{P}$  for some  $\delta > 0$  (and, in particular, Assumption  $A_j$  every  $j \in [\![1, N]\!]$ ).

As a consequence of Corollary 3.2.7 and Theorem 3.5.1, every h in the set  $\hat{\Xi}$  of parameter-dependent Hamiltonians appearing in the statement of Corollary 3.6.2 defines, when restricted to  $\mathbf{U} \times [-\delta, \delta]$ , a system which is ensemble approximately controllable between eigenstates and which satisfies the conclusion of Theorem 3.5.1.

The same conclusions as those of Theorem 3.6.1 and Corollary 3.6.2 hold when d = 2 (i.e., **U** is a subset of  $\mathbb{R}^2$ ) and the Hamiltonians H and h take values in the set of symmetric  $N \times N$  real matrices.

# 3.7 Multidimensional set of parameters

We consider in this section the situation in which the parameter on which the Hamiltonian depends varies in a set of dimension larger than one. Our technique still applies when the set of eigenvalue intersections projects onto a one-dimensional curve in the space of controls. Even if this situation is not generic, it however shows up in several physical examples, as we illustrate in Sections 3.7.1 and 3.7.2.

Let  $\Sigma$  be the set of parameters, contained in  $\mathbb{R}^m$  for some  $m \in \mathbb{N}$ . We consider the ensemble controllability problem for the Schrödinger equation on  $\mathbb{C}^N$ ,  $N \in \mathbb{N}$ ,

$$i\dot{\psi} = H^z(u(t))\psi, \qquad z \in \Sigma.$$
 (3.9)

Denote by  $(\lambda_j^z(u))_{j=1}^N$  the nondecreasing sequence of eigenvalues of  $H^z(u)$  repeated according to their multiplicities and by  $(\phi_1^z(u), \ldots, \phi_N^z(u))$  an orthonormal basis of associated eigenvectors. Let us define

$$\begin{aligned} \gamma_0 &= \emptyset, \\ \gamma_j &= \{(u, z) \in \mathbf{U} \times \boldsymbol{\Sigma} \mid \text{ such that } \lambda_j^z(u) = \lambda_{j+1}^z(u)\}, \quad j \in [\![1, N-1]\!], \\ \gamma_N &= \emptyset. \end{aligned}$$

Denote the projection of  $(u, z) \in U \times \Sigma$  onto the *u*-component by  $\pi$ .

**Assumption**  $A_j^*$ . There exists a connected component  $\hat{\gamma}_j$  of  $\gamma_j$  such that

- There exists an interval  $[z_0, z_1] \subset \mathbb{R}$  and a  $\mathcal{C}^3$  embedding  $\beta_j : [z_0, z_1] \to \mathbf{U}$  such that  $\pi(\hat{\gamma}_j) = \beta_j([z_0, z_1]);$
- $\pi(\hat{\gamma}_j)$  is contained in  $\mathbf{U} \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1}));$
- For every  $z \in \Sigma$ , there exists a unique  $u \in \pi(\hat{\gamma}_j)$  such that  $\lambda_j^z$  and  $\lambda_{j+1}^z$  have an intersection at u, which is conical.

Moreover the set  $\mathbf{U} \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_j) \cup \pi(\gamma_{j+1}))$  is pathwise connected.

Theorem 3.2.5 and Corollary 3.2.7 hold true by replacing  $[z_0, z_1] \times \mathbf{U} \ni (z, u) \mapsto H^z(u) \in \operatorname{Herm}(N)$  by  $\Sigma \times \mathbf{U} \ni (z, u) \mapsto H^z(u) \in \operatorname{Herm}(N)$  and  $A_j$  by  $A_j^*$ . The same proof works without major modifications.

### 3.7.1 Chirped pulses for two-level systems with two parameters

We consider here below an extension of the example studied in Section 3.4, in which we add an uncertain parameter also in the coupling between the control and the system. Consider the controlled equation

$$i\frac{d}{dt}\begin{pmatrix}\psi_1^z\\\psi_2^z\end{pmatrix} = \begin{pmatrix}E+\alpha & \beta\Omega(t)\\\beta\Omega^*(t) & -E-\alpha\end{pmatrix}\begin{pmatrix}\psi_1^z\\\psi_2^z\end{pmatrix},$$
(3.10)

where  $z = (\alpha, \beta) \in [\alpha_0, \alpha_1] \times [\beta_0, \beta_1] \subset \mathbb{R} \times (0, +\infty)$ . As before, E > 0 is fixed and we want to steer the system from  $\phi_1^z(0) = (1, 0)$  to  $\phi_2^z(0) = (0, 1)$ , up to phases.

The same change of variables as in Section 3.4 and the control transformation

$$\Omega(t) = u_1(t)e^{-i(2Et+\Delta(t))}, \quad u_2(t) = \frac{d\Delta}{dt}(t),$$

yield the equivalent system  $i\frac{d}{dt}\Phi^z = \mathcal{H}^z(t)\Phi^z$  with

$$\mathcal{H}^{z}(t) = \left(\begin{array}{cc} \alpha & \beta u_{1}(t) \\ \beta u_{1}(t) & -\alpha + u_{2}(t) \end{array}\right).$$

Since the eigenvalue intersections of  $\mathcal{H}^z$  only depend on  $\alpha$  and not on  $\beta$ , the entire discussion of Section 3.4 still applies in the presence of the parameter  $\beta$ . This underlines once more the robustness of the chirped pulse strategy for two-level systems.

### 3.7.2 Example 2: STIRAP

Consider the three-level system with controlled Hamiltonian

$$H^{(\alpha_1,\alpha_3,\beta_1,\beta_2)}(u_1,u_2) = \begin{pmatrix} E_1 + \alpha_1 & \beta_1 u_1 & 0\\ \beta_1 u_1 & E_2 & \beta_2 u_2\\ 0 & \beta_2 u_2 & E_3 + \alpha_3 \end{pmatrix}$$



Figure 3.9 – The sets  $\pi(\gamma_1)$  and  $\pi(\gamma_2)$  for the STIRAP process.

with  $\alpha_j \in [\alpha_{j0}, \alpha_{j1}], j = 1, 3, \beta_j \in [\beta_{j0}, \beta_{j1}], j = 1, 2, \beta_{10}, \beta_{20} > 0$  and  $\alpha_{j0}, \alpha_{j1}$  such that  $E_1 + \alpha_1 < E_2 < E_3 + \alpha_3$ . (One could clearly add a further parameter uncertainty in the level  $E_2$ , which is not relevant, since the trace of the matrix can always be shifted without modifying the dynamical properties of the system.) The controls  $u_1$  and  $u_2$  are real-valued.

Denote  $z = (\alpha_1, \alpha_3, \beta_1, \beta_2)$  and

$$\Sigma = [\alpha_{10}, \alpha_{11}] \times [\alpha_{30}, \alpha_{31}] \times [\beta_{10}, \beta_{11}] \times [\beta_{20}, \beta_{21}].$$

The eigenvalue intersections of  $H^{z}(u_{1}, u_{2})$  occur on the axes  $u_{1}$  and  $u_{2}$ . More precisely,

$$\begin{aligned} \pi(\gamma_1) &= \left\{ \left( 0, -\frac{\sqrt{(E_1 + \alpha_1 - E_3 - \alpha_3)(E_1 + \alpha_1 - E_2)}}{\beta_2} \right) \mid z \in \Sigma \right\} \cup \\ &\left\{ \left( 0, \frac{\sqrt{(E_1 + \alpha_1 - E_3 - \alpha_3)(E_1 + \alpha_1 - E_2)}}{\beta_2} \right) \mid z \in \Sigma \right\}, \\ \pi(\gamma_2) &= \left\{ \left( -\frac{\sqrt{(E_1 + \alpha_1 - E_3 - \alpha_3)(E_2 - E_3 - \alpha_3)}}{\beta_1}, 0 \right) \mid z \in \Sigma \right\} \cup \\ &\left\{ \left( \frac{\sqrt{(E_1 + \alpha_1 - E_3 - \alpha_3)(E_2 - E_3 - \alpha_3)}}{\beta_1}, 0 \right) \mid z \in \Sigma \right\}. \end{aligned}$$

In particular,  $\pi(\gamma_1)$  and  $\pi(\gamma_2)$  are both unions of two segments (see Figure 3.9), hence they are 1-dimensional smooth embedded manifolds. All hypotheses of Corollary 3.2.7 (in the multi-dimensional parameter extension discussed above) are satisfied and we conclude



Figure 3.10 – Example of STIRAP process with  $E_1 = -1$ ,  $E_2 = 0$ ,  $E_3 = 1$ ,  $\alpha_1, \alpha_3 \in [-0.1, 0.1]$ ,  $\beta_1, \beta_2 \in [0.8, 1.2]$ . In (a) we give the parametric plot of the control, in (b) the shape of its components as functions of time and in (c) the components of the wave function for the three choices of parameters  $(\alpha_1, \alpha_3, \beta_1, \beta_2) =$ (0, 0, 1, 1), (0.1, -0.1, 0.8, 1.2), (-0.1, 0.1, 1.2, 0.8). For this simulation we used  $\varepsilon = 0.05$ .

that the system is ensemble approximately controllable between eigenstates. Moreover, according to Remark 3.2.8, the precision of the transition described in Theorem 3.2.5 can be made of order  $\varepsilon$ , where  $\varepsilon$  is the velocity at which we follow the adiabatic path.

For instance, in order to steer system (3.9) from  $\phi_1^z = (1, 0, 0)$  to  $\phi_3^z = (0, 0, 1)$  we can follow the path in Figure 3.10a.

In Figure 3.10b we plot the two components of the control realising the transition. Notice that such controls are in the celebrated *counter-intuitive order*, meaning that, in order to go from state 1 to state 3, one first activates the control  $u_2$ , responsible for the transition  $2 \rightarrow 3$ , and then the control  $u_1$ , responsible for the transition  $1 \rightarrow 2$  (creating the so-called *dark state*). The approach presented in this paper gives a complete mathematical explanation of why the counter-intuitive order works and why it is so robust with respect to parameter fluctuations.

### 3.8 Extension to the infinite-dimensional case

The results of the previous sections extend, under some suitable regularity assumptions, to the case where  $\mathbb{C}^N$  is replaced by an infinite-dimensional complex separable Hilbert space  $\mathcal{H}$ .

In order to avoid excessive technicalities, we present this extension in the case where the Hamiltonian H depends affinely on the controls and where the controlled Hamiltonians are bounded. (For the general nonlinear case, one could follow the approach in [30].)

We then consider a Hamiltonian of the type

$$H^{z}(u) = H_{0}^{z} + \sum_{j=1}^{d} u_{j}H_{j}^{z},$$

with the parameter z belonging to  $\Sigma \subset \mathbb{R}^m$  for some  $m \in \mathbb{N}$ , and with  $H_j^z$ ,  $j = 0, \ldots, d$ , satisfying the following assumption.

( $\mathbf{H}_{\infty}$ ) Fix  $z_0 \in \Sigma$  and assume that:

- $H_0^{z_0}$  is self-adjoint and bounded from below;
- $H_0^z H_0^{z_0}, H_1^z, \ldots, H_d^z$  are bounded for all  $z \in \Sigma$ ;
- the map  $(z, u) \mapsto H_0^{z_0} H^z(u)$  is  $\mathcal{C}^3$  from  $\Sigma \times \mathbf{U}$  to the Banach space of bounded self-adjoint operators endowed with the operator norm;
- for all (z, u) ∈ Σ × U the eigenvalues λ<sup>z</sup><sub>1</sub>(u) ≤ ··· ≤ λ<sup>z</sup><sub>N</sub>(u) of H<sup>z</sup>(u) are such that
   spectrum(H<sup>z</sup>(u)) ∩ [λ<sup>z</sup><sub>1</sub>(u), λ<sup>z</sup><sub>N</sub>(u)] = {λ<sup>z</sup><sub>1</sub>(u), ..., λ<sup>z</sup><sub>N</sub>(u)} and λ<sup>z</sup><sub>j</sub>(u) has finite multiplicity for every j ∈ [1, N],

 $\circ \operatorname{dist}(\operatorname{spectrum}(H^{z}(u)) \setminus [\lambda_{1}^{z}(u), \lambda_{N}^{z}(u)], \{\lambda_{1}^{z}(u), \lambda_{N}^{z}(u)\}) \geq \Gamma,$ 

with  $\Gamma > 0$  independent of (z, u) (see Figure 3.11).



Figure 3.11 – The eigenvalues  $\lambda_1^z(u) \leq \lambda_2^z(u) \leq \lambda_3^z(u)$  of  $H^z(u)$  separated from the rest of the spectrum (which is contained in the shaded regions).

Under the hypothesis  $(\mathbf{H}_{\infty})$ , for every continuous control  $u : [0, T] \to \mathbf{U}$  and every initial condition, the equation

$$i\dot{\psi} = H^z(u(t))\psi \tag{3.11}$$

admits a unique solution [78, Proposition 2.1].

Let  $P^{z,u} : \mathcal{H} \to \mathcal{H}$  be the orthogonal projector onto the sum of the eigenspaces corresponding to  $\lambda_1^z(u), \ldots, \lambda_N^z(u)$ . Without loss of generality this space is of complex dimension N. Then (see, for instance, [68]), for every  $(\bar{z}, \bar{u}) \in \Sigma \times \mathbf{U}$  and every (z, u) in a neighborhood of  $(\bar{z}, \bar{u})$ ,

$$P^{z,u} = -\frac{1}{2\pi i} \int_{\gamma} (H^z(u) - c)^{-1} dc,$$

where  $\gamma$  is a Jordan curve in  $\mathbb{C}$  separating  $\lambda_1^{\bar{z}}(\bar{u}), \ldots, \lambda_N^{\bar{z}}(\bar{u})$  and spectrum $(H^{\bar{z}}(\bar{u})) \setminus [\lambda_1^{\bar{z}}(\bar{u}), \lambda_N^{\bar{z}}(\bar{u})]$ . Hence  $(z, u) \mapsto P^{z, u}$  is  $\mathcal{C}^3$  as a map from  $\Sigma \times \mathbf{U}$  to the Banach space of bounded operators on  $\mathcal{H}$  endowed with the operator norm.

Let  $I^{z,u}: \mathbb{C}^N \to \mathcal{H}$  be a linear map such that  $I^{z,u}$  is unitary between  $\mathbb{C}^N$  and the image of  $P^{z,u}$ .

Assume for now that  $(z, u) \mapsto I^{z,u}$  is globally  $\mathcal{C}^3$  on  $\Sigma \times \mathbf{U}$ . We denote by  $(I^{z,u})^{-1}$  the inverse of  $I^{z,u}$  on the image of  $P^{z,u}$ . The Hamiltonian  $(z, u) \mapsto \widehat{H}^z(u) = (I^{z,u})^{-1} H^z(u) I^{z,u} \in$ Herm(N) is  $\mathcal{C}^3$  and its eigenvalues are  $\lambda_1^z(u), \ldots, \lambda_N^z(u)$ .

When  $(z, u) \mapsto I^{z,u}$  cannot be globally defined as a  $C^3$  map, the same arguments can be applied in local charts leading to the following generalization of Theorem 3.2.5 and Corollary 3.2.7

As in the previous section, the sets  $\pi(\gamma_i)$  are defined as

$$\begin{aligned} \pi(\gamma_0) &= \emptyset, \\ \pi(\gamma_j) &= \{ u \in \mathbf{U} \mid \exists z \in \Sigma \text{ such that } \lambda_j^z(u) = \lambda_{j+1}^z(u) \}, \quad j \in [\![1, N-1]\!], \\ \pi(\gamma_N) &= \emptyset. \end{aligned}$$

**Theorem 3.8.1.** Assume that  $(H_{\infty})$  holds true. Let  $j \in [[1, N - 1]]$  be such that Assumption  $A_j^*$  of Section 3.7 is satisfied. Take  $u_0, u_1 \in \mathbf{U} \setminus (\gamma_{j-1} \cup \gamma_j \cup \gamma_{j+1})$  and consider a  $\mathcal{C}^3$  path  $u(\cdot) : [0,1] \to \mathbf{U}$  satisfying  $u(0) = u_0, u(1) = u_1$ , and such that  $u|_{[t_0,t_1]}$  is a reparameterization of  $\beta_j$  for some  $0 < t_0 < t_1 < 1$  such that  $u(t) \notin \gamma_{j-1} \cup \gamma_j \cup \gamma_{j+1}$  for every  $t \in [0,1] \setminus [t_0,t_1]$ . Assume, moreover, that  $\dot{u}(t) \neq 0$  for every  $t \in [0,1]$ .

Then there exists a constant C > 0 such that for every  $z \in \Sigma$  and  $\varepsilon > 0$  the solutions  $\psi_{\varepsilon}^{z,\pm}$  of

$$i\dot{\psi}^{z,\pm}_{\varepsilon}(t) = H^z(u(\varepsilon t))\psi^{z,\pm}_{\varepsilon}(t)$$

with initial conditions

$$\psi_{\varepsilon}^{z,+}(0) = \phi_{j}^{z}(u_{0}), \qquad \psi_{\varepsilon}^{z,-}(0) = \phi_{j+1}^{z}(u_{0})$$

satisfy

$$\|\psi_{\varepsilon}^{z,+}(1/\varepsilon) - e^{i\theta^+}\phi_{j+1}^z(u_1)\| \le C\sqrt{\varepsilon}, \qquad \|\psi_{\varepsilon}^{z,-}(1/\varepsilon) - e^{i\theta^-}\phi_j^z(u_1)\| \le C\sqrt{\varepsilon},$$

for some  $\theta^{\pm} \in \mathbb{R}$ .

Proof. An adiabatic decoupling theorem (see Theorem 3.9.1 in appendix) states that, given a  $\mathcal{C}^2$  path  $u: [0,1] \to \mathbf{U}$  as defined in the statement of the theorem and  $\hat{\psi}_0 \in \mathbb{C}^N$ , the solutions  $\psi(\cdot)$  and  $\hat{\psi}(\cdot)$  of, respectively,  $i\dot{\psi}(t) = H^z(u(\varepsilon t))\psi(t), \psi(0) = I^{z,u(0)}\hat{\psi}_0$ , and  $i\dot{\hat{\psi}}(t) = \hat{H}^z(u(\varepsilon t))\hat{\psi}(t), \ \hat{\psi}(0) = \hat{\psi}_0$ , are such that  $\psi(1/\varepsilon)$  is  $\sqrt{\varepsilon}$ -close, up to phases, to  $I^{z,u(1)}\hat{\psi}(1/\varepsilon)$ , uniformly w.r.t.  $z \in \Sigma$ . The result follows by applying Theorem 3.2.5 to the equation  $i\dot{\psi}(t) = \hat{H}^z(u(\varepsilon t))\hat{\psi}(t)$ .

**Corollary 3.8.2.** Assume that  $(H_{\infty})$  holds true. Let Assumption  $A_j^*$  be satisfied for every  $j \in [\![1, N-1]\!]$ . Then (3.11) is ensemble approximately controllable between the eigenstates  $\{\phi_1^z(u), \ldots, \phi_N^z(u)\}$  in the sense that for every  $\varepsilon > 0$ ,  $j, k \in [\![1, N]\!]$  and  $u_0, u_1 \in \mathbf{U}$  such that  $\lambda_j^z(u_0)$  and  $\lambda_k^z(u_1)$  are simple for every  $z \in \Sigma$ , there exists a control  $u(\cdot) : [0, T] \to \mathbf{U}$  such that for every  $z \in \Sigma$  the solution of (3.11) with initial condition  $\psi^z(0) = \phi_j^z(u_0)$  satisfies  $\|\psi^z(T) - e^{i\theta}\phi_k^z(u_1)\| < \varepsilon$  for some  $\theta \in \mathbb{R}$ .

**Remark 3.8.3.** When  $\Sigma$  is one-dimensional, genericity results similar to those in Section 3.6 can be obtained. For related structural stability and genericity considerations, see [22].

### 3.8.1 Example 3: Eberly–Law-like models

We study in this section a model for coupled spin-oscillator dynamics (see [61, 84, 15]). For other spin-boson models, see [20, 56, 67] and references therein. The state space is the tensor product of the state spaces of an harmonic oscillator and of a spin- $\frac{1}{2}$  particle. The system has two control parameters that we assume to be real, the first one,  $u_1$ , coupling the two levels of the spin system and the other one,  $u_2$ , producing simultaneous spin transitions and vibrational phonon excitations.

We let  $\omega$  be the gap between the levels of the harmonic oscillator and  $\delta$  be the internal gap of the spin system,  $\alpha$  (respectively,  $\beta$ ) the coupling strength between  $u_1$  (respectively,  $u_2$ ) and the system. The vector  $z = (\alpha, \beta, \omega, \delta)$  will be considered as a set of parameters whose uncertainty will be tackled by the technique proposed in the previous sections.

The model can then be represented by the infinite-dimensional controlled quantum system

$$i\dot{\psi} = H^z(u)\psi, \qquad \psi \in \ell^2,$$

where  $u = (u_1, u_2) \in \mathbb{R}^2$ ,  $z \in [\alpha_0, \alpha_1] \times [\beta_0, \beta_1] \times [\omega_0, \omega_1] \times [\delta_0, \delta_1] = \Sigma$ , and

$$H^{z}(u) = \begin{pmatrix} 0 & \alpha u_{1} & 0 & 0 & 0 & \cdots \\ \alpha u_{1} & \delta & \beta u_{2} & 0 & 0 & \ddots \\ 0 & \beta u_{2} & \omega & \alpha u_{1} & 0 & \ddots \\ 0 & 0 & \alpha u_{1} & \omega + \delta & \beta u_{2} & \ddots \\ 0 & 0 & 0 & \beta u_{2} & 2\omega & \ddots \\ \vdots & \ddots & \ddots & \ddots & \ddots & \ddots \end{pmatrix}.$$
(3.12)

We assume that  $\alpha_0, \beta_0, \omega_0, \delta_0 > 0$ . For simplicity, we also assume that  $\omega_0 > \delta_1$ , that is, the diagonal of  $H^z(u)$  is a strictly increasing sequence. Further conditions on  $\alpha_j, \beta_j, \omega_j, \delta_j$ , j = 0, 1, will be imposed below (see (3.13)).

The Hamiltonian  $H^{z}(u)$  is self-adjoint, has purely discrete spectrum, and, if both  $u_{1}$  and  $u_{2}$  are different from zero, then all eigenvalues of  $H^{z}(u)$  are non-degenerate (see [2]).

In order to apply our general strategy, let us describe the eigenvalue intersections of  $H^{z}(u)$ . This is quite simple, since, for  $u_{1} = 0$  or  $u_{2} = 0$ , the matrix describing  $H^{z}(u)$  is block-diagonal.

Let us first consider intersections along the axis  $u_2 = 0$ . The eigenvalues of  $H^z(u_1, 0)$  as a function of  $u_1$  are shown in Figure 3.12. A simple computation shows that the smallest value of  $|u_1|$  for which  $H^z(u_1, 0)$  has degenerate eigenvalues is  $u_1^* = \frac{\sqrt{\omega^2 - \delta^2}}{2\alpha}$  and the secondsmallest value is  $\bar{u}_1 = \frac{\sqrt{4\omega^2 - \delta^2}}{2\alpha}$ .

We reason similarly along the axis  $u_1 = 0$  and we get that the smallest value of  $|u_2|$  for which  $H^z(0, u_2)$  has degenerate eigenvalues different from zero is  $u_2^* = \frac{\sqrt{\delta(2\omega-\delta)}}{2\beta}$  and the second-smallest value is  $\bar{u}_2 = \frac{\sqrt{\delta(2\omega-\delta)+3\omega^2}}{2\beta}$ . The smallest value of  $|u_2|$  for which 0 is a degenerate eigenvalue for  $H^z(0, u_2)$  is  $u_{20}^* = \frac{\sqrt{\delta\omega}}{\beta}$  and the second-smallest value is  $\bar{u}_{20} = \frac{\sqrt{2\omega(\delta+\omega)}}{\beta}$ .

We assume in the following that

$$\max_{z \in \Sigma} u_1^* < \min_{z \in \Sigma} \bar{u}_1, \quad \max_{z \in \Sigma} u_2^* < \min_{z \in \Sigma} \bar{u}_2, \quad \max_{z \in \Sigma} u_{20}^* < \min_{z \in \Sigma} \bar{u}_{20}.$$
(3.13)

Let  $U = [-\eta, \eta + \max_{z \in \Sigma} u_1^*] \times [-\eta, \eta + \max_{z \in \Sigma} u_{20}^*]$  with  $\eta > 0$  small. Then Assumptions  $A_j^*, j \ge 1$ , introduced in Section 3.7, hold true with

$$\pi(\hat{\gamma}_{1}) = \{0\} \times [\min_{z \in \Sigma} u_{20}^{*}, \max_{z \in \Sigma} u_{20}^{*}],$$
  

$$\pi(\hat{\gamma}_{2j+1}) = \{0\} \times [\min_{z \in \Sigma} u_{2}^{*}, \max_{z \in \Sigma} u_{2}^{*}], \qquad j \ge 1,$$
  

$$\pi(\hat{\gamma}_{2j}) = [\min_{z \in \Sigma} u_{1}^{*}, \max_{z \in \Sigma} u_{1}^{*}] \times \{0\}, \qquad j \ge 1.$$

In particular, all eigenvalue intersections in  $\hat{\gamma}_j$ ,  $j \ge 1$ , are conical. Condition (3.13) also implies that, for every  $n \in \mathbb{N}$ , Assumption  $(\mathbf{H}_{\infty})$  is satisfied by the first 2n+1 eigenvalues



Figure 3.12 – The eigenvalues of  $H^{z}(u_{1}, 0)$  (left) and  $H^{z}(0, u_{2})$  (right).

with  $\mathbf{U} = U \cap \{u_2 < u_1 + \eta\}$ , by the first 2 eigenvalues  $\lambda_1^z, \lambda_2^z$  with  $\mathbf{U} = U \cap \{u_1 < u_2 + \eta\}$ and by the eigenvalues  $\lambda_3^z, \ldots, \lambda_{2n}^z$  with  $\mathbf{U} = U \cap \{u_1 < u_2 + \eta, u_2 < \max_{z \in \Sigma} u_2^* + \eta\}$ .

Henceforth, ensemble approximate controllability between eigenvectors holds true. Figure 3.13 shows a path in the space of controls leading to an ensemble transfer from  $\phi_1^z(0)$ to  $\phi_4^z(0)$ . The path is a loop starting and ending at u = 0 that goes through  $\pi(\hat{\gamma}_1), \pi(\hat{\gamma}_2)$ , and  $\pi(\hat{\gamma}_3)$ .



Figure 3.13 – A loop in the plane  $(u_1, u_2)$  inducing a population transfer from the first to the fourth eigenstate of the drift Hamiltonian.

# 3.9 Appendix

**Regularity of eigenvalues and eigenvectors** In this section we prove Lemma 3.3.2, which states that a time-dependent Hamiltonian  $H \in C^{k+1}(I, \text{Herm}(N))$  whose eigenvalues intersections are all conical has  $C^{k+1}$  eigenvalues and  $C^k$  eigenvectors.

Proof of Lemma 3.3.2. For each  $j \in [\![1, N]\!]$ , consider the function

$$\ell_j : \operatorname{Herm}(N) \to \mathbb{R}$$

that associates with a Hermitian matrix its *j*-th eigenvalue, where eigenvalues are counted according to their multiplicities, in such a way that  $\ell_1 \leq \cdots \leq \ell_N$ . By Rellich theorem [69], each map  $\ell_j$  is analytic on

$$\operatorname{Herm}_{i}(N) = \{h \in \operatorname{Herm}(N) \mid \ell_{i}(h) \text{ is a simple eigenvalue of } h\}.$$

Moreover, for every  $h \in \operatorname{Herm}_{j}(N)$ , there exist a neighborhood  $\mathcal{V}_{h}$  of h in  $\operatorname{Herm}_{j}(N)$  and an analytic function  $\varphi_{j} : \mathcal{V}_{h} \to \mathbb{C}^{N}$  such that  $\varphi_{j}(g)$  is a norm 1 eigenvector of g of eigenvalue  $\ell_{j}(g)$  for every  $g \in \mathcal{V}_{h}$ . Hence, for every  $j \in [\![1, N]\!]$ , the *j*-th eigenvalue and a choice of a corresponding eigenvector of norm 1 are  $\mathcal{C}^{k+1}$  on

 $\{t \in I \mid \ell_j(H(t)) \text{ is a simple eigenvalue of } H(t)\}.$ 

We are left to asses the regularity of eigenpairs in a neighborhood of a time  $\bar{t} \in I$  such that  $H(\bar{t}) \notin \operatorname{Herm}_{i}(N)$ .

Let J be a neighborhood of  $\overline{t}$  in I such that for every  $t \in J \setminus {\overline{t}}$  both  $\ell_j(H(t))$  and  $\ell_{j+1}(H(t))$  are simple. For every  $t \in J$ , let  $P^t : \mathbb{C}^N \to \mathbb{C}^N$  be the orthogonal projector on the sum of the eigenspaces corresponding to  $\ell_j(H(t))$  and  $\ell_{j+1}(H(t))$ .

Then (see, for instance, [68]),

$$P^{t} = -\frac{1}{2\pi i} \int_{\gamma} (H(t) - c)^{-1} dc$$

where  $\gamma$  is a Jordan curve in  $\mathbb{C}$  separating  $\ell_i(H(t)), \ell_{i+1}(H(t))$  and

spectrum(
$$H(t)$$
) \ { $\ell_j(H(t)), \ell_{j+1}(H(t))$ }.

Up to taking a smaller J if necessary, the curve  $\gamma$  can be taken independently of  $t \in J$ .

Hence  $t \mapsto P^t$  is  $\mathcal{C}^{k+1}$  as a map from J to the space of linear operators on  $\mathbb{C}^N$ . Then there exists a linear map  $I^t : \mathbb{C}^2 \to \mathbb{C}^N$ , depending  $\mathcal{C}^{k+1}$  on t, such that  $I^t$  is unitary between  $\mathbb{C}^2$  and the image of  $P^t$ . We denote by  $(I^t)^{-1}$  the inverse of  $I^t$  on the image of  $P^t$ .

The Hamiltonian  $h(t) = (I^t)^{-1}H(t)I^t : \mathbb{C}^2 \to \mathbb{C}^2$  is well defined on J, depends  $\mathcal{C}^{k+1}$ on t, and has  $\ell_i(H(t))$  and  $\ell_{i+1}(H(t))$  as eigenvalues. If, moreover, v is an eigenvector of h(t), then  $I^t v$  is an eigenvector of H(t). We are therefore left to prove the result for the eigenpairs of h(t).

Without loss of generality we can consider the case where the trace of h(t) is zero for every  $t \in J$ . Hence, h(t) has the form

$$h(t) = \left(\begin{array}{cc} a(t) & b(t) + ic(t) \\ b(t) - ic(t) & -a(t) \end{array}\right),$$

where  $a(\cdot), b(\cdot), c(\cdot)$  are real-valued  $\mathcal{C}^{k+1}$  functions on J.

Moreover, since  $h(\bar{t})$  has a double eigenvalue, we have  $a(\bar{t}) = b(\bar{t}) = c(\bar{t}) = 0$ . Without loss of generality  $\bar{t} = 0$ .

Let  $\alpha, \beta, \gamma: J \to \mathbb{R}$  be  $\mathcal{C}^k$  functions such that

$$a(t) = t\alpha(t), \quad b(t) = t\beta(t), \quad c(t) = t\gamma(t), \qquad t \in J.$$

Hypothesis (3.5) guarantees that  $\alpha(0) \neq 0$  or  $\beta(0) \neq 0$  or  $\gamma(0) \neq 0$ . Notice that, up to applying a unitary change of variables in  $\mathbb{C}^2$  and restricting J, we can assume that  $\beta(t) \neq 0$  for every  $t \in J$ .

The eigenvalues of h(t) are  $\pm \sqrt{a(t)^2 + b(t)^2 + c(t)^2} = \pm |t| \sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}$ . The function

$$t \mapsto \begin{cases} -\sqrt{a(t)^2 + b(t)^2 + c(t)^2} & \text{if } t < 0\\ \sqrt{a(t)^2 + b(t)^2 + c(t)^2} & \text{if } t \ge 0 \end{cases}$$

is then equal to  $t\sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}$  which is  $\mathcal{C}^{k+1}$  on J. Indeed, the only term of the (k+1)-th derivative of  $t\sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}$  involving the (k+1)-th derivative of  $\alpha$ ,  $\beta$ , and  $\gamma$  is

$$\frac{t(\alpha^{(k+1)}(t)\alpha(t) + \beta^{(k+1)}(t)\beta(t) + \gamma^{(k+1)}(t)\gamma(t))}{\sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}}.$$

Now, since  $a(t) = t\alpha(t)$  is  $\mathcal{C}^{k+1}$ , the term  $t\alpha^{(k+1)}(t)$  is continuous. The same argument holds for  $\beta$  and  $\gamma$  and the  $\mathcal{C}^{k+1}$  regularity of  $t\sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}$  is proved.

For what concerns the unit eigenvectors, a simple calculation shows that, up to phases and scaling, they are equal to

$$\left(\frac{-a(t) \pm \sqrt{a(t)^2 + b(t)^2 + c(t)^2}}{b(t) - ic(t)}, 1\right) = \left(\frac{-\alpha(t) \pm \operatorname{sign}(t)\sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}}{\beta(t) - i\gamma(t)}, 1\right)$$

for  $t \neq 0$ . As above, connecting suitably the two branches, we can rewrite the eigenvectors as

$$\left(\frac{-\alpha(t) \pm \sqrt{\alpha(t)^2 + \beta(t)^2 + \gamma(t)^2}}{\beta(t) - i\gamma(t)}, 1\right),$$

which are  $\mathcal{C}^k$  on J.

Adiabatic decoupling of dimension n. We present here a result of adiabatic decoupling. It is adapted from [78], and is interesting because the adiabatic structure is preserved for the local *n*-levels dynamics. The proof is similar to the proofs of the theorems 4.6.4 and 4.6.5 of Section 4.6.2.

**Theorem 3.9.1** (Adiabatic decoupling for parametric systems). Assume that the Hamiltonian H is such that  $(H_{\infty})$  holds true as in Section 3.8. Assume moreover that

$$\left\{\lambda_j(H^z(u)) \mid j \in K\right\},\,$$

with  $K = \{q \dots q'\} \subset \mathbb{N}$  where  $q, q' \in \mathbb{N}$ , is separated from

$$Spectrum(H^{z}(u)) \setminus \{\lambda_{j}(H^{z}(u)) \mid j \in K\},\$$

for u in a neighborhood W of 0 in  $\mathbb{R}^k$  and  $z \in [z_0, z_1]$ . Define, for every  $(u, z) \in W \times [z_0, z_1]$ ,  $I^{u,z}$  and  $\hat{H}^z(u)$  as in section 3.8. Consider a  $C^2$  regular path  $u : [0,1] \to W$  and  $\tilde{\psi}_0^z \in \mathbb{C}^2$ , for every  $z \in [z_0, z_1]$ . Let  $\ell \in \mathbb{N}$  and assume that for every  $z \in [z_0, z_1]$ , there exist  $C^\ell$ functions  $\Lambda_j^z : [0,1] \to \mathbb{R}$  for every  $j \in K$  such that for every  $t \in [0,1]$ ,  $\{\Lambda_j^z(t), | j \in K\} = \{\lambda_j^z(u(t)), | j \in K\}$  and that, for every  $z \in [z_0, z_1]$ ,  $\hat{H}^z(u(\cdot))$  admits  $C^2$  eigenvectors  $(\Phi_j^z(\cdot))_{j \in K}$  such that  $(t, z) \mapsto \frac{d\Phi_j^z(t)}{dt}$  and  $(t, z) \mapsto \frac{d^2\Phi_j^z(t)}{dt^2}$  are bounded uniformly with respect to  $(t, z) \in [0, 1] \times [z_0, z_1]$ , for every  $j \in K$ . Assume that there exists c > 0 such that

$$\left|\int_{0}^{t} e^{\frac{i}{\epsilon}\int_{0}^{s}(\Lambda_{j+1}^{z}(x)-\Lambda_{j}^{z}(x))dx}ds\right| \leq c\epsilon^{1/(\ell+1)}, \qquad \forall t \in [0,1], \qquad \forall z \in [z_{0},z_{1}], \qquad \forall j \in K.$$

$$(2.14)$$

Then the solutions  $\psi_{\epsilon}^z$  and  $\tilde{\psi}_{\epsilon}^z$  of, respectively,  $i\frac{d\psi^z}{dt} = H^z(u(\epsilon t))\psi^z, \psi^z(0) = I^{u(0),z}\tilde{\psi}_0^z$ , and  $i\frac{d\tilde{\psi}^z}{dt} = \hat{H}^z(u(\epsilon t))\tilde{\psi}^z, \tilde{\psi}^z(0) = \tilde{\psi}_0^z$  are such that  $\psi^z(1/\epsilon)$  is  $O(\epsilon^{\frac{1}{\ell+1}})$ -close, up to phases, to  $I^{u(1),z}\tilde{\psi}^z(1/\epsilon)$ , uniformly w.r.t.  $z \in [z_0, z_1]$ .

# 3.9. Appendix

# Chapter

# Ensemble control of quantum systems with two controls: non-conical case

In Chapter 3 we proposed a framework for the adiabatic ensemble control of a continuum of n-level systems with real Hamiltonian, driven by two controls and having conical intersections between the eigenvalues. The method works under the assumption that for all values of the parameter eigenvalue intersections remain conical and describe a smooth curve. These assumptions are satisfied for generic small parametric perturbations, however, for generic large perturbations it may happen that conicity of eigenvalue intersections is lost at isolated points of the curve. The goal of this chapter is to extend the analysis to this case. In particular, we

- characterize typical non-conical intersections and give normal forms for them;
- study the evolution of the system corresponding to adiabatic paths in the space of controls passing through such intersections;
- conclude on the ensemble controllability of generic 1-parameter systems presenting typical intersections (conical and non-conical).

### Contents

4.1	Intro	oduction	78
4.2	Basi	c facts and normal forms	82
	4.2.1	Generic families of 2-level Hamiltonians	82
	4.2.2	Admissible transformations	87
	4.2.3	Normal forms for the non parametric case $\ldots \ldots \ldots \ldots$	91
	4.2.4	Normal forms for the parametric case	96
4.3	Gen	eric global properties of the singular locus	99
	4.3.1	Proof of Lemma 4.1.1 and Theorem 4.1.2	99

### 4.1. Introduction

	4.3.2	Generic self-intersections of $\pi(f)$	99
4.4	Adia	batic control through a semi-conical intersection of eigen-	
	value	es	100
	4.4.1	Adiabatic dynamics	101
	4.4.2	Regularity of the eigenpairs along smooth control paths $\ . \ . \ .$	102
	4.4.3	Dynamical properties at semi-conical intersections of eigenvalues	104
4.5	Cont	trol of a continuum of systems	106
	4.5.1	Ensemble adiabatic dynamics	106
	4.5.2	Controllability properties between the eigenstates for the normal	
		forms	107
	4.5.3	The control path $(u, v)$ exits from $\pi(f)$	110
	4.5.4	Proof of Theorem 4.1.3	111
4.6	Exte	ension to <i>n</i> -level systems	111
	4.6.1	Generic assumptions on n-level Hamiltonians and adiabatic de-	
		coupling	111
	4.6.2	Adiabatic decoupling	113
	4.6.3	Semi-conical intersections for $n$ -level quantum systems $\ldots$ .	115
	4.6.4	Controllability result	117
4.7	App	endix	117
	4.7.1	Averaging theorems and estimates of oscillatory integrals	117
	4.7.2	Two useful lemmas	119

# 4.1 Introduction

Consider a two-level system driven by a zero trace real Hamiltonian, that is, an equation of the form

$$i\frac{d\psi}{dt}(t) = H(u(t), v(t))\psi(t), \qquad \psi(t) \in \mathbb{C}^2, \quad (u(t), v(t)) \in \mathbb{R}^2, \tag{4.1}$$

with

$$H(u,v) = \begin{pmatrix} f_1(u,v) & f_2(u,v) \\ f_2(u,v) & -f_1(u,v) \end{pmatrix}, \qquad f = (f_1,f_2) \in C^{\infty}(\mathbb{R}^2,\mathbb{R}^2).$$

As in Chapter 3, we restrict our attention to real Hamiltonians, which are relevant in many physical systems, for instance for Galerkin approximations of the Schrödinger equation  $i\partial_t\psi(x,t) = (-\Delta + V(x) + u(t)W(x))\psi(x,t)$ , where x belongs to a bounded set of  $\mathbb{R}^n$  and V,W are regular enough real functions. The spectrum of H(u,v) is  $\{\pm \sqrt{f_1(u,v)^2 + f_2(u,v)^2}\}$ and, in particular, it is degenerate if and only if f(u,v) = (0,0). Denote by  $\lambda^+(u,v)$  the largest eigenvalue of H(u,v), that is,  $\lambda^+(u,v) = \sqrt{f_1(u,v)^2 + f_2(u,v)^2}$ . Notice that the gap (denoted Gap(u, v)) between the two eigenvalues of H(u, v) is equal to  $2\lambda^+(u, v)$  and that  $\lambda^+(u, v) = 0$  if and only if f(u, v) = 0, that is, if (u, v) is an *eigenvalue intersection*. Such a point is said to be *conical* if

$$\chi(f) := \det(\nabla f_1, \nabla f_2)$$

is nonzero at (u, v), where  $\nabla$  denotes the gradient with respect to the variables u and v. An eigenvalue intersection (u, v) is said to be *semi-conical* if  $\nabla f_1(u, v)$  and  $\nabla f_2(u, v)$  are collinear, are not both zero, and the directional derivative  $\partial_\eta \chi(f)$  along

$$\eta = (-\partial_2 f_j(u, v), \partial_1 f_j(u, v))$$

is nonzero if  $j \in \{1,2\}$  is such that  $\eta \neq 0$ . The direction spanned by  $\eta$  is called the *non-conical direction* at (u, v). If (u, v) is a conical intersection, then

$$\frac{1}{C} \| (u', v') - (u, v) \| \le \operatorname{Gap}(u', v') \le C \| (u', v') - (u, v) \|$$
(4.2)

for some C > 0 and for all (u', v') in a neighborhood of (u, v). If, instead, (u, v) is a semi-conical intersection, then an inequality of the type (4.2) holds along any line passing through (u, v) in a direction transversal to the non-conical direction. Along the non-conical direction  $\eta$  we have

$$\frac{1}{C}t^2 \le \operatorname{Gap}((u,v) + t\eta) \le Ct^2$$

for some C > 0 and for all t in a neighborhood of 0.





Figure 4.1 – Conical intersection as a function of the controls  $(u, v) \in \mathbb{R}^2$ .

Figure 4.2 – Semi-conical intersection of eigenvalues as a function of the controls  $(u, v) \in \mathbb{R}^2$ .

We are now considering a one-parameter family of two-level systems as above, that is, the *Ensemble Schrödinger Equation* 

$$i\frac{d\psi}{dt}(t) = H(u(t), v(t), z)\psi(t), \qquad \psi(t) \in \mathbb{C}^2, \quad (u(t), v(t)) \in \mathbb{R}^2, \tag{4.3}$$

#### 4.1. Introduction

with

$$H(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix}, \qquad f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2),$$

the spectrum of H(u, v, z) is  $\{\pm \sqrt{f_1(u, v, z)^2 + f_2(u, v, z)^2}\}$  and, in particular, it is degenerate if and only if f(u, v, z) = (0, 0). In order to extend the definition of conical and semi-conical intersections for a one-parameter Hamiltonian, we need to add to the previous definitions some regularity assumptions with respect to the perturbation parameter z. Let (u, v, z) be a point such that f(u, v, z) = (0, 0). It is said to be conical for the family (*F-conical*) if (u, v) is conical for  $f(\cdot, \cdot, z)$  and  $\partial_3 f(u, v, z) \neq (0, 0)$ . It is said to be semi-conical for the family (*F-semi-conical*) if it is semi-conical for  $f(\cdot, \cdot, z)$  and f is a submersion at (u, v, z). The requirement that f is a submersion guarantees that the set

$$Z(f) = \{(u, v, z) \mid f(u, v, z) = (0, 0)\}$$

is a smooth curve in the nieghborhood of a semi-conical point. The gap  $\operatorname{Gap}(u, v, z)$  between the two eigenvalues of H(u, v, z) is equal to  $2\lambda^+(u, v, z) = 2\sqrt{f_1(u, v, z)^2 + f_2(u, v, z)^2}$ and we have  $\operatorname{Gap}(u, v, z) = 0$  if and only if  $(u, v, z) \in Z(f)$ . In the following we denote by  $Z_{\rm nc}(f)$  the set of non-conical intersections in Z(f).

**Lemma 4.1.1.** Let  $(\bar{u}, \bar{v}, \bar{z})$  be a *F*-semi-conical intersection. Then  $(\bar{u}, \bar{v}, \bar{z})$  is isolated in  $Z_{nc}(f)$ , Z(f) is locally a smooth curve near  $(\bar{u}, \bar{v}, \bar{z})$ , and its tangent is not vertical. Moreover, the non-conical direction corresponding to  $(\bar{u}, \bar{v}, \bar{z})$  is tangent to the projection on the plane (u, v) of such a curve at  $(\bar{u}, \bar{v})$ .

We focus in what follows on generic properties for systems of the type (4.3). This means that we look for properties which hold for all f in a "large" subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . For a precise definition of genericity as we intend it in this chapter, we refer to Section 4.2.1.

**Theorem 4.1.2.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , for any connected component  $\gamma$  of Z(f),

- (i)  $\gamma$  is a one-dimensional submanifold of  $\mathbb{R}^3$ ;
- (ii) The projection  $\pi(\gamma)$  of  $\gamma$  on the plane (u, v) is a  $C^{\infty}$  embedded curve of  $\mathbb{R}^2$ ;
- (iii)  $(Z(f) \setminus Z_{nc}(f)) \cap \gamma$  is made of F-conical intersections and  $Z_{nc}(f) \cap \gamma$  is made of F-semi-conical intersections only.

The following theorem resumes the main properties of the control strategy that we study in the chapter.

**Theorem 4.1.3.** Assume that Z(f) has a single connected component  $\gamma$ . Assume moreover that

- 1.  $\gamma$  satisfies properties (i)–(iii) of Theorem 4.1.2;
- 2.  $\pi(\gamma)$  has no self-intersections.

Take two conical intersections  $(u_0, v_0, z_0)$ ,  $(u_1, v_1, z_1)$  in  $\gamma$  with  $z_0 < z_1$ . Consider a regular  $C^4$  path  $(u, v) : [0, 1] \rightarrow \mathbb{R}^2$  such that  $(u, v)(t_0) = (u_0, v_0)$ ,  $(u, v)(t_1) = (u_1, v_1)$  for some  $0 < t_0 < t_1 < 1$ . Assume, moreover, that  $(u, v)(0) = (u, v)(1) =: (\bar{u}, \bar{v})$ , that  $(u, v)(t) \in \pi(\gamma)$  if and only if  $t \in [t_0, t_1]$ , that  $z \in [z_0, z_1]$  for every z and t such that  $(u(t), v(t), z) \in \gamma$ . For every  $z \in [z_0, z_1]$ , let  $\phi_-^z$  and  $\phi_+^z$  be two normalized eigenvectors of  $H(\bar{u}, \bar{v}, z)$  corresponding to  $\lambda^-(\bar{u}, \bar{v}, z)$  and  $\lambda^+(\bar{u}, \bar{v}, z)$ , respectively. Then there exists C > 0 such that for every  $z \in [z_0, z_1]$  and every  $\epsilon > 0$ , the solution  $\psi$  of  $i\dot{\psi}(t) = H_f(u(\epsilon t), v(\epsilon t), z)\psi(t)$ ,  $\psi(0) = \phi_-^z$ , satisfies

$$\left\|\psi\left(\frac{1}{\epsilon}\right) - e^{i\xi}\phi_+^z\right\| < C\epsilon^{\frac{1}{3}},$$

for some  $\xi \in \mathbb{R}$ , possibly depending on  $\epsilon$  and z.



Figure 4.3 – A curve (u, v) as in the statement of Theorem 4.1.3

The chapter is organized as follows: In Section 4.2, we set the generic hypothesis on the Hamiltonians and provide the normal forms for the different types of systems briefly exposed in the introduction. In Section 4.3, we prove Theorem 4.1.2 and we study the singularities of the projection  $\pi(f)$  of Z(f) on the control plane. In Section 4.4, we study the dynamical properties of an isolated semi-conical intersection of eigenvalues and we prove Theorem 4.1.3. Then, in Section 4.6, we extend Theorem 4.1.3 for higher dimensional systems.

## 4.2 Basic facts and normal forms

### 4.2.1 Generic families of 2-level Hamiltonians

Define the smooth function  $f = (f_1, f_2) : \mathbb{R}^k \times \mathbb{R}^l \to \mathbb{R}^2$ . Denote by  $(e_1, \ldots, e_{k+l})$  the canonical basis of  $\mathbb{R}^{k+l}$ . Given a vector  $\eta \in \mathbb{R}^{k+l}$  and a smooth function  $g : \mathbb{R}^{k+l} \to \mathbb{R}^q$ ,  $q \in \mathbb{N}$ , we write  $\partial_{\eta}g$  for the directional derivative of g in the direction v and  $\partial_i$  for  $\partial_{e_i}$ ,  $i = 1, \ldots, k+l$ . For  $x \in \mathbb{R}^{k+l}$  and  $h \in T_x \mathbb{R}^{k+l} \approx \mathbb{R}^{k+l}$ , denote the differential of f at x applied to h by  $Df_x(h)$ .

In the following, we study generic situations in the cases k = 2, l = 0 and k = 2, l = 1. The coordinates  $(x_1, x_2)$  play the role of controls, and are denoted by (u, v), while—in the case l = 1—the coordinate  $x_3$  is a parameter and is denoted by z. The space  $C^{\infty}(\mathbb{R}^{2+l}, \mathbb{R}^2)$  is endowed in what follows with the  $C^{\infty}$ -Whitney topology.

We say that a property satisfied by  $f \in C^{\infty}(\mathbb{R}^{2+l}, \mathbb{R}^2)$  is generic if it is satisfied in an open and dense subset of the space  $C^{\infty}(\mathbb{R}^{2+l}, \mathbb{R}^2)$  endowed in what follows with the  $C^{\infty}$ -Whitney topology.

### The single system case k = 2, l = 0

Consider a 2-dimensional real Hamiltonian of the form

$$H_f(u,v) = \begin{pmatrix} f_1(u,v) & f_2(u,v) \\ f_2(u,v) & -f_1(u,v) \end{pmatrix},$$

where  $f_1, f_2 : \mathbb{R}^2 \to \mathbb{R}$  are smooth functions depending on 2 control variables (u, v). Define the smooth function  $f = (f_1, f_2) : \mathbb{R}^2 \to \mathbb{R}^2$ . Denote by  $\chi(f)(u, v)$  the Jacobian of f. By a slight abuse of notations, we will confound  $\chi(f)$  and  $\chi(H_f)$ . Notice that the eigenvalues of  $H_f$  are  $\lambda^+ = \sqrt{f_1^2 + f_2^2}$  and  $\lambda^- = -\sqrt{f_1^2 + f_2^2}$ . Define  $\text{Gap} = \lambda^+ - \lambda^- = 2\sqrt{f_1^2 + f_2^2}$ .

**Definition 4.2.1.** Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be such that  $f(\bar{u}, \bar{v}) = (0, 0)$ . We say that  $(\bar{u}, \bar{v})$  is conical in direction  $\nu \in \mathbb{R}^2$  if  $\partial_{\nu} f(\bar{u}, \bar{v}) \neq (0, 0)$ .

**Definition 4.2.2.** Let us consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Let  $(\bar{u}, \bar{v})$  be such that  $f(\bar{u}, \bar{v}) = (0, 0)$ .

- We say that  $(\bar{u}, \bar{v})$  is conical for f if  $\chi(f)(\bar{u}, \bar{v}) \neq 0$ .
- We say that  $(\bar{u}, \bar{v})$  is semi-conical if  $\nabla f_1(\bar{u}, \bar{v})$  and  $\nabla f_2(\bar{u}, \bar{v})$  are collinear, are not both zero, and the directional derivative  $\partial_\eta \chi(f)(\bar{u}, \bar{v})$  along  $\eta = (-\partial_2 f_j(\bar{u}, \bar{v}), \partial_1 f_j(\bar{u}, \bar{v}))$ is nonzero if  $j \in \{1, 2\}$  is such that  $\eta \neq 0$ . The direction spanned by  $\eta$  is called the non-conical direction at  $(\bar{u}, \bar{v})$ .

**Remark 4.2.3.** The definition of conical intersection for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  given above is equivalent the one used in [22], namely, a point  $(\bar{u}, \bar{v})$  such that  $f(\bar{u}, \bar{v}) = (0, 0)$  and there exists c > 0 such that, for every  $\eta \in \mathbb{R}^2$  of norm 1 and  $\delta > 0$  small enough, we have  $\frac{Gap((\bar{u}, \bar{v}) + \delta \eta)}{\delta} \ge c$ 

### Chapter 4. Ensemble control of quantum systems with two controls: non-conical case

The following proposition states that semi-conical points are isolated zeros of f. The proof can be deduced quite easily from the definition, for simplicity the proof can be deduced directly from the normal forms given in Section 4.2.3.

**Proposition 4.2.4** (Semi-conical intersections are isolated). Consider a semi-conical point  $(\bar{u}, \bar{v})$  for f. Then there exists a neighborhood V of  $(\bar{u}, \bar{v})$  in  $\mathbb{R}^2$  such that for every  $(u, v) \in V \setminus \{(\bar{u}, \bar{v})\}, f(u, v) \neq (0, 0).$ 

We recall here that conical intersections are generic. Denote by  $J^{n_1}(\mathbb{R}^{n_2}, \mathbb{R}^{n_3})$  the set of  $n_1$ -order jets of functions from  $\mathbb{R}^{n_2}$  to  $\mathbb{R}^{n_3}$ . Define

$$S_r = \{j^1(f)(0) \in J^1(\mathbb{R}^2, \mathbb{R}^2) \mid f(0) = 0, \operatorname{rank}(Df(0)) = r\}, \quad r = 0, 1.$$

It is easy to check that  $S_0$ ,  $S_1$  are two submanifolds of  $J^1(\mathbb{R}^2, \mathbb{R}^2)$  of codimension 6 and 3, respectively. One can easily show that the algebraic subset  $S_0 \cup S_1$  of  $J^1(\mathbb{R}^3, \mathbb{R}^2)$  admits a Whitney stratification (see [42] Part I, Chapter 1) whose strata have codimension strictly larger than the dimension of  $\mathbb{R}^2$ . By Thom's transversality theorem (see, e.g., [41]) used in combination with [42, §1.3.2],

$$U = \{ f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \mid j^1(f)(\mathbb{R}^2) \cap (S_1 \cup S_2) = \emptyset \}$$
  
=  $\{ f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \mid j^1(f)(\mathbb{R}^2) \cap S_1 = \emptyset \} \cap \{ f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \mid j^1(f)(\mathbb{R}^2) \cap S_2 = \emptyset \}$ 

is an open and dense subset of  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ .

### The parametric case k = 2, l = 1

Consider a 2-dimensional real Hamiltonian of the form

$$H_f(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix},$$

where  $f_1, f_2 : \mathbb{R}^3 \to \mathbb{R}$  are smooth functions depending on two control variables (u, v)and one parameter z. Define the smooth function  $f = (f_1, f_2) : \mathbb{R}^3 \to \mathbb{R}^2$ . An *eigenvalue intersection* is a point (u, v, z) such that f(u, v, z) = (0, 0).

**Definition 4.2.5.** For  $i, j \in \{1, ..., 3\}$ , let  $\chi_{ij}(f)$  be the Jacobian of the restriction of f to the plane parallel to  $span(e_i, e_j)$ , i.e.,

$$\chi_{ij}(f)(u,v,z) = \begin{vmatrix} \partial_i f_1(u,v,z) & \partial_j f_1(u,v,z) \\ \partial_i f_2(u,v,z) & \partial_j f_2(u,v,z) \end{vmatrix}.$$

By a slight abuse of notation, we set  $\chi(f) = \chi_{12}(f)$ .

In order to extend the definition of conical and semi-conical intersections for a parametrized Hamiltonian, we need to add to the previous definitions some regularity assumptions with respect to the parameter z.

**Definition 4.2.6.** Let (u, v, z) be a point such that f(u, v, z) = (0, 0). It is said to be conical for the family (F-conical) if (u, v) is conical for  $f(\cdot, \cdot, z)$  and  $\partial_3 f(u, v, z) \neq (0, 0)$ . It is said to be semi-conical for the family (F-semi-conical) if it is semi-conical for  $f(\cdot, \cdot, z)$  and f is a submersion at (u, v, z).

**Proposition 4.2.7.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , f is a submersion at every point of Z(f) and the set  $Z(f) = \{(u, v, z) \in \mathbb{R}^3 \mid f(u, v, z) = 0\}$  is a submanifold of  $\mathbb{R}^3$  of dimension 1.

Proof. Define

$$\Sigma_r = \{ j^1(f)(0) \in J^1(\mathbb{R}^3, \mathbb{R}^2) \mid f(0) = 0, \operatorname{rank}(Df(0)) = r \}, \qquad r = 0, 1$$

Notice that  $\Sigma_0$  and  $\Sigma_1$  are smooth submanifolds of  $J^1(\mathbb{R}^3, \mathbb{R}^2)$  of codimensions 8 and 4, respectively. One can easily show that the algebraic subset  $\Sigma_0 \cup \Sigma_1$  of  $J^1(\mathbb{R}^3, \mathbb{R}^2)$  admits a Whitney stratification (see [42] Part I, Chapter 1) whose strata have codimension strictly larger than 3. By Thom's transversality theorem (see, e.g., [41]) used in combination with [42, §1.3.2],  $U = \{f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2) \mid j^1(f)(\mathbb{R}^3) \cap (\Sigma_0 \cup \Sigma_1) = \emptyset\}$  is a an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Hence, f is generically a submersion at every point  $(u, v, z) \in Z(f)$ . The proposition is proved.

In the next two propositions we provide a geometric description of the curve Z(f) and we show its links with the conicity properties of f.

**Proposition 4.2.8.** A point  $(\bar{u}, \bar{v}, \bar{z}) \in Z(f)$  is conical for  $f(\cdot, \cdot, \bar{z})$  if and only if f is a submersion at  $(\bar{u}, \bar{v}, \bar{z})$  such that Z(f) is locally near  $(\bar{u}, \bar{v}, \bar{z})$  a one dimensional submanifold transversal to the plane of  $\mathbb{R}^3$  of equation  $z = \bar{z}$ .

Proof. Let  $(\bar{u}, \bar{v}, \bar{z}) \in Z(f)$  be conical for  $f(\cdot, \cdot, \bar{z})$ . By definition, we have  $\chi(f)(\bar{u}, \bar{v}, \bar{z}) \neq 0$ , hence f is a submersion at  $(\bar{u}, \bar{v}, \bar{z})$ . It follows that Z(f) is locally near  $(\bar{u}, \bar{v}, \bar{z})$  a one dimensional submanifold of  $\mathbb{R}^3$ . Fix a local smooth regular parametrization  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$  of  $c \subset Z(f)$  such that  $c(\bar{t}) = (\bar{u}, \bar{v}, \bar{z})$ . Assume for the sake of contradiction that  $\dot{z}(\bar{t}) = 0$  where  $\bar{t} \in \mathbb{R}$ . Differentiating the condition f(c(t)) = 0 for every  $t \in \mathbb{R}$ , we have

$$\begin{cases} \dot{u}(\bar{t})\partial_1 f_1(\bar{u},\bar{v},\bar{z}) + \dot{v}(\bar{t})\partial_2 f_1(\bar{u},\bar{v},\bar{z}) = 0\\ \dot{u}(\bar{t})\partial_1 f_2(\bar{u},\bar{v},\bar{z}) + \dot{v}(\bar{t})\partial_2 f_2(\bar{u},\bar{v},\bar{z}) = 0 \end{cases}$$
(4.4)

Hence  $\chi(f)(\bar{u}, \bar{v}, \bar{z}) = 0$ , that is impossible.

Conversely, consider a submersion f at  $(\bar{u}, \bar{v}, \bar{z})$  such that Z(f) is locally near  $(\bar{u}, \bar{v}, \bar{z})$ a one dimensional submanifold transversal to the plane of  $\mathbb{R}^3$  of equation  $z = \bar{z}$ . For the sake of contradiction, assume that  $(\bar{u}, \bar{v}, \bar{z})$  is non conical for  $f(\cdot, \cdot, \bar{z})$ . By definition, there exists a direction  $\eta \in \mathbb{R}^2$  such that  $\partial_{(\eta,0)}f(\bar{u}, \bar{v}, \bar{z}) = 0$ . Fix a local smooth regular parametrization  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$  of  $c \subset Z(f)$  such that  $c(\bar{t}) = (\bar{u}, \bar{v}, \bar{z})$  where  $\bar{t} \in \mathbb{R}$ . Differentiating the condition f(c(t)) = 0 for every  $t \in \mathbb{R}$ , we have  $(\dot{u}(\bar{t}), \dot{v}(\bar{t}), \dot{z}(\bar{t})) \in$ ker  $Df_{(\bar{u},\bar{v},\bar{z})}$ . f being a submersion at  $(\bar{u}, \bar{v}, \bar{z})$ , we can deduce that  $(\dot{u}(\bar{t}), \dot{v}(\bar{t}), \dot{z}(\bar{t}))$  is colinear to  $(\eta, 0)$ . Hence we get  $\dot{z}(\bar{t}) = 0$ , that is impossible.

**Proposition 4.2.9.** Assume that f is a submersion locally near  $(\bar{u}, \bar{v}, \bar{z})$  and that  $(\bar{u}, \bar{v}, \bar{z})$ is non-conical for  $f(\cdot, \cdot, \bar{z})$  in the direction  $\eta \in \mathbb{R}^2 \setminus \{(0,0)\}$ . Fix a local smooth regular parametrization  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$  of  $c \subset Z(f)$  such that  $c(\bar{t}) = (\bar{u}, \bar{v}, \bar{z})$  and  $\dot{z}(\bar{t}) = 0$  where  $\bar{t} \in \mathbb{R}$ . Then we have the equivalence

$$\ddot{z}(\bar{t}) = 0 \iff \partial_{(\eta,0)}\chi(f)(\bar{u},\bar{v},\bar{z}) = 0$$

In particular, if  $(\bar{u}, \bar{v}, \bar{z})$  is F-semi-conical for f then  $\ddot{z}(\bar{t}) \neq 0$ .

Proof. Without loss of generality, we can assume  $(\bar{u}, \bar{v}, \bar{z}) = (0, 0, 0)$ . Under the assumption that f is a submersion, we have  $(\partial_1 f_1(0), \partial_2 f_1(0)) \neq (0, 0)$  or  $(\partial_1 f_2(0), \partial_2 f_2(0)) \neq (0, 0)$ . Without loss of generality, assume  $\partial_1 f_1(0) = r \cos(\theta)$  and  $\partial_2 f_1(0) = r \sin(\theta)$  where  $\begin{pmatrix} -\sin(\theta) & -\cos(\theta) & 0 \end{pmatrix}$ 

r > 0 and  $\theta \in [0, 2\pi]$ . Define the matrix  $R_{\theta} = \begin{pmatrix} -\sin(\theta) & -\cos(\theta) & 0\\ \cos(\theta) & -\sin(\theta) & 0\\ 0 & 0 & 1 \end{pmatrix}$ . For every

 $(u, v, z) \in \mathbb{R}^3$ , define  $\tilde{f}(u, v, z) = (f \circ R_\theta)(u, v, z)$ . By simple computations, we have  $\partial_1 \tilde{f}(0) = 0$ . Moreover, differentiating at order 2, we see that for every  $(u, v, z) \in \mathbb{R}^3$ ,  $\partial_1 \chi(\tilde{f})(u, v, z) = \partial_{(\eta, 0)} \chi(f)(u, v, z)$ . Hence, it is sufficient to prove the proposition for  $\eta = (1, 0)$ .

Assume  $\partial_1 f_1(0) = \partial_1 f_2(0) = 0$ . Fix a local smooth regular parametrization  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$  of  $c \subset Z(f)$  such that  $c(\bar{t}) = 0$  and  $\dot{z}(\bar{t}) = 0$ . For every t in [0, 1], the equality  $\frac{d}{dt}f(c(t)) = 0$  provides  $\dot{c}(\bar{t})$  is collinear to  $e_1$ . The equality  $\frac{d^2}{dt^2}f(c(t)) = 0$  can be rewritten as

$$Df_{c(t)}(\ddot{c}(t)) + D^2 f_{c(t)}(\dot{c}(t), \dot{c}(t)) = 0,$$

that is

$$\begin{cases} \ddot{y}(\bar{t})\partial_2 f_1(0) + \ddot{z}(\bar{t})\partial_3 f_1(0) + \partial_{11}f_1(0) = 0\\ \ddot{y}(\bar{t})\partial_2 f_2(0) + \ddot{z}(\bar{t})\partial_3 f_2(0) + \partial_{11}f_2(0) = 0 \end{cases}.$$
(4.5)

Assume  $\ddot{z}(\bar{t}) = 0$ . Using Equation (4.5), the linear system  $\begin{cases} u\partial_2 f_1(0) + v\partial_{11}f_1(0) = 0\\ u\partial_2 f_2(0) + v\partial_{11}f_2(0) = 0 \end{cases}$ has a non null solution in (u, v). It implies its determinant is 0. By direct computa-

tions, we have, for every (u, v, z) in  $\mathbb{R}^3$ ,  $\partial_1 \chi(f)(u, v, z) = \begin{vmatrix} \partial_{11} f_1(u, v, z) & \partial_2 f_1(u, v, z) \\ \partial_{11} f_2(u, v, z) & \partial_2 f_2(u, v, z) \end{vmatrix} +$ 

 $\begin{vmatrix} \partial_1 f_1(u, v, z) & \partial_{21} f_1(u, v, z) \\ \partial_1 f_2(u, v, z) & \partial_{21} f_2(u, v, z) \end{vmatrix}.$  Using the hypothesis  $\partial_1 f_1(0) = \partial_1 f_2(0) = 0$ , we get  $\partial_1 \chi(f)(0) = 0$ 

Conversely, assume  $\partial_1 \chi(f)(0) = 0$ . By definition of the determinant,  $(\partial_{11}f_1(0), \partial_{11}f_2(0))$ is colinear to  $Df_0(e_2)$ . Assume  $\ddot{z}(\bar{t}) \neq 0$ . Equation (4.5) provides  $Df_0(e_3)$  is colinear to  $Df_0(e_2)$ . It is impossible under the assumption that f is a submersion near 0.  $\Box$ 

Remark 4.2.10. We have the following

- If  $(\bar{u}, \bar{v}, \bar{z})$  is F-conical and  $(u(t), v(t), z(t))_{t \in \mathbb{R}}$  such that  $(u(\bar{t}), v(\bar{t}), z(\bar{t})) = (\bar{u}, \bar{v}, \bar{z})$ is a smooth and regular local parametrization of Z(f) then  $\dot{z}(\bar{t}) \neq 0$ .
- Similarly, if  $(\bar{u}, \bar{v}, \bar{z})$  is F-semi-conical, then  $\dot{z}(\bar{t}) = 0$  and  $\ddot{z}(\bar{t}) \neq 0$ .
- In particular, F-semi-conical intersections are isolated in  $\mathbb{R}^3$ .

Using the notations of the previous proposition, we are going to see that, for a generic f in  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , the height function  $t \mapsto z(t)$  of the submanifold  $c \subset Z(f)$  may admit non-degenerate singular points that do not disappear by small perturbations of f. The following propositions guarantee that for a generic f, all intersections are either F-conical or F-semi-conical.

**Proposition 4.2.11.** For a generic smooth function  $f : \mathbb{R}^3 \to \mathbb{R}^2$ , for every  $(u, v, z) \in \mathbb{R}^3$  such that (u, v, z) is a conical intersection for  $f(\cdot, \cdot, z)$ , we have that (u, v, z) is a F-conical intersection for f.

Proof. The set  $Q = \{j^1(f)(0) \in J^1(\mathbb{R}^3, \mathbb{R}^2) \mid f(0) = 0, \partial_3 f(0) = 0\}$  is a Whitney stratified subset of  $J^1(\mathbb{R}^3, \mathbb{R}^2)$  of codimension 4. By Thom's transversality theorem (see, e.g., [41]) used in combination with [42, §1.3.2], the set  $\{f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2) \mid j^1(f)(\mathbb{R}^3) \cap Q = \emptyset\}$  is an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

**Proposition 4.2.12.** For a generic smooth function  $f : \mathbb{R}^3 \to \mathbb{R}^2$ , for every  $(u, v, z) \in \mathbb{R}^3$  such that (u, v, z) is a non-conical intersection for  $f(\cdot, \cdot, z)$ , we have that (u, v, z) is a *F*-semi-conical intersection for f.

Proof. Set

$$S_j = \left\{ j^2(f)(0) \in J^2(\mathbb{R}^3, \mathbb{R}^2) \mid f(0) = (0, 0), \partial_1 f_j(0) = \partial_2 f_j(0) = 0 \right\}, \qquad j = 0, 1.$$

Then  $S_0$  and  $S_1$  are smooth subspaces of  $J^2(\mathbb{R}^3, \mathbb{R}^2)$  of codimension 4. Define

$$\eta = (-\partial_2 f_1(0), \partial_1 f_1(0), 0),$$

$$S_2 = \left\{ j^2(f)(0) \in J^2(\mathbb{R}^3, \mathbb{R}^2) \mid f(0) = 0, \ (\partial_1 f_1(0), \partial_2 f_1(0)) \neq 0, \ \chi(f)(0) = 0, \ \partial_\eta \chi(f)(0) = 0 \right\}$$

Chapter 4. Ensemble control of quantum systems with two controls: non-conical case

and

$$\tilde{S}_2 = \left\{ j^2(f)(0) \in J^2(\mathbb{R}^3, \mathbb{R}^2) \mid f(0) = 0, \ \chi(f)(0) = 0, \ \partial_\eta \chi(f)(0) = 0 \right\}$$

We are going to prove that  $S_2$  is a smooth submanifold of  $J^2(\mathbb{R}^3, \mathbb{R}^2)$  of codimension 4, that is, that the equalities f(0) = 0,  $\chi(f)(0) = 0$  and  $\partial_\eta \chi(f)(0) = 0$  define independent equations in  $J^2(\mathbb{R}^3, \mathbb{R}^2)$  under the condition  $(\partial_1 f_1(0), \partial_2 f_1(0)) \neq 0$ . The equality f(0) =0 is clearly independent from the two others. Using the property that  $\partial_1 f_1(0) \neq 0$  or  $\partial_2 f_1(0) \neq 0$ , one easily establishes that  $\chi(f)(0) = 0$  and  $\partial_\eta \chi(f)(0) = 0$  define independent equations in  $J^2(\mathbb{R}^3, \mathbb{R}^2)$ .

One then can easily prove that the algebraic subset  $S = S_0 \cup S_1 \cup \tilde{S}_2 = S_0 \cup S_1 \cup S_2$ of  $J^2(\mathbb{R}^3, \mathbb{R}^2)$  admits a Whitney stratification whose strata have a codimension strictly larger than 3. By Thom's transversality theorem (see, e.g., [41]) used in combination with [42, §1.3.2], we get that  $O = \{f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2) \mid j^2(f)(\mathbb{R}^3) \cap S = \emptyset\}$  is an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

Considering  $G = U \cap R \cap O$  where U, R, and O are defined respectively in Propositions 4.2.7,4.2.11, and 4.2.12, G is an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

### 4.2.2 Admissible transformations

The aim of this section is to define admissible transformations in order to get normal forms for the Hamiltonians  $H_f$  defined for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  by  $H_f(u, v) = \begin{pmatrix} f_1(u, v) & f_2(u, v) \\ f_2(u, v) & -f_1(u, v) \end{pmatrix}$ and for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  by  $H_f(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix}$ . Consider the Schrödinger Equation, defined for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  by

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t))\psi(t), \qquad (4.6)$$

where  $\psi(t) \in \mathbb{C}^2$ , and the Ensemble Schrödinger Equation, defined for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  by

$$i\frac{d\psi(t)}{dt} = H_f(u(t), v(t), z)\psi(t), \qquad (4.7)$$

where  $\psi(t) \in \mathbb{C}^2$ . The control functions u, v are in  $L^{\infty}(\mathbb{R}, \mathbb{R})$  and the perturbation z belongs to  $[z_0, z_1]$  where  $z_0, z_1 \in \mathbb{R}$ .

The three transformations correspond to equivalent representations of the dynamical systems (4.6) and (4.7) achieved, respectively, by time-reparameterization, state-space diffeomorphism, and independent diffeomorphic transformations of both the space of controls and the space of perturbations.

**Definition 4.2.13.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  (respectively,  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ ) are time-equivalent at 0 if there exists a nowhere-vanishing function  $\xi \in$ 

 $C^{\infty}(\mathbb{R}^2,\mathbb{R})$  such that  $\tilde{f}(u,v) = \xi(u,v)f(u,v)$  (respectively,  $\tilde{f}(u,v,z) = \xi(u,v)f(u,v,z)$ ) in a neighborhood of 0.

**Remark 4.2.14.** A time-equivalence as introduced in Definition 4.2.13 with  $\xi > 0$  corresponds to a time-change in Equation (4.6). As for the case  $\xi < 0$ , consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ ,  $\psi_0, \psi_1 \in \mathbb{C}^2$  and a control path  $(u(\cdot), v(\cdot))$  such that the solution  $\psi(\cdot)$  of Equation (4.6) such that  $\psi(0) = \psi_0$  satisfies  $\psi(T) = \psi_1$  where T > 0. Then the solution  $\tilde{\psi}$  of Equation (4.6) associated with  $(u(T - \cdot), v(T - \cdot))$  such that  $\tilde{\psi}(0) = \bar{\psi}_1$  satisfies  $\tilde{\psi}(T) = \bar{\psi}_0$  (where we denote by  $\bar{x}$  the complex-conjugate of  $x \in \mathbb{C}^2$ ). Hence the equations (4.6) and

$$i\frac{d\psi(t)}{dt} = -H_f(u(t), v(t))\psi(t), \qquad (7-)$$

where  $\psi(t) \in \mathbb{C}^2$  have the same controllability properties. Hence time-equivalence can be defined rigorously for a function  $\xi \in C^{\infty}(\mathbb{R}^2, (-\infty, 0))$ . The same argument is also valid for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

**Definition 4.2.15.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  or  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  are left-equivalent if there exists  $P \in O_2(\mathbb{R})$  independent of u, v, z such that  $H_f = PH_{\tilde{f}}P^{-1}$ .

**Remark 4.2.16.** Let f be in  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  or  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Considering

$$P_{\theta,\zeta} = \begin{pmatrix} \cos(\theta) & -\zeta \sin(\theta) \\ \sin(\theta) & \zeta \cos(\theta) \end{pmatrix} \in \mathcal{O}_2(\mathbb{R}),$$

where  $\zeta = \pm 1$ , the associated left-equivalence transforms  $f = (f_1, f_2)$  into

$$\tilde{f} = (\cos(2\theta)f_1 - \zeta\sin(2\theta)f_2, \zeta\cos(2\theta)f_2 + \sin(2\theta)f_1).$$

**Remark 4.2.17.** Let f be in  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  or  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . If  $t \mapsto \psi(t)$  is a solution of Equation (4.6) or (4.7) associated with f and with initial condition  $\psi(0) = \psi_0$ , then  $t \mapsto Y(t) = P\psi(t)$  is a solution of Equation (4.7) associated with  $\tilde{f}$  satisfying  $Y(0) = P\psi_0$ . Hence, transitions for Y between the eigenstates of  $H_{\tilde{f}} = PH_fP^{-1}$  correspond to transitions for  $\psi$  between the eigenstates of  $H_f$ .

**Definition 4.2.18.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  are right-equivalent at 0 if there exists a diffeomorphism  $\phi \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  such that  $\phi(0) = 0$  and  $\tilde{f} = f \circ \phi$  in a neighborhood of 0.

**Definition 4.2.19.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  are right-equivalent at 0 if there exists a diffeomorphism  $\phi \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^3)$  of the form

 $\phi: (u, v, z) \mapsto (\phi_1(u, v), \phi_2(u, v), \phi_3(z)),$ 

where  $\phi_1, \phi_2 \in C^{\infty}(\mathbb{R}^2, \mathbb{R})$  and  $\phi_3 \in C^{\infty}(\mathbb{R}, \mathbb{R})$ , satisfying  $\phi(0) = 0$  and  $\tilde{f} = f \circ \phi$  in a neighborhood of 0.

Chapter 4. Ensemble control of quantum systems with two controls: non-conical case

Combining the previous three definitions we introduce the following notion of equivalence.

**Definition 4.2.20.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  are equivalent at 0 if there exists  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  with  $\phi$  as in Definition 4.2.19, and  $\zeta = \pm 1$  such that for every (u, v, z) in a neighborhood of 0,

$$\begin{cases} \tilde{f}_1(u,v) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v) - \zeta\sin(2\theta)f_2 \circ \phi(u,v)), \\ \tilde{f}_2(u,v) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v) + \zeta\cos(2\theta)f_2 \circ \phi(u,v)). \end{cases}$$

**Definition 4.2.21.** We say that two elements f and  $\tilde{f}$  of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  are equivalent at 0 if there exists  $(\phi, \theta, \xi) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^3) \times [-\pi, \pi] \times C^{\infty}(\mathbb{R}^2, \mathbb{R} \setminus \{0\})$  with  $\phi$  as in Definition 4.2.19, and  $\zeta = \pm 1$  such that for every (u, v, z) in a neighborhood of 0,

$$\begin{cases} \tilde{f}_1(u,v,z) = \xi(u,v)(\cos(2\theta)f_1 \circ \phi(u,v,z) - \zeta\sin(2\theta)f_2 \circ \phi(u,v,z)), \\ \tilde{f}_2(u,v,z) = \xi(u,v)(\sin(2\theta)f_1 \circ \phi(u,v,z) + \zeta\cos(2\theta)f_2 \circ \phi(u,v,z)). \end{cases}$$

An essential feature of admissible transformations is the following proposition

- **Proposition 4.2.22.** Let  $f, \tilde{f} \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  be equivalent. Then 0 is conical for f if and only if it is conical for  $\tilde{f}$  and 0 is semi-conical for f if and only if it is semi-conical for  $\tilde{f}$ .
  - Let  $f, \tilde{f} \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  be equivalent. Then 0 is F-conical for f if and only if it is F-conical for  $\tilde{f}$  and 0 is F-semi-conical for f if and only if it is F-semi-conical for  $\tilde{f}$ .

In order to prove the proposition, we first obtain the following lemma.

**Lemma 4.2.23.** Let  $f, \tilde{f} \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  be equivalent. Then, for every  $(u, v) \in \mathbb{R}^2$  in a neighborhood of 0,  $\chi(f)(u, v) \neq 0$  if and only if  $\chi(\tilde{f})(\phi(u, v)) \neq 0$ .

*Proof.* Notice that it is sufficient to prove the result separately for each of the three type of equivalence: time-equivalence, left-equivalence, and right-equivalence.

- For a time-equivalence  $\tilde{f}$  of f associated with  $\xi$  in  $C^{\infty}(\mathbb{R}^2, \mathbb{R})$ , we have by direct computations of the determinant,  $\chi(\tilde{f})(u, v) = \xi(u, v)^2 \chi(f)(u, v)$  where  $\xi$  is non-vanishing.
- For a left-equivalence  $\tilde{f}$  of f, we have

$$\nabla \tilde{f}_1 = \cos(2\theta)\nabla f_1 - \zeta \sin(2\theta)\nabla f_2, \quad \nabla \tilde{f}_2 = \cos(2\theta)\nabla f_2 + \zeta \sin(2\theta)\nabla f_1,$$

where  $\zeta = \pm 1$ . Hence, we obtain that, for every  $(u, v) \in \mathbb{R}^2$ ,  $\chi(\tilde{f})(u, v) = \zeta \chi(f)(u, v)$ .
— For a right-equivalence  $\tilde{f} = f \circ \phi$  of f associated with  $\phi$ , we have for every  $X = (u, v) \in \mathbb{R}^2$  in a neighborhood of (0, 0), and for  $i \in \{1, 2\}$ ,

$$\nabla(f_i \circ \phi)(u, v) = \begin{pmatrix} \partial_1 f_i(\phi(X)) \partial_1 \phi_1(X) + \partial_2 f_i(\phi(X)) \partial_1 \phi_2(X) \\ \partial_1 f_i(\phi(X)) \partial_2 \phi_1(X) + \partial_2 f_i(\phi(X)) \partial_2 \phi_2(X) \end{pmatrix}.$$

Hence,

$$\begin{aligned} \chi(f \circ \phi)(X) \\ &= \det(\nabla(f_1 \circ \phi)(X), \nabla(f_2 \circ \phi)X)) \\ &= \partial_1 f_1(\phi(X)) \partial_2 f_2(\phi(X)) \partial_1 \phi_1(X) \partial_2 \phi_2(X) - \partial_1 f_2(\phi(X)) \partial_2 f_1(\phi(X)) \partial_1 \phi_1(X) \partial_2 \phi_2(X) \\ &+ \partial_2 f_1(\phi(X)) \partial_1 f_2(\phi(X)) \partial_1 \phi_2(X) \partial_2 \phi_1(X) - \partial_2 f_2(\phi(X)) \partial_1 f_1(\phi(X)) \partial_1 \phi_2(X) \partial_2 \phi_1(X). \end{aligned}$$

We deduce that

$$\chi(f \circ \phi)(X) = \chi(f)(\phi(X)) \begin{vmatrix} \partial_1 \phi_1(X) & \partial_2 \phi_1(X) \\ \partial_1 \phi_2(X) & \partial_2 \phi_2(X) \end{vmatrix},$$

where  $\phi$  is a diffeomorphism in a neighborhood of (0,0,0) satisfying  $\partial_1 \phi_3(0) = \partial_2 \phi_3(0) = 0$ , we obtain  $\begin{vmatrix} \partial_1 \phi_1(0) & \partial_2 \phi_1(0) \\ \partial_1 \phi_2(0) & \partial_2 \phi_2(0) \end{vmatrix} \neq 0.$ 

By similar arguments, we have the following lemma.

**Lemma 4.2.24.** Let  $f, \tilde{f} \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  be equivalent. Then, for every  $(u, v, z) \in \mathbb{R}^3$  in a neighborhood of 0,  $\chi(f)(u, v, z) \neq 0$  if and only if  $\chi(\tilde{f})(\phi(u, v, z)) \neq 0$ .

Proof. (Proof of Proposition 4.2.22) We prove the first claim. Consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . For a right-equivalence  $\tilde{f}$  of f, Lemma 4.2.23 proves that 0 is conical for f if and only if 0 is conical for  $\tilde{f}$ . Now assume that 0 is semi-conical for  $f \in C^{\infty}$  With the previous computations, we have seen in the proof of Lemma 4.2.23 that for every right-equivalence  $\tilde{f} = f \circ \phi$  where  $\phi : \mathbb{R}^2 \to \mathbb{R}^2$ , we have, for every  $(u, v) \in \mathbb{R}^2$  in a neighborhood of (0, 0),

$$\chi(\tilde{f})(u,v) = \chi(f)(\phi(u,v)) \begin{vmatrix} \partial_1\phi_1(u,v) & \partial_2\phi_1(u,v) \\ \partial_1\phi_2(u,v) & \partial_2\phi_2(u,v) \end{vmatrix}.$$

Hence denoting by  $\eta$  the non-conical direction, we get  $\partial_{\eta}\chi(f)(u,v) \neq 0$  if and only if  $\partial_{\tilde{\eta}}\chi(\tilde{f})(\phi(u,v)) \neq 0$  where  $\tilde{\eta} = D\phi_{\phi(u,v)}^{-1}(\eta)$ . By the computations made in Lemma 4.2.23, we get that, if  $\tilde{f}$  is a left or right equivalence of f, then  $\nabla \tilde{f}_1(0) = \nabla \tilde{f}_2(0) = 0$  if and only if  $\nabla f_1(0) = \nabla f_2(0) = 0$ , and  $\chi(\tilde{f})(0) \neq 0$  if and only if  $\chi(f)(0) \neq 0$ . Then we get the expected equivalence.

By similar computations and noticing for the F-conical case that if two elements of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  are equivalent then  $\partial_3 f(0) \neq 0$  if and only if  $\partial_3 \tilde{f}(0) \neq 0$  and for the F-semiconical case that f is a submersion at 0 if and only if  $\tilde{f}$  is a submersion at 0, we prove the second claim concerning  $f, \tilde{f} \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

4.2.3 Normal forms for the non parametric case

#### **Conical intersection**

Define  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  such that  $\chi(f)(0) \neq 0$ . In this case, f is a diffeomorphism in a neighborhood of 0. Hence f is right-equivalent to  $\mathrm{Id} : \mathbb{R}^2 \to \mathbb{R}^2$ . The normal form provides the well-known Hamiltonian  $H(u, v) = \begin{pmatrix} u & v \\ v & -u \end{pmatrix}$ , for  $u, v \in \mathbb{R}^2$ .

#### Semi-conical intersection

The main result of this section is the following theorem.

**Theorem 4.2.25.** Assume that 0 is semi-conical for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R})$ . Then f is equivalent to  $(u, v) \mapsto \begin{pmatrix} h(u)u\\ u+v^2 \end{pmatrix}$  where  $h : \mathbb{R} \to \mathbb{R}$  is a smooth function satisfying h(0) = 1.

The algorithm that we will refer as (A) to get the normal form is the following:

- **STEP 1:** By a left-equivalence we transform  $f_1$  and  $f_2$  into two functions  $\tilde{f}_1$  and  $\tilde{f}_2$  such that  $\nabla \tilde{f}_1(0) = \nabla \tilde{f}_2(0) \neq 0$ .
- STEP 2: By a right-equivalence, we bring the non-conical direction to  $\operatorname{span}(e_2)$ .
- **STEP 3:** By a right-equivalence and using Lemma 4.7.11, we transform the functions  $f_1$  and  $f_2$  into a suitable form.

#### Proof of Theorem 4.2.25: STEP 1

**Proposition 4.2.26.** Consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  having a semi-conical intersection at 0. Then there exists  $\tilde{f}$  left-equivalent to f such that  $\nabla \tilde{f}_1(0) = \nabla \tilde{f}_2(0) \neq 0$ .

*Proof.* Without loss of generality, we can assume  $\nabla f_1(0) \neq 0$ . Define  $\alpha \in \mathbb{R}$  such that  $\nabla f_2(0) = \alpha \nabla f_1(0)$ .

Denote by  $\tilde{f}$  the function obtained by applying to f the left-equivalence associated with  $\theta \in [-\pi, \pi]$  and  $\zeta = 1$  as in Remark 4.2.16. Hence,

$$\nabla \tilde{f}_1 = \cos(2\theta)\nabla f_1 - \sin(2\theta)\nabla f_2, \quad \nabla \tilde{f}_2 = \cos(2\theta)\nabla f_2 + \sin(2\theta)\nabla f_1.$$

We have  $\nabla \tilde{f}_1(0) = \nabla \tilde{f}_2(0)$  if and only if  $\langle \begin{pmatrix} \cos(2\theta) \\ \sin(2\theta) \end{pmatrix}, \begin{pmatrix} 1-\alpha \\ -(1+\alpha) \end{pmatrix} \rangle = 0$ . It is clearly possible to choose  $\theta \in [-\pi, \pi]$  satisfying the previous condition, the proposition is proved.

Until the end of this section, we assume that 0 is semi-conical for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ .

#### Proof of Theorem 4.2.25: STEP 2

**Proposition 4.2.27.** Assume that 0 is semi-conical for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . There exists a right-equivalence  $\phi : \mathbb{R}^2 \to \mathbb{R}^2$  of f such that  $\tilde{f} = f \circ \phi$  satisfies

$$\partial_2 \tilde{f}_1(0,0) = \partial_2 \tilde{f}_2(0,0) = 0.$$

Moreover, if  $\nabla f_1(0,0) \neq 0$  and  $\nabla f_2(0,0) \neq 0$ , then we have

$$\partial_1 \tilde{f}_1(0,0) \neq 0, \ \partial_1 \tilde{f}_2(0,0) \neq 0.$$

*Proof.* Consider  $r_1, r_2 \in [0, +\infty)$  and  $\beta_1 \in [0, 2\pi]$  such that

$$\partial_2 f_1(0,0) = r_1 \cos(\beta_1), \ \partial_1 f_1(0,0) = r_1 \sin(\beta_1), \ \partial_2 f_2(0,0) = r_2 \cos(\beta_1), \ \partial_1 f_2(0,0) = r_2 \sin(\beta_1)$$

Introducing the right-equivalence  $\phi(u, v) = \begin{pmatrix} -\sin(\beta_1) & \cos(\beta_1) \\ -\cos(\beta_1) & -\sin(\beta_1) \end{pmatrix} \begin{pmatrix} u \\ v \end{pmatrix}$  and  $\tilde{f} = (f \circ \phi)$ , we have  $D\tilde{f}(0,0) = \begin{pmatrix} -r_1 & 0 \\ -r_2 & 0 \end{pmatrix}$ . If we assume  $\nabla f_1(0,0) \neq 0$  and  $\nabla f_2(0,0) \neq 0$ , then we

can choose  $r_1$  and  $r_2$  which are non-zero, and we get the result.

For  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , the propositions 4.2.26 and 4.2.27 lead us to consider the next condition that we will refer to as condition (**SC2**),

$$f(0) = 0, \ \partial_2 f(0) = 0, \ \partial_1 f_1(0) = \partial_1 f_2(0) \neq 0, \ \partial_2 \chi(f)(0) \neq 0.$$
 (SC2)

#### Proof of Theorem 4.2.25: STEP 3

**Proposition 4.2.28.** Let  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  satisfy Condition (SC2). Then there exists  $h \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfying h(0) = 1 such that f is right-equivalent to  $(u, v) \mapsto \begin{pmatrix} h(u)u \\ u+v^2 \end{pmatrix}$  or  $(u, v) \mapsto \begin{pmatrix} h(u)u \\ u-v^2 \end{pmatrix}$ .

*Proof.* Because of the condition  $\partial_1 f_1(0) \neq 0$ , the map  $\Phi : (u, v) \mapsto (f_1(u, v), v)$  is a diffeomorphism in a neighborhood of 0 and  $g = f \circ \Phi^{-1}$  is right-equivalent to f. Locally near 0 we have

$$g_1(u,v) = u,$$
  $g_2(u,v) = f_2(G(u,v),v),$ 

Chapter 4. Ensemble control of quantum systems with two controls: non-conical case

where G is a smooth function satisfying  $\partial_1 G(u, v) = \frac{1}{\partial_1 f_1(G(u,v),v)}$  and  $\partial_2 G(u, v) = -\frac{\partial_2 f_1(G(u,v),v)}{\partial_1 f_1(G(u,v),v)}$ . Hence,

$$\partial_1 g_2(u,v) = \partial_1 f_2(G(u,v),v) \partial_1 G(u,v),$$
  
$$\partial_2 g_2(u,v) = \partial_1 f_2(G(u,v),v) \partial_2 G(u,v) + \partial_2 f_2(G(u,v),v).$$

The condition  $\partial_1 f_1(0) \neq 0 \neq \partial_1 f_2(0)$  implies that  $\partial_1 g_2(0) \neq 0$ . Moreover,

$$\begin{aligned} \partial_{22}g_2(u,v) = &\partial_{22}G(u,v)\partial_1 f_2(G(u,v),v) + \partial_{22}f_2(G(u,v),v) \\ &+ \partial_2 G(u,v)^2 \partial_{11}f_2(G(u,v),v) + 2\partial_2 G(u,v)\partial_{12}f_2(G(u,v),v) \end{aligned}$$

Evaluating at 0 and noticing that  $\partial_1 G(0) = \frac{1}{\partial_1 f_1(0)}$  and  $\partial_2 G(0) = 0$ , we get  $\partial_2 g_2(0) = 0$ and

$$\partial_{22}g_2(0) = \partial_{22}G(0)\partial_1f_2(0) + \partial_{22}f_2(0) = \frac{\partial_2\chi(f)(0)}{\partial_1f_1(0)} \neq 0,$$

since f satisfies Condition (**SC2**). By the Implicit Function Theorem,  $\partial_2 g_2(u, v)$  vanishes on a smooth curve  $v = \eta(u)$  in a neighborhood of 0. By Lemma 4.7.11, there exist two smooth functions  $m \in C^{\infty}(\mathbb{R}^2, \mathbb{R})$  and  $f_0 \in C^{\infty}(\mathbb{R}, \mathbb{R})$  such that  $g_2(u, v) = (v - \eta(u))^2 m(u, v) + f_0(u)$  in a neighborhood of 0. The conditions  $\partial_1 g_2(0) \neq 0$  and  $\partial_{22} g_2(0) \neq 0$ yield  $f'_0(0) \neq 0$  and  $m(0) \neq 0$ , respectively. Applying first the right-equivalence  $(u, v) \mapsto \begin{pmatrix} u \\ (v - \eta(u))\sqrt{|m(u, v)|} \end{pmatrix}$  and then  $(u, v) \mapsto \begin{pmatrix} f_0(u) \\ v \end{pmatrix}$ , we deduce that f is right-equivalent to  $(u, v) \mapsto \begin{pmatrix} h(u)u \\ u + \operatorname{sign}(m(0, 0))v^2 \end{pmatrix}$ , for some smooth function h obtained by inversion of  $f_0$ . Noticing that the quantity  $\frac{\partial_1 f_1(0)}{\partial_1 f_2(0)}$  is invariant by right-equivalence, this provides  $h(0) = \frac{\partial_1 f_1(0)}{\partial_1 f_2(0)} = 1$ .

*Proof.* (Proof of Theorem 4.2.25) We start the proof by showing that

$$f: (u,v) \mapsto \begin{pmatrix} h(u)u\\ u-v^2 \end{pmatrix}$$

is right-time-equivalent to

$$(u,v)\mapsto \begin{pmatrix} \tilde{h}(u)u\\u+v^2 \end{pmatrix},$$

where  $\tilde{h}$  is in  $C^{\infty}(\mathbb{R}, \mathbb{R})$ . Indeed, we can apply the right-equivalence  $(u, v) \mapsto (-u, v)$ , then the time-equivalence associated with  $\xi \equiv -1$ . The result follows defining  $\tilde{h}(u) = h(-u)$ . Theorem 4.2.25 then follows from Propositions 4.2.27 and 4.2.28.

#### Conditions that guarantee $h'(0) \neq 0$

In this section, we show that we can define the Algorithm (A) such we can define a surjective mapping  $f \mapsto h_f$  from the set of functions in  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  having a semi-conical intersection at 0 to the set of functions in  $C^{\infty}(\mathbb{R}, \mathbb{R})$  that are equal to 1 at 0, where for every f,  $h_f$  corresponds to the function h obtained by the Algorithm (A). Then we show that

 $V_M = \left\{ j^2(f)(0,0) \in J^2(\mathbb{R}^2,\mathbb{R}^2) \mid 0 \text{ is a semi-conical intersection for } f, \ h'_f(0) = 0 \right\}$ 

is a closed Whitney stratified set of codimension 4.

For the first step of the Algorithm (A), consider a function  $f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ having a semi-conical intersection at 0. Without loss of generality, we can assume  $\nabla f_1(0) \neq 0$ . Define  $\alpha \in \mathbb{R}$  as in 4.2.26 as the unique real number such that  $\nabla f_2(0) = \alpha \nabla f_1(0)$ , and consider the fonction  $\tilde{f} \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , left-equivalent to f, defined by

$$\tilde{f}_1(u,v) = \frac{1}{\sqrt{2+2\alpha^2}} \left( (1+\alpha)f_1(u,v) - (1-\alpha)f_2(u,v) \right)$$
$$\tilde{f}_2(u,v) = \frac{1}{\sqrt{2+2\alpha^2}} \left( (1+\alpha)f_2(u,v) + (1-\alpha)f_1(u,v) \right),$$

for every  $(u, v) \in \mathbb{R}^2$ . Consider the coordinates  $(x_{jk} = \partial_j f_k(0))_{j,k \in \{1,2\}}, (\tilde{x}_{jk} = \partial_j \tilde{f}_k(0))_{j,k \in \{1,2\}}, (y_{jk}^l = \partial_{jk}^2 f_l(0))_{j,k,l \in \{1,2\}}, (\tilde{y}_{jk}^l = \partial_{jk}^2 \tilde{f}_l(0))_{j,k,l \in \{1,2\}}, (\tilde{y}_{jk}^l = \partial_{jk}^2 \tilde{f}_l(0))_{j,k,l \in \{1,2\}}, (\tilde{y}_{jk}^l = \partial_{jk}^2 \tilde{f}_l(0))_{j,k,l \in \{1,2\}}, (\tilde{y}_{jk}^l = \partial_j \tilde{f}_k(0))_{j,k,l \in \{1,2\}}, (\tilde{y}_{jk}^l = \partial$ 

$$F: \left( (x_{j1})_{j \in \{1,2\}}, (y_{jk}^l)_{j,k,l \in \{1,2\}}, \alpha \right) \mapsto \left( (\tilde{x}_{j1})_{j \in \{1,2\}}, (\tilde{y}_{jk}^l)_{j,k,l \in \{1,2\}} \right).$$

We prove that F is a submersion from  $(\mathbb{R}^2 \setminus \{(0,0)\}) \times \mathbb{R}^7$  to  $(\mathbb{R}^2 \setminus \{(0,0)\}) \times \mathbb{R}^6$ , so that if S is a codimension k submanifold of  $\mathbb{R}^8$ , then  $F^{-1}(S)$  is a codimension k submanifold of  $\mathbb{R}^9$ . To get this, we notice that for every (u, v),

$$\begin{pmatrix} \tilde{f}_1(u,v)\\ \tilde{f}_2(u,v) \end{pmatrix} = M(\alpha) \begin{pmatrix} f_1(u,v)\\ f_2(u,v) \end{pmatrix},$$

where  $M(\alpha)$  is an invertible 2 × 2 matrix depending only on  $\alpha$ . Hence it is sufficient to prove that the mapping

 $((x_{j1})_{j\in\{1,2\}}, \alpha) \mapsto ((\tilde{x}_{j1})_{j\in\{1,2\}})$ 

is a submersion from  $(\mathbb{R}^2 \setminus \{(0,0)\}) \times \mathbb{R}$  to  $\mathbb{R}^2 \setminus \{(0,0)\}$ . For every  $(x_{11}, x_{21}, \alpha) \in \mathbb{R}^2 \setminus \{(0,0)\} \times \mathbb{R}$ , and for every  $j \in \{1,2\}$ , we have by definition of  $\alpha$ ,  $\tilde{x}_{j1} = \frac{\sqrt{1+\alpha^2}}{\sqrt{2}} x_{j1}$ . We can conclude by direct computations.

For the second step of the Algorithm (A), for every  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , define the rightequivalence of f as in Proposition 4.2.27 by  $\hat{f}(u, v) = f(\partial_1 f_1(0)u - \partial_2 f_1(0)v, -\partial_2 f_1(0)u +$   $\partial_1 f_1(0)v$ ). Consider the coordinates  $(x_{jk} = \partial_j f_k(0))_{j,k \in \{1,2\}}, (\hat{x}_{jk} = \partial_j \hat{f}_k(0))_{j,k \in \{1,2\}}, (y_{jk}^l = \partial_{jk}^2 f_l(0))_{j,k,l \in \{1,2\}}, and (\hat{y}_{jk}^l = \partial_{jk}^2 \hat{f}_l(0))_{j,k,l \in \{1,2\}}$  in  $J^2(\mathbb{R}^2, \mathbb{R}^2)$ . Notice that we have  $\hat{x}_{21} = 0$ . By direct computations, we can show that the mapping

$$\hat{F}: \left( (x_{jk})_{j,k \in \{1,2\}}, (y_{jk}^l)_{j,k,l \in \{1,2\}} \right) \mapsto \left( (\hat{x}_{jk})_{(j,k) \in \{1,2\}^2 \setminus \{(2,1)\}}, (\hat{y}_{jk}^l)_{j,k,l \in \{1,2\}} \right)$$

is a submersion from  $\mathbb{R}^{10}$  to  $\mathbb{R}^{9}$ .

For the third step of the Algorithm (A), for every  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  satisfying Condition (SC2), we apply the same right-transformation as in Proposition 4.2.28, and we get a function  $h \in C^{\infty}(\mathbb{R}, \mathbb{R})$  such that h(0) = 1.

Concatenating the three steps, we define a surjective mapping  $f \mapsto h_f$  from the set of functions in  $C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  having a semi-conical intersection at 0 to the set of functions in  $C^{\infty}(\mathbb{R}, \mathbb{R})$  that are equal to 1 at 0, where for every f,  $h_f$  corresponds to the function h obtained by the Algorithm (A).

We prove the following result concerning the third step of the Algorithm (A).

**Lemma 4.2.29.** For a function  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  satisfying Condition (SC2), consider the function  $(u, v) \mapsto \begin{pmatrix} uh(u) \\ u + v^2 \end{pmatrix}$  obtained from f by the same right-equivalence as in the proof of Proposition 4.2.28, where  $h \in C^{\infty}(\mathbb{R}, \mathbb{R})$  is such that h(0) = 1. Then we have h'(0) = 0 if and only if

$$(\partial_{11}f_2(0,0)\partial_1f_1(0,0) - \partial_1f_2(0,0)\partial_{11}f_1(0,0))\partial_2\chi(f)(0,0) = (\partial_1\chi(f)(0,0))^2.$$
(4.8)

*Proof.* By using the functions  $f_0$  and  $g_2$  and G as in the proof of Proposition 4.2.28, the following equivalences hold.

$$h'(0) = 0 \iff f_0''(0) = 0$$
  
 $\iff \partial_{11}g_2(0,0)\partial_{22}g_2(0,0) = (\partial_{12}g_2(0,0))^2.$  (E)

Now recall that  $\forall (u, v), g_2(u, v) = f_2(G(u, v), v)$ . Moreover, we have

$$\begin{aligned} \partial_2 g_2(0,0) &= 0\\ \partial_{22} g_2(0,0) &= \frac{\partial_2 \chi(f)(0,0)}{\partial_1 f_1(0,0)}\\ \partial_1 g_2(0,0) &= \frac{\partial_1 f_2(0,0)}{\partial_1 f_1(0,0)}. \end{aligned}$$

We can show that

$$\partial_{12}g_2(0,0) = \frac{\partial_1\chi(f)(0,0)}{(\partial_1f_1(0,0))^2}$$
  
$$\partial_{11}g_2(0,0) = \frac{\partial_{11}f_2(0,0)\partial_1f_1(0,0) - \partial_1f_2(0,0)\partial_{11}f_1(0,0)}{(\partial_1f_1(0,0))^3}$$

Replacing these expressions in Equation (E), we get the result.

Notice that Equality (4.8) is clearly independent from Condition (SC2). Using Lemma 4.2.29 and the fact that F and  $\hat{F}$  are submersions, we can conclude that

$$V_M = \left\{ j^2(f)(0,0) \in J^2(\mathbb{R}^2,\mathbb{R}^2) \mid 0 \text{ is a semi-conical intersection for } f, \ h'_f(0) = 0 \right\}$$

is a closed Whitney stratified set of codimension 4.

Consider now the set  $V_S = \{j^2(F)(0,0,0) \in J^2(\mathbb{R}^3,\mathbb{R}^2) \mid j^2(F(\cdot,\cdot,0))(0,0) \in V_M\}$  where  $j^2(F(\cdot,\cdot,0))$  is the 2-jet of  $f(\cdot,\cdot) = F(\cdot,\cdot,0)$  related to the variables u and v. By Thom's transversality theorem (see, e.g., [41]) used in combination with [42, §1.3.2], we can deduce that

$$\tilde{O} = \left\{ F \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2), \ j^2 F(\mathbb{R}^3) \cap V_S = \emptyset \right\}$$

is an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

#### 4.2.4 Normal forms for the parametric case

**Proposition 4.2.30.** For every  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  such that 0 is *F*-conical or *F*-semi-conical for f and  $\partial_3 f_2(0) \neq 0$ , let  $\beta(f) = \frac{\partial_3 f_1}{\partial_3 f_2}(0)$ . Then  $\beta(f)$  is invariant by right-equivalence.

*Proof.* Let  $\tilde{f}$  be right-equivalent to f and let  $\phi_1 \in C^{\infty}(\mathbb{R}^2, \mathbb{R}), \phi_2 \in C^{\infty}(\mathbb{R}^2, \mathbb{R}), \phi_3 \in C^{\infty}(\mathbb{R}, \mathbb{R})$  be such that  $\phi(u, v, z) = (\phi_1(u, v), \phi_2(u, v), \phi_3(z))$  is a right-equivalence between f and  $\tilde{f}$ . Differentiating w.r.t. z, we have

$$\frac{\partial_3 \tilde{f}_1}{\partial_3 \tilde{f}_2}(0) = \frac{\partial_3 f_1(0)\phi'_3(0)}{\partial_3 f_2(0)\phi'_3(0)} = \beta(f),$$

using the fact  $\phi'_3(0) \neq 0$  because  $\phi$  is a diffeomorphism.

#### Conical case

**Theorem 4.2.31.** Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Then 0 is *F*-conical for f if and only if there exist  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  satisfying  $h_1(0) = h_2(0) = 1$ , such that f is equivalent to

$$(u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z-u) \\ h_2(u, v, z)(z-v) \end{pmatrix}.$$

*Proof.* By the same argument as in the proof of Proposition 4.2.26, there exists  $\tilde{f}$  leftequivalent to f such that  $\partial_3 \tilde{f}_1(0) \neq 0$ ,  $\partial_3 \tilde{f}_2(0) \neq 0$  and  $\beta(\tilde{f}) = 1$ . Using the fact that  $\tilde{f}(0) = 0$ , we deduce that  $\tilde{f}_1$  and  $\tilde{f}_2$  vanish respectively on two smooth surfaces whose equations are of the form  $z = \eta_1(u, v)$  and  $z = \eta_2(u, v)$ , where  $\eta_1, \eta_2$  are smooth functions vanishing at (0, 0). By Lemma 4.7.10, there exist two smooth scalar functions  $(u, v, z) \mapsto \phi_1(u, v, z), (u, v, z) \mapsto \phi_2(u, v, z)$  such that

$$\tilde{f}_1(u,v,z) = \phi_1(u,v,z)(z-\eta_1(u,v)), \quad \tilde{f}_2(u,v,z) = \phi_2(u,v,z)(z-\eta_2(u,v)).$$

in the		1
н		
н		

Differentiating these expressions and evaluating them at 0 we get that

$$\chi(f)(0) = \phi_1(0)\phi_2(0)\chi(\eta)(0),$$

where  $\eta = (\eta_1, \eta_2)$ . Hence, by F-conicity of 0,  $\phi_1(0) \neq 0$ ,  $\phi_2(0) \neq 0$ , and  $\chi(\eta)(0) \neq 0$ . In particular,  $\eta$  is a diffeomorphism in a neighborhood of (0, 0). Then  $\tilde{f}$  is right-equivalent to

$$\tilde{f} \circ \mu^{-1}(u, v, z) = \begin{pmatrix} h_1(u, v, z)(u - z) \\ h_2(u, v, z)(v - z) \end{pmatrix},$$

where  $\mu : (u, v, z) \mapsto (\eta_1(u, v), \eta_2(u, v), z)$  and  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  satisfy  $h_1(0) \neq 0$  and  $h_2(0) \neq 0$ . By Proposition 4.2.30,  $\frac{h_1(0)}{h_2(0)} = \beta(\tilde{f}) = 1$ . By applying a time equivalence associated with  $\xi \equiv \frac{1}{h_1(0)}$ , we conclude the proof of the theorem.  $\Box$ 

#### Semi-conical case

**Theorem 4.2.32.** Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Then 0 is F-semi-conical for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ if and only if there exist  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfying  $h_1(0) = h_2(0) = 1$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfying  $m(0) \notin \{-1, 0\}$  such that f is equivalent to

$$(u,v,z) \mapsto \begin{pmatrix} h_1(u,v,z)(z-m(u)u) \\ h_2(u,v,z)(z+u+v^2) \end{pmatrix}.$$

Before proving the theorem, let us make some general considerations and provide an intermediate result in Proposition 4.2.33.

First remark that, up to a left-equivalence, we can assume that

$$\partial_1 f_1(0) \neq 0, \ \partial_1 f_2(0) \neq 0, \ \partial_3 f_1(0) = \partial_3 f_2(0) \neq 0.$$

In particular,  $\beta(f) = 1$ . In order to impose the non-conical direction to be in the span $(e_2)$ direction, we use the same right-equivalence of the plane (u, v) as in the first step of the algorithm **(A)** in Section 4.2.3 (see Proposition 4.2.27). As a result, we end up with  $\hat{f}$ equivalent to f and such that

$$f(0) = 0, \ \partial_2 f(0) = 0, \ \partial_1 f_1(0) \neq 0, \ \partial_1 f_2(0) \neq 0, \ \partial_3 f_1(0) = \partial_3 f_2(0) \neq 0,$$
  

$$\chi_{13}(f)(0) \neq 0, \ \partial_2 \chi(f)(0) \neq 0.$$
(SC2P)

Notice that the condition  $\chi_{13}(f)(0) \neq 0$  can then be rewritten as  $\partial_1 f_1(0) \neq \partial_1 f_2(0)$ . The proof of the theorem is based on the following result.

**Proposition 4.2.33.** Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfy (SC2P) at 0. Then there exist  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  non-vanishing at 0 and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  such that  $\frac{h_1(0)}{h_2(0)} = 1$ ,  $m(0) = -\frac{\partial_1 f_1(0)}{\partial_1 f_2(0)} \notin \{-1, 0\}$ , and f is right-equivalent to

$$(u,v,z) \mapsto \begin{pmatrix} h_1(u,v,z)(z-m(u)u) \\ h_2(u,v,z)(z+u\pm v^2) \end{pmatrix},$$

where the sign depends on f.

Proof. Using the fact that f(0) = 0 and the conditions  $\partial_3 f_1(0) \neq 0 \neq \partial_3 f_2(0)$ , we can deduce that  $f_1$  and  $f_2$  are smooth functions vanishing, in the neighborhood of the origin, on two smooth surfaces whose equations are, respectively,  $z = \eta_1(u, v)$  and  $z = \eta_2(u, v)$ , where  $\eta_1, \eta_2 : \mathbb{R}^2 \to \mathbb{R}$  are smooth functions vanishing at 0. By Lemma 4.7.10, there exist two smooth functions  $\phi_1, \phi_2 : \mathbb{R}^3 \to \mathbb{R}$  such that

$$f_1(u, v, z) = \phi_1(u, v, z)(z - \eta_1(u, v)), \qquad f_2(u, v, z) = \phi_2(u, v, z)(z - \eta_2(u, v)).$$

Differentiating these expressions with respect to z, we deduce that  $\phi_1(0) \neq 0 \neq \phi_2(0)$ . Differentiating  $f_1$  and  $f_2$  with respect to y, we get then from (SC2P) that  $\partial_1\eta_1(0) \neq 0 \neq \partial_1\eta_2(0)$  and  $\partial_2\eta_1(0) = \partial_2\eta_2(0) = 0$ . Applying the right-equivalence associated with the inverse of  $(u, v) \mapsto \begin{pmatrix} \eta_1(u, v) \\ v \end{pmatrix}$ , we get that f is right-equivalent to  $(u, v, z) \mapsto \begin{pmatrix} \phi_1(G(u, v), v, z)(z - u) \\ \phi_2(G(u, v), v, z)(z - \eta_2(G(u, v), v)) \end{pmatrix}$ ,

for some smooth function G such that  $\partial_1 G(0) \neq 0$ . Set  $\tilde{\eta}(u, v) = \eta_2(G(u, v), v)$ . Then  $\tilde{\eta}(0) = 0$  and

$$\partial_1 \tilde{\eta}(u,v) = \partial_1 G(u,v) \partial_1 \eta_2(G(u,v),v), \quad \partial_2 \tilde{\eta}(u,v) = \partial_1 \eta_2(G(u,v),v) \partial_2 G(u,v) + \partial_2 \eta_2(u,v) + \partial_2 \eta_2(u,v)$$

Evaluating at zero, we get that  $\partial_1 \tilde{\eta}(0) \neq 0$  and  $\partial_2 \tilde{\eta}(0) = 0$ . Differentiating once more and using the hypothesis  $\partial_2 \chi(f)(0) \neq 0$ , we have  $\partial_{22} \tilde{\eta}(0) \neq 0$ . By the same arguments as in the proof of Theorem 4.2.25, f is right-equivalent to

$$(u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z - m(u)u) \\ h_2(u, v, z)(z + u \pm v^2) \end{pmatrix}$$

where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$ . Noticing that the quantities  $\frac{\partial_1 f_1(0)}{\partial_1 f_2(0)}$  and  $\beta(f)$  are invariant by right-equivalence, we get  $\frac{h_1(0)}{h_2(0)} = \beta(f) = 1$  and  $m(0) = -\frac{\partial_1 f_1(0)}{\partial_1 f_2(0)} \notin \{-1, 0\}$  because f satisfies .(SC2P) at 0.

Proof of Theorem 4.2.32. First notice that if f is of the form

$$f: (u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z - m(u)u) \\ h_2(u, v, z)(z + u - v^2) \end{pmatrix}$$

then there exist  $\tilde{h}_1, \tilde{h}_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  and  $\tilde{m} \in C^{\infty}(\mathbb{R}, \mathbb{R})$  such that f is right-time equivalent to

$$(u,v,z) \mapsto \begin{pmatrix} \tilde{h}_1(u,v,z)(z-\tilde{m}(u)u)\\ \tilde{h}_2(u,v,z)(z+u+v^2) \end{pmatrix}.$$

Indeed, applying the right-equivalence  $(u, v, z) \mapsto (-u, v, -z)$  and the time-equivalence associated with  $\xi : (u, v) \mapsto -1$  the claim follows with  $\tilde{h}_i(u, v, z) = h_i(-u, v, -z), i \in \{1, 2\}$ , and  $\tilde{m}(u) = m(-u)$ . Theorem 4.2.32 hence follows from Proposition 4.2.33.  $\Box$ 

# 4.3 Generic global properties of the singular locus

#### 4.3.1 Proof of Lemma 4.1.1 and Theorem 4.1.2

Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ ,  $Z(f) = \{(u, v, z) \mid f(u, v, z) = (0, 0)\}$ , and denote by  $Z_{nc}(f)$  the set of non-conical intersections in Z(f). Let  $\pi(f)$  be the orthogonal projection of Z(f) onto the plane (u, v).

**Proposition 4.3.1.** Assume that  $(\bar{u}, \bar{v}, \bar{z})$  is a *F*-semi-conical intersection for  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Then  $\pi(f)$  is tangent at  $(\bar{u}, \bar{v})$  to the non-conical direction of f at  $(\bar{u}, \bar{v}, \bar{z})$ .

Proof. By Proposition 4.2.8, Z(f) is locally near  $(\bar{u}, \bar{v}, \bar{z})$  a smooth curve that we parameterize by  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$ , with  $c(0) = (\bar{u}, \bar{v}, \bar{z})$  and  $\dot{z}(0) = 0$ . The condition  $f(c(t)) \equiv 0$  implies that  $Df_{c(t)}(\dot{c}(t)) \equiv 0$ . In particular,  $(\dot{u}(0), \dot{v}(0))$  is colinear to  $\eta = \begin{pmatrix} -\partial_2 f_1(\bar{u}, \bar{v}, \bar{z}) \\ \partial_1 f_1(\bar{u}, \bar{v}, \bar{z}) \end{pmatrix}$ , which is the non-conical direction of f at  $(\bar{u}, \bar{v}, \bar{z})$ .

**Proposition 4.3.2.** Assume that  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  is a submersion at every point of Z(f). Then  $\pi(f)$  has no cuspidal point.

Proof. Fix a local smooth regular parametrization  $c(t) = (u(t), v(t), z(t))_{t \in \mathbb{R}}$  of  $c \subset Z(f)$ . It is sufficient to show that there exist no  $t \in \mathbb{R}$  such that  $\dot{u}(t) = \dot{v}(t) = 0$ . By the same arguments as those used in the proof of Proposition 4.2.8, we get that  $\dot{u}(t) = 0$  implies that  $\begin{vmatrix} \partial_2 f_1(c(t)) & \partial_3 f_1(c(t)) \\ \partial_2 f_2(c(t)) & \partial_3 f_2(c(t)) \end{vmatrix} = 0$ , while  $\dot{v}(t) = 0$  implies that  $\begin{vmatrix} \partial_1 f_1(c(t)) & \partial_3 f_1(c(t)) \\ \partial_1 f_2(c(t)) & \partial_3 f_2(c(t)) \end{vmatrix} = 0$ . If the two determinants simultaneously vanish then f is not a submersion at c(t).

Propositions 4.3.1 and 4.3.2, together with Remark 4.2.10, prove Lemma 4.1.1. As for Theorem 4.1.2, it follows from Proposition 4.3.2 together with Propositions 4.2.7, 4.2.11, and 4.2.12.

**Remark 4.3.3.** Condition (ii) in Theorem 4.1.2 is essential for the control applications by adiabatic methods. It guarantees that the only possible singularities for the curve  $\pi(f)$ in  $\mathbb{R}^2$  are self-intersections.

#### 4.3.2 Generic self-intersections of $\pi(f)$

We aim at studying more precise generic properties of  $\pi(f)$ . Following the notations in [41], let X and Y be two smooth manifolds and denote by  $J^k(X,Y)$  the set of k-jets from X to Y. Set  $X^s = X \times ... \times X$  (s-times) and

$$X^{(s)} = \{ (x_1, ..., x_s) \in X^s \mid x_i \neq x_j \text{ for } 1 \le i < j \le s \}.$$

Let  $\alpha : J^k(X, Y) \to X$  be the source map and denote by  $\alpha^s : J^k(X, Y) \to X^s$  its natural extension. The set  $J^k_s(X, Y) = (\alpha^s)^{-1}(X^{(s)})$  is called the *s*-fold *k*-jet bundle. We associate with a smooth function  $f : X \to Y$  the map

$$j_s^k f : X^{(s)} \longrightarrow J_s^k(X, Y),$$
$$(x_1, ..., x_s) \mapsto (j^k f(x_1), ..., (j^k f(x_s)).$$

**Theorem 4.3.4.** ([41, §4, Theorem 4.13] and [42, §1.3.2]) Let X and Y be smooth manifolds and W be an algebraic submanifold of  $J_s^k(X,Y)$ . Let

$$T_W = \left\{ f \in C^{\infty}(X, Y) \mid j_s^k f \pitchfork W \right\}$$

Then  $T_W$  is open and dense in  $C^{\infty}(X, Y)$ .

**Proposition 4.3.5.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ ,

- 1.  $\pi(f)$  has no triple points;
- 2. Let (u, v) and  $(\tilde{u}, \tilde{v})$  be two double points of  $\pi(f)$  and let  $z_1 \neq z_2$  and  $\tilde{z}_1 \neq \tilde{z}_2$  be such that  $f(u, v, z_1) = f(u, v, z_2) = f(\tilde{u}, \tilde{v}, \tilde{z}_1) = f(\tilde{u}, \tilde{v}, \tilde{z}_2) = 0$ . Then  $z_i \neq \tilde{z}_j$  for every  $i, j \in \{1, 2\}$ ;
- 3. Let (u, v) and  $z \neq \tilde{z}$  satisfy  $f(u, v, z) = f(u, v, \bar{z}) = 0$ . Then (u, v, z) and  $(u, v, \tilde{z})$  are *F*-conical for *f*;
- 4. Let (u, v, z) and  $(\tilde{u}, \tilde{v}, \tilde{z})$  be two non-conical intersections for f. Then  $z \neq \tilde{z}$ .

*Proof.* Here we prove the first claim, similar arguments leading to the other claims. A point (u, v) is a triple point of  $\pi(f)$  if and only if there exist  $z_1, z_2, z_3$  pairwise distinct such that  $f(u, v, z_1) = f(u, v, z_2) = f(u, v, z_3) = 0$ . Define the subspace W of  $J_3^1(\mathbb{R}^3, \mathbb{R}^2)$  by

$$W = \{ j_3^1 f(X_1, X_2, X_3) \mid f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2), \ X_1, X_2, X_3 \in \mathbb{R}^3, \ \pi(X_1) = \pi(X_2) = \pi(X_3), f(X_1) = f(X_2) = f(X_3) = 0 \}.$$

Then W is a codimension  $10 > 9 = \dim(X^{(3)})$  algebraic submanifold of  $J_3^1(X, Y)$ . Theorem 4.3.4 then implies that  $T_W = \{f \in C^{\infty}(X, Y) \mid j_3^1 f \cap W = \emptyset\}$  is an open and dense subset of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ .

# 4.4 Adiabatic control through a semi-conical intersection of eigenvalues

We consider  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  and the associated Hamiltonian  $H_f = \begin{pmatrix} f_1 & f_2 \\ f_2 & -f_1 \end{pmatrix}$ . Consider a control path  $(u(t), v(t))_{t \in [0,1]}$ . Denote by

$$\lambda^{-}(u(t), v(t)) = -\sqrt{f_1(u(t), v(t))^2 + f_2(u(t), v(t))^2}$$

Chapter 4. Ensemble control of quantum systems with two controls: non-conical case

and

$$\lambda^+(u(t), v(t)) = \sqrt{f_1(u(t), v(t))^2 + f_2(u(t), v(t))^2}$$

the smallest and largest eigenvalue of  $H_f$ , respectively.

In the following, we denote by  $\phi_{-}(u(t), v(t))$  (respectively,  $\phi_{+}(u(t), v(t))$ ) a real normalized eigenvector of  $H_f$  at (u(t), v(t)) associated with  $\lambda_{-}(u(t), v(t))$  (respectively,  $\lambda_{+}(u(t), v(t))$ ). If  $f(u(t), v(t)) \neq 0$  then  $\lambda_{-}(u(t), v(t))$  and  $\lambda_{+}(u(t), v(t))$  are simple and the choice of  $\phi_{\pm}(u(t), v(t))$  is unique up to multiplication by -1. If  $(u(t), v(t))_{t\in[0,1]}$  does not cross Z(f)then  $t \mapsto \phi_{-}(u(t), v(t)), t \mapsto \phi_{+}(u(t), v(t))$  and  $t \mapsto \lambda^{-}(u(t), v(t)), t \mapsto \lambda^{+}(u(t), v(t))$  can be chosen with the same regularity as  $(u(t), v(t))_{t\in[0,1]}$ . It is a classical fact that this may not be the case when  $(u(t), v(t))_{t\in[0,1]}$  crosses Z(f). However, we are going to prove the existence of a  $C^k$  basis of eigendirections of  $H_f$  along a  $C^{k+2}$  path  $(u(t), v(t))_{t\in[0,1]}$  passing through a semi-conical intersection in a conical or a non-conical direction.

#### 4.4.1 Adiabatic dynamics

Let  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . Consider a smooth regular control path  $(u(t), v(t))_{t \in [0,1]}$  such that there exist  $P \in C^2([0,1], \operatorname{SO}_2(\mathbb{R}))$  and  $\lambda \in C^k([0,1], \mathbb{R})$  such that  $\{\lambda(t), -\lambda(t)\}$  is the spectrum of  $H_f(u(t), v(t))$ , and the columns of P form a basis of eigenvectors of  $H_f(u(t), v(t))$ for every  $t \in [0,1]$ . We can write, for every  $t \in [0,1]$ ,  $P(t) = \begin{pmatrix} \cos(\theta(t)) & -\sin(\theta(t)) \\ \sin(\theta(t)) & \cos(\theta(t)) \end{pmatrix}$ where  $\theta \in C^2([0,1], \mathbb{R})$ .

Let us study the dynamics of

$$i\frac{d\psi_{\epsilon}(t)}{dt} = H_f(u(\epsilon t), v(\epsilon t))\psi_{\epsilon}(t), \quad \psi_{\epsilon}(0) = \tilde{\psi}_0, \tag{4.9}$$

where  $t \in [0, \frac{1}{\epsilon}]$  and  $\tilde{\psi}_0$  is independent of  $\epsilon$ .

Defining  $Y_{\epsilon}(\tau) = P(\tau)\psi_{\epsilon}(\frac{\tau}{\epsilon})$  for every  $\tau \in [0, 1]$ , we have

$$i\frac{dY_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon} \begin{pmatrix} \lambda(\tau) & 0\\ 0 & -\lambda(\tau) \end{pmatrix} + \begin{pmatrix} 0 & i\dot{\theta}(\tau)\\ -i\dot{\theta}(\tau) & 0 \end{pmatrix} \right) Y_{\epsilon}(\tau).$$
(4.10)

Thanks to the further change of variables  $\tilde{Y}_{\epsilon}(\tau) = \begin{pmatrix} e^{\frac{i}{\epsilon} \int_{0}^{\tau} \lambda(s) ds} & 0\\ 0 & e^{-\frac{i}{\epsilon} \int_{0}^{\tau} \lambda(s) ds} \end{pmatrix} Y_{\epsilon}(\tau)$ , the dynamics are transformed into

$$\frac{d\tilde{Y}_{\epsilon}(\tau)}{d\tau} = \begin{pmatrix} 0 & i\dot{\theta}(\tau)e^{\frac{2i}{\epsilon}\int_{0}^{\tau}\lambda(s)ds} \\ -i\dot{\theta}(\tau)e^{-\frac{2i}{\epsilon}\int_{0}^{\tau}\lambda(s)ds} & 0 \end{pmatrix}\tilde{Y}_{\epsilon}(\tau).$$
(4.11)

Based on Corollaries 4.7.3 and 4.7.7, one gets the following result.

**Theorem 4.4.1** (Adiabatic Theorem). Let  $k \in \mathbb{N}$  and assume that  $\lambda : [0,1] \to \mathbb{R}$  is  $C^k$ and  $\theta : [0,1] \to \mathbb{R}$  is  $C^2$ . Let  $\psi_{\epsilon} : [0,1] \to \mathbb{C}^2$  be the solution of Equation (5.3). Assume that there exists c > 0 such that

$$\left|\int_{0}^{t} e^{\frac{2i}{\epsilon} \int_{0}^{s} \lambda(x) dx} ds\right| \le c \epsilon^{1/(k+1)}, \qquad \forall t \in [0, 1].$$

$$(4.12)$$

Then  $\tilde{Y}_{\epsilon}(\tau) = \tilde{Y}_{\epsilon}(0) + O(\epsilon^{q})$  uniformly w.r.t.  $\tau \in [0, 1]$ . In particular, if  $\tilde{\psi}_{0} = \begin{pmatrix} \cos(\theta(0)) \\ \sin(\theta(0)) \end{pmatrix}$ , then  $\psi_{\epsilon}(\frac{1}{\epsilon}) = e^{i\eta} \begin{pmatrix} \cos(\theta(1)) \\ \sin(\theta(1)) \end{pmatrix} + O(\epsilon^{\frac{1}{k+1}})$ , where  $\eta$  possibly depends on  $\epsilon$ .

#### 4.4.2 Regularity of the eigenpairs along smooth control paths

The main goal of this section is to study the regularity of eigenpairs of  $H_f$  along smooth curves passing through a semi-conical intersection for f. Using the normal form obtained in Section 4.2.3, we can restrict our attention to the case where f has the form  $f: (u, v) \mapsto \begin{pmatrix} u + v^2 \\ uh(u) \end{pmatrix}$ , where  $h \in C^{\infty}(\mathbb{R}, \mathbb{R})$  is such that h(0) = 1. Notice that for the simplicity of computations, we have inverted the two components  $f_1$  and  $f_2$  with respect to the normal form found in Section 4.2.3. However,  $(f_1, f_2)$  being left-equivalent to  $(f_2, f_1)$ , the same results hold for  $f: (u, v) \mapsto \begin{pmatrix} uh(u) \\ u + v^2 \end{pmatrix}$ .

#### **Conical directions**

We recall here the following regularity result which is a special case of [31, Proposition 3.1] and [10, Lemma 3.2].

**Proposition 4.4.2** (Eigenpairs in the conical directions). Consider a Hamiltonian  $H_f$ where  $f: (u,v) \mapsto \begin{pmatrix} u+v^2 \\ uh(u) \end{pmatrix}$  and  $h \in C^{\infty}(\mathbb{R},\mathbb{R})$  is such that h(0) = 1. Let  $\ell \in \mathbb{N}$ ,  $t_0 \in (0,1)$ , and  $(u(t), v(t))_{t \in [0,1]}$  be a  $C^{\ell+1}$  path such that (u(t), v(t)) = 0 if and only if  $t = t_0$  and  $\dot{u}(t_0) \neq 0$ . Define  $\lambda_0, \lambda_1 : [0,1] \rightarrow \mathbb{R}$  by

$$\begin{split} \lambda_0(t) &= \lambda_-(u(t), v(t)), \ \lambda_1(t) = \lambda_+(u(t), v(t)), \qquad \text{for } t < t_0, \\ \lambda_0(t) &= \lambda_+(u(t), v(t)), \ \lambda_1(t) = \lambda_-(u(t), v(t)), \qquad \text{for } t \ge t_0. \end{split}$$

Then  $\lambda_0$  and  $\lambda_1$  are  $C^{\ell+1}$  on [0,1]. Moreover, there exist  $\Phi_0, \Phi_1 \in C^{\ell}([0,1], \mathbb{R}^2)$  such that  $\Phi_j(t)$  is a normalized eigenvector of  $H_f(u(t), v(t))$  corresponding to the eigenvalue  $\lambda_j(t)$  for  $j \in \{0,1\}$  and  $t \in [0,1]$ .

The following proposition states that the limit eigenvectors along a  $C^2$  curve crossing conically a semi-conical intersection do not depend on the choice of the curve.

**Proposition 4.4.3** (Limit eigenvectors along a conical direction). Consider f,  $(u(t), v(t))_{t \in [0,1]}$ ,  $t_0$ , and  $\Phi_0, \Phi_1$  as in Proposition 4.4.2, which are uniquely defined up to multiplication to -1. Then  $\Phi_0(t_0)$  and  $\Phi_1(t_0)$  depend only on the sign of  $\dot{u}(t_0)$ . More precisely,  $\Phi_0(t_0) = \frac{1}{\sqrt{1+\bar{V}^2}} \begin{pmatrix} -1\\ \bar{V} \end{pmatrix}$  and  $\Phi_1(t_0) = \frac{1}{\sqrt{1+\bar{V}^2}} \begin{pmatrix} \bar{V}\\ 1 \end{pmatrix}$  with  $\bar{V} = -(1 + \operatorname{sign}(\dot{u}(t_0))\sqrt{2})$ .

*Proof.* By definition of  $\Phi_0, \Phi_1$ , we have

$$\Phi_0(t) = \phi_-(u(t), v(t)), \ \Phi_1(t) = \phi_+(u(t), v(t)), \qquad \text{for } t < t_0.$$

Hence, up to multiplication by -1,

$$\Phi_0(t) = \frac{1}{\sqrt{1 + V(t)^2}} \begin{pmatrix} -1\\ V(t) \end{pmatrix}, \ \Phi_1(t) = \frac{1}{\sqrt{1 + V(t)^2}} \begin{pmatrix} V(t)\\ 1 \end{pmatrix}, \quad \text{for } t < t_0,$$

where V(t) = 0 if u(t)h(u(t)) = 0 and

$$V(t) = \frac{-u(t) - v(t)^2 + \sqrt{u(t)^2 h(u(t))^2 + (u(t) + v(t)^2)^2}}{u(t)h(u(t))}$$

otherwise. Then, as  $t \to t_0^-$ ,

$$V(t) = \frac{-\dot{u}(t_0)(t-t_0) + o(t-t_0) + \sqrt{\dot{u}(t_0)^2(t-t_0)^2(h(0)^2+1) + o((t-t_0)^2)}}{\dot{u}(t_0)h(0)(t-t_0)}$$
  
=  $-\frac{\dot{u}(t_0)(t-t_0) + |t-t_0||\dot{u}(t_0)|\sqrt{2}}{\dot{u}(t_0)(t-t_0)} + o(1)$   
=  $-(1 + \operatorname{sign}(\dot{u}(t_0))\sqrt{2}) + o(1).$ 

Since  $\Phi_0, \Phi_1$  are continuous on [0, 1] by Proposition 4.4.2, the conclusion follows.

#### Non-conical direction

**Proposition 4.4.4** (Continuity of the eigenstates in the non-conical direction). Let f be as in Proposition 4.4.2. Let  $\ell \in \mathbb{N}$ ,  $t_0 \in (0,1)$ , and  $(u(t), v(t))_{t \in [0,1]}$  be a  $C^{\ell+2}$  path such that (u(t), v(t)) = 0 if and only if  $t = t_0$  and  $\dot{u}(t_0) = 0$  (i.e., (u(t), v(t)) passes through 0 in the non-conical direction at  $t = t_0$  as on Figure 4.4). Then there exist  $\lambda_0, \lambda_1 \in$  $C^{\ell+2}([0,1], \mathbb{R}^2), \Phi_0, \Phi_1 \in C^{\ell}([0,1], \mathbb{R}^2)$  such that  $\lambda_0(t) = \lambda_-(t), \lambda_1(t) = \lambda_+(t), \Phi_0(t) =$  $\phi_-(t)$  and  $\Phi_1(t) = \phi_+(t)$  for every  $t \in [0, t_0) \cup (t_0, 1]$ . Moreover,  $\Phi_0(t_0) = \frac{1}{\sqrt{1+\bar{V}^2}} \begin{pmatrix} -1 \\ \bar{V} \end{pmatrix}$ and  $\Phi_1(t_0) = \frac{1}{\sqrt{1+\bar{V}^2}} \begin{pmatrix} \bar{V} \\ 1 \end{pmatrix}$  where  $\bar{V} = -\frac{\beta - \sqrt{\frac{\bar{u}(t_0)^2}{4} + \beta^2}}{\frac{\bar{u}(t_0)}{2}}$  where  $\beta = \frac{\bar{u}(t_0)}{2} + \dot{v}(t_0)^2$  if  $\ddot{u}(t_0) \neq 0$ . If  $\ddot{u}(t_0) = 0$ , we have  $\Phi_0(t_0) = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$  and  $\Phi_1(t_0) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$ . *Proof.* The condition  $\dot{u}(t_0) = 0$  provides  $u(t) + v(t)^2 = (t - t_0)^2 (\frac{\ddot{u}(t_0)}{2} + \dot{v}(t_0)^2) + o((t - t_0)^2) = \beta(t - t_0)^2 + o((t - t_0)^2)$  when  $t \to t_0$ .

By direct computations, we show that  $\frac{d^2\lambda_{\pm}(t)}{dt^2} = \pm \sqrt{\beta^2 + \ddot{u}(t_0)^2} + o(1)$  when  $t \to t_0$ . Hence  $t \mapsto \lambda_{\pm}(t)$  can be extended in a  $C^2$  function at  $t = t_0$  by fixing  $\frac{d^2\lambda_{\pm}(t_0)}{dt^2} = 2 \pm \sqrt{\beta^2 + \ddot{u}(t_0)^2}$ . The  $C^{l+2}$  regularity follows by higher order analog computations.

Define, for every  $t \in [0, 1]$  such that  $u(t)h(u(t)) \neq 0$ ,

$$V(t) = \frac{-(u(t) + v(t)^2) + \sqrt{u(t)^2 h(u(t))^2 + (u(t) + v(t)^2)^2}}{u(t)h(u(t))}$$

Reasoning as in the proof of Proposition 6.2.2, we must prove that V can be extended as a  $C^{\ell}$  function at  $t_0$  by setting  $V(t_0) = \overline{V}$ .

**First case:**  $\ddot{u}(t_0) \neq 0$ . Assuming that  $\ddot{u}(t_0) \neq 0$ , we have, as  $t \to t_0$ ,

$$V(t) = -\frac{\beta(t-t_0)^2 - \sqrt{\frac{(t-t_0)^4}{4}\ddot{u}(t_0)^2 + \beta^2(t-t_0)^4 + o((t-t_0)^4)}}{\frac{(t-t_0)^2}{2}\ddot{u}(t_0)} = -\frac{\beta - \sqrt{\frac{\ddot{u}(t_0)^2}{4} + \beta^2}}{\frac{\ddot{u}(t_0)}{2}} + o(1).$$

The continuity of V is proved in the case  $\ddot{u}(t_0) \neq 0$ . The same computations show that V is  $C^{\ell}$  at  $t_0$  if  $(u(t), v(t))_{t \in [0,1]}$  is  $C^{\ell+2}$ .

Second case:  $\ddot{u}(t_0) = 0$ . In the case  $\ddot{u}(t_0) = 0$ , consider the left equivalence  $\tilde{f}$  of f associated with  $\theta = \frac{\pi}{4}$  and  $\zeta = -1$  as in Remark 4.2.16, so that, we have  $H_{\tilde{f}} = P_{\theta,\zeta}H_fP_{\theta,\zeta}^{-1}$ . If  $f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , then we have  $\tilde{f} = (f_2, f_1)$ . Define, for every  $t \in [0, 1]$  such that  $u(t) + v(t)^2 \neq 0$ ,  $\tilde{V}(t) = \frac{-u(t)h(u(t)) + \sqrt{u(t)^2h(u(t))^2 + (u(t) + v(t)^2)^2}}{u(t) + v(t)^2}$ . We have easily  $\lim_{t \to t_0} \tilde{V}(t) = 1$ . Hence we can define continuous eigenvectors  $\tilde{\Phi_0}$  and  $\tilde{\Phi_1}$  of  $H_{\tilde{f}}$  along  $(u(t), v(t))_{t \in [0,1]}$ , respectively equal, up to phases, to  $\frac{1}{\sqrt{2}} \begin{pmatrix} -1\\ 1 \end{pmatrix}$  and  $\frac{1}{\sqrt{2}} \begin{pmatrix} 1\\ 1 \end{pmatrix}$  at  $t = t_0$ . Knowing that the eigenvectors of  $H_f$  are equal, up to phases, to  $P_{\theta,\zeta}\tilde{\Phi_j}$ , for  $j \in \{0,1\}$ , we can deduce the continuity of  $\Phi_0$  and  $\Phi_1$  at  $t = t_0$ , and that  $\Phi_0(t_0) = \begin{pmatrix} 0\\ 1 \end{pmatrix}$  and  $\Phi_1(t_0) = \begin{pmatrix} 1\\ 0 \end{pmatrix}$  in this case.

# 4.4.3 Dynamical properties at semi-conical intersections of eigenvalues

By using the previous results, we get the following adiabatic approximations along curves going through a semi-conical intersection, either along conical directions (Proposition 4.4.5) or along the non-conical direction (Proposition 4.4.6).

**Proposition 4.4.5.** Let f and  $(u(t), v(t))_{t \in [0,1]}$  be as in Proposition 4.4.2. Consider a solution  $\psi_{\epsilon}$  of Equation (5.3) such that  $\tilde{\psi}_0 = \phi_-(u(0), v(0))$ . Then  $\langle \psi_{\epsilon}(\frac{1}{\epsilon}), \phi_-(u(1), v(1)) \rangle = O(\sqrt{\epsilon})$ .

Proof. First notice that  $\lambda_{-}$  and  $\lambda_{+}$  are  $C^{2}$  separately on  $[0, t_{0}]$  and  $[t_{0}, 1]$ . Moreover, Corollary 4.7.7 proves that they satisfy (4.22) with k = 1 on  $[0, t_{0}]$  and  $[t_{0}, 1]$ . By Proposition 6.2.2, moreover, there exists a  $C^{2}$  basis of eigenvectors defined on [0, 1] with  $\lim_{t \to t_{0}^{-}} \phi_{\pm}(t) = \lim_{t \to t_{0}^{+}} \phi_{\mp}(t)$ . By applying Theorem 4.4.1 on the interval  $[0, t_{0}]$ , then on the interval  $[t_{0}, 1]$ , we get the result.

**Proposition 4.4.6.** Let f and  $(u(t), v(t))_{t \in [0,1]}$  be as in Proposition 4.4.4 with  $l \ge 2$  (see Figure 4.4). Consider a solution  $\psi_{\epsilon}$  of Equation (4.13) such that  $\psi_{\epsilon}(0) = \phi_{-}(u(0), v(0))$ . Then  $\langle \psi_{\epsilon}(\frac{1}{\epsilon}), \phi_{+}(u(1), v(1)) \rangle = O(\epsilon^{1/3})$ .

Proof. Let  $\lambda(\tau) = \lambda_{-}(\tau)$  and  $\phi(\tau) = \int_{0}^{\tau} \lambda_{-}(s) ds$  for  $\tau \in [0, 1]$ . Notice that  $\lambda$  is at least  $C^{2}$  by Proposition 4.4.4 and that  $\phi$  satisfies  $\phi(t_{0}) = \phi'(t_{0}) = \phi''(t_{0}) = 0$  and  $\phi^{(3)}(t_{0}) \neq 0$ . Hence, by Lemma 4.7.4,  $\lambda$  satisfies the estimate (4.22) with k = 2. The result follows by applying Theorem 4.4.1 in combination with Proposition 4.4.4.



Figure 4.4 – A control path passing at a semi-conical intersection in the non-conical direction as a function of the controls  $(u, v) \in \mathbb{R}^2$ .

# 4.5 Control of a continuum of systems

#### 4.5.1 Ensemble adiabatic dynamics

Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Consider a smooth regular control path  $(u(t), v(t))_{t \in [0,1]}$ . In analogy with the previous sections, denote by

$$\lambda_z^{\pm}(u(t), v(t)) = \pm \sqrt{f_1(u(t), v(t), z)^2 + f_2(u(t), v(t), z)^2}$$

the eigenvalues of  $H_f(u(t), v(t), z)$ . Similarly, let  $\phi_{\pm}^z(u(t), v(t))$  be two real normalized eigenvector of  $H_f(u(t), v(t), z)$  at (u(t), v(t)) associated with  $\lambda_z^{\pm}(u(t), v(t))$ , uniquely defined up to a sign.

Let  $V = [z_0, z_1]$  be a compact interval of  $\mathbb{R}$ . Assume that for every  $z \in V$  there exist  $P_z \in C^1([0, 1], \operatorname{SO}_2(\mathbb{R}))$  and  $\lambda_z \in C^1([0, 1], \mathbb{R})$  such that  $\{\lambda_z(t), -\lambda_z(t)\}$  is the spectrum of  $H_f(u(t), v(t), z)$  and the columns of  $P_z$  form a basis of eigenvectors of  $H_f(u(t), v(t), z)$  for every  $t \in [0, 1]$ . We can write, for every  $t \in [0, 1]$ ,  $P_z(t) = \begin{pmatrix} \cos(\theta_z(t)) & -\sin(\theta_z(t)) \\ \sin(\theta_z(t)) & \cos(\theta_z(t)) \end{pmatrix}$  where  $\theta_z \in C^1([0, 1], \mathbb{R})$ .

Let us study the dynamics of

$$i\frac{d\psi_{\epsilon}^{z}(t)}{dt} = H_{f}(u(\epsilon t), v(\epsilon t), z)\psi_{\epsilon}^{z}(t), \quad \psi_{\epsilon}^{z}(0) = \tilde{\psi}_{0}^{z}, \tag{4.13}$$

where  $t \in [0, \frac{1}{\epsilon}]$  and  $\tilde{\psi}_0^z$  is independent of  $\epsilon$ .

Defining  $Y_{\epsilon}^{z}(\tau) = P_{z}(\tau)\psi_{\epsilon}^{z}(\frac{\tau}{\epsilon})$  for every  $\tau \in [0, 1]$ , we have

$$i\frac{dY_{\epsilon}^{z}(\tau)}{d\tau} = \left(\frac{1}{\epsilon} \begin{pmatrix} \lambda_{z}(\tau) & 0\\ 0 & -\lambda_{z}(\tau) \end{pmatrix} + \begin{pmatrix} 0 & i\dot{\theta}_{z}(\tau)\\ -i\dot{\theta}_{z}(\tau) & 0 \end{pmatrix} \right) Y_{\epsilon}^{z}(\tau).$$
(4.14)

Thanks to the further change of variable  $\tilde{Y}_{\epsilon}^{z}(\tau) = \begin{pmatrix} e^{\frac{i}{\epsilon}\int_{0}^{\tau}\lambda_{z}(s)ds} & 0\\ 0 & e^{-\frac{i}{\epsilon}\int_{0}^{\tau}\lambda_{z}(s)ds} \end{pmatrix} Y_{\epsilon}^{z}(\tau)$ , the dynamics is transformed into

$$\frac{d\tilde{Y}_{\epsilon}^{z}(\tau)}{d\tau} = \begin{pmatrix} 0 & i\dot{\theta}_{z}(\tau)e^{\frac{2i}{\epsilon}\int_{0}^{\tau}\lambda_{z}(s)ds} \\ -i\dot{\theta}_{z}(\tau)e^{-\frac{2i}{\epsilon}\int_{0}^{\tau}\lambda_{z}(s)ds} & 0 \end{pmatrix} \tilde{Y}_{\epsilon}^{z}(\tau).$$
(4.15)

Based on Corollary 4.7.8, one get the following result.

**Theorem 4.5.1** (Parametric Adiabatic Theorem). Let  $k \in \mathbb{N}$ . For every  $z \in [z_0, z_1]$ , assume that  $\lambda_z : [0, 1] \to \mathbb{R}$  is  $C^k$  in [0, 1] and  $\theta_z$  is  $C^2$  in [0, 1]. Let  $\psi_{\epsilon}^z(t)$  be the solution of Equation (4.13). Assume that there exists c > 0 such that for every t in [0, 1], and for every  $z \in [z_0, z_1]$ ,

$$\left| \int_{0}^{t} e^{\frac{2i}{\epsilon} \int_{0}^{s} \lambda_{z}(x) dx} ds \right| \le c \epsilon^{\frac{1}{k+1}}, \qquad \forall \epsilon > 0,$$
(4.16)

and that  $(t,z) \mapsto \dot{\theta}_z(t)$  and  $(t,z) \mapsto \ddot{\theta}_z(t)$  are uniformly bounded with respect to  $(t,z) \in [0,1] \times [z_0,z_1]$ . Then we have  $\tilde{Y_{\epsilon}}(\tau) = \tilde{Y_{\epsilon}^z}(0) + O(\epsilon^{\frac{1}{k+1}})$ , uniformly w.r.t.  $\tau \in [0,1]$  and  $z \in [z_0,z_1]$ . In particular, if  $\tilde{\psi_0^z} = \begin{pmatrix} \cos(\theta_z(0)) \\ \sin(\theta_z(0)) \end{pmatrix}$ , then we have  $\psi_{\epsilon}^z(\frac{1}{\epsilon}) = e^{i\eta} \begin{pmatrix} \cos(\theta_z(1)) \\ \sin(\theta_z(1)) \end{pmatrix} + O(\epsilon^{\frac{1}{k+1}})$ , uniformly w.r.t.  $z \in [z_0,z_1]$ , where  $\eta$  is possibly depending on  $\epsilon, z$ .

# 4.5.2 Controllability properties between the eigenstates for the normal forms

Let  $f : (u, v, z) \mapsto \begin{pmatrix} h_1(u, v, z)(z - m(u)u) \\ h_2(u, v, z)(z + u + v^2) \end{pmatrix}$ , where  $h_1, h_1 \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfy  $h_1(0) = h_2(0) \neq 0$  and  $m(0) \notin \{-1, 0\}$ . Recall that f has a F-semiconical intersection at 0.

Consider a compact neighborhood S of 0 in  $\mathbb{R}^3$  on which the product  $h_1h_2$  does not vanish and m is different from 0 and -1. Assume that S writes  $S = U \times [z_0, z_1]$ , where U is a compact neighborhood of 0 in  $\mathbb{R}^2$  and  $z_0 < 0 < z_1$ . Define  $\mathcal{C} = Z(f) \cap S$  and, for every  $(u, v, z) \in S$ ,  $h(u, v, z) = \frac{h_1(u, v, z)}{h_2(u, v, z)}$ . Notice that  $(u, v, z) \in S$  is in  $\mathcal{C}$  if and only if  $m(u)u = z = -u - v^2$ . Up to restricting U, we can assume that  $u \to (m(u) + 1)u$  is monotone, so that

$$(m(u)+1)u = -v^2 \tag{4.17}$$

defines a smooth submanifold of U which contains the projection  $\pi(\mathcal{C})$  of  $\mathcal{C}$  onto the plane of controls  $(u, v) \in \mathbb{R}^2$ .

Without loss of generality, assume that m(0) > -1 (the case m(0) < -1 being analogous). According to (4.17), this means that  $\pi(\mathcal{C})$  lies in the intersection of U with the left half-plane. Notice that the sign of z = m(u)u on  $\mathcal{C}$  is the opposite as the sign of m(0).

#### Uniform adiabatic estimates when $(u(t), v(t))_{t \in [0,1]} \subset \pi(\mathcal{C})$

Assume that  $(u, v) : [0, 1] \to U$  is a regular  $C^{\infty}$  control path satisfying the following conditions, referred to as (**C**):

 $\begin{array}{l} - & (u,v) \subset \pi(\mathcal{C}) \\ - & (u,v)(0) = (u_0,v_0) \text{ where } (u_0,v_0,z_0) \in \mathcal{C} \text{ is a F-conical intersection for } f; \\ - & (u,v)(1) = (0,0). \end{array}$ 

Under the previous assumptions, for every  $z \in [z_0, 0]$ , we can consider  $t_z$  as the unique element in [0, 1] that satisfies  $(u(t_z), v(t_z), z) \in \mathcal{C}$ . On the other hand, for  $z \in (0, z_1]$  there exist no  $t \in [0, 1]$  such that  $(u(t), v(t), z) \in \mathcal{C}$ . By the regularity of (u, v), the application  $[z_0, 0] \ni z \mapsto t_z$  is  $C^{\infty}$ . Moreover, as a direct consequence of equation (4.17), which holds on  $\mathcal{C}$ , we have  $f_1(u(t), v(t), z) = h(u(t), v(t), z)f_2(u(t), v(t), z)$  for every  $t \in [0, 1]$  and  $z \in [z_0, z_1]$ . **Definition 4.5.2.** For any  $z \in [z_0, z_1]$ , define  $V_z : [0, 1] \to \mathbb{R}$  as follows: if  $z \in [z_0, 0)$ , let

$$V_z(t) = -(h(u(t), v(t), z) - \sqrt{1 + h(u(t), v(t), z)^2})$$

for  $t < t_z$  and  $V_z(t) = -(h(u(t), v(t), z) + \sqrt{1 + h(u(t), v(t), z)^2})$  for  $t \ge t_z$ ; if  $z \in [0, z_1]$ , let

$$V_z(t) = -(h(u(t), v(t), z) - \sqrt{1 + h(u(t), v(t), z)^2})$$

for every  $t \in [0, 1]$ .

By Propositions 4.4.2 and 4.4.4, we get the following result on the regularity of eigenpairs.

**Proposition 4.5.3** (Regularity of eigenpairs for a control path  $(u, v) \subset \pi(\mathcal{C})$ ). Let (u, v)satisfy (C). For  $z \in [0, z_1]$ , define, for every  $t \in [0, 1]$ ,  $\Phi_0^z(t) = \phi_-(u(t), v(t))$ ,  $\Phi_1^z(t) = \phi_+(u(t), v(t))$ ,  $\lambda_0^z(t) = \lambda_-^z(u(t), v(t))$ , and  $\lambda_1^z(t) = \lambda_+^z(u(t), v(t))$ . For  $z \in [z_0, 0)$ , define  $\lambda_0^z, \lambda_1^z : [0, 1] \to \mathbb{R}$  by

$$\begin{aligned} \lambda_0^z(t) &= \lambda_-^z(u(t), v(t)), \ \lambda_1^z(t) &= \lambda_+^z(u(t), v(t)), \\ \lambda_0^z(t) &= \lambda_+^z(u(t), v(t)), \ \lambda_1^z(t) &= \lambda_-^z(u(t), v(t)), \end{aligned} for \ t &\geq t_z, \end{aligned}$$

and  $\Phi_0^z, \Phi_1^z: [0,1] \to \mathbb{R}^2$  by

$$\begin{split} \Phi_0^z(t) &= \phi_-^z(u(t), v(t)), \ \Phi_1^z(t) = \phi_+^z(u(t), v(t)), \qquad \text{for } t < t_z \\ \Phi_0^z(t) &= \phi_+^z(u(t), v(t)), \ \Phi_1^z(t) = \phi_-^z(u(t), v(t)), \qquad \text{for } t \ge t_z \end{split}$$

Then, for every  $z \in [z_0, z_1]$ ,  $\lambda_0^z, \lambda_1^z$  and  $\Phi_0^z, \Phi_1^z$  are  $C^{\infty}$  on [0, 1]. Moreover,  $\Phi_0^z$  and  $\Phi_1^z$  can be written as  $\Phi_0^z = \frac{1}{\sqrt{1+V_z^2}} \begin{pmatrix} -1 \\ V_z \end{pmatrix}$  and  $\Phi_1^z = \frac{1}{\sqrt{1+V_z^2}} \begin{pmatrix} V_z \\ 1 \end{pmatrix}$ , where  $V_z \in C^{\infty}([0, 1], \mathbb{R})$  is defined as in Definition 4.5.2.

A direct corollary of Proposition 4.5.3 is the following.

**Proposition 4.5.4.** For any  $z \in [z_0, z_1]$ , let  $\theta_z = \arctan(V_z) \in C^{\infty}([0, 1], \mathbb{R})$ , where  $V_z$  is defined as in Definition 4.5.2. Then  $(t, z) \mapsto \dot{\theta}_z(t)$  and  $(t, z) \mapsto \ddot{\theta}_z(t)$  are bounded w.r.t.  $(t, z) \in [0, 1] \times [z_0, z_1]$ .

Proof. By definition of  $V_z$ , for every  $(t, z) \in [0, 1] \times [z_0, z_1]$ ,  $\dot{\theta}_z(t)$  and  $\ddot{\theta}_z(t)$  depend only on  $\frac{d}{dt}h(u(t), v(t), z)$  and  $\frac{d^2}{dt^2}h(u(t), v(t), z)$ , which are uniformly bounded w.r.t.  $(t, z) \in [0, 1] \times [z_0, z_1]$  because  $h \in C^{\infty}(\mathcal{S}, \mathbb{R})$ .

**Remark 4.5.5.** In the particular (non-generic) case in which h is constant, the curve (u, v) is non-mixing for all  $z \in [z_0, z_1]$ , in the sense developed in [22]. Non-mixing curves are characterized by an enhanced adiabatic approximation with respect to general curves passing through an eigenvalue intersection.

**Proposition 4.5.6.** The functions  $\lambda_0^z, \lambda_1^z$ , defined as in Proposition 4.5.3, satisfy (4.16) with k = 2.

*Proof.* As a first step of the proof, let us show the following local estimate: There exist  $t_1 \in [0, 1)$ , a nonempty compact neighborhood  $W \subset [z_0, z_1]$  of 0, and  $C_1 > 0$  independent of z such that for every  $t \in [t_1, 1]$ ,  $j \in \{0, 1\}$ , and  $z \in W$  we have

$$\left| \int_{t_1}^t e^{\frac{2i}{\epsilon} \int_0^s \lambda_j^z(r) dr} ds \right| \le C_1 \epsilon^{\frac{1}{3}}, \qquad \forall \epsilon > 0.$$
(4.18)

According to Corollary 4.7.8, it is enough to prove that there exist  $t_1$  and W as above such that

$$|\ddot{\lambda}_z(t)| > c, \qquad \forall z \in W, \ t \in [t_1, 1] \setminus \{t_z\},$$
(4.19)

where c > 0 is independent of  $z \in W$ . Notice that

$$\lambda_z(t) = |z - m(u(t))u(t)| \sqrt{h_1(u(t), v(t), z)^2 + h_2(u(t), v(t), z)^2}.$$

By hypothesis (C),  $u(1) = \dot{u}(1) = 0$  and  $\ddot{u}(1) = 0$ . Hence,

$$\left. \frac{d}{dt} (z - m(u(t))u(t)) \right|_{t=1} = 0, \quad \left. \frac{d^2}{dt^2} (z - m(u(t))u(t)) \right|_{t=1} = (z - m(0))\ddot{u}(1),$$

and, in particular, for z = 0 we have

$$\lim_{z \to 0} \left| \frac{d^2}{dt^2} \lambda_z(t) \right|_{t=1} = |m(0)\ddot{u}(1)| > 0.$$

Inequality (4.19), and hence the required local estimate (4.18), follow by a continuity argument. Notice that, up to restricting W or increasing  $t_1$ , we can assume that  $\{t_z \mid z \in W \cap [z_0, 0]\} = [t_1, 1]$ .

Let us now extend (4.18) to  $z \in [z_0, z_1]$  and  $t \in [0, 1]$ . For  $z \in [0, z_1] \setminus W$ , there exists  $c_1 > 0$  (independent of z) such that  $|\lambda_0^z(t)| > c_1 > 0$  for every  $t \in [0, 1]$ . Hence, by applying Lemma 4.7.9, we have  $|\int_0^t e^{\frac{2i}{\epsilon} \int_0^s \lambda_0^z(r) dr} ds| \leq C_1 \epsilon$ , where  $C_1 > 0$  is independent of  $(t, z) \in [0, 1] \times ([0, z_1] \setminus W)$ .

For every z in  $[z_0, 0)$  we have  $\dot{\lambda}_0^z(t_z) \neq 0$ . By continuity of the applications  $z \mapsto t_z$ and  $(t, z) \mapsto \dot{\lambda}_0^z(t)$ , there exist  $\alpha, c_2 > 0$  such that  $|\dot{\lambda}_0^z(t)| > c_2 > 0$  for every  $z \in [z_0, 0]$ such that  $t_z \leq (t_1 + 1)/2$  and every  $t \in [t_z - \alpha, t_z + \alpha]$ . By continuity of the application  $(t, z) \mapsto \lambda_0^z(t)$ , moreover, we get the existence of  $c_3 > 0$  such that  $|\lambda_0^z(t)| > c_3 > 0$  for every  $z \in [z_0, 0] \setminus W$  and every  $t \in [0, 1] \setminus [t_z - \alpha, t_z + \alpha]$ , also for every  $z \in [z_0, 0] \cap W$ and every  $t \in [0, t_1] \setminus [t_z - \alpha, t_z + \alpha]$ .

For  $z \in [z_0, 0] \setminus W$  and  $t \in [0, 1]$ , we write

$$\int_0^t e^{\frac{2i}{\epsilon} \int_0^s \lambda_0^z(r) dr} ds = \int_{[0,t] \cap [0,t_1-\alpha]} e^{\frac{2i}{\epsilon} \int_0^s \lambda_0^z(r) dr} ds + \int_{[0,t] \cap [t_1-\alpha,t_1+\alpha]} e^{\frac{2i}{\epsilon} \int_0^s \lambda_0^z(r) dr} ds + \int_{[0,t] \cap [t_1+\alpha,1]} e^{\frac{2i}{\epsilon} \int_0^s \lambda_0^z(r) dr} ds,$$

and we conclude, up a change of time variable, by applying Corollary 4.7.8 (on  $[0, t] \cap [t_1 - \alpha, t_1 + \alpha]$ , with k = 2) and Lemma 4.7.9 (on  $[0, t] \cap [0, t_1 - \alpha]$  and  $[0, t] \cap [t_1 + \alpha, 1]$ ).

We conclude similarly for  $z \in W \cap [z_0, 0]$ , by splitting [0, 1] in the intervals  $[0, \min(t_1, t_z - \alpha)]$ ,  $[\min(t_1, t_z - \alpha), t_1]$ , and  $[t_1, 1]$  and by applying Corollary 4.7.8, Lemma 4.7.9, and (4.18).

Proposition 4.5.4 and 4.5.6 allow us to apply Theorem 4.5.1 and deduce the following ensemble adiabatic approximation result.

**Theorem 4.5.7** (Semi-conical case). Let (u, v) be a regular  $C^{\infty}$  control path satisfying condition (C). Let  $\psi_{\epsilon}^z$  be the solution of Equation (4.13), where  $\tilde{\psi}_0^z = \phi_-^z(u(0), v(0))$  for every  $z \in (z_0, z_1]$  and  $\tilde{\psi}_0^{z_0} = \lim_{t\to 0^+} \phi_-^{z_0}(u(t), v(t))$ . Set  $T_{\epsilon}(z) = |\langle \psi_{\epsilon}^z(\frac{1}{\epsilon}), \phi_+^z(u(1), v(1)) \rangle|$ . Then

$$\lim_{\epsilon \to 0} T_{\epsilon}(z) = \begin{cases} 0 & \text{if } z \in [0, z_1] \cup \{z_0\}, \\ 1 & \text{if } z \in (z_0, 0), \end{cases}$$

the convergence being uniform w.r.t.  $z \in [z_0, z_1]$ . More precisely, we have  $T_{\epsilon}(z) = O(\epsilon^{1/3})$ for  $z \in [0, z_1] \cup \{z_0\}$ 

#### **4.5.3** The control path (u, v) exits from $\pi(f)$ .

By similar arguments as those developed in [10], we get the following proposition.

**Proposition 4.5.8** (Conical exit). Let  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ . Let  $(u_1, v_1, z_1)$  be a F-conical intersection for f. Let N be a neighborhood of  $(u_1, v_1, z_1)$  in  $\mathbb{R}^3$  such that  $Z(f) \cap N$  is made of F-conical intersections only and  $\pi(Z(f) \cap N)$  is a  $C^{\infty}$  submanifold of  $\mathbb{R}^2$ . Let  $(u_0, v_0, z_0) \in Z(f) \cap N$  be such that  $z_0 < z_1$ . Consider a regular  $C^3$  control path  $(u(t), v(t))_{t \in [0,1]}$  and a time  $t_1 \in (0,1)$  such that  $(u(0), v(0)) = (u_0, v_0)$ ,  $(u(t_1), v(t_1)) = (u_1, v_1)$ ,  $(u(t), v(t)) \in \pi(Z(f) \cap N)$  for  $t \in [0, t_1]$ , and  $(u(t), v(t)) \notin \pi(f)$  for  $t > t_1$ . For every  $z \in \mathbb{R}$ , consider  $\theta_z \in C^2([0, 1], \mathbb{R})$  and  $\lambda_z \in C^2([0, 1], \mathbb{R})$  as in Theorem 4.5.1. Then  $(t, z) \mapsto \dot{\theta}_z(t)$  and  $(t, z) \mapsto \ddot{\theta}_z(t)$  are uniformly bounded with respect to  $(t, z) \in [0, 1] \times [z_0, z_1]$ , and there exists c > 0 such that for every  $z \in (z_0, z_1]$  and for every t in [0, 1],

$$\left|\int_0^t e^{\frac{2i}{\epsilon}\int_0^s \lambda_z(x)dx}ds\right| \le c\epsilon^{\frac{1}{2}}, \qquad \forall \epsilon > 0.$$

**Corollary 4.5.9.** Let  $f_{\epsilon}(u, v)$  as in Proposition 4.5.8. Define  $\psi_{\epsilon}^{z}$  as the solution of (4.13) with  $\tilde{\psi}_{0}^{z} = \phi_{-}^{z}(u(0), v(0))$  for  $z \neq 0$  and  $\tilde{\psi}_{0}^{z_{0}} = \lim_{t\to 0} \phi_{-}^{z_{0}}(u(t), v(t))$ . Then for every  $z \in (z_{0}, z_{1}]$  and for every t in [0, 1],

$$|\langle \psi_{\epsilon}^{z}(1/\epsilon), \phi_{+}^{z}(u(1), v(1)) \rangle| = 1 + O(\epsilon^{1/2}).$$

Moreover, for every  $\overline{z} \leq z_0$ , we have, uniformly w.r.t.  $z \in [\overline{z}, z_0]$ ,

$$|\langle \psi_{\epsilon}^{z}(1/\epsilon), \phi_{+}^{z}(u(1), v(1)) \rangle| = O(\epsilon^{1/2}).$$

**Remark 4.5.10.** For such a control path, we can notice that the adiabatic approximation is not uniform for every  $z \in (z_1, z_1 + \alpha]$  for  $\alpha > 0$ .

#### 4.5.4 Proof of Theorem 4.1.3

Proof of Theorem 4.1.3. Consider a regular  $C^4$  control path  $(\eta(t))_{t\in[0,1]} = (u(t), v(t))_{t\in[0,1]}$ such that  $\eta(0) = \eta(1), \ \eta(t_0) = (u_0, v_0), \ \eta(t_1) = (u_1, v_1), \ \eta(t) \in \pi(\gamma)$  for  $t \in [t_0, t_1]$ , and  $\eta(t) \notin \pi(f)$  for  $t \notin [t_0, t_1]$ . Under these hypotheses, we can define, for every  $z \in \mathbb{R}$ ,  $\theta_z \in C^2([0, 1], \mathbb{R})$  and  $\lambda_z \in C^2([0, 1], \mathbb{R})$  along the path  $\eta$ , as required in Theorem 4.5.1.

For  $t \in [t_0, t_1]$ , the hypothesis of non-existence of self-intersections for  $\pi(\gamma)$  guarantees that we can apply the same arguments as those used in the proof of Proposition 4.5.6 for the normal form in order to get

$$\left|\int_{t_0}^t e^{\frac{2i}{\epsilon}\int_0^s \lambda_z(r)dr} ds\right| \le C\epsilon^{\frac{1}{3}}, \qquad \forall \epsilon > 0$$

where C > 0 is independent of  $(t, z) \in [t_0, t_1] \times [z_0, z_1]$ . Moreover by Proposition 4.5.4,  $(t, z) \mapsto \dot{\theta}_z(t)$  and  $(t, z) \mapsto \ddot{\theta}_z(t)$  are bounded on  $[t_0, t_1] \times [z_0, z_1]$ .

Under the assumptions that  $z \in [z_0, z_1]$  for every z and t such that  $(u(t), v(t), z) \in \gamma$ and that  $(u_1, v_1, z_1)$  is a F-conical intersection, we can apply Proposition 4.5.8 and get that  $(t, z) \mapsto \dot{\theta}_z(t)$  and  $(t, z) \mapsto \ddot{\theta}_z(t)$  are uniformly bounded with respect to  $(t, z) \in$  $[t_1, 1] \times [z_0, z_1]$ , and there exists c > 0 such that for every  $z \in [z_0, z_1]$  and for every t in  $[t_1, 1]$ ,

$$\left|\int_{t_1}^t e^{\frac{2i}{\epsilon}\int_0^s \lambda_z(x)dx}ds\right| \le c\epsilon^{\frac{1}{2}}, \qquad \forall \epsilon > 0.$$

By similar arguments for  $t \in [0, t_0]$ , we obtain that on the whole interval  $t \in [0, 1], (t, z) \mapsto \dot{\theta}_z(t)$  and  $(t, z) \mapsto \ddot{\theta}_z(t)$  are uniformly bounded with respect to  $(t, z) \in [0, 1] \times [z_0, z_1]$ , and by triangular inequality, there exists  $\tilde{C} > 0$  such that for every  $z \in [z_0, z_1]$  and for every t in [0, 1],

$$\left|\int_0^t e^{\frac{2i}{\epsilon}\int_0^s \lambda_z(x)dx}ds\right| \le \tilde{C}\epsilon^{\frac{1}{3}}, \qquad \forall \epsilon > 0.$$

We get the expected result by applying Theorem 4.5.1.

### 4.6 Extension to *n*-level systems

# 4.6.1 Generic assumptions on n-level Hamiltonians and adiabatic decoupling

In this section, we show that the study of a n-level real Hamiltonian can be reduced locally to the study of a 2-level Hamiltonian in the adiabatic regime and that such a transformation preserves the codimension of the generic conditions expressed in Section 4.2.1. Such a reduction allows us to define a semi-conical intersection model for a *n*-level real Hamiltonian.

For every  $H \in S_n(\mathbb{R})$  denote by  $(\lambda_j(H))_{j=1}^n$  the spectrum of H, where  $j \mapsto \lambda_j(H)$  is the nondecreasing sequence of eigenvalues of H repeated according to their multiplicities. We write  $(\phi_1(H), \ldots, \phi_n(H))$  to denote an orthonormal basis of associated eigenvectors.

Next lemma is a classical result of continuity of the spectrum (see, for instance [68]).

**Lemma 4.6.1.** Let  $H_0 \in S_n(\mathbb{R})$  and  $j \in \{1, ..., n-1\}$  be such that  $\lambda_j(H_0), \lambda_{j+1}(H_0)$ are separated from the rest of the spectrum of  $H_0$ . Then, there exists a neighborhood V of  $H_0$  in  $S_n(\mathbb{R})$  and a Jordan curve c in  $\mathbb{C}$  separating  $\{\lambda_q(H) \mid q \in \{j, j+1\}, H \in V\}$  from  $\cup_{H \in V}(Spectrum(H) \setminus \{\lambda_j(H), \lambda_{j+1}(H)\}).$ 

From now on, we consider  $H_0, j, c, V$  verifying Lemma 4.6.1. For all  $H \in V$ , we consider  $P_{j,j+1}(H) = \frac{1}{2i\pi} \int_c (H-\xi)^{-1} d\xi$ . Notice that,  $P_{j,j+1}(H)$  is a real matrix because H is real. By construction of  $c, V \ni H \mapsto P_H$  is smooth. Up to reducing V, for every H we can consider an orthogonal mapping  $I(H) : \mathbb{R}^2 \to \operatorname{Im}(P_{j,j+1}(H))$  such that  $V \ni H \mapsto I(H)$  is smooth. For every  $H \in V$  define  $\pi_{j,j+1}(H) = \operatorname{Im}(P_{j,j+1}(H)), I^{-1}(H)$  as the inverse of I(H) on  $\pi_{j,j+1}(H)$  and

 $F(H) = I^{-1}(H)HI(H) \in S_2(\mathbb{R}).$ 

Notice that  $I^{-1}(H) = {}^{t}I(H)$ .

Consider  $H \in C^{\infty}(\mathbb{R}^k, S_n(\mathbb{R}))$  such that  $H(0) = H_0$ , and denote by W a neighborhood of 0 in  $\mathbb{R}^k$  such that  $H(u) \in V$  for every  $u \in W$ . Define  $h \in C^{\infty}(\mathbb{R}^k, S_2(\mathbb{R}))$  such that for every  $u \in W$ ,  $h(u) = (F \circ H)(u)$ . We say that h is a *reduced Hamiltonian* for H. Notice that if  $\phi \in \mathbb{C}^2$  is an eigenvector of h(u) associated with the eigenvalue  $\lambda \in \mathbb{R}$  then  $I(H(u))\phi$  is an eigenvector of H(u) associated with the same eigenvalue  $\lambda$ . We deduce from this, as it has been already used in [10], that the regularity of the eigenpairs of Hwith respect to  $u \in W$  can be deduced from the regularity of those of h.

**Proposition 4.6.2.** *F* is a submersion from *V* to  $S_2(\mathbb{R})$ .

*Proof.* Consider  $A \in V$ . Define  $\psi_1 = I(A)e_1, \psi_2 = I(A)e_2$  where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{C}^2$ .

- Define  $H = h_{11}\psi_1 \psi_1 + h_{22}\psi_2 \psi_2 + h_{12}\psi_1 \psi_2 + h_{12}\psi_2 \psi_1$  with  $h_{11}, h_{22}, h_{12} \in \mathbb{R}$ . By direct computations, we get

$${}^{t}I(A)HI(A) = \begin{pmatrix} h_{11} & h_{12} \\ h_{12} & h_{22} \end{pmatrix}.$$

Hence, the application  $S_n(\mathbb{R}) \ni H \mapsto {}^tI(A)HI(A) \in S_2(\mathbb{R})$  is surjective.

— For  $H \in S_n(\mathbb{R})$  such that  $A + H \in V$ ,

$$F(A + H) = {}^{t}I(A + H)(A + H)I(A + H)$$
  
= F(A) + {}^{t}I(A)HI(A) + {}^{t}DI\_{A}(H)AI(A) + {}^{t}I(A)ADI\_{A}(H) + o(H).

Hence,  $\forall H \in S_n(\mathbb{R})$ ,  $DF_A(H) = {}^tI(A)HI(A) + {}^tDI_A(H)AI(A) + {}^tI(A)ADI_A(H)$ . Let us consider  $H = h_{11}\psi_1{}^t\psi_1 + h_{22}\psi_2{}^t\psi_2 + h_{12}\psi_1{}^t\psi_2 + h_{12}\psi_2{}^t\psi_1$ . Then, we have  $DF_A(H) = {}^tI(A)HI(A)$ . Hence, F is a submersion.

Using classical facts on the composition of k-jets (see [65]), we obtain the following result.

**Proposition 4.6.3.** Consider  $\mathcal{F} : J^2(W, S_n(\mathbb{R})) \to J^2(W, S_2(\mathbb{R}))$  such that for every  $u \in W$ ,  $j^2(h)(u) = \mathcal{F}(j^2(H)(u))$ . Then  $\mathcal{F}$  is a submersion.

It follows that if S is a codimension q smooth submanifold of  $J^2(W, S_2(\mathbb{R}))$ , then  $\mathcal{F}^{-1}(S)$  is a codimension q smooth submanifold of  $J^2(W, S_n(\mathbb{R}))$ . This can be used to deduce generic properties for  $H \in C^{\infty}(W, S_n(\mathbb{R}))$  from generic properties for  $h \in C^{\infty}(W, S_2(\mathbb{R}))$ .

#### 4.6.2 Adiabatic decoupling

We present here some results of adiabatic decoupling. They are adapted from [78], and are interesting because the adiabatic structure is preserved for the local two-levels dynamics.

**Theorem 4.6.4** (Adiabatic decoupling). Let  $H \in C^{\infty}(\mathbb{R}^k, S_n(\mathbb{R}))$  and  $j \in \{1, \ldots, n-1\}$ be such that  $\{\lambda_q(H(u)) \mid q \in \{j, j+1\}\}$  is separated from

$$Spectrum(H(u)) \setminus \{\lambda_q(H(u)) \mid q \in \{j, j+1\}\}$$

for u in a neighborhood W of 0 in  $\mathbb{R}^k$ . Define, for every  $u \in W$ , I(H(u)) and h(u) as in Section 4.6.1. Consider a  $C^2$  regular path  $u : [0,1] \to W$  such that there exist  $\ell \in \mathbb{N}$ and  $C^\ell$  functions  $\Lambda_j, \Lambda_{j+1} : [0,1] \to \mathbb{R}$  such that for every  $t \in [0,1], \{\Lambda_j(t), \Lambda_{j+1}(t)\} =$  $\{\lambda_j(H(u(t))), \lambda_{j+1}(H(u(t)))\}$  and that h admits  $C^2$  eigenvectors along u.

Assume that there exists c > 0 such that

$$\left|\int_{0}^{t} e^{\frac{i}{\epsilon} \int_{0}^{s} (\Lambda_{j+1}(x) - \Lambda_{j}(x)) dx} ds\right| \le c \epsilon^{1/(\ell+1)}, \qquad \forall t \in [0, 1].$$
(4.20)

Let  $\tilde{\psi}_0 \in \mathbb{C}^2$ . Then the solutions  $\psi_{\epsilon}$  and  $\tilde{\psi}_{\epsilon}$  of, respectively,

$$i\frac{d\psi}{dt} = H(u(\epsilon t))\psi, \ \psi(0) = I(H(u(0)))\tilde{\psi}_0, \quad and \quad i\frac{d\tilde{\psi}}{dt} = h(u(\epsilon t))\tilde{\psi}, \ \tilde{\psi}(0) = \tilde{\psi}_0,$$

are such that  $\psi_{\epsilon}(1/\epsilon)$  is  $O(\epsilon^{\frac{1}{\ell+1}})$ -close to  $I(H(u(1)))\tilde{\psi}_{\epsilon}(1/\epsilon)$ .

*Proof.* Define for  $q \in \{1, 2\}$ , and for every  $u \in W$ ,  $\psi_q(u) = I(H(u))e_q$ , where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{C}^2$ . Define  $\psi_{\epsilon}^{\text{eff}}(t)$  as the solution for  $t \in [0, \frac{1}{\epsilon}]$ , of

$$i\frac{d\psi_{\epsilon}^{\text{eff}}(t)}{dt} = \left(h(u(\epsilon t)) - i\epsilon \begin{pmatrix} 0 & \langle \dot{\psi}_{j}(u(\epsilon t)), \psi_{j+1}(u(\epsilon t)) \rangle \\ \langle \dot{\psi}_{j+1}(u(\epsilon t)), \psi_{j}(u(\epsilon t)) \rangle & 0 \end{pmatrix} \right) \psi_{\epsilon}^{\text{eff}}(t),$$

$$(4.21)$$

with  $\psi_{\epsilon}^{\text{eff}}(0) = \tilde{\psi}_0.$ 

By [78, Theorem 1.4], there exists C > 0, such that, for every  $t \in [0, \frac{1}{\epsilon}]$ ,

$$\|\psi_{\epsilon}(t) - I(h(u(\epsilon t)))\psi_{\epsilon}^{\text{eff}}(t)\| \le C\epsilon, \qquad \forall \epsilon > 0.$$

Under the assumptions of the theorem, we can consider a  $C^2$  basis of eigenvectors of h along u. Hence, by the same arguments as those used in Section 4.4.1 in order to prove Theorem 4.4.1, there exists c > 0 such that, for every  $t \in [0, \frac{1}{\epsilon}]$ ,

$$\|\tilde{\psi}_{\epsilon}(t) - \psi_{\epsilon}^{\text{eff}}(t)\| \le c\epsilon^{\frac{1}{\ell+1}}, \qquad \forall \epsilon > 0.$$

We get the expected result by triangular inequality.

In the parametric case, using estimates that are uniform with respect to the parameter z, we get the following extension of Theorem 4.6.4.

**Theorem 4.6.5** (Adiabatic decoupling for parametric systems). Let  $H \in C^{\infty}(\mathbb{R}^{k+1}, S_n(\mathbb{R}))$ and  $j \in \{1, ..., n-1\}$  be such that  $\{\lambda_q(H(u, z)) \mid q \in \{j, j+1\}\}$  is separated from

$$Spectrum(H(u,z)) \setminus \{\lambda_q(H(u,z)) \mid q \in \{j, j+1\}\}$$

for u in a neighborhood W of 0 in  $\mathbb{R}^k$  and  $z \in [z_0, z_1]$ . Define, for every  $(u, z) \in W \times [z_0, z_1]$ , I(H(u, z)) and h(u, z) as in Section 4.6.1. Consider a  $C^2$  regular path  $u : [0, 1] \to W$  and  $\tilde{\psi}_0^z \in \mathbb{C}^2$ , for every  $z \in [z_0, z_1]$ . Let  $\ell \in \mathbb{N}$  and assume that for every  $z \in [z_0, z_1]$ , there exist  $C^\ell$  functions  $\Lambda_j^z, \Lambda_{j+1}^z : [0, 1] \to \mathbb{R}$  such that for every  $t \in [0, 1], \{\Lambda_j^z(t), \Lambda_{j+1}^z(t)\} = \{\lambda_j^z(u(t)), \lambda_{j+1}^z(u(t))\}$  and that, for every  $z \in [z_0, z_1]$ ,  $h(u(\cdot), z)$  admits  $C^2$  eigenvectors  $\Phi_j^z, \Phi_{j+1}^z$  such that  $(t, z) \mapsto \frac{d\Phi_q^z(t)}{dt}$  and  $(t, z) \mapsto \frac{d^2\Phi_q^z(t)}{dt^2}$  are bounded uniformly with respect to  $(t, z) \in [0, 1] \times [z_0, z_1]$ , for every  $q \in \{j, j+1\}$ . Assume that there exists c > 0 such that

$$\left|\int_{0}^{t} e^{\frac{i}{\epsilon}\int_{0}^{s}(\Lambda_{j+1}^{z}(x)-\Lambda_{j}^{z}(x))dx}ds\right| \le c\epsilon^{1/(\ell+1)}, \qquad \forall t \in [0,1], \qquad \forall z \in [z_{0}, z_{1}].$$
(4.22)

Then the solutions  $\psi^z_{\epsilon}$  and  $\tilde{\psi}^z_{\epsilon}$  of, respectively,

$$i\frac{d\psi^{z}}{dt} = H(u(\epsilon t))\psi^{z}, \ \psi^{z}(0) = I(H(u(0), z))\tilde{\psi}_{0}^{z}, \quad and \quad i\frac{d\tilde{\psi}^{z}}{dt} = h(u(\epsilon t), z)\tilde{\psi}^{z}, \ \tilde{\psi}^{z}(0) = \tilde{\psi}_{0}^{z},$$

are such that  $\psi^{z}(1/\epsilon)$  is  $O(\epsilon^{\frac{1}{\ell+1}})$ -close to  $I(H(u(1),z))\tilde{\psi}^{z}(1/\epsilon)$ , uniformly w.r.t.  $z \in [z_0, z_1]$ .

#### 4.6.3 Semi-conical intersections for *n*-level quantum systems

Let  $H \in C^{\infty}(\mathbb{R}^k, S_n(\mathbb{R}))$  and  $j \in \{1, \ldots, n-1\}$  be such that  $\{\lambda_q(H(u)) \mid q \in \{j, j+1\}\}$ is separated from Spectrum $(H(u)) \setminus \{\lambda_q(H(u)) \mid q \in \{j, j+1\}\}$  for u in a neighborhood W of 0 in  $\mathbb{R}^k$ . Define, for every  $u \in W$ , I(H(u)) and h(u) as in Section 4.6.1. Define for  $q \in \{1, 2\}$ , and for every  $u \in W$ ,  $\psi_q(u) = I(H(u))e_q$ , where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{C}^2$ . Then we have the identity

$$h(u) = \begin{pmatrix} \langle \psi_1(u), H(u)\psi_1(u) \rangle & \langle \psi_1(u), H(u)\psi_2(u) \rangle \\ \langle \psi_1(u), H(u)\psi_2(u) \rangle & \langle \psi_2(u), H(u)\psi_2(u) \rangle \end{pmatrix}$$

**Definition 4.6.6.** By removing the trace of H(u), that is equivalent to changing a global phase in the dynamics, define the reduced zero-trace Hamiltonian of H as, for every  $u \in W$ ,

$$h_{\rm red}(u) = \begin{pmatrix} f_1(u) & f_2(u) \\ f_2(u) & -f_1(u) \end{pmatrix},$$
  
with  $f_1(u) = \frac{1}{2} \left( \langle \psi_1(u), H(u)\psi_1(u) \rangle - \langle \psi_2(u), H(u)\psi_2(u) \rangle \right)$  and  $f_2(u) = \langle \psi_1(u), H(u)\psi_2(u) \rangle$ 

Assume k = 2. By a slight abuse of notations, write  $u := (u, v) \in U$ , where U is a connected open neighborhood of the origin in  $\mathbb{R}^2$ . In the next proposition, we prove that conicity properties do not depend on the choice of the unitary transformation  $I(H(u, v)) : \mathbb{C}^2 \to \operatorname{Im}(P_{j,j+1}(H(u, v)))$ .

**Proposition 4.6.7.** Let  $f \in C^{\infty}(U, \mathbb{R}^2)$  and define

$$R_{\theta}(u,v) = \begin{pmatrix} \cos(\theta(u,v)) & -\sin(\theta(u,v)) \\ \sin(\theta(u,v)) & \cos(\theta(u,v)) \end{pmatrix} \in \mathrm{SO}_{2}(\mathbb{R}),$$

with  $\theta \in C^{\infty}(U, \mathbb{R})$ . Define  $\tilde{f} \in C^{\infty}(U, \mathbb{R}^2)$  such that  $H_{\tilde{f}}(u, v) = R_{\theta}(u, v)H_f(u, v)^t R_{\theta}(u, v)$ for every (u, v) in U. Then

- 0 is conical for f if and only if 0 is conical for  $\tilde{f}$ ;
- 0 is semi-conical for f if and only if 0 is semi-conical for  $\tilde{f}$ . Moreover, their non-conical directions are the same.

*Proof.* Using the fact that  $f_1(0) = f_2(0) = 0$ , we get

$$\chi(f)(0) = \chi(f)(0).$$

Hence we get the first claim.

Assume that 0 is semi-conical for f. With no loss of generality, assume that the nonconical direction for f is  $e_2$ , where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{R}^2$ . By direct computations, we have that  $\nabla \tilde{f}_1(0)$  and  $\nabla \tilde{f}_2(0)$  are not both equal to zero. Moreover, we get  $\partial_2 H_{\tilde{f}}(0) = R_{\theta}(0)\partial_2 H_f(0)^t R_{\theta}(0) = 0$ . Hence  $e_2$  is a non-conical direction for  $\tilde{f}$ . Differentiating at the second order, we get

$$\partial_2 \chi(\tilde{f})(0) = \operatorname{Det} \left( R_{2\theta}(0) \begin{pmatrix} \partial_1 f_1(0) & \partial_{22} f_1(0) \\ \partial_1 f_2(0) & \partial_{22} f_2(0) \end{pmatrix} {}^t R_{2\theta}(0) \right)$$
$$= \partial_2 \chi(f)(0) \neq 0.$$

Hence 0 is semi-conical for  $\tilde{f}$ .

The same result holds when for every  $(u, v) \in U$ ,  $S_{\theta}(u, v) \in O_2^{-}(\mathbb{R})$ , and

$$H_{\tilde{f}}(u,v) = S_{\theta}(u,v)H_f(u,v)^t S_{\theta}(u,v).$$

By similar computations, we show that in the parametric case, F-conical intersections and F-semi-conical intersections are invariant under such a orthogonal mapping, possibly depending on the parameter z. We define semi-conical intersections of eigenvalues for  $H(\cdot) \in C^{\infty}(U, S_n(\mathbb{R}))$  and F-conical (respectively F-semi-conical intersections) for  $H(\cdot) \in$  $C^{\infty}(U \times \mathbb{R}, S_n(\mathbb{R}))$  (see Section 4.2.26 for precise definitions of these notions for two level systems) as follows.

**Definition 4.6.8.** *Let*  $j \in \{1, ..., n-1\}$ *.* 

- We say that  $(\bar{u}, \bar{v}) \in U$  is a semi-conical intersection for  $H \in C^{\infty}(U, S_n(\mathbb{R}))$  between the levels j and j+1 if and only if there exists a unitary mapping  $I(H(u, v)) : \mathbb{C}^2 \to$  $\operatorname{Im}(P_{j,j+1}(H(u, v))), C^{\infty}$  with respect to  $(u, v) \in U$ , such that  $(\bar{u}, \bar{v})$  is a semi-conical intersection for the associated reduced Hamiltonian  $h_{\operatorname{red}} \in C^{\infty}(U, S_2(\mathbb{R})).$
- We say that  $(\bar{u}, \bar{v}, \bar{z}) \in U \times \mathbb{R}$  is a F-conical (respectively F-semi-conical) intersection for  $H \in C^{\infty}(U \times \mathbb{R}, S_n(\mathbb{R}))$  between the levels j and j+1 if and only if there exists a unitary mapping  $I(H(u, v, z)) : \mathbb{C}^2 \to \operatorname{Im}(P_{j,j+1}(H(u, v, z))), C^{\infty}$  with respect to  $(u, v, z) \in U \times \mathbb{R}$ , such that  $(\bar{u}, \bar{v}, \bar{z})$  is a F-conical (respectively F-semi-conical) intersection for the associated reduced Hamiltonian  $h_{\operatorname{red}} \in C^{\infty}(U \times \mathbb{R}, S_2(\mathbb{R}))$ .

By Proposition 4.6.3, we get that *F*-conical intersections and *F*-semi-conical intersections as defined in Definition 4.6.8 are generic for  $H \in C^{\infty}(\mathbb{R}^3, S_n(\mathbb{R}))$  endowed with the Whitney topology.

**Remark 4.6.9.** For  $j \in \{1, ..., n-1\}$  set  $Z_j = \{(u, v, z) \in U \times \mathbb{R} \mid \lambda_j(u, v, z) = \lambda_{j+1}(u, v, z)\}$ . By Definition 4.6.8, we have the expected result (see Proposition 4.2.9 for the same property for two-level systems) that if  $(\bar{u}, \bar{v}, \bar{z})$  is a F-semi-conical intersection between the levels j and j + 1, then  $Z_j$  is tangent to the plane  $z = \bar{z}$  at the point  $(\bar{u}, \bar{v}, \bar{z})$  and, considering a local smooth and regular parametrization  $(u(t), v(t), z(t))_{t \in [0,1]}$  of  $Z_j$  and  $\bar{t} \in [0, 1]$  such that  $(u(\bar{t}), v(\bar{t}), z(\bar{t})) = (\bar{u}, \bar{v}, \bar{z})$ , we have  $\dot{z}(\bar{t}) = 0$  and  $\ddot{z}(\bar{t}) \neq 0$ .

#### 4.6.4 Controllability result

We consider the controlled Schrödinger equation in  $\mathbb{C}^n$ ,  $n \in \mathbb{N}$ ,

$$i\frac{d\psi(t)}{dt} = H(u(t), v(t), z)\psi(t).$$
(4.23)

**Definition 4.6.10.** Let  $z_0, z_1 \in \mathbb{R}$ . We say that system (4.23) is ensemble approximately controllable between eigenstates if for every  $\epsilon > 0$ ,  $j, k \in \{1, \ldots, n\}$ , and  $(u_0, v_0), (u_1, v_1) \in$ U such that  $\lambda_j(u_0, v_0, z)$  and  $\lambda_k(u_0, v_0, z)$  are simple for every  $z \in [z_0, z_1]$ , there exists a control  $(u(\cdot), v(\cdot)) \in L^{\infty}([0, T], U)$  such that for every  $z \in [z_0, z_1]$  the solution of (4.23) with initial condition  $\psi^z(0) = \phi_j^z(u_0, v_0)$  satisfies  $\|\psi^z(T) - e^{i\theta}\phi_k^z(u_1, v_1)\| < \epsilon$  for some  $\theta \in \mathbb{R}$  (possibly depending on z and  $\epsilon$ ).

For  $j \in \{1, \ldots, n-1\}$ , let us denote by  $\gamma_j$  the set  $\{(u, v, z) \in U \times [z_0, z_1] \mid \lambda_j(u, v, z) = \lambda_{j+1}(u, v, z)\}$ . Let, moreover,  $\gamma_0 = \gamma_n = \emptyset$ . Denote by  $\pi$  the projection  $\pi : (u, v, z) \mapsto (u, v)$ .

ASSUMPTION  $A_j$ . There exists a connected component  $\hat{\gamma}_j$  of  $\gamma_j$  such that

- $\hat{\gamma}_j$  is a one-dimensional submanifold of  $\mathbb{R}^3$  made of F-conical intersections and F-semi-conical intersections only;
- There exist  $(u_0, v_0) \in U$  and  $(u_1, v_1) \in U$  such that  $(u_0, v_0, z_0), (u_1, v_1, z_1) \in \hat{\gamma}_j$  are F-conical intersections for H;
- $\pi(\hat{\gamma}_j)$  is a  $C^{\infty}$  embedded curve of  $\mathbb{R}^2$  without self-intersections, which is contained in  $U \setminus (\pi(\gamma_{j-1}) \cup \pi(\gamma_{j+1})).$

Using the control strategy proposed in Theorem 4.1.3 and the result of adiabatic decoupling proposed in Theorem 4.6.5, we get the following result.

**Theorem 4.6.11.** Consider a  $C^{\infty}$  map  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$ . Let assumption  $A_j$  be satisfied for every  $j \in \{1, \ldots, n-1\}$ . Then system (4.23) is ensemble approximately controllable between eigenstates.

**Remark 4.6.12.** If the map  $U \times [z_0, z_1] \ni (u, v, z) \mapsto H(u, v, z) \in S_n(\mathbb{R})$  is  $C^4$  then the same result holds.

# 4.7 Appendix

#### 4.7.1 Averaging theorems and estimates of oscillatory integrals

The following theorem is a quantitative version in u(n) of a more general averaging result stated in [5, Lemma 8.2]. Its proof is similar to the proof of [5, Lemma 8.2] using an explicit inequality that yields the speed of convergence of order  $O(\epsilon)$ .

#### 4.7. Appendix

**Theorem 4.7.1.** Consider A and  $(A_{\epsilon})_{\epsilon>0}$  in  $C^{\infty}([0,1], u(n))$  and assume that  $A_{\epsilon}(\tau)$  is uniformly bounded w.r.t.  $(\tau, \epsilon)$ . Denote the flow of the equation  $\frac{dx(\tau)}{d\tau} = A(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau} \in U(n)$  and the flow of the equation  $\frac{dx(\tau)}{d\tau} = A_{\epsilon}(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau}^{\epsilon} \in U(n)$ . If

$$\int_0^\tau A_\epsilon(s)ds = \int_0^\tau A(s)ds + O(\epsilon)$$

then

$$P_{\tau}^{\epsilon} = P_{\tau} + O(\epsilon),$$

both estimates being uniform w.r.t.  $\tau \in [0, 1]$ .

**Remark 4.7.2.** Notice that the hypothesis that  $||A_{\epsilon}||_{\infty}$  is bounded w.r.t.  $\epsilon$  is not explicitly mentioned in [5, Lemma 8.2], but is necessary also for the qualitative version (i.e., for concluding that  $P_{\tau}^{\epsilon} \to P_{\tau}$  as  $\epsilon \to 0$ ).

A direct consequence of Theorem 6.2.3 is the following.

**Corollary 4.7.3** (Quantum 2-level systems averaging). Let  $v, \phi : [0,1] \to \mathbb{R}$  be two smooth functions and, for every  $\epsilon > 0$ , denote by  $P^{\epsilon}_{\tau}$  the flow at time  $\tau \in [0,1]$  of

$$i\frac{dX}{d\tau} = \begin{pmatrix} 0 & v(\tau)e^{\frac{i}{\epsilon}\phi(\tau)} \\ v(\tau)e^{-\frac{i}{\epsilon}\phi(\tau)} & 0 \end{pmatrix} X(\tau).$$

If  $|\int_0^\tau v(s)e^{\frac{i}{\epsilon}\phi(s)}ds| \leq c\epsilon^q$ , where q is a positive real number and c > 0 is independent of  $\epsilon, \tau$ , then  $P_{\tau}^{\epsilon}$  satisfies  $P_{\tau}^{\epsilon} = \mathrm{Id} + O(\epsilon^q)$ .

We recall a classical result (see [76]) which is useful to estimate oscillatory integrals.

**Lemma 4.7.4** (Van Der Corput). Let  $k \in \mathbb{N}$  and  $\phi : [a, b] \to \mathbb{R}$  be smooth and such that  $|\phi^{(k)}(x)| \ge 1$  for all  $x \in [a, b]$ . Assume either that  $k \ge 2$  or that k = 1 and  $\phi'$  is monotone. Then

$$\left| \int_{a}^{b} e^{i\phi(x)/\epsilon} dx \right| \le c_k \epsilon^{1/k},$$

where  $c_k$  is independent of  $\phi$  and  $\epsilon$ .

In the case k = 1, if  $\phi'$  is not monotone we may lose the uniformity of the estimate with respect to the phase  $\phi$ . However, we can recover by a direct integration by parts the following estimate.

**Lemma 4.7.5** (The case k = 1). Let  $\phi : [a, b] \to \mathbb{R}$  be smooth and such that  $|\phi'(x)| \ge 1$  for all  $x \in [a, b]$ . Then

$$\left|\int_{a}^{b} e^{i\phi(x)/\epsilon} dx\right| \leq 2\epsilon + \epsilon \int_{a}^{b} \left|\frac{d}{dx}\frac{1}{\phi'(x)}\right| dx.$$

By integration by parts we also get the following results.

**Corollary 4.7.6.** Let  $\phi$  and k be as in Lemma 4.7.4. Let, moreover,  $v : [a, b] \to \mathbb{R}$  be smooth. Then

$$\left| \int_{a}^{b} v(x) e^{i\phi(x)/\epsilon} dx \right| \le c_k \epsilon^{1/k} \left[ |v(b)| + \int_{a}^{b} |v'(x)| dx \right]$$

where  $c_k$  is the constant obtained in Lemma 4.7.4.

**Corollary 4.7.7.** Let  $\phi$  be as in Lemma 4.7.5 and consider a smooth function  $v : [a, b] \rightarrow \mathbb{R}$ . Then

$$\left| \int_{a}^{b} v(x) e^{i\phi(x)/\epsilon} dx \right| \le c\epsilon$$

where c is independent of  $\epsilon$ .

**Corollary 4.7.8.** Consider an open subset V of  $\mathbb{R}$ . Assume that  $\phi : [a, b] \times V \to \mathbb{R}$  and  $v : [a, b] \times V \to \mathbb{R}$  are real-valued and smooth with respect to the first variable  $x \in [a, b]$ . Assume that there exists k > 1 such that  $|\frac{\partial^k \phi}{\partial x^k}(x, y)| \ge 1$  for all  $x \in [a, b]$  and  $y \in V$ . Then

$$\left| \int_{a}^{b} v(x,y) e^{i\phi(x,y)/\epsilon} dx \right| \le c_k \epsilon^{1/k} \left[ |v(b,y)| + \int_{a}^{b} |\frac{\partial v}{\partial x}(x,y)| dx \right]$$

where  $c_k$  is the constant obtained in the Lemma 4.7.4 (independent of  $\phi$ , y and  $\epsilon$ ). If we assume that v and  $\frac{\partial v}{\partial x}$  are uniformly bounded on  $[a, b] \times V$ , then

$$\left|\int_{a}^{b} v(x,y) e^{i\phi(x,y)/\epsilon} dx\right| \le d_k \epsilon^{1/k}$$

where  $d_k$  depends on v and is independent of  $\phi$ ,  $y \in V$  and  $\epsilon$ .

Next lemma is a direct consequence of Lemma 4.7.5.

**Lemma 4.7.9.** Consider a compact subset V of  $\mathbb{R}$ . Consider two real-valued and smooth functions  $\phi : [a,b] \times V \to \mathbb{R}$  and  $v : [a,b] \times V \to \mathbb{R}$ . Assume that  $\left|\frac{\partial \phi}{\partial x}(x,y)\right| \ge 1$  for all  $x \in [a,b]$  and  $y \in V$ . Then

$$\left|\int_{a}^{b} v(x,y) e^{i\phi(x,y)/\epsilon} dx\right| \le d\epsilon$$

where d depends on v and  $\phi$  and is independent of  $y \in V$  and  $\epsilon$ .

#### 4.7.2 Two useful lemmas

We recall some classical results that are derived from  $[3, \S 9]$ 

**Lemma 4.7.10.** Let  $n \in \mathbb{N}$  and let  $\mathbb{R}^n \times \mathbb{R} \ni (x, y) \mapsto F(x, y) \in \mathbb{R}$  be a smooth function vanishing on the graph  $y = \eta(x)$ , where  $\eta : \mathbb{R}^n \to \mathbb{R}$  is a smooth function. Then for every point  $x_0 \in \mathbb{R}^n$  there exist a neighborhood W of  $(x_0, \eta(x_0))$  and a smooth function  $\phi : W \to \mathbb{R}$  such that

$$\forall (x,y) \in W, \qquad F(x,y) = (y - \eta(x))\phi(x,y).$$

**Lemma 4.7.11.** Let  $n \in \mathbb{N}$  and let  $F: \frac{\mathbb{R}^n \times \mathbb{R} \to \mathbb{R}}{(x, y) \mapsto F(x, y)}$  be a smooth function such that  $\frac{\partial F}{\partial y}$  is vanishing on the smooth hypersurface  $y = \eta(x)$ . Then for every point  $x_0 \in \mathbb{R}^n$  there exist a neighborhood W of  $(x_0, \eta(x_0))$  that can be written as  $W = W_1 \times W_2$  where  $W_1$  is an open subset of  $\mathbb{R}^n$  and  $W_2$  is an open subset of  $\mathbb{R}$ , and smooth functions  $\phi: W \to \mathbb{R}$  and  $f_0: W_1 \to \mathbb{R}$  such that

$$\forall (x, y) \in U, F(x, y) = (y - \eta(x))^2 \phi(x, y) + f_0(x).$$

# Chapter 5

# Classification of the singularities of the non-mixing field

In this chapter, we classify completely the singularities of the non-mixing field for two-level systems and for generic one-parameter families of two-level systems, and we propose a new model for the avoided crossing problem. Then we continue the analysis of the singularities of the non-mixing field started in [22] for general quantum systems and we prove that it has interesting topological properties. In particular it can exhibit both singularities having a half-integer index, which are usually generic for proto line fields on  $\mathbb{R}^2$  (see Section 5.3.1), and singularities having an integer index, usually generic for  $C^{\infty}$ vector fields on  $\mathbb{R}^2$ . It opens new prospects for the construction of a topology on line-fields which allows such properties.

#### Contents

5.1	Gen	eral definition of the non-mixing field $\ldots \ldots \ldots \ldots \ldots 122$
5.2	The non-mixing field for two level systems	
	5.2.1	Classification of the singularities of the non-mixing field of a generic two-level system
	5.2.2	Bifurcations of the non-mixing field for two-level systems and the avoided crossing problem 125
	5.2.3	Parametric families of real Hamiltonians
5.3	The	non-mixing field for general quantum systems $\ldots$ 134
	5.3.1	Useful results about line fields
	5.3.2	Non-mixing field
	5.3.3	Singularities of $\chi_{j-1,j}$ at intersections $(j, j+1)$

# 5.1 General definition of the non-mixing field

Let  $\mathcal{H}$  be a separable Hilbert space, and  $\mathcal{L}_{sa}(\mathcal{H})$  be the set of essentially self-adjoint operators on  $\mathcal{H}$ . Denote the scalar product on  $\mathcal{H}$  by  $\langle, \rangle$ . Let V be an open set of  $\mathbb{R}^k$  and  $\gamma: [0,1] \to V$  be a regular smooth control path. Let  $H(\cdot) \in C_b^2(V, \mathcal{L}_{sa}(\mathcal{H}))$  such that the operators H(u) have a common dense domain  $D \subset \mathcal{H}$  for every  $u \in V$ , and are bounded from below, uniformly with respect to  $u \in V$ . For every  $u \in V$ , denote the spectrum of H(u) by  $\sigma(u)$ . Let  $j \in \mathbb{N}$ . Assume that  $\sigma_*(u) = \{\lambda_{j-1}(u), \lambda_j(u)\} \subset \sigma(u)$  is a locally discrete separated part of  $\sigma$ , for  $u \in V$ , and let  $P_{j-1,j}(u)$  be the spectral projection of H(u)onto the eigenspace associated with  $\sigma_*(u)$ . Let  $\gamma = (\gamma(\tau))_{\tau \in [0,1]}$  a smooth regular curve of  $\mathbb{R}^2$  along which the eigenvectors  $\phi_{j-1}$  and  $\phi_j$  associated with  $\lambda_{j-1}$  and  $\lambda_j$  are smooth. For every  $\tau \in [0, 1]$ , consider a unitary mapping  $\mathcal{U}(\tau)$  from  $\mathbb{C}^2$  to  $\mathrm{Im} P_{j-1,j}(\gamma(\tau))$ , which is  $C^2$  with respect to  $\tau \in [0, 1]$ , such that  $\mathcal{U}(\tau)(e_1) = \phi_{j-1}(\gamma(\tau))$  and  $\mathcal{U}(\tau)(e_2) = \phi_j(\gamma(\tau))$ , where  $(e_1, e_2)$  is the canonical basis of  $\mathbb{C}^2$ . Then the Effective Hamiltonian associated with H and the transformation  $\mathcal{U}$  reads

$$H_{\rm eff}(\gamma(\tau)) = \begin{pmatrix} \lambda_{j-1}(\gamma(\tau)) & 0\\ 0 & \lambda_j(\gamma(\tau)) \end{pmatrix} - i\epsilon \begin{pmatrix} 0 & \langle \dot{\phi}_{j-1}(\gamma(\tau)), \phi_j(\gamma(\tau)) \rangle\\ \langle \dot{\phi}_j(\gamma(\tau)), \phi_{j-1}(\gamma(\tau)) \rangle & 0 \end{pmatrix}$$
(5.1)

where  $\dot{\phi}_q(\gamma(\tau)), q \in \{j-1, j\}$  is the derivative of  $\phi_q$  along the path  $\gamma$ . Denoting the propagator of Equation

$$i\epsilon \frac{d\psi(\tau)}{d\tau} = H(\gamma(\tau))\psi(\tau)$$

by  $U^{\epsilon}(\tau)$  and the propagator of Equation

$$i\epsilon \frac{d\psi(\tau)}{d\tau} = H_{\text{eff}}(\gamma(\tau))\psi(\tau)$$

by  $U_{\text{eff}}^{\epsilon}(\tau)$ , Theorem 2.1.1 yields

$$\left|\left(U^{\epsilon}(\tau) - \mathcal{U}(\gamma(\tau))U^{\epsilon}_{\text{eff}}(\tau)\mathcal{U}^{-1}(\gamma(0))\right)P_{j-1,j}(0)\right|\right| \le C\epsilon,\tag{5.2}$$

for every  $\tau \in [0, 1]$ .

For a general Hamiltonian H depending on two real controls, the non-mixing curves between  $\lambda_{j-1}$  and  $\lambda_j$  for  $j \geq 2$  have been defined in [22] as the curves  $\gamma = (\gamma(\tau))_{t \in [0,1]}$  of  $\mathbb{R}^2$  along which  $\dot{\phi}_{j-1}(\gamma(\tau))$  is orthogonal to  $\phi_j(\gamma(\tau))$  for the scalar product  $\langle,\rangle$ , for every  $\tau \in [0,1]$ . Let  $\psi_0 = \phi_j(\gamma(0))$  and for every  $\tau \in [0,1]$ ,  $\psi_{\epsilon}(\tau) = U^{\epsilon}(\tau)\psi_0$ . By the inequality (5.2), when  $\gamma$  is a non-mixing curve, we have,  $\|\psi_{\epsilon}(\tau) - e^{i\theta}\phi_j(\gamma(\tau))\| \leq C\epsilon$ , for  $\tau \in [0,1]$ , where  $\theta$  is possibly depending on  $\epsilon$ .

The precision of the adiabatic approximation along such a curve is improved. Qualitatively speaking, the error made in the adiabatic regime along a non-mixing curve does not depend on the gap between  $\lambda_{j-1}$  and  $\lambda_j$  and on the variations of their associated eigenvectors but on the gap between  $\{\lambda_{j-1}(\cdot), \lambda_j(\cdot)\}$  and the rest of the spectrum of  $H(\cdot)$  and the variations of the eigenvectors associated with  $\operatorname{Spectrum}(H(\cdot)) \setminus \{\lambda_{j-1}(\cdot), \lambda_j(\cdot)\}$ . In particular, the error of order  $\sqrt{\epsilon}$  for a control path ending a conical intersection between  $\lambda_{j-1}$  and  $\lambda_j$  is transformed into  $O(\epsilon)$  along a control path which follows a non-mixing curve. This property has been used in [22] for a precise control of the Schrödinger Equation with real Hamiltonians and an extension has been presented in [31] for complex Hamiltonians with three real controls. In this section, we study the singularities of the non-mixing curves for two level systems, then for more general quantum systems.

# 5.2 The non-mixing field for two level systems

Let  $f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  and  $k \in \mathbb{N}$ . Consider a smooth regular control path  $\gamma(t) = (u(t), v(t))_{t \in [0,1]}$  such that there exist  $P \in C^2([0,1], \operatorname{SO}_2(\mathbb{R}))$  and  $\lambda \in C^k([0,1], \mathbb{R})$  such that  $\{\lambda(t), -\lambda(t)\}$  is the spectrum of  $H_f(u, v) = \begin{pmatrix} f_1(u, v) & f_2(u, v) \\ f_2(u, v) & -f_1(u, v) \end{pmatrix}$ , and the columns of P form a basis of eigenvectors of  $H_f(\gamma(t))$  for every  $t \in [0,1]$ . We can write, for every  $t \in [0,1]$ ,  $P(t) = \begin{pmatrix} \cos(\theta(t)) & -\sin(\theta(t)) \\ \sin(\theta(t)) & \cos(\theta(t)) \end{pmatrix}$  where  $\theta \in C^2([0,1], \mathbb{R})$ .

Let us study the dynamics of

$$i\frac{d\psi_{\epsilon}(t)}{dt} = H_f(u(\epsilon t), v(\epsilon t))\psi_{\epsilon}(t), \quad \psi_{\epsilon}(0) = \tilde{\psi}_0, \tag{5.3}$$

where  $t \in [0, \frac{1}{\epsilon}]$  and  $\tilde{\psi}_0$  is independent of  $\epsilon$ .

Defining  $Y_{\epsilon}(\tau) = P(\tau)\psi_{\epsilon}(\frac{\tau}{\epsilon})$  for every  $\tau \in [0, 1]$ , we have

$$i\frac{dY_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon} \begin{pmatrix} \lambda(\tau) & 0\\ 0 & -\lambda(\tau) \end{pmatrix} + \begin{pmatrix} 0 & i\dot{\theta}(\tau)\\ -i\dot{\theta}(\tau) & 0 \end{pmatrix} \right) Y_{\epsilon}(\tau).$$
(5.4)

Using the previous notations, we have that  $\gamma$  is a non-mixing curve if it satisfies  $\dot{\theta}(t) = 0$ , for every  $t \in [0, 1]$ . Along a non-mixing curve  $\gamma$ , the error occuring in the adiabatic theorem (see Theorem 4.4.1) is equal to 0, that is, the non-mixing curves can be followed at an arbitrary speed.

By a direct computation of the eigenvectors of  $H_f$ , we show the following

**Proposition 5.2.1.**  $\gamma$  is a non-mixing curve of  $H_f$  if and only if there exists  $(c_1, c_2) \in \mathbb{R}^2 \setminus \{(0, 0)\}$  such that

$$c_1 f_1(\gamma(t)) + c_2 f_2(\gamma(t)) = 0,$$
 (\*)

for every  $t \in [0, 1]$ .

**Proposition 5.2.2.** The non-mixing curves of  $H_f$  are the integral curves of the  $C^{\infty}$  vector field  $\chi(f) = \begin{pmatrix} f_1 \partial_2 f_2 - f_2 \partial_2 f_1 \\ f_2 \partial_1 f_1 - f_1 \partial_1 f_2 \end{pmatrix}$ , called the non-mixing field.

Proof. Let  $(\gamma(t))_{t\in[0,1]}$  be a regular smooth curve of  $\mathbb{R}^2$  and let  $t_0 \in [0,1]$ . Assume that  $f_2(\gamma(t)) \neq 0$ , for every t in a neighborhood V of  $t_0$ . Then  $\gamma$  satisfies Equation (\*) for every  $t \in V$  if and only if  $\frac{f_1(\gamma(\cdot))}{f_2(\gamma(\cdot))}$  is constant on V. Provided that  $\frac{f_1}{f_2}$  is  $C^{\infty}$  in a neighborhood of  $\gamma(t_0)$ , we deduce that the previous condition is satisfied if and only if, for every  $t \in V$ ,  $\gamma'(t)$  is orthogonal to  $\nabla \frac{f_1}{f_2}(\gamma(t))$ . Hence,  $\gamma$  is an integral curve of  $\chi(f)$ , up to a time reparametrization. The same holds if  $f_1(\gamma(t)) \neq 0$ , for every t in a neighborhood of  $t_0$ .

# 5.2.1 Classification of the singularities of the non-mixing field of a generic two-level system

The zeros of  $\chi(f)$  are defined by

$$f_1(u,v)\partial_2 f_2(u,v) - f_2(u,v)\partial_2 f_1(u,v) = 0$$
  
$$f_2(u,v)\partial_1 f_1(u,v) - f_1(u,v)\partial_1 f_2(u,v) = 0.$$

First we study zeros of  $\chi(f)$  in the set  $\Sigma = \{(u, v) \in \mathbb{R}^2 \mid f_1(u, v) = f_2(u, v) = 0\}$ . We recall that generically w.r.t.  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  an intersection of eigenvalues is conical.

**Proposition 5.2.3.** Assume that  $H_f$  admits a conical intersection of eigenvalues at (0,0). Then the non-mixing curves of  $H_f$  are locally diffeomorphic to a critical node, that is, to the integral curves of the smooth vector field  $\mathbb{R}^2 \ni (u, v) \mapsto \begin{pmatrix} u \\ v \end{pmatrix}$  (see Figure 5.1).

*Proof.* Under the assumption that (0,0) is a conical intersection for  $H_f$ , the results of Section 4.2.3 ensure that there exists a diffeomorphism  $\phi \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  such that  $(f \circ \phi)(u, v) = (u, v)$  in a neighborhood of (0, 0). The proposition follows.

Then we study zeros of  $\chi(f)$  in the set  $\mathbb{R}^2 \setminus \Sigma$ .

**Proposition 5.2.4.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ , critical points of  $\chi(f)$  belonging to  $\mathbb{R}^2 \setminus \Sigma$  are saddles or centers, that is, the non-mixing curves are locally diffeomorphic to the integral curves of, respectively,  $(u, v) \mapsto \begin{pmatrix} u \\ -v \end{pmatrix}$ , or  $(u, v) \mapsto \begin{pmatrix} -v \\ u \end{pmatrix}$ .

Proof. Assume that (0,0) is a zero of  $\chi(f)$  such that  $f_2(0) \neq 0$ . Then (0,0) is a critical point of  $g = \frac{f_1}{f_2}$ , which is  $C^{\infty}$  in a neigborhood of (0,0). Using the fact that the application  $\mathcal{F} : J^2(\mathbb{R}^2, \mathbb{R}^2) \to J^2(\mathbb{R}^2, \mathbb{R})$  such that  $\mathcal{F}(j^2(f)(0)) = j^2(g)(0)$  is a submersion and that Morse functions are open and dense in  $C^{\infty}(\mathbb{R}^2, \mathbb{R})$  (see [41, Theorem 6.2]), we deduce the result.



Figure 5.1 – Non-mixing curves at a conical intersection.

# 5.2.2 Bifurcations of the non-mixing field for two-level systems and the avoided crossing problem

#### Non-mixing curves for semi-conical intersections

Let  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  having a semi-conical intersection (see Section 4.2.2 for a precise definition) at (0,0). The non-mixing curves defined for a general function  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ having a semi-conical intersection at (0,0) are diffeomorphic to those of the normal form defined in Section 4.2.3 in a neighborhood of (0,0). More precisely, up to an admissible transformation (as defined in Section 4.2.2), it is sufficient to study the non-mixing curves of the normal form, that is, for every  $(u, v) \in \mathbb{R}^2$ ,

$$H(u,v) = \begin{pmatrix} h(u)u & u+v^2\\ u+v^2 & -h(u)u \end{pmatrix},$$

where  $h \in C^{\infty}(\mathbb{R}, \mathbb{R})$  is a smooth function such that h(0) = 1. We introduce the following condition:

$$h'(0) \neq 0. \tag{L}$$

Assume in this section that the condition  $(\mathbf{L})$  holds.

The non-mixing field is defined (see Proposition 5.2.2) as

$$\chi(u,v) = \begin{pmatrix} 2vuh(u) \\ u^2h'(u) + v^2uh'(u) + v^2h(u) \end{pmatrix}.$$

From Proposition 5.2.1, for  $f \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ ,  $\gamma$  is a non-mixing curve of  $H_f$  if and only if there exists  $(c_1, c_2) \in \mathbb{R}^2 \setminus \{(0, 0)\}$  such that

$$0 = c_1 f_1(\gamma(t)) + c_2 f_2(\gamma(t)) = c_1 u(t) h(u(t)) + c_2 (u(t) + v(t)^2), \qquad (*)$$
for every  $t \in [0, 1]$ .

Hence we can define  $\forall (u, v) \in U, F_c(u, v) = uh(u) - cu - cv^2$ . The set of the non mixing curves is the union of the curves along which  $F_c$  is equal to zero for some  $c \in \mathbb{R}$  with the parabola of equation  $u + v^2 = 0$ .

**Proposition 5.2.5.** For  $c \neq 1$ ,  $F_c(u, v) = 0 \iff u = \phi_c(v)$ , where  $\phi_c \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $\phi'_c(0) = 0$  and  $\phi''_c(0) = \frac{2c}{1-c}$ .

Proof. Assume  $c \neq 1$ . By direct computations, we have  $\frac{\partial F_c}{\partial u} = h(u) + uh'(u) - c$ . Evaluating at u = 0 and using the condition h(0) = 1, we get  $\frac{\partial F_c}{\partial u}\Big|_{(0,0)} = 1 - c \neq 0$ . The Implicit Function Theorem ensures the existence of  $\phi_c \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfying  $\phi_c(0) = 0$  such that  $F_c(u, v) = 0 \iff u = \phi_c(v)$ , for every (u, v) in a neighborhood of (0, 0). Differentiating twice the relation  $F_c(\phi_c(v), v) = 0$  w.r.t. v, we get  $\phi'_c(0) = 0$  and  $\phi''_c(0) = \frac{2c}{1-c}$ .

**Remark 5.2.6.** Notice that for every  $\alpha \in \mathbb{R}$ , there exists a non-mixing curve that can be parametrized as  $u = \phi_{\alpha}(v)$  with a smooth function  $\phi_{\alpha}$  such that  $\phi_{\alpha}(0) = \phi'_{\alpha}(0) = 0$ and  $\phi''_{\alpha}(0) = \alpha$ . This property can be used to get precise superpositions of states by using control paths passing at the singularity in the non-conical direction whose second order derivatives change (see [22] for a similar first order technique at conical intersections), such developments will be exposed in future works.

**Theorem 5.2.7.** The set  $\{(u,v) \in U \mid F_1(u,v) = 0\}$  is diffeomorphic to  $\{(0,0)\}$  or  $\{(u,v) \in U, |u| = |v|\}.$ 

*Proof.* Consider the function g defined by g(u) = h(u) - 1. Under assumption (**L**), we have g(0) = 0 and  $g'(0) = h'(0) \neq 0$ . Then we get the equivalence

$$F_1(u,v) = 0 \iff v^2 = ug(u),$$

for every  $(u, v) \in \mathbb{R}^2$ . If g'(0) < 0, the previous equation has no solution but u = v = 0. If g'(0) > 0, we can write  $g(u) = u(g'(0) + u\tilde{g}(u))$ , where  $\tilde{g} \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $\tilde{g}(0) \neq 0$ . Then we have  $F_1(u, v) = 0 \iff v^2 = u^2 \tilde{g}(u) = u^2(g'(0) + ug_1(u))$ . Applying the the right transformation  $(u, v) \mapsto \begin{pmatrix} u\sqrt{g'(0) + ug_1(u)} \\ v \end{pmatrix}$ , we have in new coordinates  $F_1(u, v) = 0 \iff v^2 = u^2$ .

We have shown that the non-mixing curves can have two different behaviours depending on the sign of h'(0):

— The case where h'(0) < 0 that we refer as the *Elliptic semi-conical singularity* (see Figure 5.2). 0 is an index 2 singularity for the vector field  $\chi$ . Every non-mixing curve passes through 0 and is tangent to the non-conical direction at 0.



Figure 5.2 – An example of *el*liptic semi-conical singularity with h(u) = 1 - u for every u. The non-conical direction is (0, 1). Every non-mixing curve passes through the singularity in the non-conical direction. The index is 2.



Figure 5.3 – A example of hyperbolic semi-conical singularity with h(u) = 1 + u for every u. The non-conical direction is (0, 1). The red curves are the only non-mixing curves passing through the singularity in conical directions. The index is 0.

— The case where h'(0) > 0 that we refer as the Hyperbolic semi-conical singularity (see Figure 5.3). 0 is an index zero singularity for the vector field  $\chi$ . Every nonmixing curve passing at the origin is tangent to the non-conical direction except the level c = 1 which passes through the origin in a conical direction (see the red curves on Figure 5.3). There exist also some non-mixing curves which do not pass through the origin. Notice that even the index of the singularity is equal to zero, the non-mixing curves are not homeomorphic to the integral curves of a non-singular vector field at 0.

We can achieve an homogeneous blow-up in the polar coordinates  $(r, \theta) \in \mathbb{R} \times \mathbb{S}^1$ . Then the origin is blown up to  $\mathbb{S}^1$ . In polar coordinates, the non-mixing field field  $\chi$  is transformed into

$$\tilde{\chi}(r,\theta) = \eta_1(r,\theta)\frac{\partial}{\partial\theta} + \eta_2(r,\theta)r\frac{\partial}{\partial r}$$

where

$$\eta_1(r,\theta) = \cos^3(\theta)h'(r\cos(\theta)) - \cos(\theta)\sin^2(\theta)h(r\cos(\theta)) + r\cos^2(\theta)\sin^2(\theta)h'(r\cos(\theta)),$$

and

$$\eta_2(r,\theta) = \cos(\theta)\sin(2\theta)h(r\cos(\theta)) + \sin(\theta)\cos^2(\theta)h(r\cos(\theta)) + \sin^3(\theta)h(r\cos(\theta)) + r\sin^3(\theta)\cos(\theta)h'(r\cos(\theta)).$$



Figure 5.4 – Elliptic semi-conical singularity in the coordinates  $(r, \theta) \in \mathbb{R} \times \mathbb{S}^1$  after desingularization with h(u) = 1 - u for every u.

By direct computations, we prove that the singularities of  $\tilde{\chi}$  on  $\{0\} \times \mathbb{S}^1$  are hyperbolic. More precisely, we can show that an elliptic semi-conical singularity can be desingularized into two nodes (one attractive and one repulsive) for a value of the angle  $\theta = \pm \frac{\pi}{2}$  (see Figure 5.4). By classical results on hyperbolic singularities of vector fields (see [36]), we deduce that the non-mixing curves are in this case homeomorphic to those obtained with h(u) = 1 - u for every u in the original coordinates  $(u, v) \in \mathbb{R}^2$ . On the other hand, an hyperbolic semi-conical singularity can be desingularized into two nodes for a value of the angle  $\theta = \pm \frac{\pi}{2}$  and four saddles for the angles  $\theta$  such that  $(\tan(\theta))^2 = h'(0)$  (see Figure 5.5). We can deduce that the non-mixing curves are in this case homeomorphic to those obtained with h(u) = 1 + u for every u, in the original coordinates  $(u, v) \in \mathbb{R}^2$ .

#### 5.2.3 Parametric families of real Hamiltonians

#### Avoided crossing model

Consider  $(H_z(u, v))_{z \in \mathbb{R}}$  a family of real symmetric matrices such that  $(u, v, z) \mapsto H_z(u, v)$  is  $C^{\infty}$ , such that, for z > 0,  $H_z$  has simple eigenvalues and  $H_0$  has an isolated double eigenvalue at  $(u, v) = (\bar{u}, \bar{v})$ , that is, the family  $(H_z(\cdot))_{z \in \mathbb{R}}$  has an avoided crossing at (0, 0) (see [49, 48] for results about the use of avoided crossings in semi-classical analysis). It results from the structural stability of conical intersections for a real Hamiltonian with two controls (see [22]) that  $(\bar{u}, \bar{v})$  is a non-conical intersection for  $H_0(\bar{u}, \bar{v})$ .



Figure 5.5 – Hyperbolic semi-conical singularity in the coordinates  $(r, \theta) \in \mathbb{R} \times \mathbb{S}^1$  after desingularization with h(u) = 1 + u for every u.

In Section 4.2.4, we have proved that the least degenerate local two level model is the model of F-semi-conical intersections, that is, for every  $(u, v, z) \in \mathbb{R}^3$  in a neighborhood of 0,

$$\tilde{H}_{z}(u,v) = \begin{pmatrix} h_{1}(u,v,z)(z-m(u)u) & h_{2}(u,v,z)(z+u+v^{2}) \\ h_{2}(u,v,z)(z+u+v^{2}) & -h_{1}(u,v,z)(z-m(u)u) \end{pmatrix},$$

where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfy  $h_1(0) = h_2(0) = 1$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $m(0) \notin \{-1, 0\}$ , where m(0) > -1, the case m(0) < -1 being similar (see Section 4.5.2).

This model has not been classified yet (see [48] for a classification of the normal forms occuring in avoided crossings) because  $\tilde{H}_0$  is not generic as a Hamiltonian depending on the two parameters  $(u, v) \in \mathbb{R}^2$ . However,  $(u, v, z) \mapsto \tilde{H}_z(u, v)$  is generic as a Hamiltonian depending on  $(u, v, z) \in \mathbb{R}^3$ .

- For z < 0, the energy levels have two conical intersections, as illustrated on Figure 5.6.
- For  $z \to 0$ , these two singular points join in the non-conical direction, their common point is a semi-conical intersection, as illustrated on Figure 5.7.
- For z > 0, the energy levels are then separated, as illustrated on Figure 5.8



Figure 5.6 – Energy levels of  $\tilde{H}_z(u,v)$  with z < 0, as a function of (u,v).

Figure 5.7 – Energy levels of  $\tilde{H}_0(u, v)$ , as a function of (u, v).



Figure 5.8 – Energy levels of  $\tilde{H}_z(u,v)$  with z > 0, as a function of (u,v).

#### Bifurcations of the non-mixing field for two-level systems

Let 
$$f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$$
 and for every  $(u, v, z) \in \mathbb{R}^3$ ,  
 $H_f(u, v, z) = \begin{pmatrix} f_1(u, v, z) & f_2(u, v, z) \\ f_2(u, v, z) & -f_1(u, v, z) \end{pmatrix}$ .

For every z in  $\mathbb{R}$ , let  $\chi_z(f)$  be the non-mixing field associated with  $f(\cdot, \cdot, z) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ . According to Chapter 4, there are two typical cases for f at intersections of eigenvalues.

**FIRST CASE:** If f has a F-conical intersection at (0,0,0), then  $H_f$  is equivalent in the sense defined in Section 4.2.2 to  $\begin{pmatrix} h_1(u,v,z)(z-u) & h_2(u,v,z)(z-v) \\ h_2(u,v,z)(z-v) & -h_1(u,v,z)(z-u) \end{pmatrix}$  for every (u,v,z) in a neighborhood of (0,0,0), where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfy  $h_1(0) = h_2(0) =$ 1. By Section 5.2, for every z the non-mixing field  $\chi_z(f)$  has a critical node singularity at the conical intersection (u,v) = (z,z).

**SECOND CASE:** If f has a F-semi-conical intersection at (0,0,0), then  $H_f$  is equivalent to  $\begin{pmatrix} h_1(u,v,z)(z-m(u)u) & h_2(u,v,z)(z+u+v^2) \\ h_2(u,v,z)(z+u+v^2) & -h_1(u,v,z)(z-m(u)u) \end{pmatrix}$  for every (u,v,z) in a neighborhood of (0,0,0), where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfy  $h_1(0) = h_2(0) = 1$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $m(0) \notin \{-1,0\}$ . Using the results of Section 4.2.3, up to reducing the open and dense subset G of  $C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  defined in Section 4.2.1, we can assume that generically w.r.t.  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ ,  $H_f(\cdot, \cdot, 0)$  is locally equivalent to

$$H(u,v) = \begin{pmatrix} uh(u) & u+v^2 \\ u+v^2 & -uh(u) \end{pmatrix}$$
(5.5)

where  $h : \mathbb{R} \to \mathbb{R}$  is a smooth function such that h(0) = 1 and  $h'(0) \neq 0$ . The case where h'(0) > 0 is referred as the *Hyperbolic bifurcation*, and the case where h'(0) < 0 is referred as the *Elliptic bifurcation*. We can show that the condition  $h'(0) \neq 0$  is satisfied when f is such that the normal form algorithm in the parametric case (see Section 4.2.4) provides three functions  $(h_1, h_2, m)$  such that  $-4\partial_1 \tilde{h}(0, 0, 0) \tilde{h}(0, 0, 0) \neq \partial_2 \tilde{h}(0, 0, 0)^2$ , where for every (u, v, z) in a neighborhood of 0 in  $\mathbb{R}^3$ ,  $\tilde{h}(u, v, z) = -\frac{h_1(u, v, z)m(u)}{h_2(u, v, z)}$ .

Under these assumptions, by Section 5.2, the non-mixing field  $\chi_0(f)$  has an elliptic or hyperbolic semi-conical singularity at (0,0). Assume m(0) > -1 (respectively m(0) < -1). Then for z > 0 (respectively, z < 0),  $f(\cdot, \cdot, z)$  has no intersection of eigenvalues. For z < 0 (respectively z > 0),  $f(\cdot, \cdot, z)$  has two conical intersections of eigenvalues, and by Section 5.2,  $\chi_z(f)$  has critical nodes at these points.

We present the next lemma, which is classical and can be deduced from  $[60, \S4.3]$ .

**Lemma 5.2.8.** For  $g \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$ , we define, for every  $(u, v) \in \mathbb{R}^2$ ,  $g_z(u, v) = g(u, v, z)$ . Generically with respect to  $g \in C^{\infty}(\mathbb{R}^3, \mathbb{R})$ , for every  $z \in \mathbb{R}$ , the critical points of  $g_z$  are such that the level-lines of  $g_z$  are locally diffeomorphic to a saddle, a center as in Proposition 5.2.4, or a cusp, that is the level lines of  $(u, v) \mapsto v^3 - u^2$ .



Figure 5.9 – Saddle-center bifurcation for  $z < \overline{z}$ ,  $\overline{z}$  being the value of z for which the bifurcation occurs.



Figure 5.10 – Saddle-center bifurcation for  $z = \overline{z}$ .

More precisely, the cusp singularity is obtained by a saddle-center bifurcation, that is  $g_z(u,v) = v^3 \pm (z-\bar{z})v - u^2$ , the bifurcation occurs for a value of the parameter  $z = \bar{z}$  (see [60, §4.3]), and it is illustrated on the Figures 5.9 and 5.10.

Now we can classify the bifurcations of singularities occuring in Theorem 2.2.7 when the parameter z varies.

**Theorem 5.2.9.** Generically with respect to  $f \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$ , for every  $z \in \mathbb{R}$ ,  $\chi_z(f)$  has the following singularities, up to a  $C^{\infty}$ -diffeomorphic coordinate change:

- Critical nodes at conical intersections of eigenvalues,
- Hyperbolic or elliptic semi-conical singularities at semi-conical intersections of eigenvalues,
- Saddles, centers or cusps at points that are not intersections of eigenvalues.

**Remark 5.2.10.** Semi-conical singularities and cusps are singularities of maximal codimension. Hence, generically, we cannot find them simultaneously for the same value of the parameter z.

**Description of the bifurcation of the non-mixing field for** *F***-semi-conical intersections** We describe the non-mixing curves for the normal form

$$\tilde{H}_{z}(u,v) = \begin{pmatrix} h_{1}(u,v,z)(z-m(u)u) & h_{2}(u,v,z)(z+u+v^{2}) \\ h_{2}(u,v,z)(z+u+v^{2}) & -h_{1}(u,v,z)(z-m(u)u) \end{pmatrix}$$

for every (u, v, z) in a neighborhood of (0, 0, 0), where  $h_1, h_2 \in C^{\infty}(\mathbb{R}^3, \mathbb{R}^2)$  satisfy  $h_1(0) = h_2(0) = 1$  and  $m \in C^{\infty}(\mathbb{R}, \mathbb{R})$  satisfies  $m(0) \notin \{-1, 0\}, m(0) > -1$ , in the generic case where  $-4\partial_1 \tilde{h}(0, 0, 0)\tilde{h}(0, 0, 0) \neq \partial_2 \tilde{h}(0, 0, 0)^2$ , where for every (u, v, z) in a neighborhood of 0 in  $\mathbb{R}^3$ ,  $\tilde{h}(u, v, z) = -\frac{h_1(u, v, z)m(u)}{h_2(u, v, z)}$ .



Figure 5.11 – Hyperbolic nonmixing curves for z < 0.



Figure 5.12 – Elliptic non-mixing curves for z < 0.

For z < 0 (see Figures 5.11 and 5.12), the non-mixing curves are the integral curves of a smooth vector field vanishing at each conical intersections which has an index equal to 1 at conical intersections. In the hyperbolic case, the two critical nodes (of index +1) are combined with two saddles (of index -1). By continuity with respect to the parameter z, it is coherent with the fact that the index of the hyperbolic semi-conical singularity vanishes. In the elliptic case, the two critical nodes join in the non-conical direction and there is no other singularity in a neighborhood of 0.

For z = 0 (see Figures 5.2 and 5.3 the singularity that we obtain has an index either equal to 2 in the elliptic case or 0 in the hyperbolic case.

For z > 0 (see Figures 5.13 and 5.14), the energy levels are separated. The non-mixing curves are the integral curves of a smooth vector field which is non-singular at 0. For  $z \to 0$ , the direction of the non-mixing curves at 0 converges to the non-conical direction for  $H_0$ . In the elliptic case, however, there are two center singularities in a neighborhood of the origin. By a continuity argument w.r.t. z, this is coherent with the fact that the index of the elliptic semi-conical singularity is equal to 2.

**Remark 5.2.11.** For a very small z > 0, one could be interested in ensuring that the system remains on the lower energy level. Following the non-mixing curves reduces the possible loss to the higher energy level, which are normally inverse proportionnal to the gap between the two levels. For two-level systems, this loss is reduced to 0.



Figure 5.13 – Hyperbolic nonmixing curves for z > 0.



Figure 5.14 – Elliptic non-mixing curves for z > 0.

#### 5.3 The non-mixing field for general quantum systems

#### 5.3.1 Useful results about line fields

We start this section by stating some results on line fields. We refer to [21] for a precise definition and classification of the singularities of line fields on two-dimensional Riemannian manifolds. Let (M, g) be a 2-dimensional Riemannian manifold. A *line field* on M is a section of  $PT(M \setminus K)$ , where K is a closed subset of M.

The next propositions (until Theorem 5.3.6) are directly taken from [21].

**Definition 5.3.1.** Let (M,g) be a 2-dimensional Riemannian manifold. A proto-linefield is a pair (X,Y) of vector fields on M. Denote by  $z_X$  and  $z_Y$  the sets of zeros of X and Y. The line field associated with (X,Y), denoted by B(X,Y), is the section of  $PT(M \setminus (z_X \cup z_Y))$  defined at a point  $p \in M \setminus (z_X \cup z_Y)$  as the line B(X(p), Y(p)) of  $T_pM$ bisecting (X(p), Y(p)) for the metric g(p).

**Proposition 5.3.2.** Let (M, g) be a 2-dimensional Riemannian manifold, K be a closed subset of M and L be a section of  $PT(M \setminus K)$ . There exist two vector fields X and Y such that L = B(X, Y).

With this definition, the Whitney topology on pairs of vector fields on M defines a topology on line fields on M, the zeros of X and Y become singularities of the associated line-field.

**Definition 5.3.3.** A one-dimensional connected immersed submanifold N of  $M \setminus (z_X \cup z_Y)$ is said to be an integral manifold of the proto-line-field (X, Y) if for any point p of N, the tangent line to N at p is given by B(X, Y). We say that a proto-line-field L has a *Darbouxian singularity* if its integral manifolds are locally homeomorphic to those of the proto-line-fields defined by  $(X_L, Y_L)$ ,  $(X_M, Y_M)$ and  $(X_S, Y_S)$  defined as follows:

— The Lemon proto-line-field (see Figure 5.15) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$  defined by

$$X_L(x,y) = \begin{pmatrix} x+y\\ y-x \end{pmatrix}, \quad Y_L(x,y) = \begin{pmatrix} 1\\ 1 \end{pmatrix}.$$

— The Monstar proto-line-field (see Figure 5.16) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$  defined by

$$X_M(x,y) = \begin{pmatrix} x \\ 3y \end{pmatrix}, \quad Y_M(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}.$$

— The Star proto-line-field (see Figure 5.17) is the pair of vector fields on  $(\mathbb{R}^2, Eucl)$  defined by

$$X_S(x,y) = \begin{pmatrix} x \\ -y \end{pmatrix}, \quad Y_S(x,y) = \begin{pmatrix} 1 \\ 0 \end{pmatrix}.$$

Darbouxian singularities have an index equal to  $\pm \frac{1}{2}$ .



Figure 5.15 – The Lemon singularity, of index  $\frac{1}{2}$ .



Figure 5.16 – The Monstar singularity, of index  $\frac{1}{2}$ .



Figure 5.17 – The Star singularity, of index  $-\frac{1}{2}$ .

**Definition 5.3.4.** Let (X, Y) be a proto-line field on (M, g), and (X', Y') be a protoline-field on (M', g'). Fix  $p \in M$  and  $p' \in M'$ . Then (X, Y) and (X', Y') are said to be topologically equivalent at p and p' if there exist two neighborhoods  $V_p$  and  $W_{p'}$  of p and p'respectively and a homeomorphism  $h: V_p \to W_{p'}$ , with h(p) = p', which takes the integral manifolds of (X, Y) onto those of (X', Y').

**Definition 5.3.5.** We say that a proto-line-field (X, Y) has a non-degenerate singularity (respectively, a hyperbolic singularity) at a point  $p \in M$  if one of the two vector fields has a non-degenerate singularity (respectively, a hyperbolic singularity) and the other is non-vanishing at p.

**Theorem 5.3.6** (Hyperbolic singularities of line fields). Let (M, g) be a two-dimensional Riemannian manifold. Hyperbolic singularities of proto line fields on M are Darbouxian.

Now we prove two technical results that will be useful for the proof of Theorem 5.3.12.

**Proposition 5.3.7** (Non-degenerate singularities of line fields). Let (M, g) be a twodimensional Riemannian manifold. Non-degenerate singularities of proto-line-fields on M are Darbouxian. Moreover, consider a proto line field (X, Y) on M such that  $\bar{p}$  is a non-degenerate singularity of (X, Y). Then there exists two sequences  $(X_n)_{n \in \mathbb{N}}$  and  $(Y_n)_{n \in \mathbb{N}}$  of smooth vector fields on M converging respectively to X and Y for the  $C^{\infty}$ -Whitney topology, such that for every  $n \in \mathbb{N}$ ,  $\bar{p}$  is an hyperbolic singularity of  $(X_n, Y_n)$ and  $B_g(X(p), Y(p)) = B_g(X_n(p), Y_n(p))$  for p in a punctured neighborhood of  $\bar{p}$ .

*Proof.* Consider a proto line field (X, Y) such that X has a non-degenerate zero at  $\bar{p} \in M$  and  $Y(\bar{p}) \neq 0$ . If  $\bar{p}$  is hyperbolic, then Theorem 5.3.6 proves the result.

Assume now that X is non-degenerate and not hyperbolic at  $\bar{p}$ . Define  $\tilde{X}$  and  $\tilde{Y}$ as the smooth vector fields which are the image of respectively, X and Y by the rotation of angle  $\frac{\pi}{2}$  for the metric g. This construction is always possible, at least locally around  $\bar{p}$ . By using local charts, one can easily show that  $\tilde{X}$  is hyperbolic at  $\bar{p}$ . Define a neighborhood V of  $\bar{p}$  such that  $\tilde{X}$  and  $\tilde{Y}$  are well defined on V and  $\bar{p}$  is the only zero of X in V, Y is non-zero on V, and set  $W = V \setminus \{\bar{p}\}$ . We can notice that for every  $p \in W$ ,  $\frac{\tilde{X}(p)}{||X(p)||} - \frac{\tilde{Y}(p)}{||Y(p)||}$  is collinear to  $B_g(X(p), Y(p))$ , hence we can deduce the equality  $B_q(X(p), Y(p)) = B_q(X(p), -Y(p))$ . By construction,  $\bar{p}$  is hyperbolic for  $(\tilde{X}, \tilde{Y})$ , hence Theorem 5.3.6 proves that  $\bar{p}$  is Darbouxian. The first claim is proved. In order to prove the second claim, define, for p in W and  $\epsilon > 0$ ,  $X_{\epsilon}(p) = X(p) + \epsilon \tilde{X}(p)$ and  $Y_{\epsilon}(p) = Y(p) - \epsilon \tilde{Y}(p)$ . By construction of  $\tilde{X}$  and  $\tilde{Y}$ , we have, for every p in W,  $\begin{aligned} ||X_{\epsilon}(p)|| &= \sqrt{1 + \epsilon^2} ||X(p)|| \text{ and } ||Y_{\epsilon}(p)|| &= \sqrt{1 + \epsilon^2} ||Y(p)||. \text{ It follows that, for } p \text{ in } W, \\ \frac{X_{\epsilon}(p)}{||X_{\epsilon}(p)||} &+ \frac{Y_{\epsilon}(p)}{||Y_{\epsilon}(p)||} &= \frac{1}{\sqrt{1 + \epsilon^2}} \left( \left( \frac{X(p)}{||X(p)||} + \frac{Y(p)}{||Y(p)||} \right) + \epsilon \left( \frac{\tilde{X}(p)}{||X(p)||} - \frac{\tilde{Y}(p)}{||Y(p)||} \right) \right). \text{ Noticing that for } p \text{ in } W, \\ \frac{\tilde{X}(p)}{||X(p)||} &- \frac{\tilde{Y}(p)}{||Y(p)||} \text{ and } \frac{X(p)}{||X(p)||} + \frac{Y(p)}{||Y(p)||} \text{ are collinear to } B_g(X(p), Y(p)), \text{ we can deduce that } \\ \frac{X_{\epsilon}(p)}{||X_{\epsilon}(p)||} + \frac{Y_{\epsilon}(p)}{||Y_{\epsilon}(p)||} \text{ is collinear to } B_g(X(p), Y(p)). \text{ Hence } B_g(X(p), Y(p)) = B_g(X_{\epsilon}(p), Y_{\epsilon}(p)), \end{aligned}$ for every  $p \in W$ . The hyperbolicity of X at  $\bar{p}$  implies that  $X_{\epsilon}$  is hyperbolic at  $\bar{p}$ . Moreover, for  $\epsilon > 0$ , we have  $Y_{\epsilon}(\bar{p}) \neq 0$ . For such  $\epsilon, \bar{p}$  is a hyperbolic singularity for the proto-linefield  $(X_{\epsilon}, Y_{\epsilon})$ . Setting for  $n \geq 1$ ,  $\epsilon_n = \frac{1}{n}$  and  $X_n = X_{\epsilon_n}, Y_n = Y_{\epsilon_n}$ , we have clearly that  $(X_n)_n$  and  $(Y_n)_n$  converge respectively to X and Y for the  $C^\infty$ -Whitney topology. The result is proved.

**Theorem 5.3.8.** Consider a two dimensional Riemannian manifold (M, g). Let L be a line field on M and A a smooth section of  $GL(TM) = \bigcup_{p \in M} GL(T_pM)$ . Denote by  $K \subset M$  the closed set of singular points of L. Let  $\tilde{L}$  be the line field defined on M by  $\tilde{L}(p) = A(p)(L(p))$  (where the equality is defined as the equality of the associated directions in  $T_pM$ ) for every  $p \in M \setminus K$ . Let  $\bar{p} \in K$ . Consider two vector fields X and Y on M such that, for every  $p \in M \setminus K$ ,  $L(p) = B_g(X(p), Y(p))$ , and satisfying the conditions  $X(\bar{p}) = 0$ ,  $DX(\bar{p})$  is non-degenerate, and  $Y(\bar{p}) \neq 0$ . Then there exists a metric  $\tilde{g}$  on M such that, for every  $p \in M \setminus K$ ,  $\tilde{L}(p) = B_{\tilde{g}}(A(p)X(p), A(p)Y(p))$ . Moreover p is non-degenerate for (AX, AY) if and only if p is non-degenerate for (X, Y).

Proof. Define the metric  $\tilde{g}$  by  $\tilde{g}(p)(x,y) = g(p)(A^{-1}(p)x, A^{-1}(p)y)$  for every  $p \in M$ and  $(x,y) \in T_pM \times T_pM$ . Then we get easily that, for every  $p \in M \setminus K$ ,  $\tilde{L}(p) = B_{\tilde{g}}(A(p)X(p), A(p)Y(p))$ . Let  $\bar{p} \in K$ . Using the conditions  $A(\bar{p}) \in GL(T_{\bar{p}}M)$ , we have  $A(\bar{p})Y(\bar{p}) \neq 0$ ,  $A(\bar{p})X(\bar{p}) = 0$ , and that  $D(AX)(\bar{p})$  is non-degenerate. The theorem is proved.  $\Box$ 

#### 5.3.2 Non-mixing field

Let  $\mathcal{H}$  be a separable Hilbert space. Let U be a connected open set of  $\mathbb{R}^2$ . Assume that for every  $(u, v) \in U$ ,  $H(u, v) = H_0 + uH_1 + vH_2$  where  $H_0, H_1, H_2$  are essentially self-adjoint operators on  $\mathcal{H}$  having a common dense domain  $\mathcal{D} \subset \mathcal{H}$  such that:

- $H_0$  has a discrete spectrum;
- $-H_1, H_2$  are bounded;
- There exists an orthonormal basis  $(b_j)_j$  of the Hilbert space  $\mathcal{H}$  such that  $\langle b_j, H_0 b_q \rangle$ ,  $\langle b_j, H_1 b_q \rangle$ ,  $\langle b_j, H_2 b_q \rangle$  are real for every j, q.

It has been proven in [22] that the non-mixing curves between two eigenvalues  $\lambda_{j-1}(\cdot)$ and  $\lambda_j(\cdot)$  of  $H(\cdot)$  are the integral curves of a line field on  $\mathbb{R}^2$ , defined up to a sign, for  $(u,v) \in \mathbb{R}^2$ , by  $\chi_{j-1,j}(u,v) = \begin{pmatrix} -\langle H_2\phi_{j-1}(u,v), \phi_j(u,v) \rangle \\ \langle H_1\phi_{j-1}(u,v), \phi_j(u,v) \rangle \end{pmatrix}$  where  $\langle, \rangle$  is the natural scalar product on  $\mathcal{H}$ .

**Remark 5.3.9.** The non-mixing curves of H between the levels  $\lambda_{j-1}(\cdot)$  and  $\lambda_j(\cdot)$  are not those of a zero trace two-level reduced Hamiltonian  $h_{red}(u, v)$ , that can be defined by removing the trace of  $I_{j-1,j}^{-1}(u, v)H(u, v)I_{j-1,j}(u, v)$  as in Section 4.6.3, where  $I_{j-1,j}(u, v)$  is a smooth orthogonal mapping between  $\mathbb{R}^2$  and the real span of  $(\phi_{j-1}(u, v), \phi_j(u, v))$  and can be defined locally for (u, v) in a neighborhood of points where  $\sigma(u, v) = \{\lambda_{j-1}(u, v), \lambda_j(u, v)\}$ is separated from the rest of the spectrum of H(u, v). Indeed, the non-mixing curves of  $h_{red}(u, v)$  depend on the choice of the mapping  $I_{j-1,j}$ , while it is not the case for those of H(u, v).

**Proposition 5.3.10.** — Let  $f = (f_1, f_2) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$ ,  $H_f = \begin{pmatrix} f_1 & f_2 \\ f_2 & -f_1 \end{pmatrix}$ , and gbe the Euclidean metric on  $\mathbb{R}^2$ . Then the real eigendirections of  $h_f$  associated with the eigenvalues  $\lambda_{\pm} = \pm \sqrt{f_1^2 + f_2^2}$  can be written as the line-fields  $L_- = B_g(\begin{pmatrix} -f_2 \\ f_1 \end{pmatrix}, \begin{pmatrix} 0 \\ 1 \end{pmatrix})$ , and  $L_+ = B_g(\begin{pmatrix} -f_2 \\ f_1 \end{pmatrix}, \begin{pmatrix} 0 \\ -1 \end{pmatrix})$ . Assume moreover that f has a conical intersection at 0, that is, f(0) = 0 and  $Df(0) \in GL_2(\mathbb{R})$ . Then the proto-

line-fields 
$$\begin{pmatrix} -f_2 \\ f_1 \end{pmatrix}, \begin{pmatrix} 0 \\ 1 \end{pmatrix}$$
 and  $\begin{pmatrix} -f_2 \\ f_1 \end{pmatrix}, \begin{pmatrix} 0 \\ -1 \end{pmatrix}$  have non-degenerate singularities at 0.

- Let  $f = (f_1, f_2, f_3) \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^3)$  and  $M_f = \begin{pmatrix} f_1 & f_2 \\ f_2 & f_3 \end{pmatrix}$ . Define  $\tilde{f} \in C^{\infty}(\mathbb{R}^2, \mathbb{R}^2)$  as  $\tilde{f} = (\frac{f_1 - f_3}{2}, f_2)$ . Then the real eigendirections of  $M_f$  are those of  $H_{\tilde{f}}$ . Moreover  $M_f$ 

has a conical intersection at 0 if and only if  $H_{\tilde{f}}$  has a conical intersection at 0.

Proof. Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be such that  $f(\bar{u}, \bar{v}) \neq 0$ . By direct computations,  $\begin{pmatrix} x \\ y \end{pmatrix} \in \mathbb{R}^2$  is an eigenvector of  $H_f(u, v)$  associated with  $\lambda_{\pm}(u, v)$  if and only if  $x(f_1(u, v) - \lambda_{\pm}(u, v)) + yf_2(u, v) = 0$ , that is,  $\begin{pmatrix} x \\ y \end{pmatrix}$  is collinear to  $\frac{1}{\lambda_{\pm}(u, v)} \begin{pmatrix} -f_2(u, v) \\ f_1(u, v) \end{pmatrix} + \begin{pmatrix} 0 \\ -1 \end{pmatrix}$ . We can deduce the first claim. Define the vector field X, for every  $(u, v) \in \mathbb{R}^2$ , by  $X(u, v) = \begin{pmatrix} -f_2(u, v) \\ f_1(u, v) \end{pmatrix}$ . The hypothesis that 0 is a conical intersection for f yields that X(0) = 0 and DX(0)is non-degenerate. The first claim is proved. The second claim is obtained by similar computations. 

We recall here some results from [22]. In a neighborhood of a conical intersection  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  between the levels j - 1 and j,

- For every  $(u, v) \neq (\bar{u}, \bar{v})$ , there exists a smooth choice of the sign of the eigenvectors  $\phi_{j-1}(u,v)$  and  $\phi_j(u,v)$  such that  $\chi_{j-1,j}$  defines a  $C^{\infty}$  vector field in a punctured neighborhood of  $(\bar{u}, \bar{v})$ ,
- The integral curves of  $\chi_{j-1,j}$  are  $C^{\infty}$  and the eigenvectors  $\phi_{j-1}$  and  $\phi_j$  are  $C^{\infty}$  along them,
- For every direction  $\eta$  of  $\mathbb{R}^2$ , there exists an integral curve  $\gamma: [0,1) \to \mathbb{R}^2$  of  $\chi_{j-1,j}$ such that  $\lim_{t \to 1^-} \gamma(t) = (\bar{u}, \bar{v}), \lim_{t \to 1^-} \frac{\dot{\gamma}(t)}{||\dot{\gamma}(t)||} = \eta.$

We say that the singularity of  $\chi_{j-1,j}$  has type (N), for node. In particular the third condition implies that the non-mixing curves are homeomorphic to the integral curves of the vector field  $(u,v) \mapsto \begin{pmatrix} u \\ v \end{pmatrix}$  locally around  $(\bar{u},\bar{v})$  and the index of  $\chi_{j-1,j}$  at  $(\bar{u},\bar{v})$  is equal to 1.

#### Singularities of $\chi_{j-1,j}$ at intersections (j, j+1). 5.3.3

Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be such that  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$ . Define a neighborhood U of  $(\bar{u}, \bar{v})$ in  $\mathbb{R}^2$  such that, for every  $(u,v) \in U \setminus \{(\bar{u},\bar{v})\}, \lambda_j(u,v) \notin \{\lambda_{j+1}(u,v),\lambda_{j-1}(u,v)\}$ . For every  $(u,v) \in U$ , let  $P_{j,j+1}(u,v) : \mathbb{R}^n \to \mathbb{R}^n$  be the orthogonal projection of H onto the eigenspace associated with  $\lambda_i(u, v)$  and  $\lambda_{i+1}(u, v)$ , which is smooth with respect to  $(u,v) \in U$ . As in Section 4.6.3, for every  $(u,v) \in U$ , define an orthogonal map  $I_{j,j+1}(u,v)$ :  $\mathbb{R}^2 \to \operatorname{Im} P_{j,j+1}(u,v)$ , smoothly depending on (u,v). Consider a normalized real eigenvector  $\phi_j(u,v)$  (respectively,  $\phi_{j+1}(u,v)$ ) of H(u,v) associated with  $\lambda_j(u,v)$  (respectively,  $\lambda_{j+1}(u,v)$ ) and a normalized real eigenvector  $\tilde{\phi}_j(u,v)$  (respectively,  $\tilde{\phi}_{j+1}(u,v)$ ) of  $h_{\operatorname{red}}(u,v)$  as defined in Section 4.6.3 associated with  $\lambda_j(u,v)$  (respectively,  $\lambda_{j+1}(u,v)$ ). Then we have  $I_{j,j+1}(u,v)(\tilde{\phi}_j(u,v)) = \eta \phi_j(u,v)$ , where  $\eta = \pm 1$ . Consider a smooth vector field X on  $\mathbb{R}^2$  such that, for every  $(u,v) \in U$ ,  $I_{j,j+1}(u,v)(X(u,v)) = P_{j,j+1}(u,v)(H_2\phi_{j-1}(u,v))$ , and a smooth vector field Y on  $\mathbb{R}^2$  such that, for every  $(u,v) \in U$ ,  $I_{j,j+1}(u,v)(H_1\phi_{j-1}(u,v))$ . Notice that the smoothness of X and Y rely on the fact that  $\phi_{j-1}(u,v)$  can be chosen smooth with respect to  $(u,v) \in U$ . Denote by g the Euclidean scalar product on  $\mathbb{R}^2$ .

**Proposition 5.3.11.** There exists  $A \in C^{\infty}(\mathbb{R}^2, M_2(\mathbb{R}))$  such that, for every  $(u, v) \in U \setminus \{(\bar{u}, \bar{v})\}, \chi_{j-1,j}(u, v)$  defines the same direction as  $A(u, v)\tilde{\phi}_j(u, v)$ .

Proof. We can write

$$\langle H_2 \phi_{j-1}(u, v), \phi_j(u, v) \rangle = \langle H_2 \phi_{j-1}(u, v), \eta I_{j,j+1}(u, v)(\phi_j(u, v)) \rangle$$
  
=  $\eta \langle P_{j,j+1}(u, v)(H_2 \phi_{j-1}(u, v)), I_{j,j+1}(u, v)(\tilde{\phi_j}(u, v)) \rangle$   
=  $\eta \langle I_{j,j+1}(u, v)(X(u, v)), I_{j,j+1}(u, v)(\tilde{\phi_j}(u, v)) \rangle$   
=  $\eta g(X(u, v), \tilde{\phi_j}(u, v)),$ 

where the last equality is obtained using that  $I_{j,j+1}(u, v)$  is an orthogonal map between  $\mathbb{R}^2$ and  $\operatorname{Im} P_{j,j+1}(u, v)$ , for every  $(u, v) \in U$ . By the same computations as previously, we have  $-\langle H_1\phi_{j-1}(u, v), \phi_j(u, v) \rangle = \eta g(Y(u, v), \tilde{\phi}_j(u, v))$ . Denoting the coordinates of X and Y in the canonical basis of  $\mathbb{R}^2$  by  $X(u, v) = \begin{pmatrix} x_1(u, v) \\ x_2(u, v) \end{pmatrix}$  and  $Y(u, v) = \begin{pmatrix} y_1(u, v) \\ y_2(u, v) \end{pmatrix}$ , we have, for every  $(u, v) \in U \setminus \{(\bar{u}, \bar{v})\}, \chi_{j-1,j}(u, v) = \eta A(u, v) \tilde{\phi}_j(u, v)$ , where, for every  $(u, v) \in U$ ,  $A(u, v) = \begin{pmatrix} x_1(u, v) & x_2(u, v) \\ y_1(u, v) & y_2(u, v) \end{pmatrix}$ , and  $\eta = \pm 1$ . We deduce the result.  $\Box$ 

Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be a conical intersection between  $\lambda_j$  and  $\lambda_{j+1}$ . We introduce the following condition:

Condition (C): The vectors  $P_{j,j+1}H_1\phi_{j-1}(\bar{u},\bar{v})$  and  $P_{j,j+1}H_2\phi_{j-1}(\bar{u},\bar{v})$  are not colinear.

We prove the following theorem.

**Theorem 5.3.12.** Let  $(\bar{u}, \bar{v}) \in \mathbb{R}^2$  be such that  $\lambda_j(\bar{u}, \bar{v}) = \lambda_{j+1}(\bar{u}, \bar{v})$  is a conical intersection. Assume that Condition (C) is satisfied at  $(\bar{u}, \bar{v})$ . Then  $\chi_{j-1,j}$  has a Darbouxian singularity at  $(\bar{u}, \bar{v})$ .

Proof. By definition  $\phi_j$  is the eigendirection associated with the eigenvalue  $\lambda_j$  of the twolevel Hamiltonian  $h_{\text{red}}$  having a conical intersection at  $(\bar{u}, \bar{v})$ . Hence Proposition 5.3.10 proves that the line field  $\phi_j$  has a non-degenerate singularity at  $(\bar{u}, \bar{v})$ . Under assumption (**C**), the matrix  $A \in C^{\infty}(\mathbb{R}^2, M_2(\mathbb{R}))$  defined in Proposition 5.3.11 is invertible. By Proposition 5.3.8, we deduce that  $\chi_{j-1,j} = A\phi_j$  (where the equality is defined as the equality of the associated directions in  $\mathbb{R}^2$ ) has a non-degenerate singularity at  $(\bar{u}, \bar{v})$ . We deduce the theorem by using Proposition 5.3.7.

We show without difficulty that Condition  $(\mathbf{C})$  is satisfied in the following case.

Corollary 5.3.13. Let  $E_1, E_2, E_3 \in \mathbb{R}$  such that  $E_1 < E_2 < E_3$  and  $H(u, v) = \begin{pmatrix} E_1 & u & 0 \\ u & E_2 & v \\ 0 & v & E_3 \end{pmatrix}$ .

Then  $\chi_{1,2}$  (respectively  $\chi_{2,3}$ ) has Darbouxian singularities at the points  $(u, v) \in \mathbb{R}^2$  such that  $\lambda_2(u, v) = \lambda_3(u, v)$  (respectively,  $\lambda_1(u, v) = \lambda_2(u, v)$ ) that are

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0) \text{ (respectively } (0, \pm \sqrt{(E_1 - E_3)(E_1 - E_2)})).$$
  
Let  $E_1, E_2, E_3 \in \mathbb{R}$  and  $H(u, v) = \begin{pmatrix} E_1 & u & 0\\ u & E_2 & v\\ 0 & v & E_2 \end{pmatrix}$ . On Figure 5.18, we have plotted

the non-mixing curves between the levels  $\lambda_1$  and  $\lambda_2$  when  $E_1 < E_2 < E_3$ . Along these curves the precision of the adiabatic approximation for a regular control path at speed  $\epsilon$  has an order equal to  $O(\epsilon)$  on a time interval of length  $\frac{1}{\epsilon}$ .

When considering the case where  $E_1 = E_2 < E_3$ , we can prove that the system admits two conical intersections between the levels  $\lambda_2$  and  $\lambda_3$  at the points

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0) = (\pm |E_1 - E_3|, 0),$$

and a semi-conical intersection between the levels  $\lambda_1$  and  $\lambda_2$  at (0,0). On Figure 5.20, we have plotted the non-mixing curves between the levels  $\lambda_1$  and  $\lambda_2$  in this case. By direct computations, we can show that Condition (C) is not satisfied at the points

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0) = (\pm |E_1 - E_3|, 0),$$

and we observe that the non-mixing field  $\chi_{12}$  between the levels  $\lambda_1$  and  $\lambda_2$  is non-singular at these points. We observe a node at the semi-conical intersection (0,0) between the levels  $\lambda_1$  and  $\lambda_2$ . In further studies, it would be interesting to prove that the singularity at the origin is a node, by studying the non-mixing field at a semi-conical intersection for *n*-level systems.

When considering the case where  $E_2 < E_1 < E_3$ , we can prove that the system admits two conical intersections between the levels  $\lambda_2$  and  $\lambda_3$  at the points

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0),$$

and admits no intersection between the levels  $\lambda_1$  and  $\lambda_2$  at (0,0). By direct computations, we can show that Condition (C) is satisfied at the points

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0).$$

As a consequence, the non-mixing field  $\chi_{12}$  between the levels  $\lambda_1$  and  $\lambda_2$  has Darbouxian singularities at these points. On Figure 5.22, we have plotted the non-mixing curves between the levels  $\lambda_1$  and  $\lambda_2$  in this case. We notice the presence of Lemon singularities at the points

$$(u, v) = (\pm \sqrt{(E_1 - E_3)(E_2 - E_3)}, 0).$$

Then we have plotted the non-mixing field between  $\lambda_1$  and  $\lambda_2$  in the three remaining cases, that require to be studied more precisely:

- The case where  $E_1 < E_2 = E_3$  on Figure 5.24. In this case, there is a semi-conical intersection between the levels  $\lambda_2 = \lambda_3$  and two conical intersections between the levels  $\lambda_1$  and  $\lambda_2$ . The non-mixing field between the levels  $\lambda_1$  and  $\lambda_2$  has two singularities of type (**N**) at the conical intersections between the levels  $\lambda_1$  and  $\lambda_2$ . Moreover, the two Star singularities that occured in the case  $E_1 < E_2 < E_3$  join at the origin (0,0) into a singularity that is a bifurcation of singularities of line-fields. Bifurcations for one-parameter families of line-fields is a future topic of study.
- The case where  $E_1 < E_3$ ,  $E_2 > E_3$ , on Figure 5.26. In this case, there is no intersection of eigenvalue between  $\lambda_2$  and  $\lambda_3$ , and two conical intersections of eigenvalues between  $\lambda_1$  and  $\lambda_2$ . Hence there are two singularities of type (**N**) at the conical intersections between  $\lambda_1$  and  $\lambda_2$ , and we notice graphically that there is a saddle at (0,0).
- The case where  $E_1 = E_2 = E_3$  on Figure 5.28. Up to a change of global phase in the dynamics, we can assume  $E_1 = E_2 = E_3 = 0$ . In this case, the eigenvectors of H(u, v) are constant functions along every half-straight line with (0, 0) as the origin of the half-straight lines.

**Remark 5.3.14.** Notice that the construction of the non-mixing field could be a first step to build a topology on line fields on dimension 2 manifolds that allows both generic singularities of integer and half-integer index.





Figure 5.18 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP of Corollary 5.3.13.

Figure 5.19 – Spectrum as a function of (u, v) for the STIRAP, where  $E_1 < E_2 < E_3$ .



Figure 5.20 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the degenerate STIRAP, where  $E_1 = E_2 < E_3$ .



Figure 5.21 – Spectrum as a function of (u, v) for the degenerate STI-RAP, where  $E_1 = E_2 < E_3$ .



Figure 5.22 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP, where  $E_2 < E_1 < E_3$ .



Figure 5.23 – Spectrum as a function of (u, v) for the STIRAP, where  $E_2 < E_1 < E_3$ .



Figure 5.24 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP, where  $E_1 < E_2 = E_3$ .



Figure 5.25 – Spectrum as a function of (u, v) for the STIRAP, where  $E_1 < E_2 = E_3$ .



Figure 5.26 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP, where  $E_1 < E_3, E_2 > E_3$ .



Figure 5.27 – Spectrum as a function of (u, v) for the STIRAP, where  $E_1 < E_3$  and  $E_2 > E_3$ .



Figure 5.28 – Non-mixing field between  $\lambda_1$  and  $\lambda_2$  for the STIRAP, where  $E_1 = E_2 = E_3$ .



Figure 5.29 – Spectrum as a function of (u, v) for the STIRAP, where  $E_1 = E_2 = E_3$ .

## Chapter 6

# Control of quantum systems with a single input

In this chapter, we discuss the compatibility between the *rotating-wave* and the *adiabatic* approximations for controlled quantum systems. We introduce the problem on two-level quantum systems, then we prove results which are valid in higher dimension. Under some suitable hypotheses on the time scales, the two approximations can be combined. As a natural consequence of this, it is possible to design control laws achieving transitions of states between two energy levels of the Hamiltonian that are robust with respect to inhomogeneities of the amplitude of the control input.

#### Contents

6.1 General framework and main results		
	6.1.1	Problem formulation
	6.1.2	$Main results \dots \dots$
6.2	$\operatorname{App}$	$ m roximation\ results\ \ .\ .\ .\ .\ .\ .\ .\ .\ .\ .\ .\ .\ $
	6.2.1	Variation formula
	6.2.2	Regularity of the eigenstates
	6.2.3	Averaging of quantum systems
	6.2.4	Perturbation of an adiabatic trajectory
	6.2.5	Parametric case
6.3	6.3 Control of two-level systems	
	6.3.1	Control strategy for two-level systems and simulations $\ldots \ldots 159$
	6.3.2	Robustness of the control strategy with respect to amplitude of
		control inhomogeneities
6.4 Control of STIRAP Process		

#### 6.1 General framework and main results

#### 6.1.1 Problem formulation

#### **Rotating frame**

Consider  $v, \varphi \in C^{\infty}([0,1], \mathbb{R})$  such that  $\varphi(0) = 0, E > 0$ , and  $\psi_0 \in \mathbb{C}^2$ . Denote by  $\psi : [0,1] \to \mathbb{C}^2$  the solution of the equation

$$i\frac{d\psi(t)}{dt} = \begin{pmatrix} E & w(t)\\ \bar{w}(t) & -E \end{pmatrix} \psi(t), \quad \psi(0) = \psi_0, \tag{6.1}$$

where  $w(t) = v(t)e^{i(2Et+\varphi(t))}$ . Define  $\eta(t) = \mathbf{U}(t)\psi(t)$  where

$$\mathbf{U}(t) = \begin{pmatrix} e^{-i\left(Et + \frac{\varphi(t)}{2}\right)} & 0\\ 0 & e^{i\left(Et + \frac{\varphi(t)}{2}\right)} \end{pmatrix}.$$

Then  $\eta(t)$  satisfies

$$i\frac{d\eta(t)}{dt} = \begin{pmatrix} -\frac{\varphi'(t)}{2} & v(t) \\ v(t) & \frac{\varphi'(t)}{2} \end{pmatrix} \eta(t), \quad \eta(0) = \psi_0.$$
(6.2)

We say that the dynamics are expressed in the *rotating frame* of speed  $E + \frac{\varphi'(t)}{2}$ . Such an equation can be controlled using several approaches, namely via the well-known Rabi oscillations and the adiabatic approach presented below (see [80] for a comparison between the two approaches).

#### Adiabatic control in the rotating frame

In order to design an adiabatic control strategy for Equation (6.2), let us add a parameter  $\epsilon$  in the control w and introduce  $w_{\epsilon}(t) = v(\epsilon t)e^{i(2Et + \frac{\varphi(\epsilon t)}{\epsilon})}$ . Consider the solution of (6.2) with initial condition  $\psi_0$  corresponding to such  $(v, \varphi)$ , that is,

$$i\frac{d\eta_{\epsilon}(t)}{dt} = \begin{pmatrix} -\frac{\varphi'(\epsilon t)}{2} & v(\epsilon t) \\ v(\epsilon t) & \frac{\varphi'(\epsilon t)}{2} \end{pmatrix} \eta_{\epsilon}(t), \quad \eta_{\epsilon}(0) = \psi_0.$$

In the variable  $\tau = \epsilon t \in [0, 1]$ , the reparameterized trajectory  $\tilde{\eta}_{\epsilon}(\tau) = \eta_{\epsilon}(\tau/\epsilon)$  satisfies

$$i\frac{d\tilde{\eta}_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon} \begin{pmatrix} -\frac{\varphi'(\tau)}{2} & v(\tau) \\ v(\tau) & \frac{\varphi'(\tau)}{2} \end{pmatrix} \tilde{\eta}_{\epsilon}(\tau), \quad \tilde{\eta}_{\epsilon}(0) = \psi_0.$$
(6.3)

Let v and  $\varphi$  be chosen so that the curve  $(v, \varphi') : [0,1] \to \mathbb{R}^2$  connects (0,-1) to (0,1) intersecting the vertical axis only at its endpoints. Then, by standard adiabatic approximation (see Theorem 2.1.1), if  $\psi_0 = (1,0)$ , then  $\tilde{\eta}_{\epsilon}(1)$  converges, up to phases, to (0,1) as  $\epsilon \to 0$ . In the literature, this control strategy is now very classical and is called *chirped adiabatic pulse*. Its robustness properties have been studied within a mathematical framework in Chapter 3.

#### Rotating wave approximation

In many applications only one real control is available. A classical strategy to duplicate the control input is the so-called *rotating wave approximation* (RWA) that works as follows. Let  $\varphi_{\tilde{\epsilon}}: [0, 1/\tilde{\epsilon}] \to \mathbb{C}^2$  be the solution of (6.1) where w is replaced by the control  $u_{\tilde{\epsilon}}(t) = 2\tilde{\epsilon}v(\tilde{\epsilon}t)\cos(2Et + \varphi(\tilde{\epsilon}t))$ . Let

$$\mathbf{U}_{\tilde{\epsilon}}(t) = \begin{pmatrix} e^{-i\left(Et + \frac{\varphi(\tilde{\epsilon}t)}{2}\right)} & 0\\ 0 & e^{+i\left(Et + \frac{\varphi(\tilde{\epsilon}t)}{2}\right)} \end{pmatrix}.$$

The RWA then states (see [70]) that  $\tau \mapsto \mathbf{U}_{\tilde{\epsilon}}(\tau/\tilde{\epsilon})\varphi_{\tilde{\epsilon}}(\tau/\tilde{\epsilon})$  converges uniformly, as  $\tilde{\epsilon} \to 0$ , to the solution of

$$i\frac{d\eta(t)}{dt} = \begin{pmatrix} -\frac{\varphi'(t)}{2} & v(t) \\ v(t) & \frac{\varphi'(t)}{2} \end{pmatrix} \eta(t), \quad \eta(0) = \psi_0.$$
(6.4)

Notice that the limit equation (6.4) coincides with (6.2), which is the original equation (6.1) with complex controls in the rotating frame. We have already described how to control (6.2) via adiabatic theory. It is not clear, however, if the RWA and the adiabatic approximations can be combined.

For this purpose, we introduce  $u_{\epsilon_1,\epsilon_2}(t) = 2\epsilon_1 v(\epsilon_1 \epsilon_2 t) \cos(2Et + \frac{1}{\epsilon_2}\varphi(\epsilon_1 \epsilon_2 t))$ , where  $\epsilon_1$ and  $\epsilon_2$  play the role of  $\tilde{\epsilon}$  and  $\epsilon$ , respectively. In order to establish in which regime the two approximations can be combined, we set  $\epsilon_1 = \epsilon^{\alpha}$ ,  $\epsilon_2 = \epsilon$  where  $\alpha \in \mathbb{R}$  and  $u_{\epsilon} = u_{\epsilon^{\alpha},\epsilon}$ . Consider the Cauchy problem

$$i\frac{d\psi_{\epsilon}(t)}{dt} = \begin{pmatrix} E & u_{\epsilon}(t) \\ u_{\epsilon}(t) & -E \end{pmatrix} \psi_{\epsilon}(t), \quad \psi_{\epsilon}(0) = \psi_{0}.$$
(6.5)

Define  $\Psi_{\epsilon}(t) = U_{\epsilon}(t)\psi_{\epsilon}(t)$  where  $U_{\epsilon}(t) = \begin{pmatrix} e^{-i\left(Et + \frac{\varphi(\epsilon^{\alpha+1}t)}{2\epsilon}\right)} & 0\\ 0 & e^{+i\left(Et + \frac{\varphi(\epsilon^{\alpha+1}t)}{2\epsilon}\right)} \end{pmatrix}$ . In the variable  $\tau = \epsilon^{\alpha+1}t \in [0, 1]$ , the reparameterized trajectory  $\tilde{\Psi}_{\epsilon}(\tau) = \Psi_{\epsilon}(\tau/\epsilon)$  satisfies,

$$\frac{d\tilde{\Psi}_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)\right)\tilde{\Psi}_{\epsilon}(\tau), \quad \tilde{\Psi}_{\epsilon}(0) = \psi_{0}, \tag{6.6}$$

where

$$A(\tau) = -i \begin{pmatrix} -\varphi'(\tau)/2 & v(\tau) \\ v(\tau) & \varphi'(\tau)/2 \end{pmatrix}, \text{ and } B_{\epsilon}(\tau) = \frac{-i}{\epsilon} \begin{pmatrix} 0 & v(\tau)e^{i(\frac{4E\tau}{\epsilon^{\alpha+1}} + \frac{2\varphi(\tau)}{\epsilon})} \\ v(\tau)e^{-i(\frac{4E\tau}{\epsilon^{\alpha+1}} + \frac{2\varphi(\tau)}{\epsilon})} & 0 \end{pmatrix}$$

The dynamics of  $\tilde{\Psi}$  are characterized by the sum of the term that we had in Equation (6.3), that corresponds to the dynamics for the complex control case in the rotating frame, and of an oscillating term  $B_{\epsilon}(\tau)$ . The RWA consists in neglecting the term  $B_{\epsilon}$ . We are going to show that this can be mathematically justified if  $\alpha > 1$ .

#### 6.1.2 Main results

In order to obtain the asymptotic analysis announced in the previous section, we show a result of approximation of adiabatic trajectories for general n-level systems under the form of Equation (6.6). Then we deduce results in the particular case of two-level systems with a drift term.

#### Adiabatic approximation result

**Definition 6.1.1.** For  $A \in C^{\infty}([0,1], u(n))$ , denote by  $j \mapsto \lambda_j(\tau)$  the nondecreasing sequence of eigenvalues of  $iA(\tau)$ . We say that A satisfies a gap condition if and only if there exists C > 0 such that

$$\forall j \neq k, \forall \tau \in [0, 1], \ |\lambda_j(\tau) - \lambda_k(\tau)| > C.$$
(GAP)

We say that A has a k-th order intersection of eigenvalues if and only if there exist  $\tau_0 \in [0,1]$  and  $j \in \{1,\ldots,n\}$  such that  $\lambda_j, \lambda_{j+1}$  is separated from the rest of the spectrum and for every  $\tau \in [0,1]$  and  $l \neq j$ ,  $\lambda_l(\tau) \neq \lambda_{l+1}(\tau)$ , and

$$\lambda_j(\tau_0) = \lambda_{j+1}(\tau_0), \quad \forall \tau \neq \tau_0, \ \lambda_j(\tau) \neq \lambda_{j+1}(\tau), \quad and \quad \frac{d^k(\lambda_{j+1} - \lambda_j)}{d\tau^k}(\tau_0) \neq 0.$$
 (k-GAP)

**Definition 6.1.2.** Let  $\alpha$  be a nonzero real number. Define by  $S(\alpha)$  the set of families  $(B_{\epsilon})_{\epsilon>0}$  of functions in  $C^{\infty}([0,1], u(n))$  such that

- $(B_{\epsilon}(\tau))_{ij} = 0$  for every  $j \in \{1, \ldots, n\}$  and every  $\tau \in [0, 1]$  and
- for every k > j there exist  $\beta_{jk} \in \mathbb{R} \setminus \{0\}$  and  $v_{jk}, h_{jk} \in C^{\infty}([0,1],\mathbb{R})$  such that  $(B_{\epsilon}(\tau))_{jk} = -\frac{i}{\epsilon} v_{jk}(\tau) e^{i(\frac{\beta_{jk}\tau}{\epsilon^{\alpha+1}} + \frac{h_{jk}(\tau)}{\epsilon})}$  for every  $\tau \in [0,1]$ .

**Theorem 6.1.3.** Consider  $A \in C^{\infty}([0,1], u(n))$  and a finite sum  $(B_{\epsilon})_{\epsilon>0}$  of elements belonging to  $S(\alpha)$  with  $\alpha > 1$ . Assume that  $A(\cdot)$  satisfies (GAP). Set  $X_0 \in \mathbb{C}^n$  independent of  $\epsilon$ . Let  $X_{\epsilon}$  be the solution of  $\frac{dX_{\epsilon}(\tau)}{d\tau} = (\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)) X_{\epsilon}(\tau)$  such that  $X_{\epsilon}(0) = X_0$  and  $\hat{X}_{\epsilon}$  be the solution of  $\frac{d\hat{X}_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon}A(\tau)\hat{X}_{\epsilon}(\tau)$  such that  $\hat{X}_{\epsilon}(0) = X_0$ . Then there exists c > 0independent of  $\tau$ ,  $\epsilon$  such that for every  $\tau \in [0, 1]$ ,

$$||X_{\epsilon}(\tau) - \hat{X}_{\epsilon}(\tau)|| \le c\epsilon^{\min(1,\alpha-1)}.$$

**Remark 6.1.4.** If A satisfies (k-GAP), then Theorem 6.1.3 remains true, replacing the error of order  $\epsilon^{\min(1,\alpha-1)}$  by  $\epsilon^{\min(\frac{1}{k+1},\alpha-1)}$ .

#### Application to two-level systems

We consider  $v, \varphi \in C^{\infty}([0,1],\mathbb{R})$  such that  $\varphi(0) = 0$  and E > 0. We consider now Equation (6.5) where  $u_{\epsilon}(t) = 2\epsilon^{\alpha}v(\epsilon^{\alpha+1}t)\cos(2Et + \frac{1}{\epsilon}\varphi(\epsilon^{\alpha+1}t))$ . In the fast time scale  $\tau = \epsilon^{\alpha+1} t \in [0, 1]$ , Equation (6.5) can be rewritten as

$$i\frac{d\psi_{\epsilon}(\tau)}{d\tau} = \begin{pmatrix} \frac{E}{\epsilon^{\alpha+1}} & u_{\epsilon}(\tau) \\ u_{\epsilon}(\tau) & -\frac{E}{\epsilon^{\alpha+1}} \end{pmatrix} \psi_{\epsilon}(\tau)$$
(6.7)

for  $\tau \in [0, 1]$  where by a slight abuse of notation, we write  $u_{\epsilon}(\tau) = \frac{2}{\epsilon}v(\tau)\cos(\frac{2E\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon}\varphi(\tau))$ . Set  $\psi_0 \in \mathbb{C}^2$  independent of  $\epsilon$ . Let  $\psi_{\epsilon}(\cdot)$  be the solution of Equation (6.7) such that  $\psi_{\epsilon}(0) = \psi_0$ . Similarly, let  $\hat{\psi}_{\epsilon}$  be the solution of

$$i\frac{d\hat{\psi}_{\epsilon}(\tau)}{d\tau} = \begin{pmatrix} \frac{E}{\epsilon^{\alpha+1}} & w_{\epsilon}(\tau)\\ \bar{w}_{\epsilon}(\tau) & -\frac{E}{\epsilon^{\alpha+1}} \end{pmatrix} \hat{\psi}_{\epsilon}(\tau), \quad \hat{\psi}_{\epsilon}(0) = \psi_{0}$$
(6.8)

for  $\tau \in [0,1]$  and  $w_{\epsilon}(\tau) = \frac{1}{\epsilon} v(\tau) e^{i(\frac{2E\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon}\varphi(\tau))}$ .

**Theorem 6.1.5.** Assume that  $\alpha > 1$ . Consider  $v, \varphi$  in  $C^{\infty}([0,1], \mathbb{R})$  such that  $\varphi(0) = 0$ and  $v^2 + \frac{\varphi'^2}{4}$  is bounded from below by C > 0. Then the solution  $\psi_{\epsilon}$  of Equation (6.7) satisfies

$$\|\psi_{\epsilon}(\tau) - \hat{\psi}_{\epsilon}(\tau)\| < c\epsilon^{\min(1,\alpha-1)}$$

where c > 0 is independent of  $(\tau, \epsilon)$ .

*Proof.* Apply the unitary transformation  $X_{\epsilon}(\tau) = U_{\epsilon}(\tau)\psi_{\epsilon}(\tau)$  where

$$U_{\epsilon}(\tau) = \begin{pmatrix} e^{-i\left(\frac{E\tau}{\epsilon^{\alpha+1}} + \frac{\varphi(\tau)}{2\epsilon}\right)} & 0\\ 0 & e^{i\left(\frac{E\tau}{\epsilon^{\alpha+1}} + \frac{\varphi(\tau)}{2\epsilon}\right)} \end{pmatrix}.$$

Then  $X_{\epsilon}$  satisfies the equation

$$\frac{dX_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)\right)X_{\epsilon}(\tau), \quad X_{\epsilon}(0) = \psi_0,$$

where 
$$A(\tau) = -i \begin{pmatrix} -\varphi'(\tau)/2 & v(\tau) \\ v(\tau) & \varphi'(\tau)/2 \end{pmatrix}$$
, and  

$$B_{\epsilon}(\tau) = \frac{-i}{\epsilon} \begin{pmatrix} 0 & v(\tau)e^{i(\frac{4E\tau}{\epsilon^{\alpha+1}} + \frac{2\varphi(\tau)}{\epsilon})} \\ v(\tau)e^{-i(\frac{4E\tau}{\epsilon^{\alpha+1}} + \frac{2\varphi(\tau)}{\epsilon})} & 0 \end{pmatrix}.$$

The condition  $v(\tau)^2 + \frac{\varphi'(\tau)^2}{4} > C$  for every  $\tau \in [0, 1]$  implies that A satisfies Condition (GAP). Let  $\hat{X}_{\epsilon} : [0, 1] \to \mathbb{C}^2$  be the solution of

$$\frac{dX_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon}A(\tau)\hat{X}_{\epsilon}(\tau), \quad \hat{X}_{\epsilon}(0) = \psi_0.$$

Theorem 6.1.3 then implies that  $||X_{\epsilon}(\tau) - \hat{X}_{\epsilon}(\tau)|| \leq c\epsilon^{\min(1,\alpha-1)}$ . Noticing that  $\tilde{\psi}_{\epsilon}(\tau) = U_{\epsilon}^{*}(\tau)\hat{X}_{\epsilon}(\tau)$ , we get the result.

Theorem 6.1.5 will be used in Section 6.3 to design control laws for two-level systems using the key fact that  $\hat{\psi}_{\epsilon}(\tau)$  follows an adiabatic evolution up to a change of frame.

#### Application to *n*-level quantum systems

Let  $E_1, \ldots, E_n \in \mathbb{R}$  such that  $|E_k - E_q| \neq |E_j - E_l|$ , for every  $k, q, j, l \in \{1, \ldots, n\}$ such that  $l \neq j$  and  $(k, q) \notin \{(j, l), (l, j)\}$ . Let  $H_0 = \text{diag}(E_j)_{j=1}^n$  and let  $H_1$  be a real symmetric matrix such that, for every  $j, k \in \{1 \ldots n\}, (H_1)_{jj} = 0$ . Define

$$u_{\epsilon}(t) = 2\epsilon^{\alpha} \sum_{k < q} v_{kq}(\epsilon^{\alpha+1}t) \cos((E_k - E_q)t + \frac{1}{\epsilon} \left(\varphi_k(\epsilon^{\alpha+1}t) - \varphi_q(\epsilon^{\alpha+1}t)\right)),$$

where, for every  $k, q \in \{1 \dots n\}$ ,  $\varphi_k, \varphi_q$  are smooth real functions such that  $\varphi_q(0) = \varphi_k(0) = 0$ , and for every k < q,  $v_{kq}$  is a smooth real function. For every  $\tau \in [0, 1]$ , define  $A(\tau) \in u(n)$  such that, for every  $j, k \in \{1, \dots, n\}$ ,  $(A(\tau))_{jj} = -i\varphi'_j(\tau)$  and for  $j \neq k$ ,  $(A(\tau))_{jk} = -iv_{jk}(\tau)(H_1)_{jk}$ .

**Theorem 6.1.6.** Set  $\psi_0 \in \mathbb{C}^n$ . Let  $\psi_{\epsilon}(\cdot)$  be the solution of

$$i\frac{d\psi_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon^{\alpha+1}}H_0 + u_{\epsilon}(\tau)H_1\right)\psi_{\epsilon}(\tau)$$
(6.9)

for  $\tau \in [0,1]$ , with  $\psi_{\epsilon}(0) = \psi_0$ , where by a slight abuse of notation, we write

$$u_{\epsilon}(\tau) = \frac{2}{\epsilon} \sum_{k < q} v_{kq}(\tau) \cos\left(\frac{(E_k - E_q)\tau}{\epsilon^{\alpha + 1}} + \frac{1}{\epsilon} \left(\varphi_k(\tau) - \varphi_q(\tau)\right)\right).$$

Similarly, let  $\hat{\psi}_{\epsilon}(\cdot)$  be the solution of

$$i\frac{d\hat{\psi}_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon^{\alpha+1}}H_0 + \Omega_{\epsilon}(\tau)\right)\hat{\psi}_{\epsilon}(\tau), \quad \hat{\psi}_{\epsilon}(0) = \psi_0, \tag{6.10}$$

for  $\tau \in [0,1]$  and, for every  $\tau \in [0,1]$  and  $\epsilon > 0$ ,  $\Omega_{\epsilon}(\tau)$  is a  $n \times n$  Hermitian matrix such that  $(\Omega_{\epsilon}(\tau))_{kq} = \frac{1}{\epsilon} v_{kq}(\tau) e^{i\left(\frac{(E_k - E_q)\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon}(\varphi_k(\tau) - \varphi_q(\tau))\right)} (H_1)_{kq}$ . Assume moreover that A satisfies Condition (GAP). Then

$$\|\psi_{\epsilon}(\tau) - \hat{\psi}_{\epsilon}(\tau)\| < c\epsilon^{\min(1,\alpha-1)}$$

where c > 0 is independent of  $(\tau, \epsilon)$ .

*Proof.* We proceed in the same way as in the proof of Theorem 6.1.5. Define  $\Psi_{\epsilon}(\tau) = U_{\epsilon}(\tau)\psi_{\epsilon}(\tau)$  and  $\hat{\Psi}_{\epsilon}(\tau) = U_{\epsilon}(\tau)\hat{\psi}_{\epsilon}(\tau)$  where  $U_{\epsilon}(\tau) = \text{diag}(e^{i(\frac{E_{k}\tau}{\epsilon^{\alpha+1}} + \frac{\varphi_{k}(\tau)}{\epsilon})})_{k \in \{1...n\}}$ . By direct computations,  $\hat{\Psi}_{\epsilon}$  satisfies

$$\frac{d\Psi_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon} A(\tau) \hat{\Psi}_{\epsilon}(\tau), \quad \hat{\Psi}_{\epsilon}(0) = \psi_0, \tag{6.11}$$

and  $\Psi_{\epsilon}$  satisfies

$$\frac{d\Psi_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)\right)\Psi_{\epsilon}(\tau), \quad \Psi_{\epsilon}(0) = \psi_{0}, \tag{6.12}$$

where, for j < l,

$$(B_{\epsilon}(\tau))_{jl} = -\frac{i}{\epsilon}(H_{1})_{jl}v_{jl}(\tau)e^{i\left(\frac{2(E_{l}-E_{j})\tau}{\epsilon^{\alpha+1}} + \frac{2(\varphi_{l}(\tau)-\varphi_{j}(\tau))}{\epsilon}\right)}\right)} -\frac{i}{\epsilon}(H_{1})_{jl}\sum_{(k$$

 $(B_{\epsilon}(\tau))_{jj} = 0$ , and  $(B_{\epsilon}(\tau))_{lj} = -(B_{\epsilon}(\tau))^*_{lj}$ . Recall that  $|E_k - E_q| \neq |E_j - E_l|$ , for every  $k, q, j, l \in \{1 \dots n\}$  such that  $(k, q) \notin \{(j, l), (l, j)\}$ . Hence we deduce that  $(B_{\epsilon})_{\epsilon>0}$  is a finite sum of elements of  $S(\alpha)$ . By applying Theorem 6.1.3, we get for every  $\tau \in [0, 1]$ ,

$$\|\Psi_{\epsilon}(\tau) - \hat{\Psi}_{\epsilon}(\tau)\| < c\epsilon^{\min(1,\alpha-1)}$$

where c > 0 is independent of  $(\tau, \epsilon)$ .

#### 6.2 Approximation results

#### 6.2.1 Variation formula

We recall here without proof a classical formula which will be useful to neglect highly oscillating parts of the dynamics.

**Proposition 6.2.1** (Variation formula). Let A, B be in  $C^{\infty}([0,1], u(n))$ . Consider the equation

$$\frac{dx(\tau)}{d\tau} = (A(\tau) + B(\tau)) x(\tau), \quad x(\tau) \in \mathbb{C}^n.$$
(6.13)

Denote the flow of the equation  $\frac{dx(\tau)}{d\tau} = A(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau} \in U(n)$  and the flow of the equation  $\frac{dx(\tau)}{d\tau} = P_{\tau}^{-1}B(\tau)P_{\tau}x(\tau)$  at time  $\tau$  by  $W_{\tau} \in U(n)$ . Then the flow of Equation (6.13) at time  $\tau$  is equal to  $Q_{\tau} = P_{\tau}W_{\tau}$ .

#### 6.2.2 Regularity of the eigenstates

We recall here a well-known regularity result.

**Lemma 6.2.2.** — Let  $A \in C^{\infty}([0,1], u(n))$  satisfy (GAP). Then the eigenvectors and the eigenvalues of  $iA(\tau)$  can be chosen  $C^{\infty}$  with respect to  $\tau$ .

- Let  $A \in C^{\infty}([0,1], u(n))$  have a k-th order intersection of eigenvalues at  $\tau_0$ . Then the eigenvectors and the eigenvalues of  $iA(\tau)$  can be chosen  $C^{\infty}$  with respect to  $\tau$ .

#### 6.2.3 Averaging of quantum systems

**Theorem 6.2.3.** Consider A and  $(A_{\epsilon})_{\epsilon>0}$  in  $C^{\infty}([0,1], u(n))$  and assume that  $A_{\epsilon}(\tau)$  is uniformly bounded w.r.t.  $(\tau, \epsilon)$ . Denote the flow of the equation  $\frac{dx(\tau)}{d\tau} = A(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau} \in U(n)$  and the flow of the equation  $\frac{dx(\tau)}{d\tau} = A_{\epsilon}(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau}^{\epsilon} \in U(n)$ . If

$$\int_0^\tau A_\epsilon(s)ds = \int_0^\tau A(s)ds + O(\epsilon)$$

then

$$P_{\tau}^{\epsilon} = P_{\tau} + O(\epsilon),$$

both estimates being uniform w.r.t.  $\tau \in [0, 1]$ .

We state Theorem 6.2.3 without proof because it is a particular case of next result, Theorem 6.2.4. In the following, we do not assume the boundedness of  $A_{\epsilon}$  with respect to  $\epsilon$ . We refer to [57, 58, 59, 66, 77] for more informations on the case of averaging of a general class of dynamical systems with non-bounded and highly oscillatory inputs. Our result provides an estimate of the error in the special case of quantum systems.

**Theorem 6.2.4.** Consider A and  $(B_{\epsilon})_{\epsilon>0}$  in  $C^{\infty}([0,1], u(n))$ . Assume that

$$\int_0^\tau B_\epsilon(s)ds = O(\epsilon)$$

and

$$\int_0^\tau |B_\epsilon(s)| \left| \int_0^s B_\epsilon(x) dx \right| ds = O(\epsilon^k)$$

uniformly w.r.t.  $\tau \in [0,1]$ , with k > 0. Set  $A_{\epsilon} = A + B_{\epsilon}$ . Denote the flow of the equation  $\frac{dx(\tau)}{d\tau} = A(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau} \in U(n)$  and the flow of the equation  $\frac{dx(\tau)}{d\tau} = A_{\epsilon}(\tau)x(\tau)$  at time  $\tau$  by  $P_{\tau}^{\epsilon} \in U(n)$ . Then we have

$$P_{\tau}^{\epsilon} = P_{\tau} + O(\epsilon^{\min(k,1)}),$$

uniformly w.r.t.  $\tau \in [0, 1]$ .

*Proof.* Under the hypotheses of the theorem, there exists K > 0 such that for every  $\tau \in [0, 1], |\int_0^\tau B_\epsilon(s)ds| < K\epsilon$ . Let  $Q_\tau^\epsilon$  be the flow associated with  $B_\epsilon$ . We have  $Q_\tau^\epsilon = \mathrm{Id} + \int_0^\tau B_\epsilon(s)Q_s^\epsilon ds$ . By integration by parts,  $Q_\tau^\epsilon = \mathrm{Id} + (\int_0^\tau B_\epsilon(s)ds) Q_\tau^\epsilon - \int_0^\tau (\int_0^s B_\epsilon(\theta)d\theta)B_\epsilon(s)Q_s^\epsilon ds$ . Moreover,  $Q_\tau^\epsilon$  is bounded uniformly w.r.t.  $(\tau, \epsilon)$ , since it evolves in U(n). By the triangular inequality, we get

$$|Q_{\tau}^{\epsilon} - \mathrm{Id}| \leq \left| \int_{0}^{\tau} B_{\epsilon}(s) ds \right| |Q_{\tau}^{\epsilon}| + \int_{0}^{\tau} \left| \int_{0}^{s} B_{\epsilon}(\theta) d\theta \right| |B_{\epsilon}(s) Q_{s}^{\epsilon}| ds$$
$$\leq C_{1}\epsilon + C_{2}\epsilon^{k},$$

where  $C_1, C_2$  are positive constants which do not depend on  $(\epsilon, \tau)$ . Hence, we deduce that  $Q_{\tau}^{\epsilon} = \mathrm{Id} + O(\epsilon^q)$  uniformly w.r.t.  $\tau \in [0, 1]$ , where  $q = \min(k, 1)$ . The variation formula (Proposition 6.2.1) provides  $P_{\tau}^{\epsilon} = Q_{\tau}^{\epsilon} W_{\tau}^{\epsilon}$ , where  $W_{\tau}^{\epsilon} \in U(n)$  is the flow of the equation  $\frac{dx(\tau)}{d\tau} = (Q_{\tau}^{\epsilon})^{-1}A(\tau)Q_{\tau}^{\epsilon}x(\tau)$  at time  $\tau$ . By the previous estimate, we have  $(Q_{\tau}^{\epsilon})^{-1}A(\tau)Q_{\tau}^{\epsilon} = A(\tau) + O(\epsilon^q)$  uniformly w.r.t.  $\tau \in [0, 1]$ . By an easy application of Gronwall's Lemma, we get that  $W_{\tau}^{\epsilon} = P_{\tau} + O(\epsilon^q)$  and we can conclude.

#### 6.2.4 Perturbation of an adiabatic trajectory

Consider  $A, B_{\epsilon} \in C^{\infty}([0, 1], u(n))$ . Fix  $\psi_0 \in \mathbb{C}^n$ . Let  $X_{\epsilon}$  be the solution of  $\frac{dX_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)\right)X_{\epsilon}(\tau)$  such that  $X_{\epsilon}(0) = X_0$  and let  $\hat{X}_{\epsilon}$  be the solution of  $\frac{d\hat{X}_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon}A(\tau)\hat{X}_{\epsilon}(\tau)$  such that  $\hat{X}_{\epsilon}(0) = X_0$ , that we call the *adiabatic trajectory associated with* A. The goal of this section is to understand under which conditions on  $B_{\epsilon}(\cdot)$  we have

$$\|X_{\epsilon}(\tau) - \hat{X}_{\epsilon}(\tau)\| \to 0 \tag{T}$$

uniformly with respect to  $\tau \in [0, 1]$ . By the variation formula (Proposition 6.2.1), one can show that if the flow of  $\frac{dx(\tau)}{d\tau} = B_{\epsilon}(\tau)x(\tau), x(\tau) \in \mathbb{C}^n$ , is equal to  $\mathrm{Id} + O(\epsilon^k)$  uniformly w.r.t.  $\tau \in [0, 1]$  with k > 1, then Property (T) is satisfied. However this condition is too conservative for our needs. We restrict our study to the class of perturbations  $(B_{\epsilon})_{\epsilon>0} \in$  $S(\alpha)$  introduced in the Definition 6.1.2. We give below a sufficient condition on  $\alpha$  such that Property (T) is satisfied for every A satisfying Condition (GAP) and every finite sum  $(B_{\epsilon})_{\epsilon>0}$  of elements of  $S(\alpha)$  (Proposition 6.2.9). Based on such a result we then provide a proof of Theorem 6.1.3.

**Lemma 6.2.5.** For every  $\alpha > 0$  and every  $a, h \in C^{\infty}([0,1],\mathbb{R})$ , we have  $\int_{0}^{\tau} a(s)e^{i(\frac{s}{\epsilon^{\alpha+1}}+\frac{h(s)}{\epsilon})}ds = O(\epsilon^{\alpha+1})$  uniformly with respect to  $\tau \in [0,1]$ .

*Proof.* Integrating by parts, for every  $\tau \in [0, 1]$ ,

$$\int_{0}^{\tau} a(s)e^{i(\frac{s}{\epsilon^{\alpha+1}}+\frac{h(s)}{\epsilon})}ds$$
  
=  $i\epsilon^{\alpha+1}\int_{0}^{\tau}e^{i\frac{s}{\epsilon^{\alpha+1}}}\left(a'(s)+i\frac{h'(s)}{\epsilon}a(s)\right)e^{i\frac{h(s)}{\epsilon}}ds$   
+  $\left[-i\epsilon^{\alpha+1}e^{i\frac{s}{\epsilon^{\alpha+1}}}a(s)e^{i\frac{h(s)}{\epsilon}}\right]_{0}^{\tau}$   
=  $-\epsilon^{\alpha}\int_{0}^{\tau}h'(s)a(s)e^{i\frac{s}{\epsilon^{\alpha+1}}}e^{i\frac{h(s)}{\epsilon}}ds + O(\epsilon^{\alpha+1}).$ 

Iterating the integration by parts on the integral term  $\lfloor \frac{1}{\alpha} \rfloor$  more times, we get

$$\int_0^\tau a(s)e^{i(\frac{s}{\epsilon^{\alpha+1}}+\frac{h(s)}{\epsilon})}ds = O(\epsilon^{\alpha+1}).$$

**Definition 6.2.6.** Let  $\alpha > 0$  and  $(B_{\epsilon})_{\epsilon>0}$  be a finite sum of element in  $S(\alpha)$ . For every  $\epsilon > 0$ ,  $P \in C^{\infty}([0,1], U(n))$ , and every diagonal matrix  $\Gamma(\tau) = \operatorname{diag}(\Gamma_j(\tau))_{j=1}^n$  with  $\Gamma_j \in C^{\infty}([0,1], \mathbb{R}), \ j = 1, \ldots, n, \ define$ 

$$M(P,\Gamma,\epsilon)(\tau) = e^{i\frac{\Gamma(\tau)}{\epsilon}}P^*(\tau)B_{\epsilon}(\tau)P(\tau)e^{-i\frac{\Gamma(\tau)}{\epsilon}}, \quad \tau \in [0,1].$$

**Lemma 6.2.7.** Let  $\alpha > 1$ . Consider  $(B_{\epsilon})_{\epsilon>0}$ , P,  $\Gamma$ , and M as in Definition 6.2.6. Then

$$\int_0^\tau M(P,\Gamma,\epsilon)(s)ds = O(\epsilon^\alpha)$$

and

$$\int_0^\tau |M(P,\Gamma,\epsilon)(s)| \left| \int_0^s M(P,\Gamma,\epsilon)(x) dx \right| ds = O(\epsilon^{\alpha-1})$$

uniformly w.r.t.  $\tau \in [0, 1]$ .

*Proof.* Define the following matrix  $C_{\epsilon}(\tau) = \frac{1}{\epsilon} v_{j\ell}(\tau) e^{i(\frac{\beta_{j\ell}\tau}{\epsilon^{\alpha+1}} + \frac{h_{j\ell}(\tau)}{\epsilon})} E_{j\ell}$  for fixed  $j, \ell \in \{1, \ldots, n\}$  where  $E_{j\ell}$  is the matrix whose coefficient  $(j, \ell)$  is equal to 1 and others are equal to 0. By direct computations, denoting  $(P(\tau))_{kq} = p_{kq}(\tau)$ , we get

$$e^{i\frac{\Gamma(s)}{\epsilon}}P^*(s)C_{\epsilon}(s)P(s)e^{-i\frac{\Gamma(s)}{\epsilon}}$$
$$=\frac{v_{j\ell}(\tau)}{\epsilon}\sum_{k,q=1}^n p_{\ell k}(\tau)\bar{p}_{jq}(\tau)e^{\frac{i}{\epsilon}(\Gamma_q(\tau)-\Gamma_k(\tau))}e^{i(\frac{\beta_{j\ell}\tau}{\epsilon^{\alpha+1}}+\frac{h_{j\ell}(\tau)}{\epsilon})}E_{qk}$$

By Lemma 6.2.5, we get for every  $q, k \in \{1, \ldots, n\}$ ,

$$\int_0^\tau v_{j\ell}(s) p_{\ell k}(s) \bar{p}_{jq}(s) e^{\frac{i}{\epsilon} (\Gamma_q(s) - \Gamma_k(s))} e^{i(\frac{\beta_{j\ell}s}{\epsilon^{\alpha+1}} + \frac{h_{j\ell}(s)}{\epsilon})} ds$$

is  $O(\epsilon^{\alpha+1})$ . Hence,

$$\int_0^\tau e^{i\frac{\Gamma(s)}{\epsilon}} P(s) C_{\epsilon}(s) P^*(s) e^{-i\frac{\Gamma(s)}{\epsilon}} ds = O(\epsilon^{\alpha}).$$

We deduce by linearity that the result is also true for  $B_{\epsilon}$ . The second part of the claim follows noticing that  $M(P, \Gamma, \epsilon)(\tau) = O(\frac{1}{\epsilon})$ .

**Lemma 6.2.8.** Let  $\alpha > 1$ . Consider  $(B_{\epsilon})_{\epsilon>0}$ , P,  $\Gamma$ , and M as in Definition 6.2.6. Then the flow of the equation  $\frac{dx(\tau)}{d\tau} = M(P, \Gamma, \epsilon)(\tau)x(\tau)$  is equal to  $Id+O(\epsilon^{\alpha-1})$ , uniformly w.r.t.  $\tau \in [0, 1]$ .

*Proof.* We apply Theorem 6.2.4 using the estimates from Lemma 6.2.7.  $\Box$ 

The next proposition, based on Lemma 6.2.8, shows that under the condition  $\alpha > 1$ , an adiabatic trajectory is robust with respect to perturbations of the dynamics by a term  $(B_{\epsilon})_{\epsilon>0}$  which is a finite sum of elements belonging to  $S(\alpha)$  for  $\epsilon$  small. **Proposition 6.2.9.** Consider  $A \in C^{\infty}([0,1], u(n))$  and  $(B_{\epsilon})_{\epsilon>0}$  be a finite sum of elements in  $S(\alpha)$ . with  $\alpha > 1$ . Assume that Condition (GAP) is satisfied. Select  $\lambda_j \in C^{\infty}([0,1],\mathbb{R}), j = 1, \ldots, n$ , and  $P \in C^{\infty}([0,1], U(n))$  such that, for  $j = 1, \ldots, n$ ,  $\lambda_j(\tau)$  and the *j*-th column of  $P(\tau)$  are, respectively, an eigenvalue of  $iA(\tau)$  and a corresponding eigenvector (the existence of  $C^{\infty}$  eigenpairs being guaranteed by Lemma 6.2.2). Define  $\Lambda(\tau) = diag(\lambda_j(\tau))_{j=1}^n, \tau \in [0,1]$ . Fix  $X_0 \in \mathbb{C}^n$  independent of  $\epsilon$ . Let  $X_{\epsilon}$  be the solution of  $\frac{dX_{\epsilon}(\tau)}{d\tau} = \left(\frac{1}{\epsilon}A(\tau) + B_{\epsilon}(\tau)\right)X_{\epsilon}(\tau)$  such that  $X_{\epsilon}(0) = X_0$ . Set

$$\Upsilon_{\epsilon}(\tau) = P(\tau) \exp\left(\frac{-i}{\epsilon} \int_{0}^{\tau} \Lambda(s) ds\right) \exp\left(\int_{0}^{\tau} D(s) ds\right) P^{*}(0),$$

where D is equal to the diagonal part of  $\frac{dP^*}{d\tau}P$ . Then

$$||X_{\epsilon}(\tau) - \Upsilon_{\epsilon}(\tau)X_{0}|| < c\epsilon^{\min(1,\alpha-1)}$$

for some constant c > 0 independent of  $\tau \in [0, 1]$  and  $\epsilon > 0$ .

*Proof.* Define  $\Gamma(\tau) = \int_0^{\tau} \Lambda(s) ds$  and  $Y_{\epsilon}(\tau) = \exp\left(\frac{i}{\epsilon} \Gamma(\tau)\right) P^*(\tau) X_{\epsilon}(\tau)$ . Then  $Y_{\epsilon}$  satisfies the equation

$$\frac{dY_{\epsilon}(\tau)}{d\tau} = \exp\left(\frac{i}{\epsilon}\Gamma(\tau)\right)\frac{dP^{*}}{d\tau}(\tau)P(\tau)\exp\left(-\frac{i}{\epsilon}\Gamma(\tau)\right)Y_{\epsilon}(\tau)$$

$$+ M(P,\Gamma,\epsilon)(\tau)Y_{\epsilon}(\tau),$$
(6.14)

where  $M(P,\Gamma,\epsilon)$  is defined as in Definition 6.2.6. In order to simplify the notations, set  $D_{\epsilon}(\tau) = \exp\left(\frac{i}{\epsilon}\Gamma(\tau)\right)\frac{dP^{*}}{d\tau}(\tau)P(\tau)\exp\left(-\frac{i}{\epsilon}\Gamma(\tau)\right)$  and denote the flow at time  $\tau$  of the equations  $\frac{dx(\tau)}{d\tau} = M(P,\Gamma,\epsilon)(\tau)x(\tau)$  and  $\frac{dx(\tau)}{d\tau} = (P_{\tau}^{\epsilon})^{-1}D_{\epsilon}(\tau)P_{\tau}^{\epsilon}x(\tau)$  by  $P_{\tau}^{\epsilon}$  and  $W_{\tau}^{\epsilon}$ , respectively. By the variation formula (Proposition 6.2.1), we get that the flow at time  $\tau$ of equation (6.14) is equal to  $Q_{\tau}^{\epsilon} = P_{\tau}^{\epsilon}W_{\tau}^{\epsilon}$ . By Lemma 6.2.8, we have  $P_{\tau}^{\epsilon} = \mathrm{Id} + O(\epsilon^{\alpha-1})$ . Hence  $(P_{\tau}^{\epsilon})^{-1}D_{\epsilon}(\tau)P_{\tau}^{\epsilon} = D_{\epsilon}(\tau) + O(\epsilon^{\alpha-1})$ . Using the gap condition (GAP), we have the estimate  $\int_{0}^{\tau} D_{\epsilon}(s)ds = \int_{0}^{\tau} D(s)ds + O(\epsilon)$  uniformly with respect to  $\tau \in [0, 1]$ . Indeed,  $(D_{\epsilon}(\tau))_{jl} = q_{jl}(\tau)e^{\frac{i}{\epsilon}\int_{0}^{\tau}(\lambda_{j}(s)-\lambda_{l}(s))ds}, j, l \in \{1, \ldots, n\}$ , where  $q_{jl}$  is  $C^{\infty}$ . Hence we get the expected estimation by a direct estimation of the integral of the oscillating term  $e^{\frac{i}{\epsilon}\int_{0}^{\tau}(\lambda_{j}(s)-\lambda_{l}(s))ds}, j, l \in \{1, \ldots, n\}$ . Moreover, since  $D_{\epsilon}$  is bounded with respect to  $\epsilon$ , Theorem 6.2.3 ensures that  $W_{\tau}^{\epsilon} = \exp\left(\int_{0}^{\tau} D(s)ds\right) + O(\epsilon^{\min(1,\alpha-1)})$ . It follows that

$$Q_{\tau}^{\epsilon} = \left( \mathrm{Id} + O(\epsilon^{\alpha - 1}) \right) \left( \exp\left( \int_{0}^{\tau} D(s) ds \right) + O(\epsilon^{\min(1, \alpha - 1)}) \right)$$
$$= \exp\left( \int_{0}^{\tau} D(s) ds \right) + O(\epsilon^{\min(1, \alpha - 1)}).$$

*Proof.* (Proof of Theorem 6.1.3) By an easy application of Theorem 6.2.3, we get the adiabatic estimate

$$\forall \tau \in [0,1], \|\hat{X}_{\epsilon}(\tau) - \Upsilon_{\epsilon}(\tau)X_0\| < C\epsilon$$

where C > 0 is independent of  $\tau, \epsilon$  and  $\Upsilon_{\epsilon}$  is defined as in Proposition 6.2.9. The result is then obtained combining the previous inequality with the estimate of Proposition 6.2.9 by triangular inequality.

#### 6.2.5 Parametric case

**Definition 6.2.10.** Let K be a compact of  $\mathbb{R}^N$ , where  $N \in \mathbb{N}$ . For  $A^{\delta}(\tau) \in u(n)$  whose dependence on  $(\tau, \delta) \in [0, 1] \times K$  is  $C^{\infty}$ , define  $\Lambda^{\delta}(\tau) = diag(\lambda_j^{\delta}(\tau))_{j \in \{1, ..., n\}}$  where  $j \mapsto \lambda_j^{\delta}(\tau)$  is the nondecreasing sequence of eigenvalues of  $iA^{\delta}(\tau)$ . We say that  $A^{\delta}$  satisfies a uniform gap condition if and only if there exists C > 0 such that

$$\forall l \neq j, \forall \delta \in K, \forall \tau \in [0, 1], \ |\lambda_l^{\delta}(\tau) - \lambda_j^{\delta}(\tau)| > C.$$
 (UGAP)

We say that  $A^{\delta}$  has a conical uniform intersection of eigenvalues if and only there exists  $j \in \{1, \ldots, n\}$  such that  $\lambda_j^{\delta}, \lambda_{j+1}^{\delta}$  is separated from the rest of the spectrum for every  $\delta \in K$ , and for every  $\delta \in K$ ,  $l \neq j$   $\lambda_l^{\delta}(\tau) \neq \lambda_{l+1}^{\delta}(\tau)$  for every  $\tau \in [0, 1]$ , and there exists  $\tau_{\delta} \in [0, 1]$  such that for every  $l \neq j$ ,

$$\lambda_{j}^{\delta}(\tau_{\delta}) = \lambda_{j+1}^{\delta}(\tau_{\delta}), \quad \forall \tau \neq \tau_{\delta}, \ \lambda_{j}^{\delta}(\tau) \neq \lambda_{j+1}^{\delta}(\tau), \quad and \quad \frac{d(\lambda_{j+1}^{\delta} - \lambda_{j}^{\delta})}{d\tau}(\tau_{\delta}) \neq 0,$$
(Conical-UGAP)

the associated eigenvectors are  $C^2$  w.r.t  $\tau \in [0, 1]$ , and their derivatives w.r.t.  $\tau$  until order 2 are bounded w.r.t.  $\delta \in K$ .

Using uniform estimates with respect to  $\delta \in K$  in the proof of the propositions 6.2.9, we get the following theorem.

**Theorem 6.2.11.** Let K = [a, b] where  $a, b \in \mathbb{R}$ . Consider  $(B_{\epsilon})_{\epsilon>0} \in S(\alpha)$  with  $\alpha > 1$ . Let  $(A^{\delta}(\tau))_{\delta \in K}$  be a family of matrices in u(n) whose dependence in  $(\tau, \delta) \in [0, 1] \times K$ is  $C^{\infty}$ . Assume that  $A^{\delta}(\tau)$  satisfies (UGAP). Fix  $X_0 \in \mathbb{C}^n$  independent of  $\epsilon$ . Let  $X_{\epsilon}(\delta, \tau)$ be the solution of  $\frac{dX_{\epsilon}(\tau)}{d\tau} = (\frac{1}{\epsilon}A^{\delta}(\tau) + \delta B_{\epsilon}(\tau)) X_{\epsilon}(\tau)$  such that  $X_{\epsilon}(0) = X_0$  and  $\hat{X}_{\epsilon}(\delta, \tau)$ be the solution of  $\frac{d\hat{X}_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon}A^{\delta}(\tau)\hat{X}_{\epsilon}(\tau)$  such that  $\hat{X}_{\epsilon}(0) = X_0$ . Then there exists c > 0independent of  $\tau, \delta, \epsilon$  such that for every  $(\tau, \delta) \in [0, 1] \times K$ ,

$$||X_{\epsilon}(\delta,\tau) - \hat{X}_{\epsilon}(\delta,\tau)|| \le c\epsilon^{\min(1,\alpha-1)}.$$

**Theorem 6.2.12.** Let K be a compact of  $\mathbb{R}^{N^2}$  and  $\delta = (\delta_{k,q})_{k,q\in\{1...N\}} \in K$ . Consider a family  $(B^{\delta}_{\epsilon})_{\epsilon>0}$  of matrices in u(n) such that, for every  $\epsilon > 0$ ,  $\tau \in [0,1]$  and  $\delta \in K$ ,  $B^{\delta}_{\epsilon}(\tau) = \sum_{k,q=1}^{N} \delta_{k,q} B^{k,q}_{\epsilon}(\tau)$ , where for every  $k, q \in \{1...N\}$ ,  $(B^{k,q}_{\epsilon})_{\epsilon>0} \in S(\alpha)$  with  $\alpha > 1$ . Let  $(A^{\delta}(\tau))_{\delta \in K}$  be a family of matrices in u(n) whose dependence in  $(\tau, \delta) \in [0, 1] \times K$ is  $C^{\infty}$ . Assume that  $A(\tau)$  satisfies (UGAP). Fix  $X_0 \in \mathbb{C}^n$ . Let  $X_{\epsilon}(\delta, \tau)$  be the solution of  $\frac{dX_{\epsilon}(\tau)}{d\tau} = (\frac{1}{\epsilon}A^{\delta}(\tau) + B^{\delta}_{\epsilon}(\tau)) X_{\epsilon}(\tau)$  such that  $X_{\epsilon}(0) = X_0$  and  $\hat{X}_{\epsilon}(\delta, \tau)$  be the solution of  $\frac{d\hat{X}_{\epsilon}(\tau)}{d\tau} = \frac{1}{\epsilon}A^{\delta}(\tau)\hat{X}_{\epsilon}(\tau)$  such that  $\hat{X}_{\epsilon}(0) = X_0$ . Then there exists c > 0 independent of  $\tau, \delta, \epsilon$ such that for every  $(\tau, \delta) \in [0, 1] \times K$ ,

$$||X_{\epsilon}(\delta,\tau) - \hat{X}_{\epsilon}(\delta,\tau)|| \le c\epsilon^{\min(1,\alpha-1)}.$$

**Remark 6.2.13.** In the case  $A^{\delta}(\tau)$  satisfies (Conical-UGAP), the theorems 6.2.11 and 6.2.12 remain valid replacing the error of order  $\epsilon^{\min(1,\alpha-1)}$  by  $\epsilon^{\min(\frac{1}{2},\alpha-1)}$ .

#### 6.3 Control of two-level systems

#### 6.3.1 Control strategy for two-level systems and simulations

Let  $v, \varphi \in C^{\infty}([0,1], \mathbb{R})$  be such that v(0) = v(1) = 0,  $\varphi(0) = 0$ ,  $\varphi'(0)\varphi'(1) < 0$ , and  $v(\tau) \neq 0$  for  $\tau \in (0,1)$ . Let  $e_1 = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$  and  $e_2 = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$ . By adiabatic approximation, the solution  $x_{\epsilon} : [0,1] \to \mathbb{C}^2$  of

$$\frac{dx_{\epsilon}(\tau)}{d\tau} = -\frac{i}{\epsilon} \begin{pmatrix} -\varphi'(\tau)/2 & v(\tau) \\ v(\tau) & \varphi'(\tau)/2 \end{pmatrix} x_{\epsilon}(\tau), \quad x_{\epsilon}(0) = e_1,$$

satisfies  $||x_{\epsilon}(1) - e^{i\xi_{\epsilon}}e_2|| \leq C\epsilon$  where C > 0 is independent of  $\epsilon$  and  $\xi_{\epsilon} \in \mathbb{R}$ . Consider the solution  $\psi_{\epsilon}(\tau)$  of Equation (6.7) such that  $\psi_0 = e_1$  and corresponding to the controls  $(v, \varphi)$ . Applying Theorem 6.1.5, we have

$$\|\psi_{\epsilon}(1) - e^{i\theta_{\epsilon}}e_2\| \le c\epsilon^{\min(1,\alpha-1)}$$

where c > 0 is independent of  $\epsilon$  and  $\theta_{\epsilon} \in \mathbb{R}$ .

On Figure 6.1, we have plotted the projection of the wave function onto  $e_2$  for

$$v(\tau) = \sin(\pi\tau), \quad \varphi(\tau) = -\frac{1}{\pi}\sin(\pi\tau),$$

with  $\epsilon = 0.01$ ,  $\alpha = 1.5$  and E = 1 in the fast time scale, that is, as a function of  $\tau \in [0, 1]$ . The total time needed by our control strategy in the variable  $t = \frac{\tau}{\epsilon^{\alpha+1}}$  is  $T = \frac{1}{\epsilon^{\alpha+1}}$ . On Figure 6.2, we have plotted the norm of the difference between  $\psi_{\epsilon}$  and the solution of Equation (6.10) with the same initial condition and parameters as a function of  $\tau \in [0, 1]$ .



Figure 6.1 – Fidelity  $|\langle \psi_{\epsilon}(\tau), e_2 \rangle|^2$  as a function of the time variable  $\tau \in$ [0, 1] with  $\epsilon = 0.01$ ,  $\alpha = 1.5$ , and E = 1.



Figure 6.2 – Squared norm of the difference between  $\psi_{\epsilon}$  and  $\hat{\psi}_{\epsilon}$  as a function of the time variable  $\tau \in [0, 1]$  with  $\epsilon = 0.01$ ,  $\alpha = 1.5$ , and E = 1.

### 6.3.2 Robustness of the control strategy with respect to amplitude of control inhomogeneities

Let U be a connected open set of  $\mathbb{R}$  containing 0.

**Theorem 6.3.1.** Let  $E \in \mathbb{R} \setminus \{0\}$ . The equation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E & \delta u\\ \delta u & -E \end{pmatrix}\psi$$
(6.15)

is approximately ensemble controllable between the eigenstates of  $H_0 = \begin{pmatrix} E & 0 \\ 0 & -E \end{pmatrix}$  uniformly with respect to  $\delta \in [a, b] \subset (0, +\infty)$  and  $u \in U$ .

Proof. Let  $\alpha > 1$  and  $v, \varphi \in C^{\infty}([0,1], \mathbb{R})$  be such that v(0) = v(1) = 0,  $\varphi(0) = 0$ ,  $\varphi'(0)\varphi'(1) < 0$ , and  $v(\tau) \neq 0$  for  $\tau \in (0,1)$ . Let us consider  $\tau = \epsilon^{\alpha+1}t \in [0,1]$  and  $u_{\epsilon}(\tau) = \frac{2}{\epsilon}v(\tau)\cos(\frac{2E\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon}\varphi(\tau))$ . For each  $\delta \in [a,b]$ , let  $\psi_{\epsilon}(\delta,\tau)$  be the solution of

$$i\frac{d\psi(\delta,\tau)}{d\tau} = \begin{pmatrix} \frac{E}{\epsilon^{\alpha+1}} & \delta u_{\epsilon}(\tau) \\ \delta u_{\epsilon}(\tau) & -\frac{E}{\epsilon^{\alpha+1}} \end{pmatrix} \psi(\delta,\tau), \quad \psi_{\epsilon}(\delta,0) = e_1.$$

Apply the unitary transformation  $X_{\epsilon}(\delta, \tau) = U_{\epsilon}(\tau)\psi_{\epsilon}(\delta, \tau)$  where

$$U_{\epsilon}(\tau) = \begin{pmatrix} e^{-i\left(\frac{E\tau}{\epsilon^{\alpha+1}} + \frac{\varphi(\tau)}{2\epsilon}\right)} & 0\\ 0 & e^{i\left(\frac{E\tau}{\epsilon^{\alpha+1}} + \frac{\varphi(\tau)}{2\epsilon}\right)} \end{pmatrix}.$$



Figure 6.3 – Fidelity  $|\langle \psi_{\epsilon}(\delta, 1), e_2 \rangle|^2$ as a function of the amplitude inhomogeneity  $\delta$ , with  $\epsilon = 0.01$ ,  $\alpha = 1.5$ , and E = 1.



Figure 6.4 – Fidelity  $|\langle \psi_{\epsilon}(E, 1), e_2 \rangle|^2$ as a function of the drift term E, with  $\epsilon = 0.01$  and  $\alpha = 1.5$ .

Then  $X_{\epsilon}(\delta, \tau)$  satisfies

$$\frac{dX_{\epsilon}(\delta,\tau)}{d\tau} = \left(\frac{1}{\epsilon}A^{\delta}(\tau) + \delta B_{\epsilon}(\tau)\right)X_{\epsilon}(\delta,\tau), \quad X_{\epsilon}(\delta,0) = e_1,$$

where

$$A^{\delta}(\tau) = -i \begin{pmatrix} -\varphi'(\tau)/2 & \delta v(\tau) \\ \delta v(\tau) & \varphi'(\tau)/2 \end{pmatrix},$$

and

$$B_{\epsilon}(\tau) = \frac{-i}{\epsilon} \begin{pmatrix} 0 & v(\tau)e^{i(\frac{4E\tau}{\epsilon\alpha+1} + \frac{2\varphi(\tau)}{\epsilon})} \\ v(\tau)e^{-i(\frac{4E\tau}{\epsilon\alpha+1} + \frac{2\varphi(\tau)}{\epsilon})} & 0 \end{pmatrix}.$$

By our choice of v and  $\varphi$ ,  $A^{\delta}(\tau)$  is  $C^{\infty}$  w.r.t  $(\tau, \delta)$  and satisfies (UGAP). Applying Theorem 6.2.11, we get that  $\|\psi_{\epsilon}(\delta, 1) - e^{i\theta_{\delta,\epsilon}}e_2\| \leq C\epsilon^{\min(1,\alpha-1)}$  where C > 0 is independent of  $\delta, \epsilon$  and  $\theta_{\delta,\epsilon} \in \mathbb{R}$ . The result follows.

Consider the same  $(v, \varphi)$  as those chosen in Section 6.3.1. For each  $\delta \in [0, 1]$ , let  $\psi_{\epsilon}(\delta, \tau)$  be the solution of Equation (6.15) with initial condition  $\psi(\delta, 0) = e_1$  and E = 1. We have plotted on Figure 6.3 the *fidelity*, that is  $|\langle \psi_{\epsilon}(\delta, 1), e_2 \rangle|^2$  for a dispersion  $\delta$  of the amplitude of the control in [0, 1]. On every sub-interval [a, b] of [0, 1] with a > 0, the fidelity converges uniformly to the constant function  $\delta \mapsto 1$  when  $\epsilon \to 0$ .

Let now  $\psi_{\epsilon}(E,\tau)$  be the solution of the equation

$$i\frac{d\psi_{\epsilon}(E,\tau)}{d\tau} = \begin{pmatrix} \frac{E}{\epsilon^{\alpha+1}} & u_{\epsilon}(\tau) \\ u_{\epsilon}(\tau) & -\frac{E}{\epsilon^{\alpha+1}} \end{pmatrix} \psi_{\epsilon}(E,\tau)$$
(6.16)
where  $u_{\epsilon}(\tau) = \frac{2}{\epsilon}v(\tau)\cos(\frac{2\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon}\varphi(\tau))$ , with initial condition  $\psi(E,0) = e_1$  for every  $E \in [\frac{1}{2}, \frac{3}{2}]$ . We have plotted on Figure 6.4 the fidelity for a dispersion of E in  $[\frac{1}{2}, \frac{3}{2}]$ . As already mentioned in the introduction of the thesis, numerical simulations suggest that our method of control is not robust w.r.t. inhomogeneities of the resonance frequency E.

## 6.4 Control of STIRAP Process

Denote the canonical basis of  $\mathbb{R}^3$  by  $(e_1, e_2, e_3)$ . We prove the following theorem.

**Theorem 6.4.1.** Let  $E_1, E_2, E_3 \in \mathbb{R}$  such that  $E_1 < E_2 < E_3$  and  $|E_k - E_q| \neq |E_j - E_l|$ , for every  $k, q, j, l \in \{1 ... 3\}$  such that  $(k, q) \notin \{(j, l), (l, j)\}$ . The equation

$$i\frac{d\psi}{dt} = \begin{pmatrix} E_1 & \delta_1 u & 0\\ \delta_1 u & E_2 & \delta_2 u\\ 0 & \delta_2 u & E_3 \end{pmatrix} \psi$$
(6.17)

is ensemble controllable between eigenstates of  $H_0 = \begin{pmatrix} E_1 & 0 & 0 \\ 0 & E_2 & 0 \\ 0 & 0 & E_3 \end{pmatrix}$  uniformly with respect to  $\delta_1 \in [a_1, b_1] \subset (0, +\infty)$  and  $\delta_2 \in [a_2, b_2] \subset (0, +\infty)$  and  $u \in U$ .

*Proof.* For  $\delta_1, \delta_2 \in \mathbb{R}$ , set  $\delta = (\delta_1, \delta_2) \in \mathbb{R}^2$ . We prove the result for a transfer of populations between  $e_1$  and  $e_3$ . Let  $\alpha > 1$  and  $v_{12}, v_{23}, (\varphi_j)_{j \in \{1...3\}} \in C^{\infty}([0, 1], \mathbb{R})$ . Define, for  $\epsilon > 0$ and  $t \in [0, \frac{1}{\epsilon^{\alpha+1}}]$ ,

$$u_{\epsilon}(t) = 2\epsilon^{\alpha} v_{12}(\epsilon^{\alpha+1}t) \cos((E_1 - E_2)t + \frac{1}{\epsilon} \left(\varphi_1(\epsilon^{\alpha+1}t) - \varphi_2(\epsilon^{\alpha+1}t)\right)) + 2\epsilon^{\alpha} v_{23}(\epsilon^{\alpha+1}t) \cos((E_2 - E_3)t + \frac{1}{\epsilon} \left(\varphi_2(\epsilon^{\alpha+1}t) - \varphi_3(\epsilon^{\alpha+1}t)\right))$$

Assume that for every  $\tau \in [0,1]$  and  $j \in \{1,\ldots,3\}$ ,  $\varphi_j(\tau) = \tilde{E}_j \tau$ , where  $(\tilde{E}_j)_{j \in \{1\ldots,3\}}$  are such that  $\tilde{E}_1 < \tilde{E}_2 < \tilde{E}_3$ . Let us consider  $\tau = \epsilon^{\alpha+1}t \in [0,1]$  and, by a slight abuse of notations,

$$u_{\epsilon}(\tau) = \frac{2}{\epsilon} v_{12}(\tau) \cos((E_1 - E_2) \frac{\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon} (\varphi_1(\tau) - \varphi_2(\tau))) + \frac{2}{\epsilon} v_{23}(\tau) \cos((E_2 - E_3) \frac{\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon} (\varphi_2(\tau) - \varphi_3(\tau))).$$

For each  $\delta \in [a, b]$ , let  $\psi_{\epsilon}(\delta, \tau)$  be the solution of

$$i\frac{d\psi(\delta,\tau)}{dt} = \begin{pmatrix} \frac{E_1}{\epsilon^{\alpha+1}} & \delta_1 u_\epsilon(\tau) & 0\\ \delta_1 u_\epsilon(\tau) & \frac{E_2}{\epsilon^{\alpha+1}} & \delta_2 u_\epsilon(\tau)\\ 0 & \delta_2 u_\epsilon(\tau) & \frac{E_3}{\epsilon^{\alpha+1}} \end{pmatrix} \psi(\delta,\tau), \quad \psi_\epsilon(\delta,0) = e_1.$$
(6.18)

Apply the unitary transformation  $X_{\epsilon}(\delta, \tau) = U_{\epsilon}(\tau)\psi_{\epsilon}(\delta, \tau)$  where

$$U_{\epsilon}(\tau) = \begin{pmatrix} e^{i\left(\frac{E_{1}\tau}{\epsilon^{\alpha+1}} + \frac{\varphi_{1}(\tau)}{2\epsilon}\right)} & 0 & 0\\ 0 & e^{i\left(\frac{E_{2}\tau}{\epsilon^{\alpha+1}} + \frac{\varphi_{2}(\tau)}{2\epsilon}\right)} & 0\\ 0 & 0 & e^{i\left(\frac{E_{2}\tau}{\epsilon^{\alpha+1}} + \frac{\varphi_{3}(\tau)}{2\epsilon}\right)} \end{pmatrix}.$$

Then  $X_{\epsilon}(\delta, \tau)$  satisfies

$$\frac{dX_{\epsilon}(\delta,\tau)}{d\tau} = \left(\frac{1}{\epsilon}A^{\delta}(\tau) + B^{\delta}_{\epsilon}(\tau)\right)X_{\epsilon}(\delta,\tau), \quad X_{\epsilon}(\delta,0) = e_1,$$

where for  $\tau \in [0, 1]$ ,  $A^{\delta}(\tau) = \begin{pmatrix} \tilde{E}_1 & \delta_1 v_{12}(\tau) & 0\\ \delta_1 v_{12}(\tau) & \tilde{E}_2 & \delta_2 v_{23}(\tau)\\ 0 & \delta_2 v_{23}(\tau) & \tilde{E}_3 \end{pmatrix}$  is  $C^{\infty}$  w.r.t  $(\tau, \delta)$ , and  $B^{\delta}_{\epsilon}(\tau)$  has the form required by Theorem 6.2.12.

Define, for 
$$(x, y) \in \mathbb{R}^2$$
,  $\tilde{A}^{\delta}(x, y) = \begin{pmatrix} \tilde{E}_1 & \delta_1 x & 0\\ \delta_1 x & \tilde{E}_2 & \delta_2 y\\ 0 & \delta_2 y & \tilde{E}_3 \end{pmatrix}$ , and its eigenvalues by  $\Lambda_j^{\delta}(x, y)$ 

for  $j \in \{1...2\}$ . Notice that we have  $A^{\delta}(\tau) = \tilde{A}^{\delta}(v_{12}(\tau), v_{23}(\tau))$ . Design a control path  $(v_{12}(\tau), v_{23}(\tau))_{\tau \in [0,1]}$  passing through the conical intersections of  $\tilde{A}^{\delta}$  between the levels  $\Lambda_1^{\delta}$  and  $\Lambda_2^{\delta}$ , then those between the levels  $\Lambda_2^{\delta}$  and  $\Lambda_3^{\delta}$ , for every  $\delta \in [a_1, b_1] \times [a_2, b_2]$  as it is made in Section 3.7.2. For every  $\delta \in [a_1, b_1] \times [a_2, b_2]$ , at a time  $\tau_{\delta}$  where the control path passes at a conical intersection,  $A^{\delta}(\tau)$  satisfies (Conical-UGAP) locally w.r.t.  $\tau$ . Applying Theorem 6.2.12, we get that  $\|\psi_{\epsilon}(\delta, 1) - e^{i\theta_{\delta,\epsilon}}e_3\| \leq C\epsilon^{\min(\frac{1}{2},\alpha-1)}$  where C > 0 is independent of  $\delta, \epsilon$  and  $\theta_{\delta,\epsilon} \in \mathbb{R}$ . Following the same steps, we can get transfers from  $e_j$  to  $e_k$ , for every  $j, k \in \{1...3\}$ . The result follows.

Set  $E_1 = 2, E_2 = 5, E_3 = 6$ . On Figure 6.5, we have illustrated the robustness of our control strategy for Equation 6.17 with respect to the parameter  $\delta_1$  defined in Theorem 6.4.1, assuming that  $\delta_2 = 1$ , for a transfer from state 1 to state 3, with a control

$$u_{\epsilon}(\tau) = \frac{2}{\epsilon} v_{12}(\tau) \cos((E_1 - E_2) \frac{\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon} (\varphi_1(\tau) - \varphi_2(\tau))) + \frac{2}{\epsilon} v_{23}(\tau) \cos((E_2 - E_3) \frac{\tau}{\epsilon^{\alpha+1}} + \frac{1}{\epsilon} (\varphi_2(\tau) - \varphi_3(\tau))),$$

where  $\tau$  is the normalized time. We have chosen piecewise affine functions  $v_{12}$  and  $v_{23}$ as on Figure 6.6, and linear functions  $(\varphi_k)_{k \in \{1...3\}}$  as in the proof of Theorem 6.4.1 with  $\tilde{E}_1 = 0, \tilde{E}_2 = 1, \tilde{E}_3 = 3$ , and we have considered the solution  $\psi_{\epsilon}(\delta, \tau)$  of Equation 6.17.

On Figure 6.7, we have illustrated the failure of our control strategy with fixed parameters  $\delta_1 = \delta_2 = 1$  and  $0 < \alpha < 1$ . On Figure 6.8, we have illustrated the failure of our control strategy with fixed parameters  $\delta_1 = \delta_2 = 1$ ,  $\alpha > 1$  and  $|E_1 - E_2| = |E_2 - E_3|$ .



Figure 6.5 – Time-evolution of  $|\langle \psi_{\epsilon}(\delta, \tau), e_3 \rangle|^2$  where  $\tau$  is the renormalized time, with  $\delta_1 = 0.5$  in blue,  $\delta_1 = 1$  in orange,  $\delta_1 = 1.5$  in green, with  $\epsilon = 0.02$  and  $\alpha = 1.5$ .



Figure 6.6 – Time-evolution of  $v_{12}(\tau)$  in blue, and of  $v_{23}(\tau)$  in orange



Figure 6.7 – Time-evolution of  $|\langle \psi_{\epsilon}(\delta, \tau), e_3 \rangle|^2$  where  $\tau$  is the renormalized time, with  $\delta_1 = \delta_2 = 1$ ,  $\epsilon = 0.02$  and  $\alpha = 0.5$ .



Figure 6.8 – Time-evolution of  $|\langle \psi_{\epsilon}(\delta, \tau), e_3 \rangle|^2$  where  $\tau$  is the renormalized time, with  $\delta_1 = \delta_2 = 1$ ,  $\epsilon = 0.02$  and  $\alpha = 1.5$ , in the resonant case  $|E_1 - E_2| = |E_2 - E_3|$ .

- R. Abraham and J. Robbin. *Transversal mappings and flows*. An appendix by Al Kelley. W. A. Benjamin, Inc., New York-Amsterdam, 1967. (Cited on page 63.)
- [2] R. Adami and U. Boscain. Controllability of the schrödinger equation via intersection of eigenvalues. In *Proceedings of the 44th IEEE Conference on Decision and Control*, pages 1080–1085, Dec 2005. (Cited on page 71.)
- [3] A. Agrachev, D. Barilari, and U. Boscain. A Comprehensive Introduction to Sub-Riemannian Geometry. 11 2019. (Cited on page 119.)
- [4] A. Agrachev, Y. Baryshnikov, and A. Sarychev. Ensemble controllability by Lie algebraic methods. *ESAIM Control Optim. Calc. Var.*, 22(4):921–938, 2016. (Cited on pages 7 and 30.)
- [5] A. A. Agrachev and Y. L. Sachkov. Control theory from the geometric viewpoint, volume 87 of Encyclopaedia of Mathematical Sciences. Springer-Verlag, Berlin, 2004. Control Theory and Optimization, II. (Cited on pages 117 and 118.)
- [6] F. Albertini and D. D'Alessandro. Notions of controllability for bilinear multilevel quantum systems. *IEEE Transactions on Automatic Control*, 48(8):1399–1403, Aug 2003. (Cited on pages 1 and 25.)
- [7] F. Albertini and D. D'Alessandro. Notions of controllability for bilinear multilevel quantum systems. *IEEE Trans. Automat. Control*, 48(8):1399–1403, 2003. (Cited on pages 21 and 45.)
- [8] G. P. Alexander, B. G.-g. Chen, E. A. Matsumoto, and R. D. Kamien. Colloquium: Disclination loops, point defects, and all that in nematic liquid crystals. *Reviews of Modern Physics*, 84:497–514, Apr. 2012. (Cited on pages xiii, xiv, 19, and 42.)
- [9] C. Altafini. Controllability of quantum mechanical systems by root space decomposition of su(n). J. Math. Phys., 43(5):2051–2062, 2002. (Cited on pages 1 and 25.)
- [10] N. Augier, U. Boscain, and M. Sigalotti. Adiabatic Ensemble Control of a Continuum of Quantum Systems. SIAM J. Control Optim., 56(6):4045–4068, 2018. (Cited on pages 13, 21, 23, 36, 45, 46, 49, 102, 110, and 112.)

- [11] J. M. Ball, J. E. Marsden, and M. Slemrod. Controllability for distributed bilinear systems. SIAM J. Control Optim., 20(4):575–597, 1982. (Cited on pages 1 and 25.)
- [12] K. Beauchard and J.-M. Coron. Controllability of a quantum particle in a moving potential well. J. Funct. Anal., 232(2):328–389, 2006. (Cited on pages 21 and 45.)
- [13] K. Beauchard, J.-M. Coron, and P. Rouchon. Controllability issues for continuousspectrum systems and ensemble controllability of Bloch equations. *Comm. Math. Phys.*, 296(2):525–557, 2010. (Cited on pages 1, 7, 25, 30, and 31.)
- [14] M. Belhadj, J. Salomon, and G. Turinici. Ensemble controllability and discrimination of perturbed bilinear control systems on connected, simple, compact Lie groups. *Eur. J. Control*, 22:23–29, 2015. (Cited on pages 7 and 31.)
- [15] A. M. Bloch, R. W. Brockett, and C. Rangan. Finite controllability of infinitedimensional quantum systems. *IEEE Trans. Automat. Control*, 55(8):1797–1805, 2010. (Cited on page 70.)
- [16] M. Born and V. Fock. Beweis des adiabatensatzes. Zeitschrift für Physik, 51(3):165–180, Mar 1928. (Cited on pages 2 and 26.)
- [17] U. Boscain, M. Caponigro, T. Chambrion, and M. Sigalotti. A weak spectral condition for the controllability of the bilinear Schrödinger equation with application to the control of a rotating planar molecule. *Comm. Math. Phys.*, 311(2):423–455, 2012. (Cited on pages 1, 21, 25, and 45.)
- [18] U. Boscain, F. C. Chittaro, P. Mason, R. Pacqueau, and M. Sigalotti. Motion planning in quantum control via intersection of eigenvalues. In 49th IEEE Conference on Decision and Control (CDC), pages 3028–3033, Dec 2010. (Cited on pages 23 and 46.)
- [19] U. Boscain, J.-P. Gauthier, F. Rossi, and M. Sigalotti. Approximate Controllability, Exact Controllability, and Conical Eigenvalue Intersections for Quantum Mechanical Systems. *Communications in Mathematical Physics*, 333(3):1225–1239, Feb. 2015. (Cited on pages 5, 6, and 29.)
- [20] U. Boscain, P. Mason, G. Panati, and M. Sigalotti. On the control of spin-boson systems. J. Math. Phys., 56(9):092101, 15, 2015. (Cited on page 70.)
- [21] U. Boscain, L. Sacchelli, and M. Sigalotti. Generic singularities of line fields on 2D manifolds. *Differential Geom. Appl.*, 49:326–350, 2016. (Cited on pages 18, 42, and 134.)
- [22] U. V. Boscain, F. Chittaro, P. Mason, and M. Sigalotti. Adiabatic control of the Schrödinger equation via conical intersections of the eigenvalues. *IEEE Trans. Automat. Control*, 57(8):1970–1983, 2012. (Cited on pages iii, iv, 1, 5, 6, 15, 18, 21, 23, 25, 29, 30, 38, 39, 41, 45, 46, 54, 70, 82, 108, 121, 122, 123, 126, 128, 138, and 139.)

- [23] F. Bullo and A. D. Lewis. Geometric control of mechanical systems, volume 49 of Texts in Applied Mathematics. Springer-Verlag, New York, 2005. (Cited on pages 22 and 45.)
- [24] T. Chambrion. Periodic excitations of bilinear quantum systems. Automatica J. IFAC, 48(9):2040–2046, 2012. (Cited on pages 22 and 45.)
- [25] T. Chambrion, P. Mason, M. Sigalotti, and U. Boscain. Controllability of the discretespectrum Schrödinger equation driven by an external field. Annales de l'I.H.P. Analyse non linéaire, 26(1):329–349, 2009. (Cited on pages 1 and 25.)
- [26] S. Chandrasekhar. *Liquid Crystals*. Cambridge University Press, 2 edition, 1992. (Cited on pages 18 and 42.)
- [27] S. Chelkowski, A. D. Bandrauk, and P. B. Corkum. Efficient molecular dissociation by a chirped ultrashort infrared laser pulse. *Phys. Rev. Lett.*, 65:2355–2358, Nov 1990. (Cited on pages 2, 26, and 56.)
- [28] C. Chen, D. Dong, R. Long, I. R. Petersen, and H. A. Rabitz. Sampling-based learning control of inhomogeneous quantum ensembles. *Phys. Rev. A*, 89:023402, Feb 2014. (Cited on pages 7 and 31.)
- [29] F. C. Chittaro and J.-P. Gauthier. Asymptotic ensemble stabilizability of the Bloch equation. Systems Control Lett., 113:36–44, 2018. (Cited on pages 7 and 31.)
- [30] F. C. Chittaro and P. Mason. Approximate controllability by adiabatic methods of the schrödinger equation with nonlinear hamiltonian. In 2015 54th IEEE Conference on Decision and Control (CDC), pages 7771–7776, Dec 2015. (Cited on pages 55 and 68.)
- [31] F. C. Chittaro and P. Mason. Approximate controllability via adiabatic techniques for the three-inputs controlled Schrödinger equation. SIAM J. Control Optim., 55(6):4202–4226, 2017. (Cited on pages 6, 15, 30, 39, 102, and 123.)
- [32] Y. Colin de Verdière. The level crossing problem in semi-classical analysis. I. The symmetric case. In Proceedings of the International Conference in Honor of Frédéric Pham (Nice, 2002), volume 53, pages 1023–1054, 2003. (Cited on pages 5 and 28.)
- [33] Y. Colin de Verdière. The level crossing problem in semi-classical analysis. II. The Hermitian case. Ann. Inst. Fourier (Grenoble), 54(5):1423–1441, xv, xx–xxi, 2004. (Cited on pages 5 and 28.)
- [34] D. D'Alessandro and M. Dahleh. Optimal control of two-level quantum systems. *IEEE Trans. Automat. Control*, 46(6):866–876, 2001. (Cited on pages 1 and 25.)
- [35] G. Dirr. Ensemble controllability of bilinear systems. Oberwolfach Rep., 9(1):661–732, 2012. (Cited on pages 7 and 31.)

- [36] F. Dumortier, J. Llibre, and J. Artés. Qualitative theory of planar differential systems. Qualitative Theory of Planar Differential Systems, 06 2007. (Cited on page 128.)
- [37] K. Fujii. Introduction to the rotating wave approximation (RWA): Two coherent oscillations. Journal of Modern Physics, 8(12):2042, 2017. (Cited on pages 22 and 45.)
- [38] U. Gaubatz, P. Rudecki, S. Schiemann, and K. Bergmann. Population transfer between molecular vibrational levels by stimulated raman scattering with partially overlapping laser fields. a new concept and experimental results. *The Journal of Chemical Physics*, 92(9):5363–5376, 1990. (Cited on pages 2 and 26.)
- [39] S. J. Glaser, U. Boscain, T. Calarco, C. P. Koch, W. Köckenberger, R. Kosloff, I. Kuprov, B. Luy, S. Schirmer, T. Schulte-Herbrüggen, D. Sugny, and F. K. Wilhelm. Training schrödinger's cat: quantum optimal control. *The European Physical Journal* D, 69(12):279, Dec 2015. (Cited on pages 7 and 30.)
- [40] S. J. Glaser, T. Schulte-Herbrüggen, M. Sieveking, O. Schedletzky, N. C. Nielsen, O. W. Sørensen, and C. Griesinger. Unitary control in quantum ensembles: Maximizing signal intensity in coherent spectroscopy. *Science*, 280(5362):421–424, 1998. (Cited on pages 7 and 30.)
- [41] M. Golubitsky and V. Guillemin. Stable mappings and their singularities. Springer-Verlag, New York-Heidelberg, 1973. Graduate Texts in Mathematics, Vol. 14. (Cited on pages 83, 84, 86, 87, 96, 99, 100, and 124.)
- [42] M. Goresky and R. MacPherson. Stratified Morse Theory. Springer Berlin Heidelberg, Berlin, Heidelberg, 1988. (Cited on pages 63, 83, 84, 86, 87, 96, and 100.)
- [43] S. Guérin. Complete dissociation by chirped laser pulses designed by adiabatic floquet analysis. *Phys. Rev. A*, 56:1458–1462, Aug 1997. (Cited on pages 2 and 26.)
- [44] S. Guérin and H. Jauslin. Two-laser multiphoton adiabatic passage in the frame of the Floquet theory. Applications to (1+1) and (2+1) STIRAP. The European Physical Journal D Atomic, Molecular, Optical and Plasma Physics, 2(2):99–113, Jun 1998. (Cited on pages 22 and 45.)
- [45] S. Guérin, R. G. Unanyan, L. P. Yatsenko, and H. R. Jauslin. Floquet perturbative analysis for STIRAP beyond the rotating wave approximation. *Opt. Express*, 4(2):84– 90, Jan 1999. (Cited on pages 22 and 45.)
- [46] S. Guérin and H. R. Jauslin. Control of Quantum Dynamics by Laser Pulses: Adiabatic Floquet Theory, chapter 3, pages 147–267. John Wiley & Sons, Ltd, 2003. (Cited on pages 5 and 29.)
- [47] G. A. Hagedorn. Adiabatic expansions near eigenvalue crossings. Ann. Physics, 196(2):278–295, 1989. (Cited on pages 2 and 26.)

- [48] G. A. Hagedorn. Classification and normal forms for quantum mechanical eigenvalue crossings. Astérisque, (210):7, 115–134, 1992. Méthodes semi-classiques, Vol. 2 (Nantes, 1991). (Cited on pages 128 and 129.)
- [49] G. A. Hagedorn and A. Joye. Molecular propagation through small avoided crossings of electron energy levels. *Rev. Math. Phys.*, 11(1):41–101, 1999. (Cited on page 128.)
- [50] U. Helmke and M. Schönlein. Uniform ensemble controllability for one-parameter families of time-invariant linear systems. Systems Control Lett., 71:69–77, 2014. (Cited on pages 7 and 30.)
- [51] E. K. Irish. Generalized rotating-wave approximation for arbitrarily large coupling. *Phys. Rev. Lett.*, 99:173601, Oct 2007. (Cited on pages 22 and 45.)
- [52] S. Jansen, M.-B. Ruskai, and R. Seiler. Bounds for the adiabatic approximation with applications to quantum computation. J. Math. Phys., 48(10):102111, 15, 2007. (Cited on page 55.)
- [53] H. Jauslin, S. Guérin, and S. Thomas. Quantum averaging for driven systems with resonances. *Physica A: Statistical Mechanics and its Applications*, 279(1):432 442, 2000. (Cited on pages 2 and 26.)
- [54] A. Joye. General adiabatic evolution with a gap condition. *Comm. Math. Phys.*, 275(1):139–162, 2007. (Cited on pages 2 and 26.)
- [55] T. Kato. On the adiabatic theorem of quantum mechanics. Journal of the Physical Society of Japan, 5(6):435–439, 1950. (Cited on pages 2 and 26.)
- [56] M. Keyl, R. Zeier, and T. Schulte-Herbrüggen. Controlling several atoms in a cavity. New Journal of Physics, 16(6):065010, jun 2014. (Cited on page 70.)
- [57] J. Kurzweil and J. Jarník. Limit processes in ordinary differential equations. Z. Angew. Math. Phys., 38(2):241–256, 1987. (Cited on page 154.)
- [58] J. Kurzweil and J. Jarnik. A convergence effect in ordinary differential equations. In Asymptotic methods in mathematical physics (Russian), pages 134–144, 301. "Naukova Dumka", Kiev, 1988. (Cited on page 154.)
- [59] J. Kurzweil and J. Jarník. Iterated Lie brackets in limit processes in ordinary differential equations. *Results Math.*, 14(1-2):125–137, 1988. (Cited on page 154.)
- [60] F. Laudenbach. Homologie de morse dans la perspective de l'homologie de floer. Mini-cours dans le cadre de la rencontre GIRAGA XIII, Yaoundé, 2010. (Cited on pages 131 and 132.)
- [61] C. K. Law and J. H. Eberly. Arbitrary control of a quantum electromagnetic field. *Phys. Rev. Lett.*, 76:1055–1058, Feb 1996. (Cited on page 70.)
- [62] Z. Leghtas, A. Sarlette, and P. Rouchon. Adiabatic passage and ensemble control of quantum systems. *Journal of Physics B: Atomic, Molecular and Optical Physics*, 44(15):154017, 2011. (Cited on pages 2, 21, 26, 45, 56, and 60.)

- [63] J.-S. Li and N. Khaneja. Ensemble control of Bloch equations. *IEEE Trans. Automat. Control*, 54(3):528–536, 2009. (Cited on pages 1, 7, 25, 30, and 31.)
- [64] J.-S. Li and J. Qi. Ensemble control of time-invariant linear systems with linear parameter variation. *IEEE Trans. Automat. Control*, 61(10):2808–2820, 2016. (Cited on pages 7 and 30.)
- [65] P. Libermann and C.-M. Marle. Symplectic geometry and analytical mechanics, volume 35 of Mathematics and its Applications. D. Reidel Publishing Co., Dordrecht, 1987. Translated from the French by Bertram Eugene Schwarzbach. (Cited on page 113.)
- [66] W. Liu. Averaging theorems for highly oscillatory differential equations and iterated Lie brackets. SIAM J. Control Optim., 35(6):1989–2020, 1997. (Cited on pages 22, 45, and 154.)
- [67] E. Paduro and M. Sigalotti. Approximate controllability of the two trapped ions system. Quantum Inf. Process., 14(7):2397–2418, 2015. (Cited on page 70.)
- [68] M. Reed and B. Simon. Methods of Modern Mathematical Physics. IV. Analysis of Operators. Academic Press, New York, 1978. (Cited on pages 69, 73, and 112.)
- [69] F. Rellich. Perturbation theory of eigenvalue problems. Assisted by J. Berkowitz. With a preface by Jacob T. Schwartz. Gordon and Breach Science Publishers, New York-London-Paris, 1969. (Cited on page 73.)
- [70] P. Rouchon. Quantum systems and control. ARIMA Rev. Afr. Rech. Inform. Math. Appl., 9:325–357, 2008. (Cited on page 149.)
- [71] J. A. Sanders, F. Verhulst, and J. Murdock. Averaging methods in nonlinear dynamical systems, volume 59 of Applied Mathematical Sciences. Springer, New York, second edition, 2007. (Cited on pages 22 and 45.)
- [72] M. Schönlein and U. Helmke. Controllability of ensembles of linear dynamical systems. Math. Comput. Simulation, 125:3–14, 2016. (Cited on pages 7 and 30.)
- [73] E. A. Shapiro, V. Milner, and M. Shapiro. Complete transfer of populations from a single state to a preselected superposition of states using piecewise adiabatic passage: Theory. *Phys. Rev. A*, 79:023422, Feb 2009. (Cited on pages 2 and 26.)
- [74] B. W. Shore. The Theory of Coherent Atomic Excitation, Volume 1, Simple Atoms and Fields. July 1990. (Cited on pages 2, 26, and 56.)
- [75] T. E. Skinner, T. O. Reiss, B. Luy, N. Khaneja, and S. J. Glaser. Application of optimal control theory to the design of broadband excitation pulses for highresolution nmr. *Journal of Magnetic Resonance*, 163(1):8 – 15, 2003. (Cited on pages 7 and 31.)

- [76] E. M. Stein. Harmonic analysis: real-variable methods, orthogonality, and oscillatory integrals, volume 43 of Princeton Mathematical Series. Princeton University Press, Princeton, NJ, 1993. With the assistance of Timothy S. Murphy, Monographs in Harmonic Analysis, III. (Cited on page 118.)
- [77] H. J. Sussmann and W. Liu. Lie Bracket Extensions and Averaging: The Single-Bracket Case, pages 109–147. Springer US, Boston, MA, 1993. (Cited on pages 22, 45, and 154.)
- [78] S. Teufel. Adiabatic perturbation theory in quantum dynamics, volume 1821 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2003. (Cited on pages 2, 3, 23, 26, 27, 46, 55, 69, 75, 113, and 114.)
- [79] L. Van Damme, Q. Ansel, S. J. Glaser, and D. Sugny. Robust optimal control of two-level quantum systems. *Phys. Rev. A*, 95:063403, Jun 2017. (Cited on pages 7 and 31.)
- [80] N. V. Vitanov, T. Halfmann, B. W. Shore, and K. Bergmann. Laser-induced population transfer by adiabatic passage techniques. *Annual Review of Physical Chemistry*, 52(1):763–809, 2001. (Cited on page 148.)
- [81] N. V. Vitanov, A. A. Rangelov, B. W. Shore, and K. Bergmann. Stimulated raman adiabatic passage in physics, chemistry, and beyond. *Rev. Mod. Phys.*, 89:015006, Mar 2017. (Cited on pages 23 and 46.)
- [82] J. von Neumann and E. P. Wigner. Über das Verhalten von Eigenwerten bei adiabatischen Prozessen, pages 294–297. Springer Berlin Heidelberg, Berlin, Heidelberg, 1993. (Cited on pages 5, 28, and 63.)
- [83] L. Yatsenko, S. Guérin, and H. Jauslin. Topology of adiabatic passage. *Physical Review A*, 65, 07 2001. (Cited on pages 2 and 26.)
- [84] H. Yuan and S. Lloyd. Controllability of the coupled spin-<sup>1</sup>/<sub>2</sub> harmonic oscillator system. *Phys. Rev. A*, 75:052331, May 2007. (Cited on page 70.)



Titre : Contrôle adiabatique des systèmes quantiques

**Mots clés :** Contrôle quantique, théorie adiabatique, approximation rotating wave, contrôle robuste, singularités, champs de direction, moyennisation de systèmes dynamiques

**Résumé :** Le but principal de la thèse est d'étudier les liens entre les singularités du spectre d'un Hamiltonien quantique contrôlé et les questions de contrôlabilité de l'équation Schrödinger associée.

La question que l'on se pose est de savoir comment contrôler une famille de systèmes quantiques dépendant d'un paramètre avec une entrée de commande commune. Grâce à l'étude des familles à un paramètre de Hamiltoniens et de leurs singularités génériques, on donne une stratégie de contrôle explicite pour le problème de contrôlabilité d'ensemble lorsque certaines conditions géométriques sur le spectre des Hamiltoniens de la famille à un paramètre sont satisfaites, en particulier l'existence d'intersections coniques ou semi-coniques, en se basant sur la théorie adiabatique.

Pour l'étude d'un système unique non-paramétrique, on se concentre sur une classe de courbes dans l'espace des contrôles, appelées les courbes nonmixantes, qui peuvent optimiser la dynamique adiabatique près des intersections coniques et semiconiques. Elles sont liées à la géométrie des espaces propres du Hamiltonien contrôlé et l'approximation adiabatique possède une meilleure précision le long de celles-ci.

On étudie ensuite la compatibilité de l'approximation adiabatique avec l'approximation rotating wave. De telles approximations sont généralement combinées par les physiciens. On montre que cela ne se justifie que dans certaines conditions sur les échelles de temps. On étudie également les questions de contrôle d'ensemble dans ce cas.

### Title : Adiabatic control of quantum systems

**Keywords :** Quantum control, adiabatic theory, rotating wave approximation, robust control, singularities, line fields, averaging of dynamical systems

**Abstract :** The main purpose of the thesis is to study the links between the singularities of the spectrum of a controlled quantum Hamiltonian and the controllability issues of the associated Schrödinger equation.

The principal issue that is developed is how to control a parameter-dependent family of quantum systems with a common control input. Thanks to the study oneparametric families of Hamiltonians and their generic singularities, we give an explicit control strategy for the ensemble controllability problem when geometric conditions on the spectrum of the Hamiltonian are satisfied, in particular the existence of conical or semiconical intersections, using adiabatic theory.

For the study of a single system, we focus on a class

of curves in the space of controls, called the nonmixing curves, that can optimize the adiabatic dynamics near conical and semi-conical intersections. They are linked to the geometry of the eigenspaces of the controlled Hamiltonian and the adiabatic approximation holds with higher precision along them.

We propose to study the compatibility of the adiabatic approximation with the rotating wave approximation. Such approximations are usually done in cascade by physicists. It is shown in this work that this is justified for finite dimensional quantum systems only under certain conditions on the time scales. We also study ensemble control issues in this case.